Extension of gyrokinetics to transport time scales

by

Félix Ignacio Parra Díaz

Aeronautical Engineer, Universidad Politécnica de Madrid (2004) S.M., Massachusetts Institute of Technology (2007)

Submitted to the Department of Aeronautics and Astronautics

in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2009

© Massachusetts Institute of Technology 2009. All rights reserved.

ARCHIVES

Author Department of Aeronautics and Astronautics May 21, 2009 Certified by ••••••• // Peter J. Catto Senior Research Scientist Ŋ Thesis Supervisor Certified by Manuel Martínez-Sánchez **Full Professor** Thesis Committee Chair \mathcal{D} Certified by... Jeffrey P. Freidberg **Full Professor** , Thésis Committee Member Accepted by David L. Darmofal MASSACHUSETTS INSTITUTE OF TECHNOLOGY Associate Department Head Chair, Committee on Graduate Students JUN 2 4 2009 LIBRARIES

Extension of gyrokinetics to transport time scales

by

Félix Ignacio Parra Díaz

Submitted to the Department of Aeronautics and Astronautics on May 21, 2009, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

In the last decade, gyrokinetic simulations have greatly improved our theoretical understanding of turbulent transport in fusion devices. Most gyrokinetic models in use are δf simulations in which the slowly varying radial profiles of density and temperature are assumed to be constant for turbulence saturation times, and only the turbulent electromagnetic fluctuations are calculated. Due to the success of these models, new massive simulations are being built to self-consistently determine the radial profiles of density and temperature. However, these new codes have failed to realize that modern gyrokinetic formulations, composed of a gyrokinetic Fokker-Planck equation and a gyrokinetic quasineutrality equation, are only valid for δf simulations that do not reach the longer transport time scales necessary to evolve radial profiles. In tokamaks, due to axisymmetry, the evolution of the axisymmetric radial electric field is a challenging problem requiring substantial modifications to gyrokinetic treatments. The radial electric field, closely related to plasma flow, is known to have a considerable impact on turbulence saturation, and any self-consistent global simulation of turbulent transport needs an accurate procedure to determine it. In this thesis, I study the effect of turbulence on the global electric field and plasma flows. By studying the current conservation equation, or vorticity equation, I prove that the long wavelength, axisymmetric flow must remain neoclassical and I show that the tokamak is intrinsically ambipolar, i.e., the radial current is zero to a very high order for any long wavelength radial electric field. Intrinsic ambipolarity is the origin of the problems with the modern gyrokinetic approach since the lower order gyrokinetic quasineutrality (if properly evaluated) is effectively independent of the radial electric field. I propose a new gyrokinetic formalism in which, instead of a quasineutrality equation, a current conservation equation or vorticity equation is solved. The vorticity equation makes the time scales in the problem explicit and shows that the radial electric field is determined by the conservation of toroidal angular momentum.

Thesis Supervisor: Peter J. Catto Title: Senior Research Scientist

Acknowledgments

First and foremost, I am deeply grateful to my wife Violeta. She is probably the only person that can truly appreciate how much effort I had to dedicate to this thesis. Without her unrelenting support I would have not been able to achieve my doctoral degree.

I am also indebted to my parents Nieves and Ignacio, and my brother Braulio. They were able to support me despite the distance.

This thesis would have not been possible without Peter Catto's judicious tutoring and Bill Dorland's support. Peter has taught me all I know about kinetic theory in plasma physics, led my work through the rough patches and, what is more important, laid down clearly with his example the path I want to follow in my future career. Bill made possible this thesis with his support and warm friendship.

I sincerely thank the members of my committee, Jeffrey Freidberg and Manuel Martínez-Sánchez, and the readers, Ian Hutchinson and Jesús Ramos, for being so patient with the initial write-up. Their questions and suggestions helped me improve the presentation of this thesis.

Finally, I am grateful to my office mates Grisha, Antoine, Susan and Matt, and to my friends Arturo, Carolina, Luis, Kate, Roberto, Ana, Joaquim, Joel, Chantal, Tanya, Andrew, Robyn, Michael, Kyle, Alex and Steve. I could not have asked for a more enjoyable and supporting group of friends.

Contents

1	Inti	oduction	17
	1.1	Turbulence in tokamaks	18
	1.2	Gyrokinetics: history and current challenges	19
	1.3	Calculating the radial electric field	23
2	Vor	ticity and intrinsic ambipolarity	30
	2.1	Orderings and assumptions	31
	2.2	Vorticity equation	36
	2.3	Radial electric field in the vorticity equation	39
3	Der	ivation of gyrokinetics	43
	3.1	Orderings	44
	3.2	Gyrokinetic variables	45
	3.3	Gyrokinetic Fokker-Planck equation	55
	3.4	Gyrokinetic quasineutrality equation	58
4	Gyı	okinetic vorticity equation	64
	4.1	Notation and assumptions	65
	4.2	General vorticity equation in gyrokinetics	69
	4.3	Transport in gyrokinetics	71
		4.3.1 Gyrokinetic equation in physical phase space	73
		4.3.2 Transport of a general function $G(\mathbf{r}, \mathbf{v}, t)$ at $k_{\perp} \rho_i \sim 1$	77
		4.3.3 Transport of particles at $k_{\perp}\rho_i \sim 1$	77

		4.3.4	Transport of momentum at $k_{\perp}\rho_i \sim 1$	81
	4.4	Vortici	ty equation for gyrokinetics	84
		4.4.1	Vorticity from quasineutrality	85
		4.4.2	Vorticity from moment description	87
	4.5	Examp	ble: quasineutrality in a θ -pinch $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	94
	4.6	Discus	sion	99
5	Solv	ving fo	r the radial electric field	101
	5.1	Ion vis	scosity and the axisymmetric potential	102
	5.2	Examp	ble: the solution of a θ -pinch \ldots \ldots \ldots \ldots \ldots \ldots	107
	5.3	Distrib	oution function and potential to second order	110
		5.3.1	Higher order ion distribution function	112
		5.3.2	Higher order electrostatic potential	115
	5.4	Discus	sion \ldots	120
c	Com	alucio		192
O	Con	ciusio	115	140
o A	Der	ivation	of the gyrokinetic variables	125
o A	Der A.1	ivation First o	order gyrokinetic variables	125 126
o A	Der A.1	ivation First of A.1.1	a of the gyrokinetic variables order gyrokinetic variables $\dots \dots \dots$	123 126 126 127
o A	Der A.1	ivation First o A.1.1 A.1.2	a of the gyrokinetic variables order gyrokinetic variables $\dots \dots \dots$	123 126 126 127 127
A	Der A.1	ivation First o A.1.1 A.1.2 A.1.3	a of the gyrokinetic variables order gyrokinetic variables $\dots \dots \dots$	123 126 127 127 128
A	Der A.1	ivation First o A.1.1 A.1.2 A.1.3 A.1.4	a of the gyrokinetic variables order gyrokinetic variables order gyrokinetic variables Useful relations for ϕ Calculation of \mathbf{R}_1 Calculation of E_1 Calculation of φ_1	123 126 127 127 128 128
A	Der A.1	ivation First o A.1.1 A.1.2 A.1.3 A.1.4 A.1.5	a of the gyrokinetic variables order gyrokinetic variables order gyrokinetic variables Useful relations for ϕ Calculation of \mathbf{R}_1 Calculation of E_1 Calculation of ϕ_1 Calculation of μ_1	123 126 127 127 128 128 129
A	Der A.1 A.2	ivation First of A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second	a of the gyrokinetic variables order gyrokinetic variables breful relations for ϕ Calculation of \mathbf{R}_1 Calculation of E_1 Calculation of ϕ_1 Calculation of μ_1 Calculation of μ_1 Calculation of μ_1	123 126 127 127 127 128 128 129 130
A	Der A.1 A.2	ivation First o A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second A.2.1	a of the gyrokinetic variables order gyrokinetic variables Useful relations for ϕ Calculation of \mathbf{R}_1 Calculation of E_1 Calculation of ϕ_1 Calculation of μ_1 Calculation of μ_1 Calculation of ϕ_1 Calculatio	123 126 127 127 128 128 128 129 130 130
A	Der A.1 A.2	ivation First o A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second A.2.1 A.2.2	The of the gyrokinetic variables order gyrokinetic variables $\dots \dots \dots$	123 126 127 127 128 128 128 129 130 130 132
A	Der A.1 A.2	ivation First of A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second A.2.1 A.2.2 A.2.3	a of the gyrokinetic variables order gyrokinetic variables Useful relations for ϕ Calculation of \mathbf{R}_1 Calculation of E_1 Calculation of ϕ_1 Calculation of μ_1 Calculation of μ_1 Calculation of \mathbf{R}_2 Calculation of \mathbf{R}_2	123 126 127 127 128 128 129 130 130 132 133
A	Der A.1 A.2	ivation First of A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second A.2.1 A.2.2 A.2.3 A.2.4	The of the gyrokinetic variables order gyrokinetic variables $\dots \dots \dots$	123 126 127 127 128 128 129 130 130 132 133 133
A	Der A.1 A.2	ivation First of A.1.1 A.1.2 A.1.3 A.1.4 A.1.5 Second A.2.1 A.2.2 A.2.3 A.2.4 Jacobi	The of the gyrokinetic variables order gyrokinetic variables Useful relations for ϕ	123 126 127 127 128 128 129 130 130 130 132 133 133

Β	Useful	gyroaverages	and	gyrophase	derivatives	1
---	--------	--------------	-----	-----------	-------------	---

\mathbf{C}	Gyr	okinetic equivalence	150
	C.1	Constant magnetic field results	151
	C.2	Time derivative of \mathbf{R}	152
	C.3	Time derivative of E	154
	C.4	Second order correction μ_2	155
	C.5	Comparisons with Dubin et al	157
D	Qua	sineutrality equation at long wavelengths	163
	D.1	Polarization density at long wavelengths	163
	D.2	Quasineutrality equation for $k_{\perp}L \sim 1$	164
\mathbf{E}	Gyr	ophase dependent piece of f_i for $k_\perp L \sim 1$	169
	E.1	General gyrophase dependent piece	169
	E.2	Gyrophase dependent piece in a θ -pinch $\ldots \ldots \ldots \ldots \ldots \ldots$	172
\mathbf{F}	Gyr	okinetic equation in physical phase space	173
	F.1	Gyrokinetic equation in \mathbf{r} , E_0 , μ_0 and φ_0 variables	173
	F.2	Conservative form in \mathbf{r} , E_0 , μ_0 and φ_0 variables $\ldots \ldots \ldots \ldots$	175
G	Det	ails of the particle and momentum transport calculation	177
	G.1	Details of the particle transport calculation	177
	G.2	Details of the momentum transport calculation	178
н	Fini	te gyroradius effects in the like-collision operator	181
I	Gyı	okinetic vorticity	186
J	Flu	x surface averaged gyrokinetic vorticity equation	189
	J.1	Limit of $\langle (cI/B) \nabla \cdot \boldsymbol{\pi}_{ig } \rangle_{\psi}$ for $k_{\perp} \rho_i \to 0$	189
	J.2	Limit of $\langle \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi - (c/B)(\nabla \cdot \overleftrightarrow{\pi}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) \rangle_{\psi}$ for $k_{\perp} \rho_i \to 0$	191
	J.3	Limit of $\langle (cI/B)\tilde{F}_{iE}\rangle_{\psi}$ for $k_{\perp}\rho_i \to 0$	193
	J .4	Limit of $\langle Zen_i \mathbf{V}_{iC} \cdot \nabla \psi - cR\mathbf{F}_{iC} \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi}$ for $k_{\perp} \rho_i \to 0$	195

K	Gyr	oviscosity in gyrokinetics	196
	K.1	Evaluation of equation (4.65)	196
	K.2	Collisional piece in up-down symmetric tokamaks	198

List of Figures

2-1	Magnetic coordinates in tokamaks	32
3-1	Gyrokinetic gyromotion and functions $\langle \phi angle$ and $\widetilde{\phi}$	49
3-2	Kinetic energy along the gyromotion	51
3-3	Geometric effects on the gyrokinetic gyrophase	53
3-4	Ion density in gyrokinetics	61
4-1	Ion diamagnetic flow	72
4-2	Gyrokinetic equation in physical phase space	74
4-3	Finite gyroradius effects in $n_i ilde{\mathbf{V}}_i$	79
4-4	Geometry of the θ -pinch	94
5-1	High flow ordering for the ion parallel velocity with $B_p/B \ll 1$	112
5-2	Procedure to obtain the radial electric field	121

List of Tables

1.1	Different methods to obtain the radial electric field	25
2.1	Typical numbers for Alcator C-Mod and DIII-D	34
4.1	Order of magnitude estimates for vorticity equation (2.9)	69
4.2	Order of magnitude estimates for particle conservation equation (4.28)	80
4.3	Order of magnitude estimates for momentum equation (4.38)	83
4.4	Order of magnitude estimates for vorticity equation (4.45)	86
4.5	Order of magnitude estimates for vorticity equation (4.53)	89

Nomenclature

Miscellaneous

$\overline{(\ldots)}$	Gyroaverage holding \mathbf{r} , $v_{ }$, v_{\perp} and t fixed.
$\langle \ldots \rangle$	Gyroaverage holding \mathbf{R} , E , μ and t fixed.
$\langle \ldots angle_\psi$	In tokamaks, flux surface average.
$\overline{ abla}$	Gradient holding E_0 , μ_0 , φ_0 and t fixed.
Greek letters	
ψ	In tokamaks, poloidal magnetic field flux, radial coordinate.
θ	In tokamaks, poloidal angle; in θ -pinches, azimuthal angle.
$\hat{oldsymbol{ heta}}$	In θ -pinches, unit vector in the azimuthal direction.
$\zeta, \hat{\boldsymbol{\zeta}}$	In tokamaks, toroidal angle and unit vector in the toroidal direction.
ϵ	In tokamaks, inverse aspect ratio a/R .
$ ho_e, ho_i$	Electron and ion gyroradii, mcv_e/eB and Mcv_i/ZeB .
Ω_e,Ω_i	Electron and ion gyrofrequencies, eB/mc and ZeB/Mc .
δ_e,δ_i	Expansion parameters $\delta_e = \rho_e/L \ll \delta_i = \rho_i/L \ll 1$.
λ_D	Debye length.
$ u_{ii}, u_{ie}, u_{ee}, u_{ei}$	Ion-ion, ion-electron, electron-electron and electron-ion Braginskii
	collision frequencies.
ω_*	Drift wave frequency.
$\omega_*^{n,T}$	Drift wave frequency in gyrokinetic equation (3.57) dependent on
	$ abla n_i ext{ and } abla T_i.$

- ϕ Electrostatic potential.
- $\langle \phi \rangle$ Electrostatic potential averaged in a gyromotion [see (3.16)].
- $\tilde{\phi}$ Difference between the potential seen by the particle and the potential averaged in a gyromotion [see (3.17)].
- $\tilde{\Phi}$ Indefinite integral of $\tilde{\phi}$ with vanishing gyroaverage [see (3.18)].
- μ Gyrokinetic magnetic moment defined to be an adiabatic invariant to higher order [see (3.33)].
- μ_{g} Gyrokinetic magnetic moment in which the explicit dependence on the potential has been subtracted.
- μ_0 Lowest order magnetic moment of the particle $v_{\perp}^2/2B$.
- μ_1 First order correction to the gyrokinetic magnetic moment [see (3.34)].
- μ_{10} First order correction to the gyrokinetic magnetic moment in which the explicit dependence on the potential has been subtracted [see (4.21)].
- φ Gyrokinetic gyrophase in which the fast time variation has been averaged out [see (3.28)].
- φ_0 Lowest order gyrophase of the particle [see (2.2)].
- φ_1 First order correction to the gyrokinetic gyrophase [see (3.29)].
- φ_{10} First order correction to the gyrokinetic gyrophase in which the explicit dependence on the potential has been subtracted [see (4.22)].
- $\dot{\pi}_i$ Ion viscosity. Its definition includes the Reynolds stress because the average velocity has not been subtracted [see (2.6)].
- $\pi_{ig||}$ Vector that gives the transport of parallel ion momentum by the $E \times B$ and magnetic drifts and the finite gyroradius drift $\tilde{\mathbf{v}}_1$ [see (4.39)].
- $\vec{\pi}_{ig\times}$ Tensor that gives the transport of perpendicular ion momentum by the parallel velocity, the $E \times B$ and magnetic drifts and the finite gyroradius drift $\tilde{\mathbf{v}}_1$ [see (4.40)].

$\overleftarrow{\pmb{\pi}}_{iG}$	Effective viscosity for gyrokinetic vorticity equation (4.53).
ϖ	Vorticity [see (2.11)].
$arpi_G$	Gyrokinetic "vorticity" [see (4.51)].
$arpi_G^{(2)}$	Higher order gyrokinetic "vorticity" [see (5.41)].
Roman	letters
a	In tokamaks, minor radius.
$\mathbf{B},B,\hat{\mathbf{b}}$	Magnetic field, magnetic field magnitude, and unit vector parallel to
	the magnetic field.
B_p	In tokamaks, magnitude of the poloidal component of the magnetic
	field.
c	Speed of light.
D_{gB}	GyroBohm diffusion coefficient $\delta_i \rho_i v_i$.
e	Electron charge magnitude.
$\hat{\mathbf{e}}_1,\hat{\mathbf{e}}_2$	Orthonormal vectors such that $\hat{\mathbf{e}}_1\times\hat{\mathbf{e}}_2=\hat{\mathbf{b}}$ used to define the gy-
	rophase φ_0 [see (2.2)].
E	Gyrokinetic kinetic energy in which the fast time variation has been
	averaged out [see (3.24)].
E_0	Kinetic energy of the particle $v^2/2$.
E_1	First order correction to the gyrokinetic kinetic energy [see (3.25)].
E_2	Second order correction to the gyrokinetic kinetic energy [see (3.26)].
E	Electric field.
f_e	Electron distribution function, dependent on \mathbf{r} , E_0 , μ_0 and φ_0 .
\overline{f}_e	Gyrophase independent piece of the electron distribution function.
$f_e - \overline{f}_e$	Gyrophase dependent piece of the electron distribution function.
f_i	Ion distribution function, usually dependent on the gyrokinetic vari-
	ables R , E and μ .
\widetilde{f}_i	Small piece of the ion distribution function that depends on the
	gyrokinetic gyrophase responsible for classical diffusion [see (3.38)].

f_{ig}	Distribution function found by replacing the gyrokinetic variables
	\mathbf{R} , E and μ in $f_i(\mathbf{R}, E, \mu, t)$ by \mathbf{R}_g , E_0 and μ_0 [see (4.3)].
f_{iG}	Distribution function found by replacing the gyrokinetic variables
	R , E and μ in $f_i(\mathbf{R}, E, \mu, t)$ by \mathbf{R}_g , E_0 and μ_g .
f_{i0}	Distribution function found by replacing the gyrokinetic variables
	R , E and μ in $f_i(\mathbf{R}, E, \mu, t)$ by r , E_0 and μ_0 .
\overline{f}_i	Gyrophase independent piece of the ion distribution function when
	written in physical phase space variables \mathbf{r} , E_0 , μ_0 and φ_0 .
$f_i - \overline{f}_i$	Gyrophase dependent piece of the ion distribution function when
	written in physical phase space variables \mathbf{r} , E_0 , μ_0 and φ_0 .
f_{Me}, f_{Mi}	Lowest order electron and ion distribution functions, assumed to be
	stationary Maxwellians.
\mathbf{F}_{ei}	Electron collisional momentum exchange with ions.
\mathbf{F}_{iB}	Change in the perpendicular momentum of the gyromotion due to
	variations in the magnetic field strength [see (4.42)].
$\mathbf{F}^{(2)}_{iB}$	Higher order version of \mathbf{F}_{iB} [see (5.37)].
\mathbf{F}_{iC}	Force due to finite gyroradius effects on collisions [see (4.43)].
$ ilde{F}_{iE}$	Change in the parallel momentum due to the short wavelength com-
	ponents of the electrostatic potential [see (4.41)].
$ ilde{F}^{(2)}_{iE}$	Higher order version of \tilde{F}_{iE} [see (5.36)].
h_i	Correction to the Maxwellian for ions.
$h_{i1},h_{i1}^{ m tb},h_{i1}^{ m nc}$	First order correction to the Maxwellian in $\delta_i,$ decomposed into two
	pieces: the piece due to turbulence and the neoclassical contribution.
$h_{i2},h_{i2}^{ m tb},h_{i2}^{ m nc}$	Second order correction to the Maxwellian in δ_i , decomposed into two
	pieces: the piece due to turbulence and the neoclassical contribution.
Ι	Function $R\mathbf{B} \cdot \hat{\boldsymbol{\zeta}}$; it only depends on ψ to lowest order.
Ī	Unit matrix.
J, J_u	Jacobian of the gyrokinetic transformation [see (3.44) and (3.47)].
J	Current density.

- \mathbf{J}_d Current density due to magnetic drifts [see (2.10)].
- $\mathbf{J}_{gd} \qquad \text{Current density due to magnetic drifts calculated integrating } \mathbf{v}_{M0} f_{ig}$ over velocity space instead of $\mathbf{v}_{M0} f_i$ [see (4.47)].
- $J_{g||}^{(2)} \qquad \text{Parallel current density calculated integrating } v_{||}f_{ig} \text{ over velocity} \\ \text{space instead of } v_{||}f_i \text{ [see (5.39)]}. \text{ The superindex (2) emphasizes} \\ \text{that } J_{||} = \mathbf{J} \cdot \hat{\mathbf{b}} \text{ and } J_{g||}^{(2)} \text{ differ only in higher order terms.} \end{cases}$

 $\tilde{\mathbf{J}}_i$ Polarization current density in gyrokinetic vorticity equation (4.45).

 $\tilde{\mathbf{J}}_{i\phi}$ Polarization current density in gyrokinetic vorticity equation (4.53).

 $\tilde{\mathbf{J}}_{i\phi}^{(2)}$ Polarization current density in gyrokinetic vorticity equation (5.40).

- k_{\parallel}, k_{\perp} Wavenumbers parallel and perpendicular to the magnetic field.
- L Characteristic length in the problem.
- m, M Electron and ion masses.

$$n_e, n_i$$
 Electron and ion densities.

- n_{ip} Gyrokinetic polarization density defined in (3.55).
- $n_{ip}^{(2)}$ Higher order gyrokinetic polarization density defined in (5.33).
- p_e, p_i Electron and ion "pressures." They are not the usual definitions because the average velocity is not subtracted.
- $p_{e||}, p_{e\perp}$ Electron parallel and perpendicular "pressures." They are not the usual definitions because the average velocity is not subtracted.
- $p_{i\parallel}, p_{i\perp}$ Ion parallel and perpendicular "pressures." They are not the usual definitions because the average velocity is not subtracted.
- $p_{ig||}, p_{ig\perp}$ Parallel and perpendicular "pressures" calculated integrating $Mv_{||}^2 f_{ig}$ and $(Mv_{\perp}^2/2)f_{ig}$ over velocity space instead of $Mv_{||}^2 f_i$ and $(Mv_{\perp}^2/2)f_i$.
- $\mathbf{\tilde{P}}_{i}$ Total ion stress tensor, including contributions due to the average ion velocity.
- q Safety factor.
- **r** Position of the particle.

- r In θ -pinches, radial coordinate.
- $\hat{\mathbf{r}}$ In θ -pinches, unit vector in the radial direction.
- R In tokamaks, the distance between the axis of symmetry and the position of the particle, also used as major radius in estimates of order of magnitude.
- **R** Position of the gyrocenter [see (3.12)].
- \mathbf{R}_{g} Position of the guiding center $\mathbf{r} + \Omega_{i}^{-1} \mathbf{v} \times \hat{\mathbf{b}}$.
- **R**₁ First order correction to the gyrocenter position $\Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$.
- \mathbf{R}_2 Second order correction to the gyrocenter position [see (3.15)].
- T_e, T_i Electron and ion temperatures.
- u Velocity of the gyrocenter parallel to the magnetic field [see (3.23)].
- u_g Parallel velocity found by replacing the gyrokinetic variables **R**, Eand μ in $u(\mathbf{R}, E, \mu)$ by \mathbf{R}_g , E_0 and μ_g . [see (5.29)].
- **v** Velocity of the particle.
- v_{\parallel} Velocity component parallel to the magnetic field.
- $v_{||0}$ Velocity parallel to the magnetic field with finite gyroradius modifications [see (4.16)].
- $\mathbf{v}_{\perp}, v_{\perp}$ Velocity perpendicular to the magnetic field and its magnitude.
- v_e, v_i Electron and ion thermal velocities, $\sqrt{2T_e/m}$ and $\sqrt{2T_i/M}$.
- \mathbf{v}_E Gyrokinetic $E \times B$ drift [see (3.21)].
- \mathbf{v}_{E0} Lowest order gyrokinetic $E \times B$ drift [see (4.18)].
- \mathbf{v}_d Gyrokinetic drift composed of \mathbf{v}_E and \mathbf{v}_M .
- \mathbf{v}_{de} Electron drift [see (3.50)].
- \mathbf{v}_{dg} Drift found by replacing the gyrokinetic variables \mathbf{R} , E and μ in $\mathbf{v}_d(\mathbf{R}, E, \mu)$ by \mathbf{R}_g , E_0 and μ_g [see (5.28)].
- \mathbf{v}_M Gyrokinetic magnetic drift [see (3.22)].
- \mathbf{v}_{M0} Standard magnetic drift [see (4.17)].
- $\tilde{\mathbf{v}}_1$ Drift due to finite gyroradius effects [see (4.19)].

- V' In tokamaks, flux surface volume element $dV/d\psi$.
- $\mathbf{V}_e, \, \mathbf{V}_i$ Electron and ion average velocities.
- \mathbf{V}_{ig} Ion average velocity calculated integrating $\mathbf{v} f_{ig}$ over velocity space instead of $\mathbf{v} f_i$.
- $$\begin{split} V_{ig||}^{(2)} & \text{Ion parallel average velocity calculated integrating } v_{||}f_{ig} \text{ over velocity} \\ & \text{space instead of } v_{||}f_i \text{ [see (5.34)]}. \text{ The superindex (2) emphasizes that} \\ & V_{i||} = \mathbf{V}_i \cdot \hat{\mathbf{b}} \text{ and } V_{ig||}^{(2)} \text{ differ only in higher order terms.} \end{split}$$
- \mathbf{V}_{iC} Ion average velocity due to finite gyroradius effects on collisions [see (4.33)].
- \mathbf{V}_{iE} , \mathbf{V}_{igd} Ion average velocities due to the gyrokinetic $E \times B$ and magnetic drifts [see (4.31) and (4.32)].
- $\tilde{\mathbf{V}}_i$ Ion average velocity due to finite gyroradius contribution $\tilde{\mathbf{v}}_1$ [see (4.30)].
- $V_{i||}^{nc}$ Neoclassical ion parallel velocity.
- Z Ion charge number.

Chapter 1

Introduction

Magnetic confinement is the most promising concept for production of fusion energy and the tokamak is the best candidate among all the possible magnetic confinement devices. However, transport of particles, energy and momentum is still not well understood in tokamaks. The transport is mainly turbulent, and modelling and predicting how it evolves is necessary to build a viable reactor.

The understanding of turbulent transport in tokamaks has greatly improved in the last decade, mainly due to more comprehensive simulations [1, 2, 3, 4, 5, 6]. These simulations employ the gyrokinetic formalism to shorten the computational time. Gyrokinetics is a sophisticated asymptotic method that keeps finite gyroradius effects without solving on the gyrofrequency time scale – a time too short to be of interest in turbulence. Unfortunately, modern formulations of gyrokinetics are still only valid for times shorter than the energy diffusion time or transport time scale. As simulations try to reach longer time scales, the gyrokinetic formalism needs to be extended. This thesis identifies the shortcomings of gyrokinetics at long transport times and solves one of the most pressing issues, namely, the calculation of the axisymmetric radial electric field, and thereby, the transport of momentum.

In this introduction, first I will review the characteristics of turbulence in tokamaks in section 1.1. This review is followed by a brief history of gyrokinetic simulations in section 1.2, with emphasis on new developments. The new codes being built will require a new formulation of gyrokinetics valid for longer time scales. In section 1.3, I will close the introduction by discussing the requirements for future gyrokinetic formalisms and summarizing the rest of the thesis.

1.1 Turbulence in tokamaks

Currently, the main part of the turbulence in the core is believed to be driven by drift waves. These waves propagate in the plasma perpendicularly to density and temperature gradients. They become unstable in tokamaks due to the curvature in the magnetic field and other inhomogeneities. There are several modes that are considered important, but the two of most interest here are the Ion Temperature Gradient mode (ITG) [7] and the Trapped Electron Mode (TEM) [8].

These modes have frequencies much smaller than the ion gyrofrequency. They are unstable at short wavelengths – the fastest growing mode wavelengths are on the order of the ion gyroradius because shorter wavelengths are stabilized by finite gyroradius effects. The measurements in tokamaks suggest turbulence correlation lengths on the order of five to ten gyroradii [9], which agrees with this idea. The anisotropy induced by the magnetic field is reflected in the spatial structure of these modes. The wavelength along the magnetic field, on the order of the characteristic size of the device, is much longer than the perpendicular wavelengths.

The ITG and TEM instabilities provide energy for the turbulence at short wavelengths. By nonlinear beating, part of the turbulence energy is deposited in a radial mode known as zonal flow [10, 11, 12]. The zonal flow is a radial structure in the radial electric field that gives rise to a sheared poloidal and toroidal $E \times B$ flow. It is a robust mode because it does not have any parallel electric field and hence electrons cannot shield it or Landau damp it. Then, any energy deposited in the zonal flow will remain there, leading to a rapid nonlinear growth of this mode. It has an impact on turbulence dynamics because the velocity shear decorrelates the turbulence at the shorter wavelengths. The statistical equilibrium of the turbulence is determined by the feedback between zonal flow and short wavelength fluctuations. The effect of zonal flow is so important that it can suppress the turbulence when the instability is not too strong [13].

The radial electric field is crucial in the saturation of turbulence. Its short wavelength radial structure is the zonal flow, whose importance in turbulence dynamics has already been discussed. It is also quite clear that it plays an important role in the pedestal of high confinement or H-mode plasmas in tokamaks [14], where the shear in the macroscopic radial electric field becomes large. Experimentally, it is observed that the radial electric field shear increases before the turbulent fluctuations are suppressed, radial transport is quenched and the gradient of density increases to form the pedestal [15]. Both zonal flow and transport barriers highlight the importance of the calculation of the radial electric field in any turbulence simulation. In this thesis, I will show that the traditional gyrokinetic approach is unable to provide the correct long wavelength axisymmetric radial electric field. This problem has gone undetected up until now because it is only noticeable at long time scales.

To summarize, the turbulence in a tokamak is characterized by electromagnetic fluctuations with wavelengths as small as the ion gyroradius. On the other hand, the frequency of these fluctuations is much smaller than the ion gyrofrequency, making the timescales of gyromotion and turbulence so disparate that both can be treated independently. It is this scale separation that gyrokinetics exploits by "averaging out" the gyromotion, while keeping the finite gyroradius effects. However, as formulated, the traditional gyrokinetic model does not contain enough physics to provide the self-consistent long wavelength axisymmetric radial electric field. Since the radial electric field affects turbulent transport, we need to extend the gyrokinetic formalism to calculate it.

1.2 Gyrokinetics: history and current challenges

The gyrokinetic model is a more suitable way of writing the Fokker-Planck equation for low frequencies and short perpendicular wavelengths. The idea of gyrokinetics is to define new variables to replace the position \mathbf{r} and velocity \mathbf{v} of the particle. The gyrokinetic variables are constructed such that the the gyromotion is decoupled from the slowly varying electromagnetic fluctuations. This approach is especially convenient in turbulence simulation because retaining the gyromotion is unnecessary. The gyrokinetic variables are the appropriate variables to solve the problem since they retain wavelengths on the order of the ion gyroradius and ignore the high frequencies.

Gyrokinetics had its roots in reduced kinetic techniques used to analyze stability problems with finite gyroradius effects. The early works of Rutherford and Frieman [16] and Taylor and Hastie [17] treated small perpendicular wavelengths in stability calculations for general magnetic field geometries by using an eikonal approximation. Years later, Catto [18, 19] formulated the gyrokinetic approach by introducing the gyrokinetic change of variables.

The gyrokinetic formulation eventually evolved to a nonlinear model. Frieman and Chen [20] developed a nonlinear theory for perturbations of small amplitude over the distribution function in general magnetic field geometry. Their work was extended later for a full distribution function in a slab geometry by Dubin *et al* [21]. Hahm *et al* extended the work of Dubin *et al* to electromagnetic perturbations [22], and toroidal geometry [23]. These nonlinear gyrokinetic equations were found employing a Hamiltonian formulation and Lie transforms [24, 25].

Based on these seminal nonlinear models, Lee developed the first gyrokinetic code for investigation of the drift wave turbulence [26, 27]. This first approach was a primitive Particle-In-Cell (PIC) δf model. The δf models avoid solving for the full distribution function, which would require much computational time. Instead, the distribution function is assumed to be Maxwellian to lowest order, and a nonlinear equation for δf retaining small fluctuations is solved. The assumption is that the time it takes the turbulence to saturate is much shorter than the diffusion time. Then, the density and temperature profiles are given as an input and do not change in time. In these codes, the turbulent fluctuations evolve and saturate, and from their saturated value we can calculate the radial particle and heat fluxes. The modern, less noisy δf models originate in the ideas put forth by Kotschenreuther in [28].

Several δf codes, both continuum, like GS2 [1], GENE [2] and GYRO [3], and PIC, like GEM [4], PG3EQ [5] and GTC [6], have been developed and benchmarked.

It is based on these codes that most of the recent advances in tokamak turbulence theory have occurred.

As it was already pointed out, the δf codes are only useful to compute the particle and heat fluxes once profiles for density and temperature are given. It is necessary to develop a new generation of models capable of self-consistently calculating and evolving those profiles. It is not obvious that it can be done with the current gyrokinetic formalism. For the δf models it was enough to run the codes until the turbulence had saturated, but in order to let the profiles relax to their equilibrium, runs on the order of the transport time scale are needed. This extension is both a costly numerical task and an unsolved physical problem. Models that reach transport time scales need to take into account phenomena that were negligible when looking for the turbulence saturation. In gyrokinetics, corrections to the velocity of the particles small in a ion gyroradius over scale length are neglected. However, as run times become longer, these terms must be retained since a small velocity correction gives a considerable contribution to the total particle motion.

In recent years, several groups have begun to build codes that evolve the full distribution function, without splitting it into a slowly varying Maxwellian and a fast, fluctuating piece. These simulations, known as full f models, are employing the traditional gyrokinetic formulation. In this thesis, I will argue that this approach is inadequate since it is unable to solve for the self-consistent radial electric field that is crucial for the turbulence.

Before getting into details, I will briefly review the four main efforts in this field: GYSELA [29], ELMFIRE [30], XGC [31] and TEMPEST [32]. All these models are electrostatic. ELMFIRE and XGC are PIC simulations, and GYSELA and TEM-PEST are continuum codes. GYSELA and XGC calculate the full ion distribution function, but they adopt a fluid model for electrons that assumes an adiabatic response along the magnetic field lines. ELMFIRE and TEMPEST solve kinetically for both ions and electrons. Importantly, all four models find the electrostatic potential from a gyrokinetic Poisson's equation [27]. This gyrokinetic Poisson's equation just imposes that the ion and electron density must be equal. It looks like Poisson's equation because there is a piece of the ion density, known as polarization density, that can be written explicitly as a Laplacian of the potential. Regardless of its appearance, the gyrokinetic Poisson's equation is no more than a lower order quasineutrality condition. It is lower order because the density is calculated from the gyrokinetic equation in which higher order terms have been neglected. This is the most problematic part of these models, as I will demonstrate in this thesis. Interestingly, GYSELA, ELMFIRE and XGC have reported an extreme sensitivity to the initialization.

Operationally, the polarization density depends on the velocity space derivatives of the distribution function (the polarization density is presented in section 3.4). It is difficult to evaluate directly. In GYSELA, XGC and TEMPEST, the wavelengths are taken to be longer than the ion gyroradius and the distribution function is assumed to be close to a Maxwellian to obtain a simplified expression. In an attempt to circumvent this problem, ELMFIRE employs the gyrokinetic variables proposed by Sosenko *et al* [33]. These variables include a polarization drift that largely removes the polarization density. With the polarization velocity, it is possible to use implicit numerical schemes that give the dependence of the ion density with the electrostatic potential.

To summarize, gyrokinetic modelling has been successfully used for studying turbulence in the past decade. Codes based on δf formulations, especially the continuum ones, have provided valuable insights into tokamak anomalous transport. Currently, there is an interest in extending these simulations to transport timescales, and that requires careful evaluation of both physical and numerical issues. As a result, several groups are building and testing full f simulations. In these codes, solving for the axisymmetric radial electric field is crucial because it determines the poloidal and toroidal flows, and those flows strongly affect and, near marginality, control the turbulence level. Unfortunately, the full f community has failed to realize that a straightforward extension of the equations valid for δf codes are unable to provide the long wavelength radial electric field. The objective of this thesis is exposing this problem and proposing a solution.

1.3 Calculating the radial electric field

There are several problems that a gyrokinetic formulation has to face before it is satisfactory for long time scales. The main issue is the missing higher order terms in the gyrokinetic Fokker-Planck equation. The transport of particles, momentum and energy from one flux surface to the next is slow compared to the typical turnover time of turbulent eddies, the characteristic time scale for the traditional gyrokinetic formulation. To see this, recall that the typical structures in the turbulence are of the size of the ion gyroradius $\rho_i = Mcv_i/ZeB$, with $v_i = \sqrt{2T_i/M}$ the ion thermal velocity, Ze, M and T_i the ion charge, mass and temperature, B the magnetic field magnitude, and e and c the electron charge magnitude and the speed of light. Then, eddies are of ion gyroradius size, requiring many eddies – and hence many eddy turnover times – for a particle to diffuse out of the tokamak.

The gyrokinetic Fokker-Planck equation is derived to an order adequate for simulation of turbulence saturation, i.e., for time scales on the order of the eddy turnover time. This equation is too low of an order for transport time scales because flows and fluxes that were neglected as small now have enough time to contribute to the motion of the particles. In other words, a higher order distribution function and hence a higher order Fokker-Planck equation are required. For this reason, the extension of gyrokinetics to transport time scales must draw from the experience developed in neoclassical theory [34, 35]. Not only can neoclassical transport compete with the turbulent fluxes in some limited cases, but the tools and techniques developed in neoclassical theory become extremely useful because they require only a lower order distribution function to determine higher order radial fluxes of particles, energy and momentum. The application of neoclassical tools in gyrokinetic simulations is described in [36] and extended herein.

The physics in which the modern gyrokinetic formulation is especially flawed is the calculation of the long wavelength radial electric field. In this case, the comparison between neoclassical theory and gyrokinetics is striking. In neoclassical theory, the tokamak is intrinsically ambipolar due to its axisymmetry [37, 38], i.e., the plasma

remains quasineutral for any value of the radial electric field unless the distribution function is known to higher order than second in an expansion on the ion gyroradius over the scale length. The reason for this is that the radial electric field is related to the toroidal velocity through the $E \times B$ drift. Due to axisymmetry, the evolution of the toroidal velocity only depends on the small off-diagonal terms of the viscosity, making impossible the self-consistent calculation of the radial electric field unless the proper off-diagonal terms are included. The distribution function required to directly obtain the viscosity is higher order than second; the order at which intrinsic ambipolarity is maintained. The axisymmetric radial electric field has only been recently found in the Pfirsch-Schlüter regime [39, 40, 41, 42], and there has been some incomplete work on the banana regime for high aspect ratio tokamaks [43, 44].

In gyrokinetics, however, the electric field is found from a lower order gyrokinetic quasineutrality equation [21, 26] rather than from the transport of toroidal angular momentum. Implicitly, it is assumed that the tokamak is not intrinsically ambipolar in the presence of turbulence. In this thesis, I prove that even turbulent tokamaks are intrinsically ambipolar in the gyrokinetic ordering. Consequently, if the radial electric field is to be retrieved from a quasineutrality equation, the distribution function must be found to a hopelessly high order. The physics that determine the radial electric field, namely, the transport of angular momentum, enters the quasineutrality condition only in higher order terms, making the gyrokinetic quasineutrality equation inadequate for the calculation.

In this thesis, I pay special attention to the evolution of the long wavelength axisymmetric radial electric field in the presence of drift wave turbulence. Employing a current conservation equation or vorticity equation, I assess the feasibility of different methods to find the long wavelength axisymmetric radial electric field. Each method requires the ion Fokker-Planck equation to a different order in $\delta_i = \rho_i/L \ll 1$ and $B/B_p \gtrsim 1$, with ρ_i the ion gyroradius, L a characteristic size in the machine, typically the minor radius a, B the magnitude of the magnetic field and B_p the magnitude of its poloidal component. The different methods explored in this thesis are summarized in table 1.1. In this table, I give the chapter in which the method is presented and the

	Order of f_i	Chapter	
Gyrokinetic quasi	neutrality equation	$\delta_i^4 f_{Mi}$	3
Radial transport	Evaluated directly from f_i	$\delta_i^3 f_{Mi}$	2
of toroidal angular	Moment equation	$\delta_i^2 f_{Mi}$	5
momentum	Moment equation and $B/B_p \gg 1$	$(B/B_p)\delta_i^2 f_{Mi}$	5

Table 1.1: Comparison of different methods to obtain the long wavelength axisymmetric radial electric field.

order of magnitude to which the ion distribution function must be known compared to the zeroth order distribution function f_{Mi} . To better explain the classification by required order of magnitude of f_i , I present here a very simplified heuristic study of the first method in table 1.1. This method, known as the gyrokinetic quasineutrality equation, will be rigourously described at the end of chapter 3. In the gyrokinetic quasineutrality equation, used in modern gyrokinetics, the electric field is adjusted so that the ion and electron densities satisfy $Zen_i = en_e$, and the densities are calculated by direct integration of the distribution functions, i.e., $n_i = \int d^3v f_i$ and $n_e = \int d^3v f_e$. The ion and electron Fokker-Planck equations used to solve for the distribution functions f_i and f_e are approximate. In chapter 3, I will describe in more detail the formalism to obtain these approximate equations. For now, it is enough to consider a heuristic form of the long wavelength limit of these equations in which only the motion of the guiding center **R** is considered. Then, schematically the Fokker-Planck equations for ions and electrons are

$$\frac{\partial f_i}{\partial t} + \dot{\mathbf{R}}_i \cdot \nabla f_i + \ldots = C_{i,\text{eff}}\{f_i, f_e\}$$
(1.1)

and

$$\frac{\partial f_e}{\partial t} + \dot{\mathbf{R}}_e \cdot \nabla f_e + \ldots = C_{e,\text{eff}} \{ f_e, f_i \}, \qquad (1.2)$$

where \mathbf{R}_i and \mathbf{R}_e are the ion and electron guiding center positions, $\dot{\mathbf{R}}_i = v_{||i} \hat{\mathbf{b}} + \mathbf{v}_{di} + \dots$ and $\dot{\mathbf{R}}_e = v_{||e} \hat{\mathbf{b}} + \mathbf{v}_{de} + \dots$ are the drifts of those guiding centers, and $C_{i,\text{eff}}$ and $C_{e,\text{eff}}$ are the effective collision operators. The lower order drifts are the parallel velocities $v_{||i} \sim v_i$ and $v_{||e} \sim v_e$, and the perpendicular drifts $\mathbf{v}_{di} = \mathbf{v}_{Mi} + \mathbf{v}_E \sim \delta_i v_i$ and $\mathbf{v}_{de} = \mathbf{v}_{Me} + \mathbf{v}_E \sim \delta_e v_e \sim \delta_i v_i$, with \mathbf{v}_{Mi} and \mathbf{v}_{Me} the magnetic drifts, $\mathbf{v}_E =$ $(c/B)\mathbf{E} \times \hat{\mathbf{b}}$ the $E \times B$ drift and \mathbf{E} the electric field. The ion drifts are expanded in the small parameter $\delta_i = \rho_i/L \ll 1$, and the electron drifts are expanded in the small parameter $\delta_e = \rho_e/L \sim \delta_i \sqrt{m/M} \ll \delta_i$, with $\rho_e = mcv_e/eB$ the electron gyroradius, $v_e = \sqrt{2T_e/m}$ the electron thermal speed, and $T_e \sim T_i$ and m the electron temperature and mass. The next order corrections in δ_i to \mathbf{R}_i may be written as $\dot{\mathbf{R}}_{i} = v_{||i}\hat{\mathbf{b}} + \mathbf{v}_{di} + \dot{\mathbf{R}}_{i}^{(2)} + \dot{\mathbf{R}}_{i}^{(3)} + \dots$, with $\dot{\mathbf{R}}_{i}^{(2)} = O(\delta_{i}^{2}v_{i}), \ \dot{\mathbf{R}}_{i}^{(3)} = O(\delta_{i}^{3}v_{i})\dots$ The operators $C_{i,\text{eff}}$ and $C_{e,\text{eff}}$ are asymptotic expansions in δ_i as well. The next order corrections for $\dot{\mathbf{R}}_e$ are small in the parameter $\delta_e \ll \delta_i$. For this heuristic introduction, I will drop the next order corrections to $\dot{\mathbf{R}}_e$ exploiting the scale separation $\delta_e \ll \delta_i$ to find $\dot{\mathbf{R}}_e \simeq v_{||e} \hat{\mathbf{b}} + \mathbf{v}_{de}$. This is not rigorous, but it does not change the final result and simplifies the derivation. Under all these assumptions, the ion and electron distribution functions can be solved for perturbatively, giving $f_i = f_i^{(0)} + f_i^{(1)} + f_i^{(2)} + \dots$ and $f_e = f_e^{(0)} + f_e^{(1)} + f_e^{(2)} + \dots$, with $f_i^{(0)} = f_{Mi}, f_i^{(1)} = O(\delta_i f_{Mi}), f_i^{(2)} = O(\delta_i^2 f_{Mi}) \dots$ and $f_e^{(0)} = f_{Me}, f_e^{(1)} = O(\delta_i f_{Me}), f_e^{(2)} = O(\delta_i^2 f_{Me})...$

In this thesis I prove that to find the long wavelength radial electric field in axisymmetric configurations using the gyrokinetic quasineutrality equation, it is necessary to solve the Fokker-Planck equations to fourth order because equations (1.1) and (1.2)satisfy the condition

$$\frac{\partial}{\partial t} \langle Zen_i - en_e \rangle_{\psi} \equiv -\left\langle \nabla \cdot \left[\int d^3 v \left(Zef_i \dot{\mathbf{R}}_i - ef_e \dot{\mathbf{R}}_e \right) \right] \right\rangle_{\psi} + \left\langle \int d^3 v \left(ZeC_{i,\text{eff}} \{ f_i, f_e \} + eC_{e,\text{eff}} \{ f_e, f_i \} \right) \right\rangle_{\psi} = O(\delta_i^4 en_e v_i/L).$$
(1.3)

Here, $\langle \ldots \rangle_{\psi}$ is the flux surface average. For now, it is only important to know that $\langle \ldots \rangle_{\psi}$ makes the non-axisymmetric pieces vanish [see chapter 2]. Since the axisymmetric radial electric field adjusts so that the axisymmetric pieces of the ion and electron densities $\langle n_i \rangle_{\psi}$ and $\langle n_e \rangle_{\psi}$ satisfy quasineutrality, equation (1.3) requires that terms of order $\delta_i^4 f_{Mi} v_i / L$ be kept in equations (1.1) and (1.2) to obtain the self-

consistent radial electric field. Gyrokinetic codes solve a Fokker-Planck equation only through $O(\delta_i f_{Mi} v_i/L)$, leaving the radial electric field as a free parameter in the best case (intrinsic ambipolarity), or finding an unphysical result in the worst scenario. From an ion Fokker-Planck equation of order $\delta_i^4 f_{Mi} v_i/L$ it is possible in principle (but not in practice) to obtain a distribution function good to order $\delta_i^4 f_{Mi}$. For this reason, in table 1.1 the order to which the distribution function is required is $\delta_i^4 f_{Mi}$. Interestingly, equation (1.3) simplifies considerably because of the flux surface average, giving

$$\frac{\partial}{\partial t} \langle Zen_{i} - en_{e} \rangle_{\psi} \equiv -Ze \left\langle \nabla \cdot \left[\int d^{3}v \left(f_{i}^{(2)} \dot{\mathbf{R}}_{i}^{(2)} + f_{i}^{(1)} \dot{\mathbf{R}}_{i}^{(3)} + f_{i}^{(0)} \dot{\mathbf{R}}_{i}^{(4)} \right) \right] \right\rangle_{\psi} \\
+ \left\langle \int d^{3}v \left(ZeC_{i,\text{eff}}^{(4)} \{ f_{i}^{(1)}, f_{i}^{(2)} \} - eC_{e,\text{eff}}^{(4)} \{ f_{i}^{(1)}, f_{i}^{(2)} \} \right) \right\rangle_{\psi} = O(\delta_{i}^{4}en_{e}v_{i}/L). \quad (1.4)$$

Notice that, after flux surface averaging and integrating over velocity space, the difference of the fourth order pieces of the collision operators only depends on $f_i^{(1)}$ and $f_i^{(2)}$, and the terms

$$\langle \nabla \cdot (J_{||}^{(4)}\hat{\mathbf{b}}) \rangle_{\psi} \equiv \left\langle \nabla \cdot \left[\int d^3 v \left(Z e f_i^{(4)} v_{||i} \hat{\mathbf{b}} - e f_e^{(4)} v_{||e} \hat{\mathbf{b}} \right) \right] \right\rangle_{\psi} = 0$$
(1.5)

and

$$\langle \nabla \cdot \mathbf{J}_{d}^{(3)} \rangle_{\psi} \equiv \left\langle \nabla \cdot \left[\int d^{3}v \left(Zef_{i}^{(3)} \mathbf{v}_{di} - ef_{e}^{(3)} \mathbf{v}_{de} \right) \right] \right\rangle_{\psi} = 0$$
(1.6)

exactly vanish in axisymmetric configurations. The term (1.5) is the contribution of the parallel current density $J_{||}$ to the radial current. Since the radial current is perpendicular, this contribution is obviously zero. The term (1.6) is more subtle. It is the contribution of the current density \mathbf{J}_d due to the magnetic drifts. This contribution vanishes in axisymmetric configurations because to lowest order the net radial displacement due to magnetic drifts is zero. In chapter 2, I will show that the radial component of the current density \mathbf{J}_d is related to the parallel and perpendicular pressures $p_{||}$ and p_{\perp} in the momentum conservation equation, and these pressures finally do not enter in the calculation of the toroidal rotation, the quantity that determines the radial electric field.

Importantly, according to equation (1.4), a distribution function f_i good to $O(\delta_i^2 f_{Mi})$ is enough to calculate the radial electric field although we need the higher order corrections $\dot{\mathbf{R}}_{i}^{(2)}$, $\dot{\mathbf{R}}_{i}^{(3)}$ and $\dot{\mathbf{R}}_{i}^{(4)}$; terms never employed in gyrokinetic codes. In this thesis, I exploit the fact that only the second order correction of f_i is needed to obtain the long wavelength axisymmetric electric field. Moreover, I will not need to compute the higher order terms $\dot{\mathbf{R}}_i^{(2)}$, $\dot{\mathbf{R}}_i^{(3)}$ and $\dot{\mathbf{R}}_i^{(4)}$ explicitly. It is possible to circumvent this calculation by employing the radial transport of toroidal angular momentum, introduced in chapter 2. By doing so we gain two orders in δ_i , i.e., we make explicit that according to (1.4) we only need an ion distribution function good to order $\delta_i^2 f_{Mi}$. It is not necessary to obtain the higher order corrections to $\dot{\mathbf{R}}_i$ because the toroidal angular momentum conservation equation is obtained from a full Fokker-Planck equation where no approximation for small δ_i has been made. Importantly, the procedure used to evaluate the radial transport of toroidal angular momentum makes a considerable difference. As given in table 1.1, direct evaluation from the ion distribution function requires the ion distribution function to $O(\delta_i^3 f_{Mi})$, whereas the moment approach presented in chapter 5 only requires a distribution function good to $O(\delta_i^2 f_{Mi})$. The moment approach works because the radial transport of toroidal angular momentum depends only on the third order gyrophase dependent piece of the ion distribution function. The third order gyrophase dependent piece of the ion distribution function can be expressed as a function of the second order piece by using the full Fokker-Planck equation, and the moment approach is a simple way to write that relation.

The second order distribution function is still an order higher than usual gyrokinetic codes are built for, but there is a possible simplification listed last in table 1.1. The idea is exploiting the usually largish parameter $B/B_p \sim 10$. The new method, described at the end of chapter 5, is advantageous because conventional gyrokinetic Fokker-Planck equations can provide, with only a few modifications in the implementation, the ion distribution function to order $(B/B_p)\delta_i^2 f_{Mi}$, high enough order to self-consistently determine the long wavelength axisymmetric radial electric field for $B/B_p \gg 1$. Along with the long wavelength axisymmetric radial electric field, I investigate the evolution of the axisymmetric flows in drift wave turbulence. Under the assumptions explained at the beginning of chapter 4, the long wavelength axisymmetric flows remain neoclassical even in turbulent tokamaks. Moreover, it is possible to prove that in the modern gyrokinetic formalism the axisymmetric components of the flows and the radial electric field with radial wavelengths above $\sqrt{\rho_i L}$ are unreliable due to intrinsic ambipolarity. This result is related to the unrealistic higher order distribution functions needed to obtain the long wavelength axisymmetric radial electric field from the quasineutrality equation.

The rest of the thesis is organized as follows. In chapter 2, the calculation of the radial electric field is formulated in terms of a current conservation equation or vorticity equation. I show that, for the axisymmetric radial electric field, the vorticity equation reduces to the radial transport of toroidal angular momentum. The vorticity equation must be evaluated in the presence of turbulence, and the natural formulation for drift wave turbulence is the gyrokinetic formalism, presented in chapter 3. In chapter 4, the gyrokinetic formulation is applied to the vorticity equation. The resulting equation makes explicit the time scales involved in the evolution of the axisymmetric flows and the axisymmetric radial electric field. With this formulation, turbulent tokamaks are proven to be intrinsically ambipolar to the same order as neoclassical theory. In chapter 5, I present a different approach based on the radial transport of toroidal angular momentum that only requires an $O(\delta_i^2 f_{Mi})$ distribution function, and by exploiting an expansion in $B/B_p \gg 1$, I formulate a relatively simple model capable of self-consistently evolving the axisymmetric radial electric field in the core of a tokamak. By employing an example with simplified geometry in chapters 4 and 5, I illustrate the problems that arise from the use of the gyrokinetic quasineutrality equation, and how a new approach can solve them. Finally, in chapter 6, I summarize the findings in this thesis, and I describe a program to gradually implement a new gyrokinetic formulation in current codes that can solve for the axisymmetric radial electric field.

Chapter 2

Vorticity and intrinsic ambipolarity

In this chapter, I study the quasineutrality equation in an axisymmetric configuration. The time derivative of quasineutrality, also known as vorticity equation, makes the time scales in the problem explicit. With this equation, it is possible to show that the radial current is zero to a very high order independently of the axisymmetric, long wavelength radial electric field, i.e., both ions and electrons drift radially in an intrinsically ambipolar manner even in the presence of turbulence. Moreover, if the radial current is calculated to high enough order, I can also show that forcing it to vanish is equivalent to solving the toroidal angular momentum conservation equation.

The chapter is organized as follows. In section 2.1, I explain and justify the assumptions necessary to simplify the problem. In section 2.2, I derive a vorticity equation from the full Fokker-Planck equation. This equation cannot be usefully implemented as it is written in this chapter because some of the terms are difficult to evaluate. This issue will be addressed in chapter 4. However, this vorticity equation is useful because it makes the study of the evolution of the radial electric field easier. In section 2.3, I flux surface average the vorticity equation to determine the radial electric field. The radial electric field adjusts so that the total radial current in the tokamak vanishes, and the flux surface averaged vorticity equation is equivalent to imposing that the total radial current is zero. I show here that the flux surface averaged

vorticity equation is the conservation equation for toroidal angular momentum. Then, by estimating the size of the term that contains the transport of angular momentum in the vorticity equation, I can argue that the radial current is zero to a very high order and hence the plasma is intrinsically ambipolar to the same order as neoclassical theory.

2.1 Orderings and assumptions

To simplify the calculations, and for the rest of this thesis, I assume that the electric field is electrostatic, i.e., $\mathbf{E} = -\nabla \phi$, with ϕ the electrostatic potential. In general, the electromagnetic turbulent fluctuations are important and should be considered [45, 46]. However, keeping electromagnetic effects would obscure derivations that are already quite involved. Furthermore, in the Coulomb gauge, the axisymmetric radial electric field is purely derived from the potential. Thus, calculating the axisymmetric radial electric field is fundamentally an electrostatic problem. The electrostatic formulation presented in this thesis will offer a solution that can be extended later to electromagnetic turbulence.

To be consistent with the electrostatic electric field, the magnetic field \mathbf{B} is assumed to be constant in time. In addition, it has a characteristic length of variation much larger than the ion gyroradius. I use an axisymmetric magnetic field,

$$\mathbf{B} = I\nabla\zeta + \nabla\zeta \times \nabla\psi, \tag{2.1}$$

with ψ and ζ the magnetic flux and toroidal angle coordinates. The vector $\nabla \zeta = \hat{\boldsymbol{\zeta}}/R$ with $\hat{\boldsymbol{\zeta}}$ the unit vector in the toroidal direction and R the radial distance to the symmetry axis of the torus. I use a poloidal angle θ as the third coordinate, and employ the unit vector $\hat{\mathbf{b}} = \mathbf{B}/B$ with $B = |\mathbf{B}|$. The coordinates ψ , ζ and θ are shown in figure 2-1. The toroidal magnetic field, $B_{\zeta} = I/R$, is determined by the function I that only depends on the radial variable ψ to zeroth order.

The results in this thesis are expansions in the small parameter $\delta_i = \rho_i/L \ll 1$,



Figure 2-1: Magnetic coordinates in tokamaks. (a) Three-dimensional view of the tokamak where the magnetic field lines are the solid black lines and the axis of symmetry is the chain-dot line. (b) Poloidal plane where the flux surfaces are schematically represented as concentric circles. Here, ζ points out of the paper, ψ labels different flux surfaces, and θ defines the position within the flux surface.

with L the characteristic length in the problem. Collisions and non-neutral effects must be ordered with respect to δ_i . To do so, I will use the characteristic tokamak values given in table 2.1. In this table, the minor and major radii and the magnetic field strength are taken from [47] for Alcator C-Mod and [48] for DIII-D. The electron density n_e and the electron temperature T_e are given in [47] for Alcator C-Mod, and in [49, 50] for DIII-D. The ion temperature T_i is difficult to measure, but in general can be assumed to be of the order of the electron temperature T_e . The average ion velocity V_i is taken from [51, 52] for Alcator C-Mod, and from [49, 53] for DIII-D. The average velocities in DIII-D depend strongly on the neutral beam injection, ranging from 20 km/s when the neutral beams are turned off or their net momentum input is zero, to 200 km/s in cases that the beams drive large rotation. Alcator C-Mod, on the other hand, does not have neutral beam injection and tends to rotate at lower speeds. The rest of the quantities in table 2.1 are calculated from the measured values. These quantities are the electron and ion thermal velocities $v_e = \sqrt{2T_e/m}$ and $v_i = \sqrt{2T_i/M}$; the electron and ion gyrofrequencies $\Omega_e = eB/mc$ and $\Omega_i = ZeB/Mc$; the electron and ion gyroradii, $\rho_e = v_e/\Omega_e$ and $\rho_i = v_i/\Omega_i$; the plasma frequency $\omega_p = \sqrt{4\pi e^2 n_e/m}$ and Debye length $\lambda_D = v_e/\omega_p$; the electronelectron, electron-ion, ion-electron and ion-ion Coulomb collision frequencies $\nu_{ee} =$ $(4\sqrt{2\pi}/3) \times (e^4 n_e \ln \Lambda/\sqrt{m}T_e^{3/2}), \nu_{ei} = Z\nu_{ee}, \nu_{ie} = (2Z^2m/M)\nu_{ee}$ and $\nu_{ii} = (4\sqrt{\pi}/3) \times$ $(Z^3e^4n_e \ln \Lambda/\sqrt{M}T_i^{3/2})$, and the corresponding mean free paths $\lambda_{ee} = v_e/\nu_{ee}, \lambda_{ei} =$ $v_e/\nu_{ei}, \lambda_{ie} = v_i/\nu_{ie}$ and $\lambda_{ii} = v_i/\nu_{ii}$.

In table 2.1, $V_i/v_i \sim 0.3$ for shots with neutral beam injection, $V_i/v_i \sim 0.1$ in the pedestal region, and $V_i/v_i \sim 0.03$ in the core in the absence of neutral beam injection. In general, in the core, $V_i \sim V_e < v_i$ can be assumed. Moreover, in the absence of neutral beam injection, the average ion velocity is comparable to $V_i \sim \delta_i v_i$, with $\delta_i = \rho_i/L \sim 5 \times 10^{-3} \ll 1$ and L the characteristic length in the problem, in this case the minor radius a. Ordering $V_i \sim \delta_i v_i \sim \delta_e v_e \sim V_e$, known as the low flow or drift ordering, is then justified. This ordering allows the electric field to compete with the pressure gradient by making the $E \times B$ flow, $(cn_i/B)\mathbf{E} \times \hat{\mathbf{b}}$, and diamagnetic flow, $(c/ZeB)\hat{\mathbf{b}} \times \nabla p_i$, comparable. According to the drift ordering, the electric field is $\mathbf{E} = -\nabla \phi \sim T_e/eL$, giving a total electrostatic potential drop across the core of order T_e/e . The cases with neutral beam injection can be recovered by employing the drift ordering and then sub-expanding in $eL|\nabla\phi|/T_e\gg 1$, giving a velocity $V_i\sim$ $(eL|\nabla\phi|/T_e)\delta_i v_i \gg \delta_i v_i$. In the pedestal, on the other hand, the gradients are large, making $L \ll a$, and the average velocity is closer to the thermal speed, $V_i/v_i \sim 0.1$. Interestingly, it is possible to find an ordering similar to the low flow ordering because the pressure gradient and the electric field must be allowed to compete [54]. In this thesis, I focus in the core, where the drift ordering $V_i \sim \delta_i v_i \sim \delta_e v_e \sim V_e$ is valid, but it is possible that the formalism I will develop could be extended to the pedestal region.

To evaluate the importance of non-neutral effects, I need to compare the turbulence frequencies and wavelengths with the plasma frequency ω_p and the Debye length λ_D . I am interested in the long wavelength, axisymmetric radial electric field and its evolution in the presence of drift wave turbulence. In section 1.1, I explained that the drift wave turbulence spectrum extends from the minor radius to the ion gyroradius.

	Alcator C-Mod		DIII-D	
	Core	Separatrix	Core	Separatrix
B(T)	5	5	2	2
$n_e ({ m m}^{-3})$	$3 imes 10^{20}$	10^{20}	$3 imes 10^{19}$	10^{19}
$T_e \sim T_i \ ({\rm keV})$	2	0.1	2	0.1
$V_i ~({ m km/s})$	20	20	20 - 200	20
$v_e ~({\rm km/s})$	27000	5900	27000	5900
$v_i \; (\rm km/s)$	620	140	620	140
$\Omega_e ~({ m GHz})$	880	880	350	350
$\Omega_i ~({ m GHz})$	0.48	0.48	0.19	0.19
$ ho_e \ (\mu { m m})$	31	6.7	77	17
$ ho_i \; (\mu { m m})$	920	290	3300	740
$\omega_p \; ({ m GHz})$	980	560	310	180
$\lambda_D \; (\mu { m m})$	28	11	87	33
$\nu_{ee} \sim \nu_{ei} \; (\rm kHz)$	150	3600	16	400
$ u_{ii} \ (\mathrm{kHz}) $	2.4	60	0.26	6.5
ν_{ie} (kHz)	0.16	4.0	0.017	0.44
$\lambda_{ii} \sim \lambda_{ei} \sim \lambda_{ee} \ (m)$	260	2.3	2400	22
λ_{ie} (m)	3800	35	35000	320
Minor radius a (m)	0.21		0.67	
Major radius R (m)	0.67		1	66

Table 2.1: Typical numbers for Alcator C-Mod [47, 51, 52] and DIII-D [49, 48, 50, 53].

Theoretical studies [1, 55] suggest that the turbulent spectrum can also reach wavelengths of the order of the electron gyroradius. Since my research focuses on the long wavelength part of the electric field, I restrict myself to perpendicular wavelengths between the ion gyroradius and the minor radius. In this range, the Debye length is small, $\lambda_D/\rho_i \sim 0.03 \ll 1$. Moreover, the typical frequencies are of the order of the drift wave frequency $\omega_* = k_{\perp}cT_e/eBL_n \sim k_{\perp}\rho_i v_i/L \leq v_i/L \sim \delta_i\Omega_i \ll \Omega_i$, with $L_n = |\nabla \ln n_e|^{-1}$. The plasma frequency is then very high, $\omega_*/\omega_p \sim 3 \times 10^{-6}$, and the plasma may be assumed quasineutral. There are exceptions in which non-neutral effects have to be considered. I will point them out as they appear, but in general I will ignore these corrections to simplify the derivation. They are easy to implement and they could be added in the future with the electromagnetic corrections.

Collisional mean free paths range from very long in the center of the tokamak, $R/\lambda_{ii} \sim R/\lambda_{ee} \sim R/\lambda_{ei} \sim 10^{-3} \sim \delta_i$, to comparable to the major radius in the pedestal, $R/\lambda_{ii} \sim R/\lambda_{ee} \sim R/\lambda_{ei} \sim 0.01 - 0.1$. Notice that the mean free path is compared to the typical length along the magnetic field, proportional to the major radius R. I will order collisions as $\nu_{ie} \ll \nu_{ii} \sim v_i/L$ and $\nu_{ee} \sim \nu_{ei} \sim v_e/L$, and the low collisionality case can be obtained by then sub-expanding in the small parameter $qR/\lambda_{ii} \sim qR/\lambda_{ee} \sim qR/\lambda_{ei} \ll 1$, where $q \gtrsim 1$ is the safety factor. In this manner, collisions are kept in the derivation. This collisional ordering implies that particles are confined long enough to become a Maxwellian to lowest order. Then, the lowest order ion and electron distribution functions are the stationary Maxwellians f_{Mi} and f_{Me} . They are assumed to be stationary to be consistent with the drift ordering.

To summarize, electromagnetic effects and non-neutral effects are dropped. The typical frequency is assumed to be $\omega_* \leq v_i/L$. Collisions are ordered as $\nu_{ie} \ll \nu_{ii} \sim v_i/L$ and $\nu_{ee} \sim \nu_{ei} \sim v_e/L$. The ion and electron distribution functions are stationary Maxwellians, f_{Mi} and f_{Me} , to lowest order. Since L is the characteristic length in the problem, $\nabla f_i \sim f_{Mi}/L$ and $\nabla f_{Me} \sim f_{Me}/L$. The average velocities are ordered as small in δ_i , $V_i \sim \delta_i v_i \sim \delta_e v_e \sim V_e$, and consequently the electric field is $O(T_e/eL)$. In chapters 3 and 4, I will extend these assumptions to the shorter turbulent wavelengths. For this chapter, intended as an introduction to the properties of the long

wavelength axisymmetric radial electric field, it will be enough to consider only the longer wavelengths, of order L. Thus, in this chapter, the gradients are $\nabla \sim 1/L$.

Finally, in the derivation it will be useful to split the velocity of the particles into components parallel and perpendicular to the magnetic field, with $v_{||} = \mathbf{v} \cdot \hat{\mathbf{b}}$ the parallel component and $\mathbf{v}_{\perp} = \mathbf{v} - v_{||}\hat{\mathbf{b}}$ the perpendicular. The perpendicular velocity is determined by its magnitude $v_{\perp} = |\mathbf{v}_{\perp}|$ and the gyrophase φ_0 , defined such that

$$\mathbf{v}_{\perp} = v_{\perp}(\hat{\mathbf{e}}_1 \cos \varphi_0 + \hat{\mathbf{e}}_2 \sin \varphi_0), \qquad (2.2)$$

where the unit vectors $\hat{\mathbf{b}}(\mathbf{r})$, $\hat{\mathbf{e}}_1(\mathbf{r})$ and $\hat{\mathbf{e}}_2(\mathbf{r})$ are an orthonormal system such that $\hat{\mathbf{e}}_1 \times \hat{\mathbf{e}}_2 = \hat{\mathbf{b}}$. Notice that $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$ depend on the position \mathbf{r} because $\hat{\mathbf{b}}$ depends on \mathbf{r} . For a general magnetic field, $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$ may be chosen to be the normal $\hat{\mathbf{e}}_1 = \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} / |\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}|$ and binormal $\hat{\mathbf{e}}_2 = \hat{\mathbf{b}} \times \hat{\mathbf{e}}_1$ of the magnetic field line. In a tokamak, they could be defined as $\hat{\mathbf{e}}_1 = \nabla \psi / |\nabla \psi|$ and $\hat{\mathbf{e}}_2 = (\hat{\mathbf{b}} \times \nabla \psi) / |\nabla \psi|$.

The distinction between the gyrophase independent and dependent pieces of the distribution function will be important. I will denote the gyroaverage, or average over the gyrophase φ_0 , holding \mathbf{r} , $v_{||}$, v_{\perp} and t fixed, by $\overline{(\ldots)}$. It is important which variables are held fixed because when the gyrokinetic variables are defined, they will have their own distinct gyroaverage. Notice that this average is not weighted with the distribution function, i.e., $\int d^3v f \overline{Q} \neq \int d^3v f Q$, where f is the distribution function and $Q(\varphi_0)$ is some given function of the gyrophase φ_0 .

2.2 Vorticity equation

To obtain the electrostatic potential and build in the quasineutrality condition, but also make explicit the time scales that enter the problem, I work with the current conservation or vorticity equation,

$$\nabla \cdot \mathbf{J} = 0, \tag{2.3}$$

where $\mathbf{J} = Zen_i \mathbf{V}_i - en_e \mathbf{V}_e$ is the current density, and $n_i = \int d^3 v f_i$, $n_e = \int d^3 v f_e$,
$n_i \mathbf{V}_i = \int d^3 v \, \mathbf{v} f_i$ and $n_e \mathbf{V}_e = \int d^3 v \, \mathbf{v} f_e$ are the ion and electron densities, and the ion and electron average flows. The functions f_i and f_e are the ion and electron distribution functions, respectively. The parallel current $J_{||} = \mathbf{J} \cdot \hat{\mathbf{b}}$ can be obtained to the requisite order by integrating over the ion and electron distribution functions as discussed in more detail in chapter 4.

The perpendicular current $\mathbf{J}_{\perp} = \mathbf{J} - J_{||} \hat{\mathbf{b}}$ is given by the perpendicular component of the total momentum conservation equation. The total momentum conservation equation is

$$\frac{\partial}{\partial t}(n_i M \mathbf{V}_i) + \nabla \cdot \left[\int d^3 v \ (M f_i + m f_e) \mathbf{v} \mathbf{v} \right] = \frac{1}{c} \mathbf{J} \times \mathbf{B}.$$
(2.4)

I neglect the inertia of electrons because their mass is much smaller than the mass of the ions. The total stress tensor can be rewritten as

$$\int d^3 v \, (Mf_i + mf_e) \mathbf{v} \mathbf{v} = p_{\perp} (\mathbf{\vec{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}) + p_{||}\mathbf{\hat{b}}\mathbf{\hat{b}} + \mathbf{\vec{\pi}}_i, \tag{2.5}$$

where $p_{\perp} = p_{i\perp} + p_{e\perp} = \int d^3v \ (Mf_i + mf_e)v_{\perp}^2/2$ is the total perpendicular "pressure", $p_{||} = p_{i||} + p_{e||} = \int d^3v \ (Mf_i + mf_e)v_{||}^2$ is the total parallel "pressure", and

$$\vec{\pi}_{i} = M \int d^{3}v \ f_{i}(\mathbf{v}\mathbf{v} - \overline{\mathbf{v}\overline{\mathbf{v}}}) = M \int d^{3}v \ f_{i}\left[\mathbf{v}\mathbf{v} - \frac{v_{\perp}^{2}}{2}(\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) - v_{\parallel}^{2}\hat{\mathbf{b}}\hat{\mathbf{b}}\right]$$
(2.6)

is the ion "viscosity." The electron viscosity is neglected because it is m/M smaller, as I will prove in chapter 4. Here, $\vec{\mathbf{I}}$ is the unit dyad, and $\overline{\mathbf{vv}} = (v_{\perp}^2/2)(\vec{\mathbf{I}} - \hat{\mathbf{bb}}) + v_{\parallel}^2 \hat{\mathbf{bb}}$ is the gyroaverage of \mathbf{vv} holding \mathbf{r} , v_{\parallel} , v_{\perp} and t fixed. The definitions of the "pressures" p_{\perp} and p_{\parallel} and the "viscosity" $\vec{\pi}_i$ differ from the usual in that the average velocity is not subtracted. The usual perpendicular and parallel pressures are $p'_{\perp} = p_{\perp} - n_i M V_{i\perp}^2/2$ and $p'_{\parallel} = p_{\parallel} - n_i M V_{i\parallel}^2$, and the usual viscosity is $\vec{\pi}'_i = \vec{\pi}_i$ $-n_i M [\mathbf{V}_i \mathbf{V}_i - (V_{i\perp}^2/2)(\vec{\mathbf{I}} - \hat{\mathbf{bb}}) - V_{i\parallel}^2 \hat{\mathbf{bb}}]$. Notice that $\vec{\pi}_i$ contains the turbulent Reynolds stress. In the drift ordering, the pressures and viscosity used here are more convenient than the more common definitions.

Obtaining the perpendicular current \mathbf{J}_{\perp} from (2.4), substituting it into (2.3) and

employing

$$\frac{c}{B}\hat{\mathbf{b}} \times \nabla p_{\perp} = -\nabla \times \left(\frac{cp_{\perp}}{B}\hat{\mathbf{b}}\right) + \frac{cp_{\perp}}{B^2}\hat{\mathbf{b}} \times \nabla B + \frac{cp_{\perp}}{B}\nabla \times \hat{\mathbf{b}}$$
(2.7)

and

$$\nabla \times \hat{\mathbf{b}} = \hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \hat{\mathbf{b}} \times \boldsymbol{\kappa}, \qquad (2.8)$$

with $\boldsymbol{\kappa} = \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}$ the curvature of the magnetic field lines, gives the vorticity equation

$$\frac{\partial \boldsymbol{\varpi}}{\partial t} = \nabla \cdot \left[J_{||} \hat{\mathbf{b}} + \mathbf{J}_d + \frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \, \vec{\boldsymbol{\pi}}_i) \right], \tag{2.9}$$

where

$$\mathbf{J}_{d} = \frac{cp_{\perp}}{B}\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{cp_{\perp}}{B^{2}}\hat{\mathbf{b}} \times \nabla B + \frac{cp_{\parallel}}{B}\hat{\mathbf{b}} \times \boldsymbol{\kappa}$$
(2.10)

is the current due to the magnetic drifts and

$$\boldsymbol{\varpi} = \nabla \cdot \left(\frac{Zen_i}{\Omega_i} \mathbf{V}_i \times \hat{\mathbf{b}} \right)$$
(2.11)

is the "vorticity", with $\Omega_i = ZeB/Mc$ the ion gyrofrequency. The quantity ϖ has dimensions of charge density, but it is traditionally called vorticity because for constant magnetic fields it is proportional to the parallel component of the curl of the ion flow, $\varpi \propto \hat{\mathbf{b}} \cdot \nabla \times (n_i \mathbf{V}_i)$.

Vorticity equations like equation (2.9) have been used in the past to solve for the electrostatic potential in fluid models for turbulence [56, 57]. The vorticity ϖ is evolved in time, and the potential can be obtained from ϖ by employing the lowest order result

$$\varpi \simeq \nabla \cdot \left[\frac{Zecn_i}{B\Omega_i} \nabla_\perp \phi + \frac{c}{B\Omega_i} (\nabla \cdot \stackrel{\leftrightarrow}{\mathbf{P}}_i)_\perp \right].$$
(2.12)

To find this equation, I substitute in (2.11) the lowest order perpendicular flow

$$n_i \mathbf{V}_{i\perp} \simeq \frac{cn_i}{B} \hat{\mathbf{b}} \times \nabla \phi + \frac{c}{ZeB} \hat{\mathbf{b}} \times (\nabla \cdot \overleftrightarrow{\mathbf{P}}_i), \qquad (2.13)$$

where $\dot{\mathbf{P}}_i = \int d^3 v \, M \mathbf{v} \mathbf{v} f_i$. The lowest order perpendicular ion flow (2.13) is obtained

from the conservation of ion momentum in the same way that the perpendicular current density \mathbf{J}_{\perp} was found from the total momentum equation (2.4). In the lowest order result (2.13), I have neglected the ion inertia, $\partial(n_i M \mathbf{V}_i)/\partial t \sim \delta_i p_i/L$, and the ion-electron friction force, $\mathbf{F}_{ei} \sim \sqrt{m/M} \delta_i p_i/L$.

2.3 Radial electric field in the vorticity equation

In chapter 4, I will show that in general $\nabla \cdot \mathbf{J}_d$ dominates or at least is comparable to the other terms in (2.9). However, in axisymmetric configurations, the physics that determines the radial electric field is an exception in that $\nabla \cdot \mathbf{J}_d$ no longer dominates and only the viscosity term $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_i)$ matters.

The radial electric field adjusts so that the radial current and hence the flux surface average of the vorticity equation vanish. To see that the flux surface average of the vorticity equation is equivalent to forcing the radial current to vanish, recall that the vorticity is $\nabla \cdot \mathbf{J} = 0$, and the flux surface average of this equation gives $\langle \mathbf{J} \cdot \nabla \psi \rangle_{\psi} = 0$, i.e., the total radial current out of a flux surface must be zero. The flux surface average of equation (2.9) is

$$\frac{\partial}{\partial t} \langle \boldsymbol{\varpi} \rangle_{\psi} = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \mathbf{J}_{d} \cdot \nabla \psi - \frac{cI}{B} (\nabla \cdot \vec{\boldsymbol{\pi}}_{i}) \cdot \hat{\mathbf{b}} \right\rangle_{\psi} + \frac{1}{V'} \frac{\partial^{2}}{\partial \psi^{2}} V' \langle cR \hat{\boldsymbol{\zeta}} \cdot \vec{\boldsymbol{\pi}}_{i} \cdot \nabla \psi \rangle_{\psi}, \quad (2.14)$$

where $\langle \ldots \rangle_{\psi} = (V')^{-1} \int d\theta \, d\zeta(\ldots) / (\mathbf{B} \cdot \nabla \theta)$ is the flux surface average and $V' \equiv dV/d\psi = \int d\theta \, d\zeta (\mathbf{B} \cdot \nabla \theta)^{-1}$ is the flux surface volume element. To simplify, I have used

$$\langle \nabla \cdot (\ldots) \rangle_{\psi} = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \langle \nabla \psi \cdot (\ldots) \rangle_{\psi},$$
 (2.15)

$$\hat{\mathbf{b}} \times \nabla \psi = I\hat{\mathbf{b}} - RB\hat{\boldsymbol{\zeta}}$$
(2.16)

and $R(\nabla \cdot \vec{\pi}_i) \cdot \hat{\boldsymbol{\zeta}} = \nabla \cdot (R \vec{\pi}_i \cdot \hat{\boldsymbol{\zeta}})$. Equations (2.15) and (2.16) are obtained from the definition of flux surface average and equation (2.1), respectively. The flux surface average of $\mathbf{J}_d \cdot \nabla \psi$ is conveniently rewritten using (2.8), $(\nabla \times \hat{\mathbf{b}}) \cdot \nabla \psi = \nabla \cdot (\hat{\mathbf{b}} \times \nabla \psi)$

and (2.16) to find

$$\mathbf{J}_{d} \cdot \nabla \psi = \frac{cp_{||}}{B} \nabla \cdot (I\hat{\mathbf{b}}) - \frac{cIp_{\perp}}{B^{2}} \hat{\mathbf{b}} \cdot \nabla B, \qquad (2.17)$$

where I use that $\nabla \cdot (RB\hat{\boldsymbol{\zeta}}) = 0 = \hat{\boldsymbol{\zeta}} \cdot \nabla B$ due to axisymmetry. The flux surface average of this expression is

$$\left\langle \mathbf{J}_{d} \cdot \nabla \psi \right\rangle_{\psi} = -\left\langle \frac{cI}{B} [\hat{\mathbf{b}} \cdot \nabla p_{||} + (p_{||} - p_{\perp}) \nabla \cdot \hat{\mathbf{b}}] \right\rangle_{\psi}, \qquad (2.18)$$

where I have integrated by parts and used $\hat{\mathbf{b}} \cdot \nabla \ln B = -\nabla \cdot \hat{\mathbf{b}}$. Substituting this result into equation (2.14), using the parallel component of (2.4) to write

$$\hat{\mathbf{b}} \cdot \nabla p_{||} + (p_{||} - p_{\perp}) \nabla \cdot \hat{\mathbf{b}} + (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_i) \cdot \hat{\mathbf{b}} = -\frac{\partial}{\partial t} M n_i \mathbf{V}_i \cdot \hat{\mathbf{b}}$$
(2.19)

and employing

$$\langle \boldsymbol{\varpi} \rangle_{\psi} - \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{Zen_i I}{\Omega_i} \mathbf{V}_i \cdot \hat{\mathbf{b}} \right\rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \langle cn_i R M \mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi}$$
(2.20)

gives the conservation equation for toroidal angular momentum

$$\frac{\partial}{\partial t} \langle n_i R M \mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \langle R \hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_i \cdot \nabla \psi \rangle_{\psi}, \qquad (2.21)$$

where I have integrated once in ψ assuming that there are no sources or sinks of momentum. Equation (2.21) shows that setting the total radial current to zero is equivalent to the toroidal angular momentum conservation equation. Equation (2.21) includes both turbulent and neoclassical effects. In a model in which the transport time scale is not reached, as is the usual case in gyrokinetics, there is not enough time for the angular momentum to diffuse from one flux surface to the next, keeping then the long wavelength toroidal velocity constant and equal to its initial value. Consequently, the long wavelength radial electric field, related to the toroidal velocity by the $E \times B$ velocity, must not evolve and must be determined by the initial condition. The vorticity equation makes this fact explicit by including the radial current density $(c/B)[\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)] \cdot \nabla \psi$. Equation (2.21) can be generalized by flux surface averaging the toroidal angular momentum conservation equation with the charge density and full current density retained. Writing the electromagnetic force as the Maxwell stress gives

$$\frac{\partial}{\partial t} \langle n_i R M \mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle R \hat{\boldsymbol{\zeta}} \cdot (\vec{\boldsymbol{\pi}}_i + \vec{\boldsymbol{\pi}}_e) \cdot \nabla \psi - \frac{1}{4\pi} R \hat{\boldsymbol{\zeta}} \cdot (\mathbf{EE} + \mathbf{BB}) \cdot \nabla \psi \right\rangle_{\psi}.$$
(2.22)

In equation (2.21), the components of the toroidal angular momentum transport due to the electron viscosity and the Maxwell stress are neglected. In subsequent chapters I will argue that the ion viscosity term $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi}$ is of order $\delta_i^3 p_i R |\nabla \psi|$. The other terms must be compared to this order of magnitude estimate to asses their importance. In chapter 4, I will show that $\dot{\pi}_e \sim \delta_e^2 p_e \sim (m/M) \delta_i^2 p_i$, negligible compared to $\dot{\pi}_i$ because $m/M = 5 \times 10^{-4} \ll \delta_i$. The magnetic portion of the Maxwell stress vanishes because the calculation is electrostatic and the magnetic field is unperturbed and given by (2.1); in particular, it satisfies $\mathbf{B} \cdot \nabla \psi = 0$. The electric part of the Maxwell stress is a non-neutral contribution. Using $\mathbf{E} = -\nabla \phi \sim T_e/eL$, I find $R\hat{\boldsymbol{\zeta}} \cdot \mathbf{E}\mathbf{E} \cdot \nabla \psi/4\pi \sim (\lambda_D/L)^2 p_e R |\nabla \psi|$, with $\lambda_D = \sqrt{T_e/4\pi e^2 n_e}$ the Debye length. This is one of the instances when the non-neutral effects may contribute to the final result. In both Alcator C-Mod and DIII-D, the non-neutral transport of toroidal angular momentum $R\hat{\boldsymbol{\zeta}} \cdot \mathbf{EE} \cdot \nabla \psi/4\pi$ is comparable to the viscosity contribution, albeit somewhat smaller. For the rest of the thesis, I will drop the non-neutral contribution to simplify the presentation, but it is important to remark that it could be easily added if necessary.

The fact that $\langle \mathbf{J} \cdot \nabla \psi \rangle_{\psi} = 0$ is equivalent to equation (2.21) means that whether a configuration is intrinsically ambipolar or not depends then on the size of the radial current density $(c/B)[\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)] \cdot \nabla \psi$. If this current density is small, as I will prove it to be in chapter 4, the radial current is effectively zero. Then, if the plasma is quasineutral initially, it stays so independently of the radial electric field, that is, the configuration remains intrinsically ambipolar.

I will finish this chapter with some simple estimates that I will prove in chapters 4 and 5. In neoclassical calculations of the radial electric field [39, 40, 42, 43, 44], the

flux surface average $\langle R\hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_i \cdot \nabla \psi \rangle_{\psi}$ is of order $\delta_i^3 p_i R |\nabla \psi|$. The radial current density $(c/B)[\hat{\mathbf{b}} \times (\nabla \cdot \overleftrightarrow{\pi}_i)] \cdot \nabla \psi$ associated with this piece of the viscosity is tiny, of order $\delta_i^4 e n_e v_i |\nabla \psi|$. The estimate from neoclassical theory is not necessarily applicable to turbulent transport of toroidal angular momentum, but it is suggestive. Indeed, the same order of magnitude is recovered if the transport of toroidal angular momentum is at the gyroBohm level and the average velocities in the tokamak are small compared to the ion thermal velocity by δ_i , i.e., $V_i \sim \delta_i v_i$. The gyroBohm diffusion coefficient is obtained from turbulence fluctuations employing a simple random walk argument as follows. In chapter 1, I pointed out that the typical eddy size is $\Delta_{\text{eddy}} \sim \rho_i$. The eddy turnover time is determined by the average velocity $V_i \sim \delta_i v_i$, giving $\tau_{\text{eddy}} \sim \Delta_{\text{eddy}}/V_i \sim L/v_i$. With these estimates, the gyroBohm diffusion coefficient is calculated to be $D_{gB} = \Delta_{\text{eddy}}^2 / \tau_{\text{eddy}} \sim \delta_i \rho_i v_i$. Multiplying this diffusivity by the macroscopic gradient of momentum, $\nabla(n_i M V_{i\zeta}) \sim \delta_i n_e M v_i / L$, I find the same result as the neoclassical calculation, namely, the transport of momentum is $\delta_i^3 p_i$ and the associated radial current density is $\delta_i^4 e n_e v_i$. In chapter 5, I will argue in favor of this estimate more strongly. In any case, in current gyrokinetic formalisms, the flux surface averaged radial current should remain equal to zero independently of the long wavelength radial electric field.

Chapter 3

Derivation of gyrokinetics

Gyrokinetics is a kinetic formalism that is especially adapted to model drift wave turbulence. By defining more convenient variables to replace the position \mathbf{r} and velocity \mathbf{v} , it "averages out" the fast gyromotion time scales and keeps finite gyroradius effects, permitting perpendicular wavelengths on the order of the ion gyroradius. Currently, there is no other model that allows simulation of short wavelength turbulence in magnetized plasmas in reasonable computational times.

In this chapter, I derive the gyrokinetic Fokker-Planck equation and the gyrokinetic quasineutrality equation. The gyrokinetic Fokker-Planck equation models the response of the plasma to fluctuations in the electrostatic potential with wavelengths that may be as small as the ion gyroradius. The gyrokinetic quasineutrality equation, or gyrokinetic Poisson's equation, is the equation used so far to determine the self-consistent electrostatic potential. This equation is given here for completeness, but in chapter 4 I will show that it is flawed at long wavelengths.

The rest of the chapter is organized as follows. In section 3.1, I lay out the assumptions I need to derive the gyrokinetic equation. In section 3.2, I find the gyrokinetic variables with a new nonlinear approach, based on the linear derivation of [58, 59]. I have already published this derivation in [60]. In section 3.3, I give the Fokker-Planck equation in these new variables. This gyrokinetic Fokker-Planck equation does not have the fast gyromotion scale, yet it retains finite gyroradius effects, as desired. I finish by deriving the traditional gyrokinetic quasineutrality in

section 3.4. The algebraic details of the calculation are relegated to Appendices A - E.

3.1 Orderings

The assumptions are the same as in section 2.1. The characteristic frequency of the processes of interest is assumed to be the drift wave frequency $\omega \sim \omega_* = k_{\perp}cT_e/eBL_n \sim k_{\perp}\rho_i v_i/L$, with $L_n^{-1} = |\nabla \ln n_e|$. To treat arbitrary collisionality, the ion-ion collision frequency ν_{ii} is assumed to be of the order of the transit time of ions, $\nu_{ii} \sim v_i/L$. Consequently, the electron-electron and the electron-ion collision frequencies are of the order of the electron transit time, $\nu_{ee} \sim \nu_{ei} \sim v_e/L$. With this assumption, I can treat low collisionality cases by sub-expanding the final results in the parameter $\nu_{ii}L/v_i \sim \nu_{ee}L/v_e \sim \nu_{ei}L/v_e \ll 1$.

The average velocities are assumed to be in the low flow or drift ordering, where the $E \times B$ drift is of order $\delta_i v_i$. Therefore, the electric field is of order $\mathbf{E} = -\nabla \phi \sim T_e/eL$, and the total electrostatic potential drop across the characteristic length L is of order T_e/e . The spatial gradients of the distribution functions are assumed to be $\nabla f_i \sim f_{Mi}/L$ and $\nabla f_e \sim f_{Me}/L$, where f_{Mi} and f_{Me} are the zeroth order distribution functions. Since I am primarily interested in the core plasma in tokamaks, I will assume that the zeroth order distribution functions are stationary Maxwellians, with ion and electron temperatures of the same order, $T_i \sim T_e$. The Maxwellians are stationary to be consistent with the drift ordering.

To include the turbulence, I allow wavelengths perpendicular to the magnetic field that are on the order of the ion gyroradius, $k_{\perp}\rho_i \sim 1$. At the same time, due to low flow or drift ordering, I assume $\nabla \phi \sim T_e/eL$, $\nabla f_i \sim f_{Mi}/L$ and $\nabla f_e \sim f_{Me}/L$. This ordering requires that the pieces of the potential and the distribution functions with short perpendicular wavelengths ϕ_k , $f_{i,k}$ and $f_{e,k}$ be small in size, in particular

$$\frac{f_{i,k}}{f_{Mi}} \sim \frac{f_{e,k}}{f_{Me}} \sim \frac{e\phi_k}{T_e} \sim \frac{1}{k_\perp L} \lesssim 1, \qquad (3.1)$$

with $k_{\perp}\rho_i \lesssim 1$. According to (3.1), the pieces of the distribution function that have wavelengths on the order of the ion gyroradius are next order in the expansion in δ_i . Notice that $k_{\perp}f_{i,k} \sim \nabla_{\perp}f_{i,k} \sim \nabla f_{Mi} \sim f_{Mi}/L$, and since $\partial/\partial t \sim k_{\perp}\rho_i v_i/L$, $\partial f_{i,k}/\partial t \sim \delta_i f_{Mi} v_i/L$. I could allow wavelengths on the order of the electron gyroradius following a similar ordering, but I ignore these small wavelengths to simplify the presentation. Unlike the perpendicular wavelengths, the wavelengths along the magnetic field, $k_{||}^{-1}$, are taken to be on the order of the larger scale L. Moreover, except for initial transients, the variations along the magnetic field of f_i , f_e and ϕ are slow, i.e., in general $\hat{\mathbf{b}} \cdot \nabla f_i \sim \delta_i f_{Mi}/L$, $\hat{\mathbf{b}} \cdot \nabla f_e \sim \delta_i f_{Me}/L$ and $\hat{\mathbf{b}} \cdot \nabla \phi \sim \delta_i T_e/eL$.

Both the potential and the distribution function may be viewed as having a piece with slow spatial variations (representing the average value in the plasma) plus some rapid oscillations of small amplitude. The zonal flow, for example, will be included in the small piece if its characteristic wavelength is comparable to the gyroradius, but its amplitude may be larger for longer wavelengths. This ordering implies that the rapid spatial potential fluctuations seen by a particle in its gyromotion are small compared to the average energy of the particle. Then, the gyromotion remains almost circular, and the distribution function of the gyrocenters is equal, to zeroth order, to the distribution function of the particles. The difference, coming from the rapidly oscillating pieces, is small in our ordering. Notice that the δf codes [1, 2, 3, 4, 5, 6] explicitly adopt this treatment for the components of ϕ , f_i and f_e that satisfy $k_{\perp}\rho_i \sim$ 1, and, as in this thesis, they order them as $O(\delta_i)$.

3.2 Gyrokinetic variables

In this section, I derive the gyrokinetic variables for ions, the only species where the finite gyroradius effects matter, since $k_{\perp}\rho_i \sim 1$. It is possible to find a similar gyrokinetic equation for electrons, but for this thesis the drift kinetic equation is all that is required. I begin by defining the Vlasov operator for ions in the usual \mathbf{r} , \mathbf{v} variables for an electrostatic electric field as the following total derivative

$$\frac{d}{dt} = \frac{\partial}{\partial t} + \mathbf{v} \cdot \nabla + \left(-\frac{Ze}{M} \nabla \phi + \Omega_i \mathbf{v} \times \hat{\mathbf{b}} \right) \cdot \nabla_v, \qquad (3.2)$$

where $\Omega_i = ZeB/Mc$ is the gyrofrequency. The Fokker–Planck equation for ions is then simply

$$\frac{df_i}{dt} = C\{f_i\},\tag{3.3}$$

where $C\{f_i\}$ is the relevant Fokker-Planck collision operator for the ions. The ionelectron collision operator is small in $\sqrt{m/M}$, leaving only the ion-ion collision operator.

The objective of gyrokinetics is to change the Fokker-Planck equation to gyrokinetic variables, defined such that the gyromotion time scale disappears from equation (3.3). The nonlinear gyrokinetic variables to be employed are the guiding center location **R**, the kinetic energy E, the magnetic moment μ , and the gyrophase φ . These variables will be defined to higher order than is customary by employing an extension of the procedure presented in [58, 59]. The general idea is to construct the gyrokinetic variables to higher order by adding in δ_i corrections such that the total derivative of a generic gyrokinetic variable Q is gyrophase independent to the desired order, and we may safely employ

$$\frac{dQ}{dt} \simeq \left\langle \frac{dQ}{dt} \right\rangle, \tag{3.4}$$

where the gyroaverage $\langle \ldots \rangle$ is performed holding **R**, E, μ and t fixed. This gyrokinetic gyroaverage can be understood as a fast time average where $\tau = -\varphi/\Omega_i$ is the fast gyromotion time and t is the slow time of the turbulent fluctuations (ion gyromotion is such that $d\varphi/dt < 0$, hence the sign difference between τ and φ). The gyrokinetic variable Q is expanded in powers of δ_i ,

$$Q = Q_0 + Q_1 + Q_2 + \dots, (3.5)$$

where Q_0 is the lowest order gyrokinetic variable (kinetic energy, magnetic moment, etc.), and $Q_1 = O(\delta_i Q_0)$, $Q_2 = O(\delta_i^2 Q_0)$... are the order δ_i , δ_i^2 ... corrections. The first correction Q_1 is constructed so that $dQ/dt = \langle dQ/dt \rangle + O(\delta_i^2 \Omega_i Q)$, while the second correction Q_2 is evaluated such that $dQ/dt = \langle dQ/dt \rangle + O(\delta_i^3 \Omega_i Q)$. In principle this process can be continued indefinitely. Any Q_k can be found once the functions Q_1 , Q_2 ..., Q_{k-1} are known. The functions $Q_1, Q_2 \dots, Q_{k-1}$ are constructed so that

$$\frac{dQ}{dt} \simeq \frac{d}{dt}(Q_0 + \ldots + Q_{k-1}) = \left\langle \frac{d}{dt}(Q_0 + \ldots + Q_{k-1}) \right\rangle + O(\delta_i^k \Omega_i Q_0).$$
(3.6)

Adding Q_k to (3.5) means adding dQ_k/dt to (3.6). To lowest order, $dQ_k/dt \simeq -\Omega_i \partial Q_k/\partial \varphi$, which to the requisite order leads to an equation for Q_k ,

$$\frac{dQ}{dt} \simeq \frac{d}{dt}(Q_0 + \ldots + Q_{k-1}) - \Omega_i \frac{\partial Q_k}{\partial \varphi} = \left\langle \frac{d}{dt}(Q_0 + \ldots + Q_{k-1}) \right\rangle, \quad (3.7)$$

where $\langle \partial Q_k / \partial \varphi \rangle = 0$ is employed. Notice that the gyrophase derivative is holding the gyrokinetic variables fixed, and not \mathbf{r} , $v_{||}$ and v_{\perp} fixed (in some cases these two distinct derivatives with respect to gyrophase are almost equal). Using (3.7), $Q_k = O(\delta_i^k Q_0)$ is found to be periodic in gyrophase and given by

$$Q_k = \frac{1}{\Omega_i} \int^{\varphi} d\varphi' \left[\frac{d}{dt} (Q_0 + \ldots + Q_{k-1}) - \left\langle \frac{d}{dt} (Q_0 + \ldots + Q_{k-1}) \right\rangle \right].$$
(3.8)

More explicitly, through the first two orders, Q_1 and Q_2 are determined to be

$$Q_1 = \frac{1}{\Omega_i} \int^{\varphi} d\varphi' \left(\frac{dQ_0}{dt} - \left\langle \frac{dQ_0}{dt} \right\rangle \right)$$
(3.9)

and

$$Q_2 = \frac{1}{\Omega_i} \int^{\varphi} d\varphi' \left[\frac{d}{dt} (Q_0 + Q_1) - \left\langle \frac{d}{dt} (Q_0 + Q_1) \right\rangle \right].$$
(3.10)

By adding Q_1 and Q_2 , the total derivative of the gyrokinetic variable $Q = Q_0 + Q_1 + Q_2$ is

$$\frac{dQ}{dt} = \left\langle \frac{d}{dt} (Q_0 + Q_1) \right\rangle + O(\delta_i^3 \Omega_i Q_0).$$
(3.11)

In the remainder of this subsection, I present the gyrokinetic variables that result

from this process. I begin with the guiding center position expanded as

$$\mathbf{R} = \mathbf{R}_0 + \mathbf{R}_1 + \mathbf{R}_2, \tag{3.12}$$

where $\mathbf{R}_0 = \mathbf{r}$, $|\mathbf{R}_1| = O(\rho_i)$ and $|\mathbf{R}_2| = O(\delta_i \rho_i)$. I construct \mathbf{R}_1 and \mathbf{R}_2 such that the gyrocenter position time derivative is gyrophase independent to order $\delta_i v_i$,

$$\frac{d\mathbf{R}}{dt} = \left\langle \frac{d\mathbf{R}}{dt} \right\rangle + O(\delta_i^2 v_i). \tag{3.13}$$

The explicit details of the calculation are presented in sections A.1 and A.2 of Appendix A. To first order, I find the usual result [18]

$$\mathbf{R}_1 = \frac{1}{\Omega_i} \mathbf{v} \times \hat{\mathbf{b}}.$$
 (3.14)

The gyromotion is approximately circular, as sketched in figure 3-1(a), even in the presence of fluctuations with wavelengths of the order of the ion gyroradius. The ordering in (3.1) that bounds the electric field to be $O(T_e/eL)$ leads to this significant simplification. To next order, I obtain

$$\mathbf{R}_{2} = \frac{1}{\Omega_{i}} \left[\left(v_{||} \hat{\mathbf{b}} + \frac{1}{4} \mathbf{v}_{\perp} \right) (\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}}) \left(v_{||} \hat{\mathbf{b}} + \frac{1}{4} \mathbf{v}_{\perp} \right) \right] \stackrel{\cdot}{\times} \nabla \left(\frac{\hat{\mathbf{b}}}{\Omega_{i}} \right) \\ + \frac{v_{||}}{\Omega_{i}^{2}} \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{b}} + \frac{\hat{\mathbf{b}}}{\Omega_{i}^{2}} \left\{ v_{||} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} + \frac{1}{8} [\mathbf{v}_{\perp} \mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}})] : \nabla \hat{\mathbf{b}} \right\} \\ - \frac{c}{B\Omega_{i}} \nabla_{\mathbf{R}} \tilde{\Phi} \times \hat{\mathbf{b}}, \quad (3.15)$$

which is the same as [58] except for the nonlinear term given last. My vector conventions are $\mathbf{xy}: \mathbf{M} = \mathbf{y} \cdot \mathbf{M} \cdot \mathbf{x}$ and $\mathbf{xy} \cdot \mathbf{M} = \mathbf{x} \times (\mathbf{y} \cdot \mathbf{M})$. One of the terms in the second order correction \mathbf{R}_2 includes the function $\tilde{\Phi}(\mathbf{R}, E, \mu, \varphi, t)$. It will appear several times during derivations, as will the functions $\langle \phi \rangle$ and $\tilde{\phi}$. These are functions related to the electrostatic potential that depend on the new gyrokinetic variables. Their definitions are

$$\langle \phi \rangle \equiv \langle \phi \rangle (\mathbf{R}, E, \mu, t) = \frac{1}{2\pi} \oint d\varphi \, \phi(\mathbf{r}(\mathbf{R}, E, \mu, \varphi, t), t),$$
 (3.16)



Figure 3-1: (a) Gyrokinetic gyromotion $\mathbf{r} = \mathbf{R} - \mathbf{R}_1 - \mathbf{R}_2$ (dashed line) in a turbulent potential. The solid circle $\mathbf{R} - \mathbf{R}_1$ describes the gyromotion to first order. (b) Potential ϕ along gyromotion as a function of φ . The average $\langle \phi \rangle$ is the dashed line, and the fast time variation $\tilde{\phi}$ is the difference $\phi - \langle \phi \rangle$.

$$\widetilde{\phi} \equiv \widetilde{\phi}(\mathbf{R}, E, \mu, \varphi, t) = \phi(\mathbf{r}(\mathbf{R}, E, \mu, \varphi, t), t) - \langle \phi \rangle(\mathbf{R}, E, \mu, t)$$
(3.17)

and

$$\widetilde{\Phi} \equiv \widetilde{\Phi}(\mathbf{R}, E, \mu, \varphi, t) = \int^{\varphi} d\varphi' \, \widetilde{\phi}(\mathbf{R}, E, \mu, \varphi', t), \qquad (3.18)$$

such that $\langle \tilde{\Phi} \rangle = 0$. These are definitions similar to those used by Dubin [21]. I have discussed the subtle differences between them in [61], as summarized in Appendix C. From their definitions (3.16) and (3.17), we see that the function $\langle \phi \rangle$ is the fast time average of the potential in a gyromotion, and $\tilde{\phi}$ is the difference between the potential at the position of the particle and $\langle \phi \rangle$. Both are represented in figure 3-1(b) as a function of the gyrophase φ . The function $\tilde{\Phi}$ is proportional to the fast time integral of $\tilde{\phi}$, with $\tau = -\varphi/\Omega_i$ the fast time variation. The term $-(c/B\Omega_i)\nabla_{\mathbf{R}}\tilde{\Phi} \times \hat{\mathbf{b}}$ in (3.15) is the correction to the gyromotion due to the fast time component of the $E \times B$ drift $-(c/B)\nabla_{\mathbf{R}}\tilde{\phi} \times \hat{\mathbf{b}}$. This fast time contribution integrated over fast times $\tau = -\varphi/\Omega_i$ gives the correction to the gyrocenter position. The other corrections in \mathbf{R}_2 are fast time contributions due to magnetic geometry.

It is important to comment on the size of functions $\langle \phi \rangle$, ϕ and Φ . Both ϕ and $\langle \phi \rangle$

are of the same order as the temperature for long wavelengths, but small for short wavelengths [recall (3.1)]. However, $\tilde{\phi}$ is always small as it accounts for the variation in the electrostatic potential that a particle sees as it moves in its gyromotion. Of course, since the potential is small for short wavelengths, the variation observed by the particle is also small. For long wavelengths, even though the potential is comparable to the temperature, the particle motion is small compared to the wavelength, and the variations that it sees in its motion are small. Therefore, $\tilde{\phi} \sim \delta_i T_e/e$ for all wavelengths in my ordering, making $\tilde{\Phi}$ small as well.

The Vlasov operator acting on \mathbf{R} gives

$$\frac{d\mathbf{R}}{dt} = \left\langle \frac{d\mathbf{R}}{dt} \right\rangle + O(\delta_i^2 v_i) = u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d + O(\delta_i^2 v_i), \tag{3.19}$$

where \mathbf{v}_d is the total drift velocity,

$$\mathbf{v}_d = \mathbf{v}_E + \mathbf{v}_M,\tag{3.20}$$

composed of the $E\times B$ drift

$$\mathbf{v}_E = -\frac{c}{B(\mathbf{R})} \nabla_{\mathbf{R}} \langle \phi \rangle \times \hat{\mathbf{b}}(\mathbf{R})$$
(3.21)

and the magnetic drift

$$\mathbf{v}_M = \frac{u^2}{\Omega_i(\mathbf{R})} \hat{\mathbf{b}}(\mathbf{R}) \times \boldsymbol{\kappa}(\mathbf{R}) + \frac{\mu}{\Omega_i(\mathbf{R})} \hat{\mathbf{b}}(\mathbf{R}) \times \nabla_{\mathbf{R}} B(\mathbf{R}).$$
(3.22)

In the preceding equations, u is the gyrocenter parallel velocity defined by

$$\frac{u^2}{2} + \mu B(\mathbf{R}) = E.$$
(3.23)

Notice that in (3.19), (3.21), (3.22) and (3.23), all the terms are given as a function of the new gyrokinetic variables, \mathbf{R} , E and μ .

Following the same procedure as for the gyrocenter position \mathbf{R} , the gyrokinetic



Figure 3-2: Kinetic energy $E_0 = v^2/2$ along the gyromotion. The dashed line is the average value $E = \langle E_0 \rangle$, and the difference $E - E_0$ is the correction E_1 . Notice that the variation of the kinetic energy and the variation of the potential have opposite signs because maxima of potential correspond to minima of kinetic energy.

kinetic energy is defined as

$$E = E_0 + E_1 + E_2, (3.24)$$

where $E_0 = v^2/2$, $E_1 = O(\delta_i v_i^2)$ and $E_2 = O(\delta_i^2 v_i^2)$. The details of the calculation are given in sections A.1 and A.2 of Appendix A. I find

$$E_1 = \frac{Ze\tilde{\phi}}{M} \tag{3.25}$$

and

$$E_2 = \frac{c}{B} \frac{\partial \tilde{\Phi}}{\partial t}.$$
(3.26)

The Vlasov operator acting on E is shown in section A.2 of Appendix A to give

$$\frac{dE}{dt} = \left\langle \frac{dE}{dt} \right\rangle + O\left(\delta_i^2 \frac{v_i^3}{L}\right) = -\frac{Ze}{M} \left[u \hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M \right] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + O\left(\delta_i^2 \frac{v_i^3}{L}\right).$$
(3.27)

The corrections E_1 and E_2 are necessary because otherwise the kinetic energy would vary in the fast gyromotion time scale. Figure 3-1 shows how the potential changes rapidly along the gyromotion due to the short wavelength turbulent structures. The kinetic energy variations are accordingly rapid, as sketched in figure 3-2. The gyrokinetic energy is the average kinetic energy $E = \langle E_0 \rangle$, where the fast time variation has been extracted.

The gyrokinetic gyrophase is obtained in a similar way as the energy and the gyrocenter position by defining

$$\varphi = \varphi_0 + \varphi_1, \tag{3.28}$$

with φ_0 the original gyrophase. The details are again in Appendix A, in section A.1. Notice that only the first order correction φ_1 is calculated since gyrokinetics will make the gyrophase dependence weak and hence next order corrections unnecessary. The first order correction is

$$\varphi_{1} = -\frac{Ze}{MB}\frac{\partial\tilde{\Phi}}{\partial\mu} - \frac{1}{\Omega_{i}}\mathbf{v}_{\perp} \cdot \left[\nabla\ln B + \frac{v_{||}^{2}}{v_{\perp}^{2}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}} - \hat{\mathbf{b}}\times\nabla\hat{\mathbf{e}}_{2}\cdot\hat{\mathbf{e}}_{1}\right] \\ -\frac{v_{||}}{4\Omega_{i}v_{\perp}^{2}}[\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})]:\nabla\hat{\mathbf{b}}, \qquad (3.29)$$

where $\tilde{\Phi}$ is defined in (3.18). With this correction, $d\varphi/dt$ is gyrophase independent to order $O(v_i/L)$, that is,

$$\frac{d\varphi}{dt} = \left\langle \frac{d\varphi}{dt} \right\rangle + O(\delta_i^2 \Omega_i) = -\overline{\Omega}_i + O(\delta_i^2 \Omega_i), \tag{3.30}$$

where

$$\overline{\Omega}_{i} = \Omega_{i}(\mathbf{R}) + \frac{v_{||}}{2}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} - v_{||}\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{e}}_{2} \cdot \hat{\mathbf{e}}_{1} + \frac{Z^{2}e^{2}}{M^{2}c}\frac{\partial\langle\phi\rangle}{\partial\mu}.$$
(3.31)

The function $\overline{\Omega}_i$ is equal to Ω_i to lowest order. It might be surprising that the gyrophase φ_0 needs to be corrected by φ_1 . The gyrokinetic gyrophase φ is in reality the fast time variation $\tau = -\varphi/\overline{\Omega}_i$. The correction φ_1 is necessary because $d\varphi_0/dt$ changes in the fast gyromotion time scale, making the dependence of φ_0 on τ non-trivial. There are two effects that contribute to the fast variation of $d\varphi_0/dt$, namely, the electromagnetic fields and geometric corrections. The electromagnetic fields contribute through the magnetic field strength $B(\mathbf{r})$ in $\Omega_i(\mathbf{r})$, giving the correction $-\Omega_i^{-1}\mathbf{v}_{\perp}\cdot\nabla \ln B$ in (3.29), and through the short wavelength turbulent structures



Figure 3-3: Geometric effects on the gyrokinetic gyrophase. The circle represents the gyromotion of a given particle. The variations of $\hat{\mathbf{e}}_1$, $\hat{\mathbf{e}}_2$ and $\hat{\mathbf{b}}$ are exaggerated.

of the potential. The particle feels a rapidly changing potential along its gyromotion. The perpendicular velocity then has a variation similar to the kinetic energy [recall figure 3-2], and the gyromotion accelerates and decelerates accordingly, giving the correction $-(Ze/MB)(\partial \tilde{\Phi}/\partial \mu)$ in (3.29). The other contributions to the fast variation of $d\varphi_0/dt$ are geometrical. The gyrophase φ_0 is defined with respect to the vectors $\hat{\mathbf{e}}_1(\mathbf{r})$, $\hat{\mathbf{e}}_2(\mathbf{r})$ and $\hat{\mathbf{b}}(\mathbf{r})$ at the position \mathbf{r} of the particle, not the gyrocenter position \mathbf{R} . Figure 3-3 shows that to find the gyrophase $\varphi_{0,A}$ that corresponds to the particle position A, the position vector $\mathbf{r} - \mathbf{R}$ must be projected onto the plane defined by the local vectors $\hat{\mathbf{e}}_{1,A}$ and $\hat{\mathbf{e}}_{2,A}$; $\varphi_{0,A}$ is the angle of the projection with respect to $\hat{\mathbf{e}}_{2,A}$ [recall from (2.2) that φ_0 is the angle of \mathbf{v}_{\perp} with respect to the vector $\hat{\mathbf{e}}_1$ and hence it is the angle of $\mathbf{r} - \mathbf{R} = -\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}}$ with respect to $\hat{\mathbf{e}}_2$]. In figure 3-3, the geometric construction to determine φ_0 at point A is sketched in red, and the corresponding construction at another point B is given in blue, making explicit the distinction between the gyrokinetic gyrophases φ_A and φ_B , and the "local" gyrophases $\varphi_{0,A}$ and $\varphi_{0,B}$. Notice that both variation in $\hat{\mathbf{b}}$, that determines the "local" plane of $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$,

and rotation of $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$ within that plane change the value of φ_0 . For this reason, both $\nabla \hat{\mathbf{b}}$ and $\nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1$ enter in φ_1 and $d\varphi/dt$.

Finally, the gyrokinetic magnetic moment variable is dealt with somewhat differently since we want to construct it to remain an adiabatic invariant order by order. The condition for the magnetic moment is not only that its derivative must be gyrophase independent, but that $\langle d\mu/dt \rangle$ must vanish order by order. In this thesis, I will define the magnetic moment μ such that $d\mu/dt$ has a gyrophase dependent component of order $\delta_i v_i^3/BL$ but satisfies

$$\left\langle \frac{d\mu}{dt} \right\rangle = O\left(\delta_i^2 \frac{v_i^3}{BL}\right) \simeq 0,$$
(3.32)

for μ to remain an adiabatic invariant. This variable μ is

$$\mu = \mu_0 + \mu_1, \tag{3.33}$$

where $\mu_0 = v_{\perp}^2/2B$ is the usual lowest order result, and $\mu_1 = O(\delta_i \mu_0)$. The correction $\mu_2 = O(\delta_i^2 \mu_0)$ is not necessary because the distribution function is assumed to be a stationary Maxwellian to zeroth order, making the dependence on μ weak. For μ to remain an adiabatic invariant, μ_1 must contain gyrophase independent contributions such that $\langle d\mu/dt \rangle = 0$ to the requisite order, given in (3.32). Solving for μ_1 as outlined in section A.1 of Appendix A gives

$$\mu_{1} = \frac{Ze\tilde{\phi}}{MB(\mathbf{R})} - \frac{1}{B}\mathbf{v}_{\perp} \cdot \mathbf{v}_{M} - \frac{v_{\parallel}}{4B\Omega_{i}} [\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}}) + (\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}] : \nabla\hat{\mathbf{b}} - \frac{v_{\parallel}v_{\perp}^{2}}{2B\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}. \quad (3.34)$$

To keep μ an adiabatic invariant, $\langle \mu_1 \rangle = -(v_{\parallel}v_{\perp}^2/2B\Omega_i)(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) \neq 0$. In subsection A.2.4 of Appendix A, $\langle \mu_1 \rangle$ is proven to make μ an adiabatic invariant to next order.

The procedure presented in this section is compared to the Lie transform techniques [25] in Appendix C. Both methods yield the same results, although the final equations look somewhat different. These apparent discrepancies are shown to be due to subtleties in the definitions of the functions $\langle \phi \rangle$, $\tilde{\phi}$ and $\tilde{\Phi}$. Finally, I also compare this derivation with drift kinetics. In particular, in Appendix E, I show that with the higher order corrections \mathbf{R}_2 and E_2 it is possible to recover the drift kinetic gyrophase dependent portion of f_i up to order $\delta_i^2 f_{Mi}$.

3.3 Gyrokinetic Fokker-Planck equation

The Fokker-Planck equation (3.3) becomes

$$\frac{\partial f_i}{\partial t} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E} \frac{\partial f_i}{\partial E} + \dot{\mu} \frac{\partial f_i}{\partial \mu} + \dot{\varphi} \frac{\partial f_i}{\partial \varphi} = C\{f_i\}$$
(3.35)

when written in gyrokinetic variables, where $\dot{Q} \equiv dQ/dt$, and Q is any of the gyrokinetic variables. The gyroaverage of this equation is

$$\frac{\partial \langle f_i \rangle}{\partial t} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} \langle f_i \rangle + \dot{E} \frac{\partial \langle f_i \rangle}{\partial E} = \langle C\{f_i\} \rangle, \qquad (3.36)$$

where $\langle f_i \rangle \equiv \langle f_i \rangle (\mathbf{R}, E, \mu, t)$ is the gyroaveraged ion distribution function. Here, I have used that E and \mathbf{R} are defined such that their time derivatives are gyrophase independent to the orders given by (3.19) and (3.27). The term $\dot{\mu}(\partial f_i/\partial \mu)$ is neglected because the magnetic moment μ is defined such that $d\mu/dt = O(\delta_i v_i^3/BL)$ and the zeroth order distribution function is assumed to be a stationary Maxwellian, making $\partial f_i/\partial \mu = O(\delta_i f_{Mi} B/v_i^2)$. Therefore, in (3.36) I have neglected pieces that are $O(f_{Mi}\delta_i^2 v_i/L)$. I have also neglected the term $\langle \dot{\varphi} \partial f_i/\partial \varphi \rangle = O(\tilde{f}_i \delta_i v_i/L)$, where $\tilde{f}_i = f_i - \langle f_i \rangle$ is the gyrophase dependent piece of the distribution function. I will prove in the next paragraph that \tilde{f}_i is $O(f_{Mi} \delta_i \nu_{ii}/\Omega_i)$, making all the neglected terms comparable to or smaller than $f_{Mi} \delta_i^2 v_i/L$, and the distribution function gyrophase independent to first order, $f_i \simeq \langle f_i \rangle$. Notice that, due to the missing pieces, I can only obtain contributions to the distribution function that are $O(\delta_i f_{Mi})$, as well as all terms with $k_{\perp}\rho_i \sim 1$.

The explicit equation for the gyrophase dependent part of the distribution function is obtained by subtracting from the full Fokker-Planck equation (3.35) its gyroaverage, giving to lowest order

$$-\Omega_i \frac{\partial \tilde{f}_i}{\partial \varphi} = C\{f_i\} - \langle C\{f_i\} \rangle.$$
(3.37)

Therefore, the collisional term is the one that sets the size of \tilde{f}_i . Since the distribution function is a Maxwellian to zeroth order, the collision operator vanishes to zeroth order, $C\{f_i\} = O(\delta_i \nu_{ii} f_{Mi})$, giving $C\{f_i\} - \langle C\{f_i\} \rangle = O(\delta_i \nu_{ii} f_{Mi})$. As a result, \tilde{f}_i is

$$\widetilde{f}_i \simeq -\frac{1}{\Omega_i} \int^{\varphi} d\varphi' \left(C\{f_i\} - \langle C\{f_i\} \rangle \right) = O\left(\frac{\delta_i \nu_{ii}}{\Omega_i} f_{Mi}\right), \qquad (3.38)$$

where $\nu_{ii}/\Omega_i \ll 1$.

Using the values of $d\mathbf{R}/dt$ from (3.19) and dE/dt from (3.27), and using $\langle f_i \rangle \simeq f_i$, the equation for f_i in gyrokinetic variables is

$$\frac{\partial f_i}{\partial t} + \left[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d\right] \cdot \left(\nabla_{\mathbf{R}} f_i - \frac{Ze}{M} \nabla_{\mathbf{R}} \langle \phi \rangle \frac{\partial f_i}{\partial E}\right) = \langle C\{f_i\}\rangle, \tag{3.39}$$

where $\langle \phi \rangle$ is defined in (3.16), and $f_i \equiv f_i(\mathbf{R}, E, \mu, t)$ is gyrophase independent.

The gyrokinetic equation can be also written in conservative form. To do so, the Jacobian of the gyrokinetic transformation is needed. Conservation of particles in phase space requires the Jacobian of the transformation, $J = \partial(\mathbf{r}, \mathbf{v})/\partial(\mathbf{R}, E, \mu, \varphi)$, to satisfy

$$\frac{\partial J}{\partial t} + \nabla_{\mathbf{R}} \cdot (\dot{\mathbf{R}}J) + \frac{\partial}{\partial E}(\dot{E}J) + \frac{\partial}{\partial \mu}(\dot{\mu}J) + \frac{\partial}{\partial \varphi}(\dot{\varphi}J) = 0.$$
(3.40)

[This is the equality $\nabla \cdot \dot{\mathbf{r}} + \nabla_v \cdot \dot{\mathbf{v}} = 0$ written in gyrokinetic variables]. Employing this property, equation (3.35) can be written in conservative form by multiplying it by J to obtain

$$\frac{\partial}{\partial t}(Jf_i) + \nabla_{\mathbf{R}} \cdot \left(\dot{\mathbf{R}}Jf_i\right) + \frac{\partial}{\partial E}\left(\dot{E}Jf_i\right) + \frac{\partial}{\partial \mu}\left(\dot{\mu}Jf_i\right) + \frac{\partial}{\partial\varphi}\left(\dot{\varphi}Jf_i\right) = JC\{f_i\}.$$
 (3.41)

The gyroaverage of this equation is

$$\frac{\partial}{\partial t}(Jf_i) + \nabla_{\mathbf{R}} \cdot (\dot{\mathbf{R}}Jf_i) + \frac{\partial}{\partial E}(\dot{E}Jf_i) = J\langle C\{f_i\}\rangle.$$
(3.42)

Here, I have taken into account that the Jacobian J is independent of φ to the order of interest, as can be seen by using (3.40). The equation for the gyrophase dependent part of the Jacobian is obtained by subtracting from (3.40) its gyroaverage. Notice that $J - \langle J \rangle$ depends on the differences $\dot{\mathbf{R}} - \langle \dot{\mathbf{R}} \rangle$, $\dot{E} - \langle \dot{E} \rangle$..., and those differences are small by definition of the gyrokinetic variables. The gyrophase-dependent part of the Jacobian is estimated to be $J - \langle J \rangle = O(\delta_i^2 B/v_i)$. Finally, I substitute $d\mathbf{R}/dt$ and dE/dt in (3.42) to get

$$\frac{\partial}{\partial t}(Jf_i) + \nabla_{\mathbf{R}} \cdot \{Jf_i[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d]\} - \frac{\partial}{\partial E} \left\{Jf_i\frac{Ze}{M}[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}}\langle\phi\rangle\right\} = J\langle C\{f_i\}\rangle.$$
(3.43)

The calculation of the Jacobian is described in section A.3 of Appendix A. The final result is

$$J = \frac{\partial(\mathbf{r}, \mathbf{v})}{\partial(\mathbf{R}, E, \mu, \varphi)} = \frac{B(\mathbf{R})}{u} + \frac{Mc}{Ze}\hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \times \hat{\mathbf{b}}(\mathbf{R}).$$
(3.44)

In section A.3, it is also proven that J satisfies the gyroaverage of (3.40).

Similar gyrokinetic equations to (3.39) and (3.43) can be found for the gyrokinetic variables **R**, u and μ , where u is defined by (3.23). Combining equations (3.19), (3.23), (3.27) and $d\mu/dt \simeq 0$ to obtain

$$\dot{u} = -\left[\hat{\mathbf{b}}(\mathbf{R}) + \frac{u}{\Omega_i}\hat{\mathbf{b}}(\mathbf{R}) \times \boldsymbol{\kappa}(\mathbf{R})\right] \cdot \nabla_{\mathbf{R}} \left[\mu B(\mathbf{R}) + \frac{Ze\langle\phi\rangle}{M}\right]$$
(3.45)

gives the gyrokinetic equation

$$\frac{\partial f_i}{\partial t} + [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} f_i + \dot{u} \frac{\partial f_i}{\partial u} = \langle C\{f_i\}\rangle.$$
(3.46)

This gyrokinetic equation can be written in conservative form by noticing that the new Jacobian is given by

$$J_{u} = \frac{\partial(\mathbf{r}, \mathbf{v})}{\partial(\mathbf{R}, u, \mu, \varphi)} = \frac{\partial(\mathbf{r}, \mathbf{v})}{\partial(\mathbf{R}, E, \mu, \varphi)} \frac{\partial E}{\partial u} = B(\mathbf{R}) + \frac{Mcu}{Ze} \hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \times \hat{\mathbf{b}}(\mathbf{R}). \quad (3.47)$$

Using the new Jacobian, the gyrokinetic equation may be written as

$$\frac{\partial}{\partial t}(J_u f_i) + \nabla_{\mathbf{R}} \cdot \{J_u f_i[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d]\} + \frac{\partial}{\partial u}(J_u f_i \dot{u}) = J_u \langle C\{f_i\}\rangle.$$
(3.48)

The ion gyrokinetic Fokker-Planck equations (3.39), (3.43), (3.46) and (3.48) have their counterpart for electrons. However, since in this thesis the wavelengths shorter than the ion gyroradius are not considered, the electron gyrokinetic equation reduces to the drift kinetic equation

$$\frac{\partial \overline{f}_{e}}{\partial t} + (v_{||}\hat{\mathbf{b}} + \mathbf{v}_{de}) \cdot \left(\overline{\nabla} \overline{f}_{e} + \frac{e}{m} \nabla \phi \frac{\partial \overline{f}_{e}}{\partial E_{0}}\right) = C\{\overline{f}_{e}\}, \tag{3.49}$$

with $\overline{f}_e = \overline{f}_e(\mathbf{r}, E_0, \mu_0, t)$ the gyrophase independent piece of the distribution function, $\overline{\nabla}$ the gradient holding E_0, μ_0, φ_0 and t fixed, and

$$\mathbf{v}_{de} = -\frac{\mu_0}{2\Omega_e} \hat{\mathbf{b}} \times \nabla B - \frac{v_{||}^2}{\Omega_e} \hat{\mathbf{b}} \times \boldsymbol{\kappa} - \frac{c}{B} \nabla \phi \times \hat{\mathbf{b}}$$
(3.50)

the electron drifts. Here, $\Omega_e = eB/mc$ is the electron gyrofrequency. The total distribution function for electrons contains the gyrophase independent piece \overline{f}_e and the gyrophase dependent piece

$$f_e - \overline{f}_e = -\frac{f_{Me}}{\Omega_e} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \left[\frac{\nabla n_e}{n_e} - \frac{e}{T_e} \nabla \phi + \left(\frac{mE_0}{T_e} - \frac{3}{2} \right) \frac{\nabla T_e}{T_e} \right].$$
(3.51)

In the rest of this thesis, the gyrokinetics variables to be used are **R**, E and μ for ions and **r**, E_0 and μ_0 for electrons. Then, the relevant equations are (3.39) for ions, and (3.49) for electrons.

3.4 Gyrokinetic quasineutrality equation

Modern gyrokinetics employs a low order quasineutrality condition to calculate the electrostatic potential. The ion and electron distribution functions f_i and f_e are calculated using the lower order equations (3.39) and (3.49). These distribution func-

tions are integrated over velocity space to obtain the densities n_i and n_e that are substituted into the quasineutrality condition $Zn_i = n_e$, from which the potential is solved. Since equations (3.39) and (3.49) give distribution functions good to order $\delta_i f_{Mi}$ and $\delta_e f_{Me}$, respectively, the quasineutrality equation is correct only to order $\delta_i n_e$. In chapter 4, I will show that this is not enough to solve for the long wavelength axisymmetric radial electric field, and employing such a low order quasineutrality equation can lead to unphysical results. In this section, I will derive the modern gyrokinetic quasineutrality equation to the order that is usually implemented, so I can demonstrate later, in chapter 4, that it cannot provide the correct long wavelength axisymmetric radial electric field.

I will begin with Poisson's equation to explicitly show that the quasineutral approximation is valid for the range of wavelengths of interest. The distribution function f_i in Poisson's equation,

$$-\nabla^2 \phi = 4\pi e \left[Z \int d^3 v f_i(\mathbf{R}, E, \mu, t) - n_e(\mathbf{r}, t) \right], \qquad (3.52)$$

is obtained from the Fokker-Planck equation (3.39). Therefore, it is known as a function of the gyrokinetic variables. The distribution function can be rewritten more conveniently as a function of $\mathbf{r} + \Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}}$, E_0 and μ_0 by Taylor expanding. However, it is important to remember that there are missing pieces of order $\delta_i^2 f_{Mi}$ in the distribution function since terms of this order must be neglected to derive (3.39). Thus, the expansion can only be carried out to the order where the distribution function is totally known, resulting in

$$f_i(\mathbf{R}, E, \mu, t) = f_i(\mathbf{R}_g, E_0, \mu_0, t) + E_1 \frac{\partial f_i}{\partial E_0} + \mu_1 \frac{\partial f_i}{\partial \mu_0} + O(\delta_i^2 f_{Mi}).$$
(3.53)

Notice that $f_i(\mathbf{R}_g, E_0, \mu_0, t)$, with $\mathbf{R}_g \equiv \mathbf{r} + \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$, cannot be Taylor expanded around \mathbf{r} because $k_{\perp}\rho_i \sim 1$. In the higher order terms proportional to E_1 and μ_1 , the function f_i is valid only to lowest order, i.e., $f_i \simeq f_{Mi}$, $\partial f_i / \partial E_0 \simeq (-M/T_i) f_{Mi}$ and $\partial f_i / \partial \mu_0 \simeq 0$. Moreover, according to the ordering in (3.1), the corrections arising from using **r** instead of \mathbf{R}_g are small by δ_i because, even though small wavelengths are allowed, the amplitude of the fluctuations with small wavelengths is assumed to be of the next order. Therefore, in the higher order integrals, only the long wavelength distribution function f_{Mi} depending on $\mathbf{R}_g \simeq \mathbf{r}$ need be retained.

Since the turbulent wavelengths are much larger than the Debye length, the term in the left side of Poisson's equation (3.52) may be neglected. The resulting quasineutrality equation reduces to

$$Zn_{ip}\{\phi\} \simeq n_e(\mathbf{r}, t) - Z\hat{N}_i(\mathbf{r}, t), \qquad (3.54)$$

with $n_e = \int d^3 v \, \overline{f}_e$ the electron density,

$$n_{ip}\{\phi\} = -\int d^3v \, \frac{Ze\tilde{\phi}}{T_i} f_{Mi} \tag{3.55}$$

the ion polarization density that depends explicitly in the potential, and

$$\hat{N}_{i}(\mathbf{r},t) = \int d^{3}v \ f_{i}(\mathbf{R}_{g}, E_{0}, \mu_{0}, t)$$
(3.56)

the ion guiding center density. In equation (3.54), terms of order $\delta_i^2 n_e$ and $n_e k_\perp \lambda_D^2/L \ll \delta_i k_\perp \rho_i n_e$ have been neglected, with $\lambda_D = \sqrt{T_e/4\pi e^2 n_e}$ the Debye length.

In the quasineutrality equation (3.54), the ion density is composed of two terms, the guiding center density \hat{N}_i and the polarization density $n_{ip}\{\phi\}$. Both terms have a clear physical meaning. In figure 3-4, the ion density calculation is sketched. In gyrokinetics, ions are substituted by rings of charge with radius the ion gyroradius. The ion density at a point is then calculated by counting the rings that pass through that point. In figure 3-4, the ion density in the black square is computed by summing all the gyrocenters whose rings of charge cross the square. In the figure, two of them, \mathbf{R}_A and \mathbf{R}_B , are shown. Importantly, the charge density along a ring is not constant, but depends on the potential. In figure 3-1 we saw that the potential changes rapidly along the gyromotion. There is a Maxwell-Boltzmann response to this variation, $-Ze\tilde{\phi}/T_e$, that gives a varying ion charge density along the ring, as indicated in



Figure 3-4: Ion density in the black square calculated using a gyrokinetic distribution function. The electrostatic potential is the background contour plot.

figure 3-4, where the left region of the ring \mathbf{R}_A has lower density than the right side. In the ring, therefore, the average charge position is displaced towards the lower potential values, i.e., the charge moves in the same direction as the local electric field, hence the name polarization. For this reason, the ion density is separated into the guiding center density \hat{N}_i , due to the average charge in the ring, and the polarization density n_{ip} that originates in the non-uniform density along the ring that is induced by the short wavelength pieces of the potential.

Equation (3.54) is used to calculate ϕ for wavelengths of the order of the gyroradius, including zonal flow, in δf turbulence codes such as GS2 [1], GENE [2] or GYRO [3]. In most cases, the distribution function is obtained from the gyrokinetic equation (3.39) written for $f_i = f_{Mi} + h_i$, with $|h_i(\mathbf{R}, E, \mu, t)| \ll f_{Mi}$ and f_{Mi} only depending on ψ . The resulting equation is

$$\frac{\partial h_i}{\partial t} + [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} h_i - \left\langle C^{(\ell)} \left\{ h_i - \frac{Ze\tilde{\phi}}{T_i} f_{Mi} \right\} \right\rangle = \frac{Ze}{T_i} f_{Mi} \left\{ i\omega_*^{n,T} \langle \phi \rangle - [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle \right\},$$
(3.57)

with $i = \sqrt{-1}$ and

$$\omega_*^{n,T} \equiv -i\frac{cT_i}{ZeB}(\hat{\mathbf{b}} \times \nabla\psi) \cdot \nabla_{\mathbf{R}} \ln\langle\phi\rangle \left[\frac{1}{n_i}\frac{dn_i}{d\psi} + \left(\frac{ME}{T_i} - \frac{3}{2}\right)\frac{1}{T_i}\frac{dT_i}{d\psi}\right]$$
(3.58)

the drift wave frequency. In equation (3.57), ϕ appears nonlinearly in $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_i$ and linearly on the right side of the equation. The linear terms are usually solved implicitly. Then, h_i has a linear dependence on ϕ that will appear as a linear dependence in \hat{N}_i , and can be used to solve for ϕ in equation (3.54). The ion polarization density (3.55) also depends linearly on ϕ . However, at long wavelengths n_{ip} becomes too small to be important. For $\delta_i \ll k_{\perp}\rho_i \ll 1$, the polarization density is

$$n_{ip} \simeq \nabla \cdot \left(\frac{cn_i}{B\Omega_i} \nabla_\perp \phi\right) = O(\delta_i k_\perp \rho_i n_e). \tag{3.59}$$

The details of this calculation are given in section D.1 of Appendix D. To estimate the size of n_{ip} , I have used (3.1) to order $\nabla_{\perp}\phi \sim T_e/eL$ and $\nabla_{\perp}\nabla_{\perp}\phi \sim k_{\perp}T_e/eL$. With this estimate, for $k_{\perp}L \sim 1$ equation (3.54) becomes

$$Z\hat{N}_i(\mathbf{r},t) = n_e(\mathbf{r},t), \qquad (3.60)$$

where terms of order $\delta_i^2 n_e$ have been neglected.

It is possible to obtain a higher order long wavelength quasineutrality equation for a non-turbulent plasma if the ion distribution function is assumed to be known to high enough order. The resulting equation is

$$\nabla \cdot \left(\frac{Zcn_i}{B\Omega_i}\nabla_{\perp}\phi\right) - \frac{ZMc^2n_i}{2T_iB^2}|\nabla_{\perp}\phi|^2 = n_e - Z\hat{N}_i,\tag{3.61}$$

where \hat{N}_i must be defined to higher order,

$$\hat{N}_{i}(\mathbf{r},t) = \int d^{3}v f_{i}(\mathbf{r},E_{0},\mu_{0},t) \left(1 + \frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}\right) + (\mathbf{\vec{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) : \frac{\nabla\nabla p_{i}}{2M\Omega_{i}^{2}}.$$
 (3.62)

The derivation of (3.61) is shown in section D.2 of Appendix D. Even though equation (3.61) is correct, it is only useful if we are able to evaluate the missing $O(\delta_i^2 f_{Mi})$ pieces

in f_i that are of the same order as the left side in (3.61). Equation (3.39) misses these pieces. Equation (3.61) will only serve to demonstrate the problems that arise from the use of equation (3.54).

In chapter 4, I will prove that neither equation (3.60) or its higher order version (3.61) are able to provide the self-consistent, long wavelength radial electric field. I will even give an example in section 4.5 in which the higher order equation (3.61) leaves the radial electric field undetermined.

Chapter 4

Gyrokinetic vorticity equation

In this chapter, I rewrite the vorticity equation (2.9) in a convenient form for gyrokinetics. The gyrokinetic change of variables found in chapter 3 is especially well suited for simulation of drift wave turbulence. However, the gyrokinetic quasineutrality equation, traditionally used to calculate the electrostatic potential, has problems. By writing the vorticity equation (2.9) in gyrokinetic form, I can study the behavior of the quasineutrality equation at different time scales and wavelengths. In particular, I am able to prove that the radial current vanishes to a very high order for any radial electric field, i.e., the radial drift of ions and electrons is intrinsically ambipolar.

In section 4.1, I explain the notation employed in this chapter and I list the assumptions. These assumptions restrict the treatment to turbulence that has reached a statistical equilibrium and, therefore, only has small variations along magnetic flux surfaces. Under the assumptions of sections 3.1 and 4.1, I evaluate the size of the terms in the vorticity equation (2.9) and in the toroidal angular momentum conservation equation (2.21). The size of the different contributions to the vorticity equation depends on the perpendicular wavelength of interest, and this dependence makes some terms important for structures of the size of the ion gyroradius and negligible at wavelengths on the order of the minor radius of the device. In section 4.2, it will become clear that direct evaluation of the terms in equation (2.9) is too difficult to be of interest. Then, in the rest of the chapter, a different approach is taken. The equations for particle and momentum conservation are derived from the gyrokinetic equation in section 4.3. These conservation equations are then combined in section 4.4 to derive two different vorticity equations equivalent to (2.9) to lowest order. Importantly, these equations are easier to study and to evaluate numerically. With them, I show that the dependence of quasineutrality on the long wavelength radial electric field is not meaningful for gyrokinetic codes to retain since gyrokinetics will be shown to be intrinsically ambipolar as already stressed in section 2.3. The problems that arise from quasineutrality are exposed in a simplified example in section 4.5. I finish this chapter with a discussion in section 4.6. All the details of the calculation are relegated to Appendices F-K.

4.1 Notation and assumptions

In this chapter, I work in both the gyrokinetic phase space $\{\mathbf{R}, E, \mu, \varphi\}$ and the "physical" phase space $\{\mathbf{r}, \mathbf{v}\}$. I refer to it as physical phase space because spatial and velocity coordinates do not get mixed as they do in gyrokinetic phase space. I use the variables \mathbf{r} , E_0 , μ_0 and φ_0 to describe this physical phase space. Whenever I write $\partial/\partial E_0$, it is implied that \mathbf{r} , μ_0 , φ_0 and t are held fixed, and similarly for $\partial/\partial\mu_0$ and $\partial/\partial\varphi_0$. The gradient holding E_0 , μ_0 , φ_0 and t fixed will be written as ∇ . In addition, any derivative with respect to a gyrokinetic variable is performed holding the other gyrokinetic variables constant. The partial derivative with respect to the time variable t deserves a special mention since it is necessary to indicate which variables are kept fixed. In this formulation, the time derivative holding \mathbf{r} and \mathbf{v} fixed is equivalent to holding \mathbf{r} , E_0 , μ_0 and φ_0 fixed because the magnetic field is constant in time. Also, a gyroaverage holding \mathbf{r} , E_0 , μ_0 and t fixed is denoted as $\overline{(\ldots)}$, as opposed to the gyrokinetic gyroaverage $\langle \ldots \rangle$ performed holding \mathbf{R} , E, μ and t fixed.

All the assumptions in section 3.1 are applicable here. Then, the zeroth order ion and electron distribution functions are assumed to be stationary Maxwellians, f_{Mi} and f_{Me} . In this chapter, the only spatial dependence allowed for these zeroth order solutions is in the radial variable ψ . Therefore, $\hat{\mathbf{b}} \cdot \nabla f_{Mi} = \hat{\mathbf{b}} \cdot \nabla f_{Me} = 0$. I assume that the radial gradients of f_{Mi} and f_{Me} are O(1/L), with L of the order of the minor radius of the tokamak. The zeroth order potential ϕ works in a similar fashion, depending only on ψ and with a radial gradient on the longer scale L.

As in section 3.1, I allow wavelengths perpendicular to the magnetic field that are on the order of the ion gyroradius, $k_{\perp}\rho_i \sim 1$. The pieces of the potential and the distribution function with short perpendicular wavelengths are small in size, following the ordering in (3.1). Except for initial transients, I assume that the variation along the magnetic field of f_i , f_e and ϕ is slow, i.e., in general $\hat{\mathbf{b}} \cdot \nabla f_i \sim \delta_i f_{Mi}/L$, $\hat{\mathbf{b}} \cdot \nabla f_e \sim$ $\delta_i f_{Me}/L$ and $\hat{\mathbf{b}} \cdot \nabla \phi \sim \delta_i T_e/eL$.

The ion distribution function $f_i(\mathbf{R}, E, \mu, t)$ is found employing the gyrokinetic equation (3.39) [the gyrophase dependent piece is $O(\delta_i f_{Mi} \nu_{ii} / \Omega_i)$ and given by (3.38)]. After the initial transient, equation (3.39) becomes $u\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} f_i = \langle C\{f_i\} \rangle$ to zeroth order. At long wavelengths, this requires that f_i approach a Maxwellian f_{Mi} with $\mathbf{\hat{b}} \cdot \nabla f_{Mi} = 0$, giving $f_{Mi} \equiv f_{Mi}(\psi, E)$. The assumed long wavelength piece of the distribution function satisfies this condition. Importantly, $\hat{\mathbf{b}} \cdot \nabla f_{Mi} = 0$ does not impose any condition on the radial dependence of f_{Mi} . Consequently, the density and temperature in f_{Mi} may have short wavelength components as long as they satisfy the orderings in (3.1), i.e., $\nabla_{\perp} n_i \sim k_{\perp} n_{i,k} \sim n_{i,k=0}/L$, $\nabla_{\perp} T_i \sim k_{\perp} T_{i,k} \sim T_{i,k=0}/L$ and $\nabla_{\perp} \nabla_{\perp} f_{Mi,k} \sim k_{\perp} f_{Mi,k=0}/L$. Solving for the next order correction $f_i - f_{Mi}$ in equation (3.39) gives $f_i - f_{Mi} \sim \delta_i f_{Mi}$. Then, the average velocity $\mathbf{V}_i = n_i^{-1} \int d^3 v \mathbf{v} f_i$ is of order $\delta_i v_i$. Furthermore, any variation of the distribution function within a flux surface is due to $f_i - f_{Mi}$, and thus small by δ_i as compared to the long wavelength piece of f_{Mi} . This means that when we consider average velocities or the gradients $\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} f_i$ and $\hat{\boldsymbol{\zeta}} \cdot \nabla_{\mathbf{R}} f_i$, it will be useful to think about the distribution function as it is done in δf codes where $f_i = f_{Mi} + h_i$, with $h_i \sim \delta_i f_{Mi} \ll f_{Mi}$. Comparing the estimate for $f_i - f_{Mi}$ with the orderings in (3.1), I find that the gradients of f_i and ϕ parallel to the flux surfaces are smaller than the maximum allowed in gyrokinetics, i.e., $\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} f_i \sim \delta_i f_{Mi} / L \lesssim \hat{\boldsymbol{\zeta}} \cdot \nabla_{\mathbf{R}} f_i \sim k_{\perp} (f_i - f_{Mi}) \sim k_{\perp} \rho_i f_{Mi} / L \lesssim f_{Mi} / L$ and $\hat{\mathbf{b}} \cdot \hat{\mathbf{b}} = k_{\perp} \hat{\boldsymbol{\beta}} \cdot \hat{$ $\nabla \phi \sim \delta_i T_e/eL \lesssim \hat{\boldsymbol{\zeta}} \cdot \nabla \phi \sim k_\perp \rho_i T_e/eL \lesssim T_e/eL$. These estimates may fail for the initial transient, but I am interested in the electric field evolution at long times, when the transient has died away.

Interestingly, these assumptions imply that the long wavelength axisymmetric flows are neoclassical. At long wavelengths, the ion distribution function $f_i(\mathbf{R}, E, \mu, t)$ can be Taylor expanded around \mathbf{r} , E_0 and μ_0 . Then, the gyroaveraged piece \overline{f}_i is approximately $f_i(\mathbf{r}, E_0, \mu_0, t)$, where \mathbf{R} , E and μ have been replaced by \mathbf{r} , E_0 and μ_0 . This gyroaveraged piece of the ion distribution function \overline{f}_i satisfies the ion drift kinetic equation

$$\frac{\partial \overline{f}_i}{\partial t} + (v_{||}\hat{\mathbf{b}} + \mathbf{v}_M) \cdot \left(\overline{\nabla} \overline{f}_i - \frac{Ze}{M} \nabla \phi \frac{\partial \overline{f}_i}{\partial E_0}\right) + \mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i = C\{\overline{f}_i\}.$$
(4.1)

This equation is obtained from (3.39) by realizing that the functional dependence of f_i on the gyrokinetic variables **R**, E and μ is the same as the dependence of \overline{f}_i on **r**, E_0 and μ_0 . Then, equation (4.1) is derived from (3.39) by replacing **R**, E and μ by **r**, E_0 and μ_0 , and employing that $\langle \phi \rangle \simeq \phi$ for long wavelengths. The difference between the long wavelength ion equation (4.1) and a drift kinetic equation [see, for example, the electron equation (3.49)] is in the nonlinear term $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$. In this term, the short wavelength components of f_i and ϕ beat together to give a long wavelength contribution. Due to the presence of these short wavelength pieces, f_i cannot be Taylor expanded and $\langle \phi \rangle \neq \phi$. Importantly, this term gives a negligible contribution to the axisymmetric piece of \overline{f}_i . For $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ to have an n = 0 toroidal mode number, the beating components $f_{i,n}$ and $\nabla_{\mathbf{R}} \langle \phi \rangle_{-n}$ must have toroidal mode numbers of the same magnitude and opposite sign. Moreover, these components must have $n \neq 0$ because otherwise $\nabla_{\mathbf{R}} \langle \phi \rangle$ is parallel to $\nabla_{\mathbf{R}} f_i$ and $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ vanishes exactly. Thus, only the non-axisymmetric components of $\langle \phi \rangle$ and f_i contribute to $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ and these are of order $\delta_i T_e/e$ and $\delta_i f_{Mi}$, respectively. Writing $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i =$ $-(c/B)\mathbf{\hat{b}}\cdot\nabla_{\mathbf{R}}\times(f_i\nabla_{\mathbf{R}}\langle\phi\rangle),$ it is easy to see that $\mathbf{v}_E\cdot\nabla_{\mathbf{R}}f_i\sim(c/B)k_{\perp}f_{i,n}|\nabla_{\mathbf{R}}\langle\phi\rangle_{-n}|,$ with k_{\perp} the radial wavenumber of the axisymmetric piece of \overline{f}_i . Since $f_{i,n} \sim \delta_i f_{Mi}$ and $(c/B)|\nabla_{\mathbf{R}}\langle\phi\rangle_{-n}| \sim \delta_i v_i$, the largest possible size for $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ is $\delta_i k_\perp \rho_i f_{Mi} v_i / L$. Consequently, at long wavelengths, $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ is negligibly small compared with the other terms in (4.1), and the equation for the axisymmetric component of \overline{f}_i is the neoclassical drift kinetic equation. For this reason, the long wavelength axisymmetric

flows must be neoclassical, i.e., they are given by

$$n_i \mathbf{V}_i = -\frac{cR}{Ze} \left(\frac{\partial p_i}{\partial \psi} + Zen_i \frac{\partial \phi}{\partial \psi} \right) + U(\psi) \mathbf{B}, \tag{4.2}$$

where $U(\psi)$ is proportional to $\partial T_i/\partial \psi$ in neoclassical theory [34, 35]. For $k_{\perp}\rho_i \sim \delta_i$, the turbulent term $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ is of order $\delta_i^2 f_{Mi} v_i/L$. By comparing its size with the smaller term in the drift kinetic equation, usually the collisional operator $C\{f_i\} \simeq$ $C^{(\ell)}\{f_i - f_{Mi}\} \sim \delta_i f_{Mi} \nu_{ii}$, I obtain that the turbulent correction to the neoclassical flow $U(\psi)$ is of order $\delta_i v_i/L\nu_{ii}$.

To finish this section, I will present the different contributions to the distribution function. To order $\delta_i f_{Mi}$, the gyrokinetic distribution function can be written as

$$f_i \equiv f_i(\mathbf{R}, E, \mu, t) \simeq f_{ig} - \frac{Ze\tilde{\phi}}{T_i} f_{Mi}, \qquad (4.3)$$

where

$$f_{ig} \equiv f_i(\mathbf{R}_g, E_0, \mu_0, t) \tag{4.4}$$

and

$$\mathbf{R}_g = \mathbf{r} + \frac{1}{\Omega_i} \mathbf{v} \times \hat{\mathbf{b}}.$$
(4.5)

In equation (4.3), I have Taylor expanded $f_i(\mathbf{R}, E, \mu, t)$ around E_0 and μ_0 , and I have used the zeroth order Maxwellian f_{Mi} in the higher order terms. In the function ϕ in (4.3), it is enough to use the lowest order variables \mathbf{R}_g , μ_0 and φ_0 instead of \mathbf{R} , μ and φ (the dependence of ϕ on E is weak). The piece f_{ig} of the distribution function will be useful in section 4.3 to obtain moment equations from the gyrokinetic equation (3.39). However, the pieces of the distribution function given in (4.3) are not useful to evaluate terms in physical phase space since the variable \mathbf{R}_g still mixes spatial and velocity space variables. In physical phase space, it is useful to distinguish between the gyrophase independent piece of the distribution function \overline{f}_i and the gyrophase dependent piece. At long wavelengths, $f_{ig} \equiv f_i(\mathbf{R}_g, E_0, \mu_0, t)$ can be Taylor expanded

Term	Order of magnitude
$\overline{\omega}$	$\delta_i k_\perp ho_i e n_e$
$ abla \cdot (J_{ } \hat{\mathbf{b}} + \mathbf{J}_d)$	$\delta_i e n_e v_i / L$
$ abla \cdot [(c/B)\hat{\mathbf{b}} imes (abla \cdot \overleftarrow{\boldsymbol{\pi}}_i)]$	$\delta_i (k_\perp ho_i)^2 e n_e v_i / L$

Table 4.1: Order of magnitude estimates for vorticity equation (2.9).

around **r**, and I can employ the lowest order result $\tilde{\phi} \simeq \Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$, giving

$$\overline{f}_i = f_{i0} + O(\delta_i k_\perp \rho_i f_{Mi}) \tag{4.6}$$

and

$$f_i - \overline{f}_i = \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \left[\frac{\nabla p_i}{p_i} + \frac{Ze\nabla\phi}{T_i} + \left(\frac{Mv^2}{2T_i} - \frac{5}{2} \right) \frac{\nabla T_i}{T_i} \right] f_{Mi} + O(\delta_i k_\perp \rho_i f_{Mi}), \quad (4.7)$$

with

$$f_{i0} \equiv f_i(\mathbf{r}, E_0, \mu_0, t).$$
 (4.8)

Notice that f_{i0} differs from f_{ig} in that the gyrocenter position **R** in $f_i(\mathbf{R}, E, \mu, t)$ is replaced by the particle position **r** and not the intermediate variable \mathbf{R}_{g} .

4.2 General vorticity equation in gyrokinetics

In this section, I evaluate the size of the different terms in the vorticity equation (2.9) and the toroidal momentum equation (2.21). For these estimates I will use the assumptions in sections 3.1 and 4.1. The estimates are summarized in table 4.1. The final result will be that the vorticity equation (2.9) is not the most convenient to evaluate the electric field and a new vorticity equation is needed.

In equation (2.9), $\varpi \sim \delta_i k_{\perp} \rho_i e n_e$ since $\mathbf{V}_i \sim \delta_i v_i$ and $\nabla_{\perp} \sim k_{\perp}$. Both currents \mathbf{J}_d and $J_{||}$ are of order $\delta_i e n_e v_i$. The divergence of \mathbf{J}_d is of order $\delta_i e n_e v_i/L$ because, according to the orderings in (3.1), $\nabla p_i \sim k_{\perp} p_{i,k} \sim p_{i,k=0}/L$. The divergence of $J_{||}\hat{\mathbf{b}}$ is also of order $\delta_i e n_e v_i/L$, but in this case it is due to the small parallel gradients.

With these estimates, all the terms $\partial \varpi / \partial t$, $\nabla \cdot \mathbf{J}_d$ and $\nabla \cdot (J_{||} \hat{\mathbf{b}})$ compete with each other to determine the electric field. The remaining term, $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_i)]$, is more difficult to evaluate.

The ion viscosity $\dot{\pi}_i$, given by (2.6), is of order $O(\delta_i k_\perp \rho_i p_i)$. It only depends on the gyrophase dependent part of the distribution function, and at long wavelengths, the gyrophase dependent piece is given by (4.7), making $\dot{\pi}_i \sim \delta_i k_\perp \rho_i p_i$ because the lowest order gyrophase dependent piece vanishes. The estimate is also valid for electrons, for which it is assumed that the shortest wavelength is of the order of the ion gyroradius, giving an electron viscosity m/M times smaller than the ion viscosity, thereby justifying its neglect.

With this estimate for $\vec{\pi}_i$, the term $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)]$ in (2.9) is formally of order $(k_{\perp}\rho_i)^3 en_e v_i/L$, while the rest of the terms are of order $\delta_i en_e v_i/L$. However, I will prove in section 4.4 that the formal estimate is too high, and in reality

$$\nabla \cdot \left[\frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_i)\right] \sim \delta_i (k_\perp \rho_i)^2 e n_e v_i / L.$$
(4.9)

This term is the only one that enters in the equation for the radial electric field, as proven by (2.21). Since it becomes small for long wavelengths, the vorticity equation and hence its time integral, quasineutrality, are almost independent of the long wavelength radial electric field, i.e., the tokamak is intrinsically ambipolar even in the presence of turbulence. The order of magnitude estimate in equation (4.9) is the proof of intrinsic ambipolarity.

Except for the flux surface averaged vorticity equation, dominated by the term $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)] \sim \delta_i (k_\perp \rho_i)^2 e n_e v_i/L$, the lower order terms in the vorticity equation are always of order $\delta_i e n_e v_i/L$. Then, since the vorticity ϖ is $O(\delta_i k_\perp \rho_i e n_e)$, becoming small for the longer wavelengths, the time evolution of equation (2.9) requires a decreasing time step as $k_\perp \rho_i \to 0$, or an implicit numerical method that will ensure that the right side of (2.9) vanishes for long wavelengths. Solving implicitly for the potential is routinely done in gyrokinetic simulations [3, 62].

The flux surface averaged vorticity equation gives toroidal angular momentum

conservation equation (2.21). According to the estimates in previous paragraphs, the radial toroidal viscosity would be $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi} \sim \delta_i k_{\perp} \rho_i p_i R | \nabla \psi |$. Then, the characteristic time derivative for the toroidal velocity is $\partial/\partial t \sim k_{\perp} \rho_i v_i / L$. This estimate is in contradiction with neoclassical calculations [39, 40, 42, 43, 44] and the random walk estimate at the end of section 2.3, where the long wavelength toroidalradial component of viscosity is found to be of order $\delta_i^3 p_i R | \nabla \psi |$. This discrepancy probably only occurs transiently for short periods of time. It is to be expected that for longer times, the transport of toroidal angular momentum $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi} \sim$ $\delta_i k_{\perp} \rho_i p_i R | \nabla \psi |$ gives a net zero contribution, and the time averaged toroidal-radial component of viscosity is actually of order $\delta_i^3 p_i R | \nabla \psi |$. The size of the time averaged $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi}$ is discussed in chapter 5.

To summarize, the vorticity equation (2.9) has the right physics, and makes explicit the different times scales (from the fast turbulence times to the slow radial transport time). However, I show in section 4.4 that the divergence of $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)$ is an order smaller than its formal estimate suggests for $k_{\perp}\rho_i \ll 1$, going from $(k_{\perp}\rho_i)^3 en_i v_i/L$ to $\delta_i (k_{\perp}\rho_i)^2 en_i v_i/L$. In other words, $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)$ has a large divergence free piece. This difference between the real size and the formal ordering makes theoretical studies cumbersome and it may lead to numerical problems upon implementation. The rest of this chapter is devoted to finding a more convenient vorticity equation. In section 4.3, I will derive the particle and momentum conservation equations from the gyrokinetic equation (3.39). In the same way that particle and momentum conservation equations were used in section 2.2, I will employ the gyrokinetic conservation equations to find two gyrokinetic vorticity equations in section 4.4.

4.3 Transport in gyrokinetics

It is necessary to understand the transport of particles and momentum at wavelengths that are of the order of the ion gyroradius. It is at those wavelengths that the divergence of the viscosity becomes as important as the gradient of pressure, and we need to determine which one dominates in the vorticity equation. The gyrokinetic



Figure 4-1: Ion diamagnetic flow due to a gradient in ion density.

equation (3.39) is especially well suited to this task. In this section, I derive moment equations from the gyrokinetic Fokker-Planck equation, in particular, conservation equations for particles and momentum. They will provide powerful insights, but we have to remember that the gyrokinetic equation is correct only to $O(\delta_i f_{Mi} v_i/L)$. Then, the conservation equations for particles and momentum are missing terms of order $\delta_i^2 n_e v_i/L$ and $\delta_i^2 p_i/L$.

With the conservation equations for particles and momentum derived in this section, I will obtain two gyrokinetic vorticity equations in section 4.4. These new vorticity equations will be equivalent to equation (2.9) up to, but not including, $O(\delta_i^2 en_e v_i/L)$. They will have the advantage of explicitly cancelling the problematic divergence free component of the current density $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)$ discussed in section 4.2. The simplification originates in the fact that the ion distribution function is gyrophase independent in gyrokinetic variables. Physically, ions are replaced by rings of charge, thereby eliminating divergence free terms due to the particle gyromotion. A simple example is the diamagnetic current $-\nabla \times (cp_{i\perp}\hat{\mathbf{b}}/B)$. The physical mechanism responsible for the ion diamagnetic flow in the presence of a gradient of density is sketched in figure 4-1. In the figure, the ion density is higher in the bottom half. Then, due to the gyration, in the middle of the figure there are more ions moving towards the right than towards the left, giving a divergence free ion flow. Its divergence
vanishes because the ions just gyrate around the fixed guiding centers and there is no net ion motion. In gyrokinetics, the gyromotion velocities are not considered because the gyromotion is replaced by rings of charge, removing the divergence free terms automatically and leaving only the net gyrocenter drifts.

The conservation equations for particles and momentum are of order $\delta_i n_e v_i/L$ and $\delta_i p_i/L$, respectively, and they miss terms of order $\delta_i^2 n_e v_i/L$ and $\delta_i^2 p_i/L$. In these equations, it is possible to study what happens for wavelengths longer than the ion gyroradius, $k_{\perp}\rho_i \ll 1$. Different terms will have different scalings in $k_{\perp}\rho_i$, and these scalings will define which terms dominate at longer wavelengths. For this reason, I will determine the scalings along with the conservation equations. It is important to keep in mind that there are missing terms of order $\delta_i^2 n_e v_i/L$ and $\delta_i^2 p_i/L$, and any terms from a subsidiary expansion in $k_{\perp}\rho_i$ are not meaningful in this limit. In particular, I will show in section 4.4 that the terms that determine the axisymmetric radial electric field are too small at long wavelengths to be determined by this subsidiary expansion.

In this section, I present the general method to obtain conservation equations from gyrokinetics. In subsection 4.3.1, I derive the gyrokinetic equation in the physical phase space variables \mathbf{r} , E_0 , μ_0 and φ_0 , and I write it in a conservative form that is convenient for deriving moment equations. The details of the calculation are contained in Appendix F. In subsection 4.3.2, I derive the general moment equation for a quantity $G(\mathbf{r}, \mathbf{v}, t)$. I will apply this general equation to obtain particle and momentum transport in subsections 4.3.3 and 4.3.4, respectively. The details of the calculations are in Appendix G. In Appendix H, I show how to treat the effect of the finite gyroradius on collisions.

4.3.1 Gyrokinetic equation in physical phase space

The distribution function shows a simpler structure when written in gyrokinetic variables, namely, it is independent of the gyrophase except for the piece in (3.38) responsible for classical collisional transport. The goal of this subsection is writing the Fokker-Planck equation, $df_i/dt = C\{f_i\}$, in the physical phase space variables \mathbf{r} , E_0 , μ_0 and φ_0 , while preserving the simple form obtained by employing the gyrokinetic



Figure 4-2: Gyrokinetic equation in physical phase space.

variables. The relation between the full Fokker-Planck equation and the gyrokinetic equation written in physical phase space variables is sketched in figure 4-2. They differ in two aspects. On the one hand, the gyrokinetic equation (3.39) misses terms of order $\delta_i^2 f_{Mi} v_i/L$. On the other hand, gyrokinetics is not only a change of variables, but it also implies a time scale separation between the gyromotion and the evolution of the slowly varying electrostatic potential. The ion distribution function is then gyrophase independent in the gyrokinetic phase space, i.e., the motion of the particles may be replaced by drifting rings of charge. The advantage of this distribution function is the lack of divergence free terms in the moment equations constructed from it, as explained in the introduction to this section.

I write the Fokker-Planck equation to order $\delta_i f_{Mi} v_i/L$, the order to which the gyrokinetic equation is deduced, by starting with

$$\frac{df_i}{dt} \equiv \left. \frac{\partial f_i}{\partial t} \right|_{\mathbf{r},\mathbf{v}} + \mathbf{v} \cdot \nabla f_i + \mathbf{a} \cdot \nabla_v f_i \simeq \left. \frac{\partial f_i}{\partial t} \right|_{\mathbf{R},E,\mu,\varphi} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E} \frac{\partial f_i}{\partial E} + \dot{\varphi} \frac{\partial f_i}{\partial \varphi}, \quad (4.10)$$

where $\mathbf{a} = -Ze\nabla\phi/M + \Omega_i(\mathbf{v}\times\hat{\mathbf{b}})$ is the acceleration of particles and I have written the Vlasov operator d/dt in both \mathbf{r} , \mathbf{v} and gyrokinetic variables. The term $\dot{\mu}(\partial f_i/\partial\mu)$ does

not appear in equation (4.10) because I assume that f_i is a stationary Maxwellian to zeroth order and $\dot{\mu}$ is small by definition of μ . The derivative respect to the gyrokinetic gyrophase φ is small and related to the collision operator by (3.38),

$$\dot{\varphi}\frac{\partial f_i}{\partial \varphi} = C\{f_i\} - \langle C\{f_i\}\rangle. \tag{4.11}$$

The difference between time derivatives of f_i can be written as

$$\frac{\partial f_i}{\partial t}\Big|_{\mathbf{R},E,\mu,\varphi} - \left.\frac{\partial f_i}{\partial t}\right|_{\mathbf{r},\mathbf{v}} \simeq -\frac{Ze}{M} \left.\frac{\partial \widetilde{\phi}}{\partial t}\right|_{\mathbf{r},\mathbf{v}} \frac{\partial f_i}{\partial E},\tag{4.12}$$

where I have employed (3.25) and that **B** is independent of time. Combining equations (4.10), (4.11) and (4.12), the Fokker-Planck equation $df_i/dt = C\{f_i\}$ becomes, to $O(\delta_i f_{Mi} v_i/L)$,

$$\left. \frac{\partial f_i}{\partial t} \right|_{\mathbf{r}, \mathbf{v}} + \left. \frac{\partial}{\partial t} \left(\frac{Z e \tilde{\phi}}{T_i} f_{Mi} \right) \right|_{\mathbf{r}, \mathbf{v}} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E} \frac{\partial f_i}{\partial E} = \langle C\{f_i\} \rangle.$$
(4.13)

Here, I have used that the time evolution of $\partial f_i/\partial E \simeq (-M/T_i)f_{Mi}$ is slow.

It is necessary to rewrite equation (4.13) in the variables \mathbf{r} , E_0 , μ_0 and φ_0 . Using equation (4.3) and considering that both the zeroth order distribution function and the zeroth order potential are almost constant along magnetic field lines, I can rewrite part of equation (4.13) in terms of the variables \mathbf{r} , E_0 , μ_0 and φ_0 . The details are in section F.1 of Appendix F, and the final result is

$$\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E} \frac{\partial f_i}{\partial E} \simeq \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \cdot \left(\overline{\nabla} f_{ig} - \frac{Ze}{M} \overline{\nabla} \langle \phi \rangle \frac{\partial f_{Mi}}{\partial E_0} \right), \tag{4.14}$$

where f_{ig} is missing the piece proportional to $\tilde{\phi}$ [see equation (4.3)]. The gradient $\nabla_{\mathbf{R}_g}$ is taken with respect to \mathbf{R}_g holding E_0 , μ_0 , φ_0 and t fixed, and $\overline{\nabla}$ is the gradient with respect to \mathbf{r} holding E_0 , μ_0 , φ_0 and t fixed. The quantity $\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r}$ is given by

$$\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} = v_{\parallel 0} \hat{\mathbf{b}} + \mathbf{v}_{M0} + \mathbf{v}_{E0} + \tilde{\mathbf{v}}_1, \qquad (4.15)$$

 with

$$v_{\parallel 0} = v_{\parallel} + \frac{v_{\perp}^{2}}{2\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{2v_{\parallel}}{\Omega_{i}}\boldsymbol{\kappa} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) - \frac{v_{\parallel}}{2B\Omega_{i}}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B + \frac{v_{\parallel}}{\Omega_{i}}\mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{e}}_{2} \cdot \hat{\mathbf{e}}_{1} + \frac{1}{4\Omega_{i}}[\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp}] : \nabla \hat{\mathbf{b}}, \qquad (4.16)$$

$$\mathbf{v}_{M0} = \frac{\mu_0}{\Omega_i} \hat{\mathbf{b}} \times \nabla B + \frac{v_{||}^2}{\Omega_i} \hat{\mathbf{b}} \times \boldsymbol{\kappa}, \qquad (4.17)$$

$$\mathbf{v}_{E0} = -\frac{c}{B}\overline{\nabla}\langle\phi\rangle \times \hat{\mathbf{b}}$$
(4.18)

and using equation (F.6)

$$\tilde{\mathbf{v}}_1 = \frac{v_{||}}{\Omega_i} \overline{\nabla} \times \mathbf{v}_{\perp}. \tag{4.19}$$

In equation (4.14), it is important to be aware of higher order terms (like $\mathbf{v}_{M0} \cdot \overline{\nabla} f_{ig}$), in which the full distribution function, not just f_{Mi} , must be retained. In these terms, the steep perpendicular gradients make the higher order pieces of the distribution function important [recall the orderings in (3.1)].

Equation (4.14) can be written in conservative form, more convenient for transport calculations. The details of this calculation are in section F.2 of Appendix F, and the result is

$$\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_{i} + \dot{E} \frac{\partial f_{i}}{\partial E} \simeq \frac{v_{||}}{B} \bigg[\overline{\nabla} \cdot \left(\frac{B}{v_{||}} f_{ig} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \right) - \frac{\partial}{\partial \mu_{0}} \left(f_{Mi} \mathbf{B} \cdot \overline{\nabla} \mu_{10} \right) \\ - \frac{\partial}{\partial \varphi_{0}} \left(f_{Mi} \mathbf{B} \cdot \overline{\nabla} \varphi_{10} \right) - \frac{\partial}{\partial E_{0}} \left(\frac{B}{v_{||}} f_{Mi} \frac{Ze}{M} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \cdot \overline{\nabla} \langle \phi \rangle \bigg) \bigg], \qquad (4.20)$$

where B/v_{\parallel} is the Jacobian $\partial(\mathbf{v})/\partial(E_0, \mu, \varphi_0)$, and the quantities μ_{10} and φ_{10} are the pieces of the first order corrections μ_1 and φ_1 that do not depend on the potential. They are given by

$$\mu_{10} = \mu_1 - \frac{Ze\phi}{MB} \tag{4.21}$$

and

$$\varphi_{10} = \varphi_1 + \frac{Ze}{MB} \frac{\partial \tilde{\Phi}}{\partial \mu}.$$
(4.22)

The definitions of φ_1 and μ_1 are in equations (3.29) and (3.34), respectively.

Finally, substituting equation (4.20) into equation (4.13), I find

$$\frac{\partial f_{ig}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + \frac{v_{||}}{B} \bigg[\overline{\nabla} \cdot \left(\frac{B}{v_{||}} f_{ig} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \right) - \frac{\partial}{\partial \mu_{0}} \left(f_{Mi} \mathbf{B} \cdot \overline{\nabla} \mu_{10} \right) \\ - \frac{\partial}{\partial \varphi_{0}} \left(f_{Mi} \mathbf{B} \cdot \overline{\nabla} \varphi_{10} \right) - \frac{\partial}{\partial E_{0}} \left(\frac{B}{v_{||}} f_{Mi} \frac{Ze}{M} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \cdot \overline{\nabla} \langle \phi \rangle \bigg) \bigg] = \langle C\{f_{i}\} \rangle.$$
(4.23)

Here, for $\langle \phi \rangle$ and $\tilde{\phi}$, it is enough to consider the dependence on the lowest order variables, i.e., \mathbf{R}_g , μ_0 and φ_0 (the dependence of $\langle \phi \rangle$ and $\tilde{\phi}$ on E is weak).

4.3.2 Transport of a general function $G(\mathbf{r}, \mathbf{v}, t)$ at $k_{\perp}\rho_i \sim 1$

Multiplying equation (4.23) by a function $G(\mathbf{r}, \mathbf{v}, t)$ and integrating over velocity space, I find the conservation equation for that function G to be

$$\frac{\partial}{\partial t} \left(\int d^3 v \, G f_{ig} \right) + \nabla \cdot \left[\int d^3 v \, f_{ig} \left(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \right) G \right] = \int d^3 v \, f_{ig} K\{G\} + \int d^3 v \, G \langle C\{f_i\} \rangle, \qquad (4.24)$$

with

$$K\{G\} = \left. \frac{\partial G}{\partial t} \right|_{\mathbf{r},\mathbf{v}} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \cdot \left(\overline{\nabla} G - \frac{Ze}{M} \overline{\nabla} \langle \phi \rangle \frac{\partial G}{\partial E_0} \right) - v_{||} \hat{\mathbf{b}} \cdot \left(\overline{\nabla} \mu_{10} \frac{\partial G}{\partial \mu_0} + \overline{\nabla} \varphi_{10} \frac{\partial G}{\partial \varphi_0} \right).$$
(4.25)

In the next two subsections, I will use this formalism to study the transport of particles and momentum at short wavelengths.

4.3.3 Transport of particles at $k_{\perp}\rho_i \sim 1$

Particle transport for electrons is easy to obtain since I only need to consider $k_{\perp}\rho_{e} \ll$ 1. In this limit, drift kinetics is valid giving

$$\frac{\partial n_e}{\partial t} + \nabla \cdot \left(n_e V_{e||} \hat{\mathbf{b}} + n_e \mathbf{V}_{ed} - \frac{c n_e}{B} \nabla \phi \times \hat{\mathbf{b}} \right) = 0, \qquad (4.26)$$

with $n_e \mathbf{V}_{e||} = \int d^3 v f_e v_{||}$ and

$$n_e \mathbf{V}_{ed} = -\frac{cp_e}{eB} \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} - \frac{cp_e}{eB^2} \hat{\mathbf{b}} \times \nabla B - \frac{cp_e}{eB} \hat{\mathbf{b}} \times \boldsymbol{\kappa}.$$
 (4.27)

The same result may be deduced from equation (4.24) by neglecting electron pressure anisotropy terms that are small by a factor $\sqrt{m/M}$ compared to the ion pressure anisotropy. Obviously, the ion particle transport must be exactly the same as for the electrons due to quasineutrality. Nonetheless, we still must obtain the particle transport equation for ions to be able to calculate the electric field by requiring that both ions and electrons have the same density.

The conservation equation for ion particle number is given by equation (4.24) with G = 1. Employing section G.1 of Appendix G, it can be written as

$$\frac{\partial}{\partial t}\left(n_{i}-n_{ip}\right)+\nabla\cdot\left(n_{i}V_{i\parallel}\hat{\mathbf{b}}+n_{i}\mathbf{V}_{igd}+n_{i}\mathbf{V}_{iE}+n_{i}\tilde{\mathbf{V}}_{i}+n_{i}\mathbf{V}_{iC}\right)=0,\qquad(4.28)$$

where n_{ip} is the polarization density, defined in (3.55), the parallel flow is

$$n_i V_{i||} = \int d^3 v \, f_i v_{||} = \int d^3 v \, f_{ig} v_{||}, \qquad (4.29)$$

the term $n_i \tilde{\mathbf{V}}_i$ is a perpendicular flow that originates in finite gyroradius effects, given by

$$n_i \tilde{\mathbf{V}}_i = \int d^3 v \, f_{ig} \tilde{\mathbf{v}}_1 = \int d^3 v \, f_{ig} \frac{v_{||}}{\Omega_i} \overline{\nabla} \times \mathbf{v}_\perp, \qquad (4.30)$$

and the flows due to the $E \times B$ and magnetic drifts are

$$n_i \mathbf{V}_{iE} = \int d^3 v \, f_{ig} \mathbf{v}_{E0} = -\frac{c}{B} \int d^3 v \, f_{ig} \overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}}$$
(4.31)

and

$$n_i \mathbf{V}_{igd} = \frac{cp_{ig\perp}}{ZeB} \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{cp_{ig\perp}}{ZeB^2} \hat{\mathbf{b}} \times \nabla B + \frac{cp_{ig\parallel}}{ZeB} \hat{\mathbf{b}} \times \boldsymbol{\kappa}, \qquad (4.32)$$

with $p_{ig||} = \int d^3v f_{ig} M v_{||}^2$ and $p_{ig\perp} = \int d^3v f_{ig} M v_{\perp}^2/2$. The collisional flow $n_i \mathbf{V}_{iC}$ is evaluated in Appendix H and is caused by ion-ion collisions due to finite gyroradius



Figure 4-3: Finite gyroradius effects in $n_i \tilde{\mathbf{V}}_i$. (a) The magnetic field line along which the gyrocenter lies is the line that guides the parallel motion, making it possible for parallel motion to transport particles across magnetic field lines. (b) As the size of the gyromotion changes, a particle that spent time on both sides of the blue plane is now only on one of its sides, leading to an effective particle transport.

effects. It is given by

$$n_i \mathbf{V}_{iC} = -\frac{\gamma}{\Omega_i} \int d^3 v \, \left(\langle \mathbf{\Gamma} \rangle \times \hat{\mathbf{b}} - \frac{1}{v_{\perp}^2} \langle \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \rangle \mathbf{v} \times \hat{\mathbf{b}} \right), \tag{4.33}$$

with $\gamma = 2\pi Z^4 e^4 \ln \Lambda / M^2$ and

$$\mathbf{\Gamma} = \int d^3v' f_{Mi} f'_{Mi} \nabla_g \nabla_g g \cdot \left[\nabla_v \left(\frac{f_i}{f_{Mi}} \right) - \nabla_{v'} \left(\frac{f'_i}{f'_{Mi}} \right) \right].$$
(4.34)

Here, $f = f(\mathbf{v}), f' = f(\mathbf{v}'), \mathbf{g} = \mathbf{v} - \mathbf{v}', g = |\mathbf{g}|$ and $\nabla_g \nabla_g g = (g^2 \stackrel{\leftrightarrow}{\mathbf{I}} - \mathbf{gg})/g^3$.

In the presence of potential structure on the order of the ion gyroradius, the contributions to $n_i \tilde{\mathbf{V}}_i$ no longer average to zero in a gyration since they can add coherently. In the integration $n_i \tilde{\mathbf{V}}_i = \int d^3 v f_{ig}(v_{||}/\Omega_i) \overline{\nabla} \times \mathbf{v}_{\perp}$ only two terms contribute to its divergence so that $\nabla \cdot (n_i \tilde{\mathbf{V}}_i) = \nabla \cdot (n_i \tilde{\mathbf{V}}_{i0})$ with

$$n_i \tilde{\mathbf{V}}_{i0} = \int d^3 v \, f_{ig} \left[\frac{v_{||}}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} + \frac{v_{||} \hat{\mathbf{b}} \cdot \nabla B}{2B\Omega_i} \mathbf{v} \times \hat{\mathbf{b}} \right].$$
(4.35)

In section G.1 of Appendix G, I prove that all the other terms in $\tilde{\mathbf{v}}_1$ can be neglected.

Term	Order of magnitude
n_{ip}	$\delta_i k_\perp ho_i n_e$
$\nabla \cdot (n_i V_{i } \hat{\mathbf{b}} + n_i \mathbf{V}_{igd} + n_i \mathbf{V}_{iE})$	$\delta_i n_e v_i/L$
$ abla \cdot (n_i ilde{\mathbf{V}}_i)$	$\delta_i (k_\perp ho_i)^2 n_e v_i/L$
$ abla \cdot (n_i \mathbf{V}_{iC})$	$\delta_i (k_\perp ho_i)^2 n_e u_{ii}$

Table 4.2: Order of magnitude estimates for ion particle conservation equation (4.28).

The physical origin of $n_i \tilde{\mathbf{V}}_{i0}$ is sketched in figure 4-3. The drift $(v_{||}/\Omega_i)(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}$ is presented in figure 4-3(a), where there is difference between the direction of the magnetic field at the gyrocenter $\hat{\mathbf{b}}(\mathbf{R})$, and the direction of the magnetic field at the real position of the particle $\hat{\mathbf{b}}(\mathbf{r})$. Due to this difference, the parallel motion of the gyrocenter drives part of the gyromotion, plotted in red in the figure, across the blue plane. Notice that the magnetic field line $\mathbf{B}(\mathbf{r})$ lies in the plane, leading to the "paradoxical" parallel motion across magnetic field lines. The other term in $n_i \tilde{\mathbf{V}}_{i0}$, $(v_{||} \hat{\mathbf{b}} \cdot \nabla B/2B\Omega_i)\mathbf{v} \times \hat{\mathbf{b}}$, is explained in figure 4-3(b). Here, the change in the size of the gyroradius due to the change in magnitude of the magnetic field, $dB/dt = v_{||} \hat{\mathbf{b}} \cdot \nabla B$, drives part of the gyromotion, plotted in red, across the blue plane.

In section 4.4, I will obtain a vorticity equation for ϕ by imposing $Zn_i = n_e$. Equations (4.26) and (4.28) will provide the time evolution of $Zn_i - n_e$. It will be useful to know the size of the different terms in equation (4.28). I summarize the estimates of order of magnitude in table 4.2. These estimates include the scaling with $k_{\perp}\rho_i$ for long wavelengths. The wavenumber \mathbf{k}_{\perp} is the overall perpendicular wavenumber, i.e., given \mathbf{k}_{\perp} , the corresponding Fourier component for nonlinear terms like $A(\mathbf{r}) \times B(\mathbf{r})$ is $\int d^2k'_{\perp}\tilde{A}(\mathbf{k}'_{\perp}) \times \tilde{B}(\mathbf{k}_{\perp} - \mathbf{k}'_{\perp})$, with \tilde{A} and \tilde{B} the Fourier transforms of functions A and B. The divergence of the drift flow, $\nabla \cdot (n_i \mathbf{V}_{igd} + n_i \mathbf{V}_{iE})$, is of order $\delta_i n_e v_i / L$ since $\nabla f_{Mi} \sim \nabla f_{ik}$ [recall (3.1)]. The flow $n_i \tilde{\mathbf{V}}_i$ is of order $\delta_i^2 k_{\perp} \rho_i n_e v_i$ because, for $k_{\perp}\rho_i \ll 1$, the gyrophase dependent piece of f_{ig} , similar to (4.7), is even in v_{\parallel} to zeroth order, making the integral vanish. Its divergence, $\nabla \cdot (n_i \tilde{\mathbf{V}}_i)$, is then of order $\delta_i (k_{\perp} \rho_i)^2 n_e v_i / L$. The divergence of the collisional flow, $\nabla \cdot (n_i \mathbf{V}_i C)$, is of order $\delta_i(k_{\perp}\rho_i)^2 \nu_{ii} n_e$, as proven in Appendix H. The polarization density n_{ip} is of order $\delta_i k_{\perp} \rho_i n_e$, as shown in (3.59). This result means that for long wavelengths, the polarization term becomes unimportant. Therefore, at long wavelengths only the balance between the time evolution of the density, the parallel flow, and the magnetic and $E \times B$ drifts matter.

4.3.4 Transport of momentum at $k_{\perp}\rho_i \sim 1$

From electron momentum conservation, I will only need the parallel component, given by

$$\hat{\mathbf{b}} \cdot \nabla p_{e||} + (p_{e||} - p_{e\perp}) \nabla \cdot \hat{\mathbf{b}} = e n_e \hat{\mathbf{b}} \cdot \nabla \phi + F_{ei||}, \qquad (4.36)$$

where $F_{ei||} = m \int d^3 v v_{||} C_{ei} \{f_e\}$ is the collisional parallel momentum exchange. The time derivative of $n_e m \mathbf{V}_e$ and the viscosity have been neglected because they are small by a factor of m/M. In this equation, there are terms of two different orders of magnitude. The dominant terms are $\hat{\mathbf{b}} \cdot \nabla p_{e||} \simeq \hat{\mathbf{b}} \cdot \nabla p_e$ and $en_e \hat{\mathbf{b}} \cdot \nabla \phi$, both of order $O(\delta_i p_e/L)$ for turbulent fluctuations at $k_{\perp}\rho_i \sim 1$. The friction force $F_{ei||}$ and the terms that contain the pressure anisotropy $p_{e||} - p_{e\perp} \sim \delta_e p_e$ are an order $\sqrt{m/M}$ smaller than the dominant terms. However, these smaller terms are crucial because they provide the non-adiabatic behavior and hence allow radial transport of particles. In the vorticity equation (2.9), the non-adiabatic electron response is kept in the integral $\int d^3 v f_e v_{||}$ in $J_{||}$. Thus, for most purposes, equation (4.36) can be simplified to

$$\hat{\mathbf{b}} \cdot \nabla p_e = e n_e \hat{\mathbf{b}} \cdot \nabla \phi. \tag{4.37}$$

Equation (4.36) can also be recovered by using equation (4.24) and neglecting terms small by m/M.

For the ions, using equation (4.24) with $G = \mathbf{v}$ and employing section G.2 of Appendix G, I find the momentum conservation equation

$$\frac{\partial}{\partial t} (n_i M \mathbf{V}_{ig}) + \hat{\mathbf{b}} [\hat{\mathbf{b}} \cdot \nabla p_{ig||} + (p_{ig||} - p_{ig\perp}) \nabla \cdot \hat{\mathbf{b}} + \nabla \cdot \boldsymbol{\pi}_{ig||}] + \nabla \cdot \boldsymbol{\overleftarrow{\pi}}_{ig\times} = -Zen_i \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \phi + \tilde{F}_{iE} \hat{\mathbf{b}} + \mathbf{F}_{iB} + \mathbf{F}_{iC}, \quad (4.38)$$

where $n_i \mathbf{V}_{ig} = \int d^3 v f_{ig} \mathbf{v}$ is the average gyrocenter velocity; the vector $\boldsymbol{\pi}_{ig||}$ is the parallel momentum transported by the drifts and is given by

$$\boldsymbol{\pi}_{ig||} = \int d^3 v \, f_{ig}(\mathbf{v}_{M0} + \mathbf{v}_{E0} + \tilde{\mathbf{v}}_1) M v_{||}; \qquad (4.39)$$

the tensor $\dot{\pi}_{ig\times}$ gives the transport of perpendicular momentum by the parallel velocity and the drifts,

$$\dot{\boldsymbol{\pi}}_{ig\times} = \int d^3 v \, f_{ig}(v_{||} \hat{\mathbf{b}} + \mathbf{v}_{M0} + \mathbf{v}_{E0} + \tilde{\mathbf{v}}_1) M \mathbf{v}_{\perp}; \qquad (4.40)$$

the vector $\tilde{F}_{iE}\hat{\mathbf{b}}$ is a correction due to the short wavelengths of the electric field with

$$\tilde{F}_{iE} = Ze \int d^3v \, f_{Mi}(\hat{\mathbf{b}} + \Omega_i^{-1} \overline{\nabla} \times \mathbf{v}_\perp) \cdot \overline{\nabla} \widetilde{\phi}; \qquad (4.41)$$

the vector \mathbf{F}_{iB} contains the effect on the gyromotion of the variation in the magnetic field and is given by

$$\mathbf{F}_{iB} = \int d^3 v \, M f_{ig} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} + \int d^3 v \, \frac{Mc}{B} f_{Mi} (\overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}; \qquad (4.42)$$

and the finite gyroradius effects on collisions are included in

$$\mathbf{F}_{iC} \equiv M \int d^{3}v \mathbf{v} \langle C\{f_{i}\}\rangle = -M\gamma \int d^{3}v \left(\langle \mathbf{\Gamma} \rangle \cdot \hat{\mathbf{b}}\hat{\mathbf{b}} + \frac{1}{v_{\perp}^{2}} \langle \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \rangle \mathbf{v}_{\perp} \right) + \nabla \cdot \left[\frac{M\gamma}{\Omega_{i}} \int d^{3}v \left(\langle \mathbf{\Gamma} \rangle \times \hat{\mathbf{b}} - \frac{1}{v_{\perp}^{2}} \langle \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \rangle \mathbf{v} \times \hat{\mathbf{b}} \right) \mathbf{v} \right].$$
(4.43)

The force \tilde{F}_{iE} originates in the change in the parallel velocity magnitude due to potential structures on the size of the ion gyroradius. The force \mathbf{F}_{iB} accounts for the change in perpendicular velocity due to variation in the magnetic field that the particle feels during its motion. Interestingly, the parallel component of equation (4.38) is simply

$$\frac{\partial}{\partial t}(n_i M V_{i||}) + \hat{\mathbf{b}} \cdot \nabla p_{ig||} + (p_{ig||} - p_{ig\perp}) \nabla \cdot \hat{\mathbf{b}} + \nabla \cdot \boldsymbol{\pi}_{ig||} = -Zen_i \hat{\mathbf{b}} \cdot \nabla \phi$$

Term	Order of magnitude
$n_i M \mathbf{V}_{ig}$	$\delta_i n_e M v_i$
$\hat{\mathbf{b}} \cdot \nabla p_{ig }, (p_{ig } - p_{ig\perp}) \nabla \cdot \hat{\mathbf{b}}, Zen_i \hat{\mathbf{b}} \cdot \nabla \phi$	$\delta_i p_i/L$
$ abla \cdot oldsymbol{\pi}_{ig }, abla \cdot \overleftrightarrow{oldsymbol{\pi}}_{ig imes}, ilde{F}_{iE}, \mathbf{F}_{iB}$	$\delta_i k_\perp ho_i p_i/L$
\mathbf{F}_{iC}	$\delta_i k_\perp ho_i u_{ii} n_e M v_i$

Table 4.3: Order of magnitude estimates for ion momentum conservation equation (4.38).

$$+\tilde{F}_{iE}+\mathbf{F}_{iC}\cdot\hat{\mathbf{b}}.$$
 (4.44)

The parallel components of $\nabla \cdot \overleftarrow{\pi}_{ig\times}$ and \mathbf{F}_{iB} cancel each other, as proven in section G.2 of Appendix G.

Equation (4.38) will be used in section 4.4 to get one of the forms of the vorticity equation. Thus, it is useful to estimate the size of the different terms in it. The estimates of order of magnitude are summarized in table 4.3. The pressure terms, $\hat{\mathbf{b}} \cdot \nabla p_{ig||} + (p_{ig||} - p_{ig\perp}) \nabla \cdot \hat{\mathbf{b}}$, and the electric field term, $Zen_i \hat{\mathbf{b}} \cdot \nabla \phi$, are $O(\delta_i p_i/L)$. The terms \tilde{F}_{iE} and \mathbf{F}_{iB} are of order $\delta_i k_{\perp} \rho_i p_i / L$. These estimates are obvious for the integrals $(Ze/\Omega_i) \int d^3v f_{Mi}(\overline{\nabla} \times \mathbf{v}_{\perp}) \cdot \overline{\nabla} \widetilde{\phi} \sim (Mc/B) \int d^3v f_{Mi}(\overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}$ since $\overline{\nabla} \phi \sim k_{\perp} \rho_i T_e / eL$ due to $e \phi / T_e \sim \delta_i$. The integral $Ze \int d^3 v f_{Mi} \hat{\mathbf{b}} \cdot \overline{\nabla} \phi$ would seem to be of order $\delta_i p_i/L$ since $\hat{\mathbf{b}} \cdot \overline{\nabla} \tilde{\phi} \sim \delta_i T_e/eL$ but it is an order $k_{\perp} \rho_i$ smaller because the integral of $\tilde{\phi}$ in the gyrophase φ_0 vanishes to zeroth order. The integral $M \int d^3v f_{ig} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp}$ is of order $\delta_i k_{\perp} \rho_i p_i / L$ because the leading order gyrophase dependent piece of f_{ig} is even in $v_{||}$ [recall (4.7)]. The collisional force, \mathbf{F}_{iC} is order $\delta_i k_{\perp} \rho_i \nu_{ii} n_e M v_i$, as proven in Appendix H. The vector $\boldsymbol{\pi}_{ig||}$ is $O(\delta_i^2 p_i)$ because f_{ig} is even in $v_{||}$ and \mathbf{v}_{\perp} up to order $\delta_i f_{Mi}$. The matrix $\overleftarrow{\pi}_{ig\times}$ has three different pieces: the integral $\int d^3 v f_{ig}(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) M \mathbf{v}_{\perp}$ also of order $\delta_i^2 p_i$, the integral $\int d^3v f_{ig} M v_{||} \hat{\mathbf{b}} \mathbf{v}_{\perp}$ of order $\delta_i k_{\perp} \rho_i p_i$, and the integral $\int d^3v f_{ig} M \mathbf{v}_{E0} \mathbf{v}_{\perp}$ of order $\delta_i p_i$. The integral $\int d^3 v f_{ig} M v_{\parallel} \hat{\mathbf{b}} \mathbf{v}_{\perp}$ is of order $\delta_i k_{\perp} \rho_i p_i$ because the leading order gyrophase dependent piece of f_{ig} is even in $v_{||}$ as in (4.7). On the other hand, the size of the integral $\int d^3v f_{ig} M \mathbf{v}_{E0} \mathbf{v}_{\perp}$ is estimated by employing $\mathbf{v}_{E0} = -(c/B)\overline{\nabla}\langle\phi\rangle \times \hat{\mathbf{b}} =$

 $-(c/B)\nabla\phi\times\hat{\mathbf{b}}+(c/B)\overline{\nabla}\phi\times\hat{\mathbf{b}}.$ Then, I find that $(Mc/B)\int d^3v f_{ig}(\nabla\phi\times\hat{\mathbf{b}})\mathbf{v}_{\perp}\sim\delta_i^2 p_i$ since f_{ig} is gyrophase dependent at $O(\delta_i f_{Mi})$, and $(Mc/B)\int d^3v f_{ig}(\overline{\nabla}\phi\times\hat{\mathbf{b}})\mathbf{v}_{\perp}\lesssim\delta_i p_i$. It is difficult to refine this last estimate because it is a nonlinear term and short wavelength pieces of f_{ig} and ϕ can beat to give a long wavelength result. The divergences of $\pi_{ig||}$ and $\tilde{\pi}_{ig\times}$ are both $O(\delta_i k_{\perp} \rho_i p_i/L)$. Importantly, the divergence of $\tilde{\pi}_{ig\times} \lesssim \delta_i p_i$ is not of order $k_{\perp} \rho_i p_i/L$ but of order $\delta_i k_{\perp} \rho_i p_i/L$. The divergence of $\int d^3v f_{ig} Mv_{||} \hat{\mathbf{b}} \mathbf{v}_{\perp}$ is of order $\delta_i k_{\perp} \rho_i p_i/L$ because it only contains a parallel gradient, and $\nabla \cdot [(Mc/B) \int d^3v f_{ig}(\overline{\nabla}\phi\times\hat{\mathbf{b}})\mathbf{v}_{\perp}] = -\nabla \cdot [Mc \int dE_0 d\mu_0 d\varphi_0 \phi \overline{\nabla} \times (\hat{\mathbf{b}} f_{ig} \mathbf{v}_{\perp}/v_{||})] \sim$ $\delta_i k_{\perp} \rho_i p_i/L$, where I have used $d^3v = (B/v_{||}) dE_0 d\mu_0 d\varphi_0$ and $\overline{\nabla} \cdot [\overline{\nabla} \times (\ldots)] = 0$ [since $(f_{ig}/v_{||})(\overline{\nabla}\phi\times\hat{\mathbf{b}})\mathbf{v}_{\perp} = \overline{\nabla} \times (\hat{\mathbf{b}}\phi f_{ig}\mathbf{v}_{\perp}/v_{||}) - \phi \overline{\nabla} \times (\hat{\mathbf{b}} f_{ig}\mathbf{v}_{\perp}/v_{||})].$

4.4 Vorticity equation for gyrokinetics

In section 4.2, I showed that the term that contains the ion viscosity in the vorticity equation (2.9) seems to dominate at short wavelengths. However, in reality it is smaller than $\nabla \cdot \mathbf{J}_d$, as I demonstrate in this section. As a result, the viscosity must be evaluated carefully; otherwise, spurious terms may appear in numerical simulations. Here, I propose two different vorticity equations that avoid this numerical problem and are valid for short wavelengths. Long wavelength, transport time scale phenomena, like the self-consistent calculation of the radial electric field, can be included, but this will be the subject of chapter 5.

The vorticity equation (2.9) provides a way to temporally evolve the electric field perpendicular to the magnetic field. However, the parallel electric field strongly depends on the parallel electron dynamics, hidden in the parallel current $J_{||}$ in equation (2.9). Fortunately, it is enough to use the integral $J_{||} = Ze \int d^3v v_{||} f_{ig} - e \int d^3v v_{||} f_e$ for the parallel current since $J_{||}$ does not alter the higher order calculation of the radial electric field determined by equation (2.21). In several codes [3, 62], the electron and ion distribution functions are solved implicitly in the potential. These implicit solutions are then substituted in $J_{||}$ to find the potential in the next time step from the vorticity equation. In subsection 4.4.1, a vorticity equation is derived directly from gyrokinetic quasineutrality. The advantage of this form is its close relation to previous algorithms, but it differs greatly from the general vorticity equation (2.9). In subsection 4.4.2, I present a modified vorticity equation that has more similarities with equation (2.9). I ensure that both forms are equivalent and satisfy the desired condition at long wavelengths, namely, that they provide a fixed toroidal velocity.

4.4.1 Vorticity from quasineutrality

The first version of the vorticity equation is obtained by taking the time derivative of the gyrokinetic quasineutrality $(Zn_i = n_e)$. In other words, I find the time evolution of ion and electron density and ensure that its difference is constant in time. This is equivalent to subtracting equation (4.26) from Z times (4.28) to obtain

$$\frac{\partial}{\partial t}(Zen_{ip}) = \nabla \cdot \left(J_{||}\hat{\mathbf{b}} + \mathbf{J}_{gd} + \tilde{\mathbf{J}}_i + Zen_i\tilde{\mathbf{V}}_i + Zen_i\mathbf{V}_{iC} \right),$$
(4.45)

where $\tilde{\mathbf{J}}_i$ is the polarization current

$$\tilde{\mathbf{J}}_{i} \equiv \frac{Zec}{B} \left(\nabla \phi \times \hat{\mathbf{b}} \int d^{3}v \, f_{i} - \int d^{3}v \, f_{ig} \overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}} \right) = \frac{Zec}{B} \left(\int d^{3}v \, f_{ig} \overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}} - \nabla \phi \times \hat{\mathbf{b}} \int d^{3}v \, \frac{Ze\widetilde{\phi}}{T_{i}} f_{Mi} \right)$$
(4.46)

and \mathbf{J}_{gd} is the drift current

$$\mathbf{J}_{gd} \equiv Zen_i \mathbf{V}_{igd} - en_e \mathbf{V}_{ed} = \frac{cp_{g||}}{B} \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{cp_{g\perp}}{B^2} \hat{\mathbf{b}} \times \nabla B + \frac{cp_{g||}}{B} \hat{\mathbf{b}} \times \boldsymbol{\kappa}, \quad (4.47)$$

with $p_{g||} = p_{ig||} + p_e$ and $p_{g\perp} = p_{ig\perp} + p_e$. Here, to write the second form of $\tilde{\mathbf{J}}_i$, I use $f_i - f_{ig} = -(Ze\tilde{\phi}/T_i)f_{Mi}$, given in (4.3). In equation (4.45), the ion polarization density, $n_{ip} = -\int d^3v (Ze\tilde{\phi}/T_i)f_{Mi}$ is advanced in time, and the electric field is solved from n_{ip} .

It is necessary to check if equation (4.45) satisfies the right conditions at long wavelengths. In the present form, though, it is a tedious task. To perform this check,

Term	Order of magnitude
Zen_{ip}	$\delta_i k_\perp ho_i e n_e$
$ abla \cdot (J_{ } \hat{\mathbf{b}} + \mathbf{J}_{gd})$	$\delta_i e n_e v_i/L$
$ abla \cdot (ilde{\mathbf{J}}_i + Zen_i ilde{\mathbf{V}}_i)$	$\delta_i (k_\perp ho_i)^2 e n_e v_i / L$
$ abla \cdot (Zen_i \mathbf{V}_{iC})$	$\delta_i (k_\perp ho_i)^2 e n_e u_{ii}$

Table 4.4: Order of magnitude estimates for vorticity equation (4.45).

I will use the much more convenient form in subsection 4.4.2, that I will prove is equivalent.

Finally, I give estimates for the size of all the terms in (4.45) in table 4.4. These estimates will be useful in subsection 4.4.2 to study the behavior of the toroidal angular momentum for $k_{\perp}\rho_i \ll 1$. The size of most of the terms in equation (4.45) can be obtained from the estimates given in table 4.2, giving $\nabla \cdot \mathbf{J}_{gd} \sim \delta_i en_e v_i/L$, $\nabla \cdot (Zen_i \tilde{\mathbf{V}}_i) \sim \delta_i (k_{\perp}\rho_i)^2 en_e v_i/L$ and $\nabla \cdot (Zen_i \mathbf{V}_{iC}) \sim \delta_i (k_{\perp}\rho_i)^2 \nu_{ii} en_e$. The size of $\nabla \cdot \tilde{\mathbf{J}}_i$ requires more work. Even though a cancellation between the drift kinetic $E \times B$ flow, $(cn_i/B)\nabla\phi \times \hat{\mathbf{b}}$, and the corresponding gyrokinetic flow, $(c/B)\int d^3v f_{ig}\nabla\langle\phi\rangle \times \hat{\mathbf{b}}$, is expected, due to the nonlinear character of these terms, where the short wavelength components of f_{ig} and ϕ can beat to give a long wavelength term, I can only give a bound for the size of $\tilde{\mathbf{J}}_i$. Given the definition of $\tilde{\mathbf{J}}_i$ in equation (4.46), its size is bounded by $|\tilde{\mathbf{J}}_i| \lesssim \delta_i en_e v_i$. Then, it would seem that its divergence must be $|\nabla \cdot \tilde{\mathbf{J}}_i| \lesssim k_{\perp}\rho_i en_e v_i/L$, but the lowest order terms contain $\nabla \times \nabla \phi = 0$ and $\nabla \times \nabla \langle \phi \rangle = 0$, leading to $|\nabla \cdot \tilde{\mathbf{J}}_i| \lesssim \delta_i en_e v_i/L$. To refine this bound, I use that $\nabla \cdot \tilde{\mathbf{J}}_i = \int dE_0 d\mu_0 d\varphi_0 \nabla \cdot$ $\tilde{\mathbf{J}}_i$, where $\nabla \cdot \tilde{\mathbf{J}}_i$ is found from equation (4.46) to be

$$\overline{\nabla} \cdot \tilde{\mathbf{j}}_{i} \equiv \overline{\nabla} \cdot \left[\frac{Zec}{v_{||}} \left(f_{i} \nabla \phi \times \hat{\mathbf{b}} - f_{ig} \overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}} \right) \right] = -\overline{\nabla} \cdot \left[\tilde{\phi} \overline{\nabla} \times \left(\hat{\mathbf{b}} \frac{Zec}{v_{||}} f_{ig} \right) + \frac{Zec}{v_{||}} \frac{Ze\tilde{\phi}}{T_{i}} f_{Mi} \nabla \phi \times \hat{\mathbf{b}} \right] \sim \delta_{i} k_{\perp} \rho_{i} e f_{Mi} B / L.$$
(4.48)

I have used $d^3v = (B/v_{||})dE_0 d\mu_0 d\varphi_0$ and $\overline{\nabla} \cdot (v_{||}^{-1}f_{ig}\overline{\nabla}\widetilde{\phi} \times \hat{\mathbf{b}}) = -\overline{\nabla} \cdot [\widetilde{\phi} \,\overline{\nabla} \times (\hat{\mathbf{b}}v_{||}^{-1}f_{ig})]$ to find this result. The second form of (4.48) is useful to estimate the size of $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i$ because I can use $e\tilde{\phi}/T_e \sim \delta_i$ and $\overline{\nabla} f_{ig} \sim f_{Mi}/L$ to find $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i \sim \delta_i k_\perp \rho_i e f_{Mi} B/L$. In $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i$, given in (4.48), there are short wavelength components of f_{ig} and ϕ that beat together to give a long wavelength component. These short wavelength components of $f_{ig} \equiv f_i(\mathbf{R}_g, E_0, \mu_0, t)$ and $\tilde{\phi}(\mathbf{R}_g, \mu_0, \varphi_0, t)$ cannot be expanded around \mathbf{r} , but the long wavelength component of $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i$ can be expanded as a whole to find

$$\overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{R}_g, E_0, \mu_0, \varphi_0) \simeq \overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{r}, E_0, \mu_0, \varphi_0) + \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} (\overline{\nabla} \cdot \tilde{\mathbf{j}}_i).$$
(4.49)

The difference between $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{r}, E_0, \mu_0, \varphi_0)$ and $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{R}_g, E_0, \mu_0, \varphi_0)$ is negligible in the higher order term. The interesting property of equation (4.49) is that the velocity integral of the zeroth order term $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{r}, E_0, \mu_0, \varphi_0)$ can be done because the gyrophase dependence in \mathbf{R}_g has disappeared. Employing the second form of equation (4.48) for $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{r}, E_0, \mu_0, \varphi_0)$, I find $\int dE_0 d\mu_0 d\varphi_0 \overline{\nabla} \cdot \tilde{\mathbf{j}}_i(\mathbf{r}, E_0, \mu_0, \varphi_0) =$ $-\nabla \cdot [(Zec/B) \int d^3v (Ze\tilde{\phi}/T_i) f_{Mi} \overline{\nabla} \tilde{\phi} \times \hat{\mathbf{b}}] \sim \delta_i^2 k_\perp \rho_i en_e v_i/L$ since $\tilde{\phi} \overline{\nabla} \tilde{\phi} = \overline{\nabla}(\tilde{\phi}^2/2)$. This result is negligible compared with the term $\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla}(\overline{\nabla} \cdot \tilde{\mathbf{j}}_i)$ in (4.49), which gives $\nabla \cdot \tilde{\mathbf{J}}_i \sim \delta_i (k_\perp \rho_i)^2 en_e v_i/L$ since $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i \sim \delta_i k_\perp \rho_i ef_{Mi} B/L$. Using this result, I find that the $k_\perp \rho_i \ll 1$ limit of $\nabla \cdot \tilde{\mathbf{J}}_i$ is

$$\nabla \cdot \tilde{\mathbf{J}}_{i} \simeq \nabla \cdot \left[\int dE_{0} \, d\mu_{0} \, d\varphi_{0} \frac{1}{\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \overline{\nabla} \cdot \tilde{\mathbf{j}}_{i} \right], \tag{4.50}$$

where I use that $\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla}(\overline{\nabla} \cdot \tilde{\mathbf{j}}_i) \simeq \overline{\nabla} \cdot [\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}})(\overline{\nabla} \cdot \tilde{\mathbf{j}}_i)]$ because the gradient of $\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}}$ is of order 1/L and the gradient of $\overline{\nabla} \cdot \tilde{\mathbf{j}}_i$ is of order k_{\perp} .

4.4.2 Vorticity from moment description

Equation (4.45) has the advantage of having a direct relation with the gyrokinetic quasineutrality. However, its relation with the full vorticity equation (2.9) and the evolution of toroidal angular momentum is not explicit. For those reasons, I next derive an alternative vorticity equation.

I define a new useful function

$$\varpi_G = \nabla \cdot \left(\frac{Ze}{\Omega_i} \int d^3 v \, f_{ig} \mathbf{v} \times \hat{\mathbf{b}}\right) - \int d^3 v \, \frac{Z^2 e^2 \tilde{\phi}}{T_i} f_{Mi} \tag{4.51}$$

that I will call gyrokinetic "vorticity" because, for $k_{\perp}\rho_i \ll 1$, it tends to $\varpi = \nabla \cdot [(Ze/\Omega_i)\mathbf{V}_i \times \hat{\mathbf{b}}]$. To see this, I use $-\int d^3v \, (Ze\tilde{\phi}/T_i)f_{Mi} \to \nabla \cdot [(cn_i/B\Omega_i)\nabla_{\perp}\phi]$ and $\int d^3v \, f_{ig}\mathbf{v} \times \hat{\mathbf{b}} \to (c/ZeB)\nabla_{\perp}p_i$ to obtain

$$\varpi_G \simeq \nabla \cdot \left(\frac{Zecn_i}{B\Omega_i} \nabla_\perp \phi + \frac{c}{B\Omega_i} \nabla_\perp p_i \right), \tag{4.52}$$

as given by $\overline{\omega}$ in equation (2.12) to first order. The new version of the vorticity equation will evolve in time the gyrokinetic "vorticity" (4.51). The advantage of this new equation is that it will tend to a form similar to the moment vorticity equation (2.9) for $k_{\perp}\rho_i \ll 1$. It is important to point out that the new version of the vorticity equation, to be given in (4.53), and equations (2.9) and (4.45) are totally equivalent up to $O(\delta_i^2 e n_e v_i/L)$. The advantage of the new version is a similarity with the full vorticity equation (2.9). This similarity helps to study the size of the term $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)]$ and the behavior of the transport of toroidal angular momentum for $k_{\perp}\rho_i \ll 1$. I will prove that the toroidal velocity varies slowly, as expected. Additionally, since the new vorticity equation is derived from quasineutrality and the gyrokinetic equation, it is equivalent to the gyrokinetic quasineutrality equation and provides a way to study its limitations.

The new version of the vorticity equation is obtained by adding equations (4.45) and $\nabla \cdot \{(c/B) | \text{equation } (4.38) | \times \hat{\mathbf{b}} \}$ to obtain

$$\frac{\partial \varpi_G}{\partial t} = \nabla \cdot \left[J_{||} \hat{\mathbf{b}} + \mathbf{J}_{gd} + \tilde{\mathbf{J}}_{i\phi} + \frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_{iG}) + Zen_i \mathbf{V}_{iC} - \frac{c}{B} \hat{\mathbf{b}} \times \mathbf{F}_{iC} \right].$$
(4.53)

The terms $\tilde{\mathbf{J}}_i$, $Zen_i\tilde{\mathbf{V}}_i$, $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_{ig\times})$ and $(c/B)\hat{\mathbf{b}} \times \mathbf{F}_{iB}$ are recombined to give $\tilde{\mathbf{J}}_{i\phi}$, $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_{iG})$ and some other terms that have vanished because they are divergence free. The details on how to obtain equation (4.53) are in Appendix I. Here

Term	Order of magnitude
$arpi_G$	$\delta_i k_\perp ho_i e n_e$
$ abla \cdot (J_{ } \hat{\mathbf{b}} + \mathbf{J}_{gd})$	$\delta_i e n_e v_i/L$
$\nabla \cdot [\tilde{\mathbf{J}}_{i\phi} + (c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_{iG})]$	$\delta_i (k_\perp ho_i)^2 e n_e v_i / L$
$\nabla \cdot [Zen_i \mathbf{V}_{iC} - (c/B)\hat{\mathbf{b}} \times \mathbf{F}_{iC}]$	$\delta_i (k_\perp ho_i)^2 e n_e u_{ii}$

Table 4.5: Order of magnitude estimates for vorticity equation (4.53).

I have defined the new viscosity tensor

and the new polarization current

$$\tilde{\mathbf{J}}_{i\phi} = \tilde{\mathbf{J}}_i - \frac{Zec}{B\Omega_i} \hat{\mathbf{b}} \times \int d^3 v \, f_{Mi} (\overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_\perp, \qquad (4.55)$$

with $\tilde{\mathbf{J}}_i$ given in (4.46). The derivation of equation (4.53) implies that both equations (4.45) and (4.53) are equivalent as long as the perpendicular component of equation (4.38) is satisfied, and any property proved for one of them is valid for the other.

Equation (4.53) gives the evolution of ϖ_G and the potential is then found by solving equation (4.51). Equation (4.53) does not contain terms that are almost divergence free, as was the case of $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_i)$ in equation (2.9). This will ease implementation in existing simulations.

It is important to know the size of the different terms in (4.53) for implementation purposes. The order of magnitude of the different terms is summarized in table 4.5. In subsection 4.4.1, in table 4.4, I showed $\nabla \cdot \mathbf{J}_{gd} \sim \delta_i en_e v_i/L$ and $\nabla \cdot (Zen_i \mathbf{V}_{iC}) \sim \delta_i (k_\perp \rho_i)^2 \nu_{ii} en_e$. The term $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times \mathbf{F}_{iC}]$ is of order $\delta_i (k_\perp \rho_i)^2 \nu_{ii} en_e$ according to the results in Appendix H. For the flow $(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_{iG})$, there are two different pieces in $\vec{\pi}_{iG}$, given in (4.54), namely, $M \int d^3 v f_{ig} \mathbf{v}_\perp v_{||} \hat{\mathbf{b}}$ and $\vec{\pi}_{ig\times}$ as defined in (4.40). The first component gives $(Mc/B)\hat{\mathbf{b}} \times [\nabla \cdot (\int d^3 v f_{ig} \mathbf{v}_\perp v_{||} \hat{\mathbf{b}})] = (Ze/\Omega_i)\hat{\mathbf{b}} \times (\int d^3 v f_{ig} v_{||} \mathbf{v}_\perp \cdot \nabla \hat{\mathbf{b}}) \sim$ $\delta_i^2 k_{\perp} \rho_i en_e v_i$, where I have used that the lowest order gyrophase dependent piece of f_{ig} is even in $v_{||}$ [recall (4.7)]. The divergence $\nabla \cdot \overleftarrow{\pi}_{ig\times}$ is of order $\delta_i k_{\perp} \rho_i p_i / L$ as proven in subsection 4.3.4, giving $\nabla \cdot [(c/B) \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_{iG})] \sim \delta_i (k_{\perp} \rho_i)^2 en_e v_i / L$. Finally, $\widetilde{\mathbf{J}}_{i\phi}$ is also composed of two pieces shown in (4.55). The divergence of $\nabla \cdot \widetilde{\mathbf{J}}_i$ was already found in (4.50), and it is of order $\delta_i (k_{\perp} \rho_i)^2 en_e v_i / L$. The second term in (4.55) is of order $\delta_i^2 k_{\perp} \rho_i en_e v_i$, giving $\nabla \cdot \widetilde{\mathbf{J}}_{i\phi} \sim \delta_i (k_{\perp} \rho_i)^2 en_e v_i / L$. Interestingly, employing equations (4.48) and (4.50), and the definition of $\widetilde{\mathbf{J}}_{i\phi}$ in (4.55), I find that for $k_{\perp} \rho_i \ll 1$, $\nabla \cdot \widetilde{\mathbf{J}}_{i\phi}$ tends to

$$\nabla \cdot \tilde{\mathbf{J}}_{i\phi} = \nabla \cdot \left\{ \frac{c}{B} \hat{\mathbf{b}} \times \left[\nabla \cdot \left\{ \int d^3 v \left(f_{ig} \frac{c}{B} \overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}} - f_i \frac{c}{B} \nabla \phi \times \hat{\mathbf{b}} \right) M \mathbf{v}_{\perp} \right\} \right] \right\}.$$
(4.56)

To obtain this expression I neglect

$$\nabla \cdot \left\{ \frac{Ze}{\Omega_i} \hat{\mathbf{b}} \times \left[\int d^3 v \left(f_i - f_{ig} \right) \frac{c}{B} (\nabla \phi \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_\perp \right] \right\} \sim \delta_i^2 k_\perp \rho_i e n_e v_i / L.$$
(4.57)

Equation (4.56) will be useful in the $k_{\perp}\rho_i \ll 1$ limit worked out in Appendix J.

The estimates in table 4.5 are useful to determine the size of the problematic term $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overleftrightarrow{\pi}_i)]$. Subtracting equation (2.9) from (4.53) gives

$$\nabla \cdot \left[\frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{i})\right] = \frac{\partial}{\partial t}(\boldsymbol{\varpi} - \boldsymbol{\varpi}_{G}) + \nabla \cdot \left[\mathbf{J}_{gd} - \mathbf{J}_{d} + \widetilde{\mathbf{J}}_{i\phi} + \frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{iG}) + Zen_{i}\mathbf{V}_{iC} - \frac{c}{B}\hat{\mathbf{b}} \times \mathbf{F}_{iC}\right] \sim \delta_{i}(k_{\perp}\rho_{i})^{2}en_{e}v_{i}/L.$$
(4.58)

This result was anticipated in equation (4.9) and proves that turbulent tokamaks are intrinsically ambipolar!

Most of the estimates for the terms in equation (4.58) are obtained from table 4.5. Only $\varpi - \varpi_G$ and $\nabla \cdot (\mathbf{J}_{gd} - \mathbf{J}_d)$ need clarification. The long wavelength limit of $\varpi_G \sim \delta_i k_{\perp} \rho_i e n_e$, given in (4.52), is the same as the long wavelength limit of ϖ in (2.12). Thus, they can only differ in the next order in $k_{\perp} \rho_i$, giving $\varpi - \varpi_G \sim \delta_i (k_{\perp} \rho_i)^2 e n_e$. The difference $\mathbf{J}_{gd} - \mathbf{J}_d$ can be rewritten using $f_i - f_{ig} = -(Ze\tilde{\phi}/T_i)f_{Mi}$ to obtain

$$\mathbf{J}_{gd} - \mathbf{J}_{d} = \int d^{3}v \, \frac{Z^{2} e^{2} \widetilde{\phi}}{T_{i}} f_{Mi} \left(\frac{v_{\perp}^{2}}{2\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{v_{\perp}^{2}}{2B\Omega_{i}} \hat{\mathbf{b}} \times \nabla B + \frac{v_{\parallel}^{2}}{\Omega_{i}} \hat{\mathbf{b}} \times \boldsymbol{\kappa} \right). \quad (4.59)$$

The size of $\mathbf{J}_{gd} - \mathbf{J}_d$ is $\delta_i^2 k_{\perp} \rho_i e n_e v_i$ since the integral of $\tilde{\phi}$ in the gyrophase vanishes to zeroth order. Then, $\nabla \cdot (\mathbf{J}_{gd} - \mathbf{J}_d) \sim \delta_i (k_{\perp} \rho_i)^2 e n_e v_i / L$. Employing these estimates, I find that $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)]$ is of order $\delta_i (k_{\perp} \rho_i)^2 e n_e v_i / L$, as given in (4.58) and asserted in equation (4.9) in section 4.2. Therefore, there is a piece of the viscosity of order $\delta_i k_{\perp} \rho_i p_i$ that vanishes to zeroth order in $\nabla \cdot [(c/B)\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_i)]$. In equations (4.45) and (4.53) this piece has already been cancelled.

Finally, I study the evolution of the toroidal velocity implicit in equation (4.53). To do so, I flux surface average equation (4.53) as I did in section 2.3 for equation (2.9). The result is

$$\frac{\partial}{\partial t} \langle \varpi_G \rangle_{\psi} = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \mathbf{J}_{gd} \cdot \nabla \psi + \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi + Zen_i \mathbf{V}_{iC} \cdot \nabla \psi - \frac{c}{B} (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{iG} - \mathbf{F}_{iC}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) \right\rangle_{\psi}.$$
(4.60)

The term $\mathbf{J}_{gd} \cdot \nabla \psi$ can be manipulated in the same way as the term \mathbf{J}_d in equation (2.14) to give $\langle \mathbf{J}_{gd} \cdot \nabla \psi \rangle_{\psi} = -\langle (cI/B) [\hat{\mathbf{b}} \cdot \nabla p_{g||} + (p_{g||} - p_{g\perp}) \nabla \cdot \hat{\mathbf{b}}] \rangle_{\psi}$. Employing the parallel momentum equation for ions, given by (4.44), and electrons, given by (4.37), to write

$$\hat{\mathbf{b}} \cdot \nabla p_{g||} + (p_{g||} - p_{g\perp}) \nabla \cdot \hat{\mathbf{b}} = -\frac{\partial}{\partial t} (n_i M V_{i||}) - \nabla \cdot \boldsymbol{\pi}_{ig||} + \tilde{F}_{iE} + \mathbf{F}_{iC} \cdot \hat{\mathbf{b}}, \quad (4.61)$$

I find

$$\frac{\partial}{\partial t} \left(\langle \varpi_G \rangle_{\psi} - \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{ZeI}{\Omega_i} n_i V_{i||} \right\rangle_{\psi} \right) = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{cI}{B} (\nabla \cdot \boldsymbol{\pi}_{ig||} - \tilde{F}_{iE}) + \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi - \frac{c}{B} (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) + Zen_i \mathbf{V}_{iC} \cdot \nabla \psi - cR \mathbf{F}_{iC} \cdot \hat{\boldsymbol{\zeta}} \right\rangle_{\psi}, \quad (4.62)$$

where I have employed $(I/B)\mathbf{F}_{iC}\cdot\hat{\mathbf{b}} - B^{-1}\mathbf{F}_{iC}\cdot(\hat{\mathbf{b}}\times\nabla\psi) = R\mathbf{F}_{iC}\cdot\hat{\boldsymbol{\zeta}}$ [recall (2.16)].

Taking the limit $k_{\perp}\rho_i \ll 1$, for which $\varpi_G \to \nabla \cdot [(Zen_i/\Omega_i)\mathbf{V}_i \times \hat{\mathbf{b}}]$, equation (4.62) can be shown to give

$$\frac{\partial}{\partial t} \langle Rn_i M \mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \langle R \hat{\boldsymbol{\zeta}} \cdot \overleftrightarrow{\boldsymbol{\pi}}_i^{(0)} \cdot \nabla \psi \rangle_{\psi}, \qquad (4.63)$$

where I have integrated once in ψ . The details of the calculation are in Appendix J. The zeroth order off-diagonal viscosity is given by

$$\langle R\hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\boldsymbol{\pi}}_{i}^{(0)} \cdot \nabla \psi \rangle_{\psi} = \left\langle \int d^{3}v \, f_{i} RM(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) \left(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_{1} - \frac{c}{B} \nabla \phi \times \hat{\mathbf{b}} \right) \cdot \nabla \psi \right\rangle_{\psi}.$$
 (4.64)

The distribution function $f_i = f_{ig} - (Ze\tilde{\phi}/T_i)f_{Mi}$ has both the adiabatic and the non-adiabatic pieces. The viscosity in (4.64) includes the nonlinear Reynolds stress, describing the $E \times B$ transport of toroidal angular momentum, and the transport due to the magnetic drifts \mathbf{v}_{M0} and finite gyroradius effects $\tilde{\mathbf{v}}_1$. In the absence of collisions, only the Reynolds stress gives a non-vanishing contribution as the other terms correspond to the gyroviscosity. In section K.1 of Appendix K, I prove that

$$\left\langle \int d^3 v \, f_i R M(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) (\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \nabla \psi \right\rangle_{\psi} = \frac{\partial}{\partial t} \left\langle \frac{1}{2B\Omega_i} (|\nabla \psi|^2 p_{i\perp} + I^2 p_{i\parallel}) \right\rangle_{\psi} - \left\langle \frac{M}{2B\Omega_i} \int d^3 v \, C\{f_i\} \left(|\nabla \psi|^2 \frac{v_\perp^2}{2} + I^2 v_{\parallel}^2 \right) \right\rangle_{\psi}. \quad (4.65)$$

It becomes apparent that when statistical equilibrium is reached and the net radial transport of energy is slow so that $\partial/\partial t \simeq 0$, the magnetic drifts only provide momentum transport proportional to the collision frequency. Since collisions are usually weak, this term will tend to be small. Moreover, in section K.2 of Appendix K, I show that this collisional piece vanishes exactly in up-down symmetric tokamaks, leaving only the Reynolds stress,

$$\langle R\hat{\boldsymbol{\zeta}}\cdot\stackrel{\leftrightarrow}{\boldsymbol{\pi}}_{i}^{(0)}\cdot\nabla\psi\rangle_{\psi}\simeq -\left\langle\int d^{3}v\,f_{i}RM(\mathbf{v}\cdot\hat{\boldsymbol{\zeta}})\frac{c}{B}(\nabla\phi\times\hat{\mathbf{b}})\cdot\nabla\psi\right\rangle_{\psi}.$$
(4.66)

In any case, the zeroth order viscosity is of order $\delta_i^2 p_i$, and the corresponding piece in the vorticity equation (4.53) is of order $\delta_i (k_\perp \rho_i)^2 e n_e v_i / L$, which becomes of order $\delta_i^3 e n_e v_i / L$ as $k_\perp \to 1/L$.

An important conclusion that can be derived from equation (4.63) is that the rate of change of the toroidal velocity is $\partial/\partial t \sim k_{\perp} \rho_i v_i/L$, becoming slower and slower as we reach longer wavelengths. This behavior must be reproduced by any equation used to calculate the radial electric field. Equations (4.45) and (4.53) satisfy this condition, but in addition they have the advantage of showing this property explicitly. The terms that determine the radial electric field turn out to be $\tilde{\mathbf{J}}_i \cdot \nabla \psi$, $Zen_i \tilde{\mathbf{V}}_i \cdot \nabla \psi$ and $Zen_i \mathbf{V}_{iC} \cdot \nabla \psi$ in equation (4.45), and $\tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi$, $(c/B)[\hat{\mathbf{b}} \times (\nabla \cdot \overleftrightarrow{\boldsymbol{\pi}}_{iG})]$)] · $\nabla \psi$, $Zen_i \mathbf{V}_{iC} \cdot \nabla \psi$ and $(c/B)(\hat{\mathbf{b}} \times \mathbf{F}_{iC}) \cdot \nabla \psi$ in equation (4.53). Additionally, it is necessary to keep the terms $\nabla \cdot \boldsymbol{\pi}_{ig||}$, \tilde{F}_{iE} and $\mathbf{F}_{iC} \cdot \hat{\mathbf{b}}$ in the parallel momentum equation (4.44). Any simulation must make sure that these terms have the correct behavior at long wavelengths and give equation (4.63). In the traditional gyrokinetic approach, the terms $\tilde{\mathbf{J}}_i \cdot \nabla \psi$, $Zen_i \tilde{\mathbf{V}}_i \cdot \nabla \psi$ and $Zen_i \mathbf{V}_{iC} \cdot \nabla \psi$ of equation (4.45) can be tracked back to terms in the gyrokinetic Fokker-Planck equation. They correspond to the difference between the ion and electron gyroaveraged $E \times B$ flows, $(c/B)(f_i \nabla \phi \times$ $\hat{\mathbf{b}} - f_{ig} \nabla_{\mathbf{R}} \langle \phi \rangle \times \hat{\mathbf{b}}$), and the finite gyroradius effects that make $\hat{\mathbf{b}}(\mathbf{R}_g) \neq \hat{\mathbf{b}}(\mathbf{r}), \nabla_{\mathbf{R}_g} \neq \overline{\nabla}$ and $\langle C\{f_i\}\rangle \neq C\{f_i\}$. This identification is the advantage of equation (4.45) since it allows easier analysis of existing simulations. Equation (4.45) can be used to check if the simulations reproduce the correct transport of toroidal angular momentum.

Since the vorticity equations (4.45) and (4.53) give equation (4.63) for $k_{\perp}\rho_i \ll 1$, it may seem that they provide the correct radial electric field at long wavelengths. Moreover, I have deduced these vorticity equations employing only the gyrokinetic Fokker-Planck equation and the corresponding quasineutrality, making it tempting to argue that the traditional gyrokinetic method is good enough to find the radial electric field. This argument is flawed because there are missing terms of order $\delta_i^2 e n_e v_i/L$ in equations (4.45) and (4.53). Then, the transport of toroidal angular momentum (4.63), that corresponds to a term of order $\delta_i(k_{\perp}\rho_i)^2 e n_e v_i/L$, will remain correct only if $(k_{\perp}\rho_i)^2 \gg \delta_i$. Consequently, the gyrokinetic quasineutrality should provide the correct radial electric field up to wavelengths of order $\sqrt{\rho_i L}$. For longer wavelengths, there will be missing terms. This estimate only considers the terms that the gyroki-



Figure 4-4: Geometry of the θ -pinch.

netic equation is missing and neglects possible numerical inaccuracies.

In the next section, I will show with a simplified example that gyrokinetic indeed has problems determining the radial electric field in axisymmetric configurations. This example is intended to illustrate the difficulties that arise from the use of gyrokinetic Fokker-Planck equation together with the gyrokinetic quasineutrality equation (3.54).

4.5 Example: quasineutrality in a θ -pinch

In this section, I try to find the solution to the non-turbulent, axisymmetric θ -pinch. Without turbulence, the perpendicular wavelengths are of the order of the characteristic size of the θ -pinch, $k_{\perp}L \sim 1$. This section is intended only as an example, and neglecting the turbulence greatly simplifies the calculation without fundamentally changing the properties of quasineutrality. In this simplified problem, I find that current gyrokinetic treatments, even if extended to a higher order in $\delta_i = \rho_i/L \ll 1$ than in chapter 3, do not yield a solution for the long wavelength radial electric field, leaving it as a free parameter. The gyrokinetic Fokker-Planck equation and the quasineutrality equation are intrinsically ambipolar and cannot determine the radial electric field. In chapter 5, I will show that the radial electric field is recovered if a different approach is employed.

In the θ -pinch, the magnetic field is given by $\mathbf{B} = B(r)\hat{\mathbf{b}}$, where here $\hat{\mathbf{b}}$ is a constant unit vector in the axial direction, and r is the radial coordinate in cylindrical geometry.

The geometry is sketched in figure 4-4. For long wavelengths, the gyrokinetic equation can be found to order $\delta_i^2 f_{Mi} v_i / L$. The simplified geometry of the magnetic field yields more manageable expressions for the gyrokinetic variables, i.e., μ_1 and \mathbf{R}_2 become

$$\mu_1 = \frac{Ze\tilde{\phi}}{MB} - \frac{v_\perp^2}{2B^2\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B \tag{4.67}$$

and

$$\mathbf{R}_{2} = \frac{1}{4B\Omega_{i}^{2}} [\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})] \cdot \nabla B, \qquad (4.68)$$

where the term $(c/B\Omega_i)\nabla_{\mathbf{R}}\tilde{\Phi} \times \hat{\mathbf{b}} \sim \delta_i^2 k_{\perp} \rho_i L$ has been neglected because I assume that $k_{\perp}L \sim 1$. Using \mathbf{R}_2 , the gyroaverage of $\dot{\mathbf{R}}$ is calculated to be

$$\langle \dot{\mathbf{R}} \rangle \simeq \langle v_{||} \rangle \hat{\mathbf{b}} + \left\langle \frac{\mu_0}{\Omega_i} \hat{\mathbf{b}} \times \nabla B - \frac{c}{B} \nabla \phi \times \hat{\mathbf{b}} + \mathbf{v}_\perp \cdot \nabla \mathbf{R}_2 - \frac{Ze}{M} \nabla \phi \cdot \nabla_v \mathbf{R}_2 \right\rangle, \quad (4.69)$$

where I have used that in a θ -pinch $\hat{\mathbf{b}} \cdot \nabla B = 0$ to write $\mathbf{v} \cdot \nabla \mathbf{R}_2 = \mathbf{v}_{\perp} \cdot \nabla \mathbf{R}_2$. The gyroaverages are performed by employing the long wavelength approximation $\nabla \phi \simeq \nabla_{\mathbf{R}} \phi - \Omega_i^{-1} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \phi$ and the relation $\langle \mathbf{v}_{\perp} \mathbf{v}_{\perp} \mathbf{v}_{\perp} \rangle = 0$ to get

$$\langle \dot{\mathbf{R}} \rangle \simeq \langle v_{||} \rangle \dot{\mathbf{b}} + \frac{\mu}{\Omega_i(\mathbf{R})} \dot{\mathbf{b}} \times \nabla_{\mathbf{R}} B - \frac{c}{B(\mathbf{R})} \nabla_{\mathbf{R}} \phi \times \dot{\mathbf{b}}.$$
 (4.70)

The gyroaverage of \dot{E} is found by using (A.17) to write

$$\dot{E} = -\frac{Ze}{M} \left(\mathbf{v} \cdot \nabla \phi - \frac{d\tilde{\phi}}{dt} \right) \simeq -\frac{Ze}{M} \left(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + \dot{\mu} \frac{\partial \langle \phi \rangle}{\partial \mu} - \frac{\partial \tilde{\phi}}{\partial t} \right), \tag{4.71}$$

where I employ that $\partial/\partial t \sim \delta_i v_i/L$ for long wavelengths and $\partial \phi/\partial E = O(\delta_i^3 M/e)$ in the θ -pinch. Considering that $\langle \dot{\mu} \rangle = O(\delta_i^2 v_i^3/BL)$ and $\partial \langle \phi \rangle / \partial \mu = O(\delta_i MB/e)$, the gyroaverage of (4.71) is calculated to be

$$\langle \dot{E} \rangle \simeq -\frac{Ze}{M} \langle \dot{\mathbf{R}} \rangle \cdot \nabla_{\mathbf{R}} \langle \phi \rangle.$$
 (4.72)

Thus, the gyrokinetic equation to order $O(\delta_i^2 f_{Mi} v_i/L)$ is

$$\frac{\partial f_i}{\partial t} + \langle \dot{\mathbf{R}} \rangle \cdot \left(\nabla_{\mathbf{R}} f_i - \frac{Ze}{M} \nabla_{\mathbf{R}} \langle \phi \rangle \frac{\partial f_i}{\partial E} \right) = \langle C\{f_i\} \rangle.$$
(4.73)

I have neglected the derivative $\partial f_i / \partial \mu$ because the distribution function is Maxwellian to zeroth order and $\langle \dot{\mu} \rangle$ is already small by definition of μ . For an axisymmetric steady state solution, the terms on the left side of (4.73) vanish, the second term because the gyrocenter parallel and perpendicular drifts, $\langle \dot{\mathbf{R}} \rangle$, remain in surfaces of constant f_i and ϕ (for this reason, $\langle v_{||} \rangle$ need not be evaluated to second order). Thus, equation (4.73) becomes $\langle C\{f_i\} \rangle = 0$. Such an equation can be solved for a simplified collision operator. I use a Krook operator, $C\{f_i\} = -\nu(f_i - f_M)$, with constant collision frequency ν and a shifted Maxwellian,

$$f_M = n_i \left(\frac{M}{2\pi T_i}\right)^{3/2} \exp\left[-\frac{M(\mathbf{v} - \mathbf{V}_i)^2}{2T_i}\right],\tag{4.74}$$

where n_i , T_i and \mathbf{V}_i are functions of the position \mathbf{r} . I assume that the parallel average velocity, $V_{i||} = \hat{\mathbf{b}} \cdot \mathbf{V}_i$, is zero and I order \mathbf{V}_i as $O(\delta_i v_i)$ to obtain

$$f_M \simeq f_{M0} \left[1 + \frac{M \mathbf{v}_{\perp} \cdot \mathbf{V}_i}{T_i} + \frac{M^2 (\mathbf{v}_{\perp} \cdot \mathbf{V}_i)^2}{2T_i^2} - \frac{M V_i^2}{2T_i} \right],$$
(4.75)

with

$$f_{M0} = n_i \left(\frac{M}{2\pi T_i}\right)^{3/2} \exp\left(-\frac{Mv^2}{2T_i}\right).$$
(4.76)

With the Krook operator, the gyrokinetic solution is

$$f_i = \langle f_M \rangle = \langle f_{M0} \rangle + \left\langle \frac{M \mathbf{v}_\perp \cdot \mathbf{V}_i}{T_i} f_{M0} \right\rangle + \frac{M^2 v_\perp^2 V_i^2}{4T_i^2} F_M - \frac{M V_i^2}{2T_i} F_M, \qquad (4.77)$$

where I have used that in the higher order terms, $f_{M0} \simeq F_M$, with

$$F_M = n_i(\mathbf{R}) \left[\frac{M}{2\pi T_i(\mathbf{R})} \right]^{3/2} \exp\left[-\frac{ME}{T_i(\mathbf{R})} \right].$$
(4.78)

For the first two terms in equation (4.77), it is necessary to Taylor expand $f_{M0}(\mathbf{r}, E_0)$,

 $\mathbf{V}_i(\mathbf{r})$ and $\mathbf{v}_{\perp} = \sqrt{2\mu_0 B(\mathbf{r})} [\hat{\mathbf{e}}_1(\mathbf{r}) \cos \varphi_0 + \hat{\mathbf{e}}_2(\mathbf{r}) \sin \varphi_0]$ around \mathbf{R} , E, μ and φ . The final result is

$$\begin{aligned} f_{i} &= \langle f_{M} \rangle = F_{M} \left[1 - \frac{x_{\perp}^{2}}{n_{i}} \nabla \cdot \left(\frac{cn_{i}}{B\Omega_{i}} \nabla_{\perp} \phi \right) + \frac{Mc^{2}}{2T_{i}B^{2}} (2 - x_{\perp}^{2}) |\nabla_{\perp} \phi|^{2} + \frac{MV_{i}^{2}}{2T} (x_{\perp}^{2} - 1) \right. \\ &+ \frac{x_{\perp}^{2}}{2M\Omega_{i}^{2}} \left(x^{2} - \frac{5}{2} \right) \nabla_{\perp}^{2} T_{i} - \frac{x_{\perp}^{2}}{2n_{i}M\Omega_{i}^{2}} \nabla_{\perp}^{2} p_{i} + \frac{x_{\perp}^{2}}{2T_{i}M\Omega_{i}^{2}} \left(\frac{35}{4} - 7x^{2} + x^{4} \right) |\nabla_{\perp} T_{i}|^{2} \\ &+ \frac{2x_{\perp}^{2}}{MB\Omega_{i}^{2}} \left(\frac{5}{2} - x^{2} \right) \nabla_{\perp} B \cdot \nabla_{\perp} T_{i} + \frac{c}{T_{i}B\Omega_{i}} \nabla_{\perp} \phi \cdot \nabla_{\perp} T_{i} \left(\frac{5}{2} - x_{\perp}^{2} - x^{2} \right) \right], \end{aligned}$$

$$(4.79)$$

where $x^2 = Mv^2/2T_i \simeq ME/T_i$, $x_{\perp}^2 = Mv_{\perp}^2/2T_i \simeq M\mu B/T_i$ and I have employed

$$\mathbf{V}_{i} = \frac{1}{n_{i}M\Omega_{i}}\hat{\mathbf{b}} \times \nabla p_{i} - \frac{c}{B}\nabla \phi \times \hat{\mathbf{b}}.$$
(4.80)

The distribution function in (4.79) has been calculated by using a gyrokinetic equation that is correct to order $\delta_i^2 f_{Mi} v_i / L$ for both the Vlasov operator and the gyroaveraged collision operator. Using the definitions $\mathbf{R} = \mathbf{r} + \mathbf{R}_1 + \mathbf{R}_2$, $E = E_0 + E_1 + E_2$ and $\mu = \mu_0 + \mu_1$, and the gyrophase dependent collisional piece \tilde{f}_i given in (3.38), I can find the distribution function f_i in \mathbf{r} , \mathbf{v} variables to order $O(\delta_i^2 f_{Mi})$. As a check, the same solution has been also obtained without resorting to gyrokinetics to order $O(\delta_i^2 f_{Mi})$. This check is omitted here.

If I had gyroaveraged $C\{f_i\}$ only to order δ_i , as most gyrokinetic models do, the solution would have been simply

$$f_i \simeq F_M. \tag{4.81}$$

Substituting this solution into the higher order gyrokinetic quasineutrality equation (3.61), I find the inconsistent result

$$\nabla \cdot \left(\frac{Zcn_i}{B\Omega_i}\nabla_{\perp}\phi\right) - \frac{ZMc^2n_i}{2T_iB^2}|\nabla_{\perp}\phi|^2 = n_e - Zn_i - \frac{Z}{2M\Omega_i^2}\nabla_{\perp}^2p_i.$$
 (4.82)

However, this quasineutrality equation is very different from the one we obtain by

using the full $O(\delta_i^2 f_{Mi})$ solution in (4.79), which simply gives

$$Zn_i = n_e. (4.83)$$

Therefore, the gyrokinetic quasineutrality equation reduces to the quasineutrality condition when the exact $O(\delta_i^2 f_{Mi})$ distribution function of (4.79) is employed. Equation (4.82) is wrong because the $O(\delta_i f_{Mi})$ result of (4.81) is either inducing an $O(\delta_i^2 e n_e)$ charge difference or imposing the non-physical condition

$$\nabla \cdot \left(\frac{Zcn_i}{B\Omega_i}\nabla_{\perp}\phi\right) - \frac{ZMc^2n_i}{2T_iB^2}|\nabla_{\perp}\phi|^2 = -\frac{Z}{2M\Omega_i^2}\nabla_{\perp}^2p_i.$$
(4.84)

The difference between (4.82) and (4.83), given by (4.84), originates in $O(\delta_i^2 n_e)$ terms that should have been cancelled by pieces of the distribution function of the same order.

The θ -pinch example illustrates the problem of using a lower order gyrokinetic equation than needed, but it also highlights another issue. The potential does not appear in the quasineutrality equation (4.83), and, therefore, it cannot be found using it. In a computer simulation, the potential is obtained from the gyrokinetic quasineutrality equation (3.61), and the distribution function is evolved employing the gyrokinetic equation (4.73). A possible initial condition $f_{i,t=0}(\mathbf{R}, E, \mu)$ is the stationary solution (4.79), where the potential $\phi(r)$ is a free function that is set to be $\phi_{t=0}(r)$. Equation (4.83) proves that the solution to the gyrokinetic quasineutrality equation (3.61) at t = 0 must be $\phi(r, t = 0) = \phi_{t=0}(r)$, where $\phi_{t=0}(r)$ is the free function that I chose for the initial condition. Since the initial condition $f_{i,t=0}(\mathbf{R}, E, \mu)$ is a stationary solution, I find that the solution for all times is $f_i = f_{i,t=0}$ and $\phi = \phi_{t=0}$, and the radial electric field is solely determined by the initial condition. If there were any numerical errors that made the solution invalid to order $\delta_i^2 f_{Mi}$, the radial electric field would suffer a non-physical evolution. In modern gyrokinetics, the radial electric field is then determined by the initial condition, in particular, by a piece of order $\delta_i^2 f_{Mi}$ of the initial condition. This result is not surprising since in this chapter I have proved that, in axisymmetric configurations, the axisymmetric piece of the vorticity equation, or time derivative of $Zn_i - n_e$, is of order $\delta_i^3 n_e v_i/L$ at the most, as given by (4.9). The gyrokinetic equation is only calculated to order $\delta_i^2 f_{Mi} v_i/L$ in this section (and only to order $\delta_i f_{Mi} v_i/L$ in codes), leading to an effectively constant $Zn_i - n_e$ and hence a radial electric field dependent only on the initial condition. In section 5.2, I will show that in reality the time derivative of $Zn_i - n_e$ is even smaller than $\delta_i^3 n_e v_i/L$ by a factor of ν/Ω_i .

4.6 Discussion

In this chapter, I have shown how the vorticity equation recovers the physics of quasineutrality and at the same time retains the effect of the transport of toroidal angular momentum in the radial electric field. I have proposed two possible vorticity equations, (4.45) and (4.53). With these two equations, I estimate the size of the term that determines the radial electric field, given by equation (4.9). In this manner, I prove that, with the usual gyrokinetic equation, setting the radial current to zero, $\langle \mathbf{J} \cdot \nabla \psi \rangle_{\psi} = 0$, cannot determine the long wavelength axisymmetric radial electric field. Therefore, modern gyrokinetic formulations are intrinsically ambipolar, and thereby unable to determine the long wavelength axisymmetric radial electric field.

I illustrated the problems that arise from a failure to satisfy intrinsic ambipolarity with a simplified problem in section 4.5. In the example, the long wavelength, axisymmetric radial electric field was left undetermined by the gyrokinetic quasineutrality equation even if the distribution was calculated to an order higher than current codes can achieve. More importantly, if there is an error in the density as small as $\delta_i^2 n_e$, the gyrokinetic quasineutrality equation yields an erroneous long wavelength radial electric field. This feature places a strong requirement on the accuracy of any code that calculates the radial electric field. The vorticity equation, on the other hand, makes the dependence of the radial electric field on the toroidal transport of angular momentum explicit. Both vorticity equations (4.45) and (4.53) would yield a long wavelength radial electric field constant for the short turbulence saturation time.

Between the two vorticity equations, equation (4.45) is closer to the gyrokinetic

quasineutrality and is probably the best candidate to implement and compare with existing results. In fact, a similar, but less complete, vorticity equation has already been implemented in the PIC gyrokinetic code GEM [63]. On the other hand, equation (4.53) is similar to the traditional vorticity equation (2.9), making the study of conservation of toroidal angular momentum straightforward.

These vorticity equations are valid for short wavelengths on the order of the ion gyroradius. They must be supplemented with long wavelength physics to be extended to wavelengths longer than $\sqrt{\rho_i L}$. Only then will the transport of toroidal angular momentum be correctly described. The extension to longer wavelengths is treated in chapter 5.

Finally, any numerical implementation of either of the vorticity equations needs to make sure that the properties derived and discussed are satisfied, namely, the scaling of the different terms with $k_{\perp}\rho_i$ should be ensured, and the cancellations that take place due to the flux surface average should also be maintained in codes. It is for this reason that I give all the details of the analytical calculations including detailed appendices.

Chapter 5

Solving for the radial electric field

In this chapter, I describe the method that I propose to calculate the radial electric field. As I showed in section 2.3, the flux surface averaged vorticity equation reduces to the transport of toroidal angular momentum. Then, to obtain the radial electric field, it is necessary to solve the conservation equation for the transport of toroidal angular momentum.

In section 5.1, I obtain an equation for the toroidal-radial component of the ion viscosity to order $\delta_i^3 p_i$ that only requires the ion distribution function to order $\delta_i^2 f_{Mi}$. I already argued in section 2.3 that $\delta_i^3 p_i$ is the order to which the ion viscosity should be found to recover gyroBohm transport of angular momentum. In this chapter, I give some arguments that suggest that the radial transport of angular momentum is indeed of order $\delta_i^3 p_i$. Interestingly, the transport of toroidal angular momentum found in (4.64), of order $\delta_i k_\perp \rho_i p_i$, should then vanish for long wavelengths. In reality, the transport probably is of order $\delta_i^2 p_i$ at each time step yet its time average vanish to that order. In other words, there might be fast local exchange of toroidal angular momentum, leading to zonal flow structure, but the irreversible transport of angular momentum from the edge to the core is much slower. In section 5.2, I apply the equation for the ion viscosity obtained in section 5.1 to solve for the radial electric field in the example presented in section 4.5.

Even with the convenient equation that gives the radial transport of toroidal angular momentum to order $\delta_i^3 p_i$ with only an $O(\delta_i^2 f_{Mi})$ distribution function, this still makes it necessary to find the ion distribution function and the potential to an order higher than the order to which they are usually calculated. At the end of this chapter, in section 5.3, I prove that under certain assumptions, the lowest order full f gyrokinetic equation (3.39) is valid even to calculate the second order distribution function, and the vorticity equations (4.45) and (4.53) are easily extended to yield a higher order potential. Finally, I discuss all the results of this chapter in section 5.4.

5.1 Ion viscosity and the axisymmetric potential

The evolution of the toroidal velocity, given in (2.21), is determined by the flux surface averaged toroidal-radial component of the ion viscosity $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi}$. According to the gyroBohm estimates at the end of section 2.3, the ion viscosity has to be obtained to order $\delta_i^3 p_i$. If the ion viscosity is to be determined directly from its definition in (2.6), the ion distribution function must be calculated to order $\delta_i^3 f_{Mi}$, too high of an order to be practical or implementable.

To avoid direct evaluation of the ion viscosity, I propose using moments of the Fokker-Planck equation. This is the approach followed in drift kinetics [64] and to formulate a hybrid gyrokinetic-fluid description [36]. The ion viscosity can be solved from the Mvv moment of the Fokker-Planck equation, given by

$$\Omega_i(\vec{\pi}_i \times \hat{\mathbf{b}} - \hat{\mathbf{b}} \times \vec{\pi}_i) = \vec{\mathbf{K}},\tag{5.1}$$

with

$$\vec{\mathbf{K}} = \frac{\partial \vec{\mathbf{P}}_i}{\partial t} + \nabla \cdot \left(M \int d^3 v \, f_i \mathbf{v} \mathbf{v} \mathbf{v} \right) + Zen_i (\nabla \phi \mathbf{V}_i + \mathbf{V}_i \nabla \phi) - M \int d^3 v \, C\{f_i\} \mathbf{v} \mathbf{v}. \tag{5.2}$$

Here, $\vec{\mathbf{P}}_i = M \int d^3 v f_i \mathbf{v} \mathbf{v}$. From the moment equation (5.1), the off diagonal elements of $\vec{\pi}_i$ can be solved for as a function of $\vec{\mathbf{K}}$. Additionally, equation (5.1) contains the energy conservation equation, $\operatorname{Trace}(\vec{\mathbf{K}}) = 0$, and the parallel pressure equation, $\hat{\mathbf{b}} \cdot \vec{\mathbf{K}} \cdot \hat{\mathbf{b}} = 0$.

To solve for the toroidal-radial component $R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\dot{\pi}}_i \cdot \nabla \psi$, I pre-multiply and post-

multiply equation (5.1) by $R\hat{\boldsymbol{\zeta}}$, giving

$$R\hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_{i} \cdot \nabla \psi = \frac{Mc}{2Ze} \frac{\partial}{\partial t} (R^{2} \hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\mathbf{P}}_{i} \cdot \hat{\boldsymbol{\zeta}}) + \frac{M^{2}c}{2Ze} \nabla \cdot \left[\int d^{3}v \, \mathbf{v} f_{i} R^{2} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2} \right] + Mcn_{i} R^{2} (\hat{\boldsymbol{\zeta}} \cdot \nabla \phi) (\mathbf{V}_{i} \cdot \hat{\boldsymbol{\zeta}}) - \frac{M^{2}c}{2Ze} \int d^{3}v \, C\{f_{i}\} R^{2} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2}, \qquad (5.3)$$

where I use $R(\hat{\mathbf{b}} \times \hat{\boldsymbol{\zeta}}) = \nabla \psi / B$ and $\nabla (R\hat{\boldsymbol{\zeta}}) = (\nabla R)\hat{\boldsymbol{\zeta}} - \hat{\boldsymbol{\zeta}}(\nabla R)$. Flux surface averaging this expression gives

$$\langle R\hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_{i} \cdot \nabla \psi \rangle_{\psi} = \frac{M^{2}c}{2Ze} \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \int d^{3}v \left(f_{i} - \overline{f}_{i}\right) (\mathbf{v} \cdot \nabla \psi) R^{2} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2} \right\rangle_{\psi} + \langle Mcn_{i}R^{2} (\hat{\boldsymbol{\zeta}} \cdot \nabla \phi) (\mathbf{V}_{i} \cdot \hat{\boldsymbol{\zeta}}) \rangle_{\psi} + \frac{Mc}{2Ze} \frac{\partial}{\partial t} \langle R^{2} \hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\mathbf{P}}_{i} \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} - \frac{M^{2}c}{2Ze} \left\langle \int d^{3}v C\{f_{i}\} R^{2} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2} \right\rangle_{\psi}.$$
(5.4)

In the first term of the right side, I use that $(\mathbf{v} \cdot \nabla \psi)(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^2 = 0$ to write the integral only as a function of the gyrophase dependent piece of the distribution function. Equation (5.4) has the advantage that a distribution function correct to order $\delta_i^2 f_{Mi}$ gives a viscosity good to order $\delta_i^3 p_i$! The method by which we have gained an order in δ_i is similar to the calculation of the perpendicular ion flow employing the momentum equation. To evaluate the perpendicular ion flow $n_i \mathbf{V}_{i\perp} = \int d^3 v f_i \mathbf{v}_{\perp}$ to order $\delta_i n_e v_i$ by direct integration over velocity space, the distribution function f_i must be correct to order $\delta_i f_{Mi}$. Instead, it is possible to use the ion perpendicular momentum equation to order p_i/L , where the Lorentz force $(Ze/c)n_i\mathbf{V}_i \times \mathbf{B}$ balances the perpendicular pressure gradient $\nabla_{\perp} p_i$ and the perpendicular electric field $Zen_i\nabla_{\perp}\phi$, giving the ion flow $n_i\mathbf{V}_{i\perp} = (c/ZeB)\hat{\mathbf{b}} \times \nabla p_i - (cn_i/B)\nabla\phi \times \hat{\mathbf{b}}$. Notice that only the lowest order Maxwellian f_{Mi} has been used to find p_i . We have gained an order in δ_i .

If the gyroBohm estimates done at the end of section 2.3 are correct, the ion viscosity must identically vanish to order $\delta_i^2 p_i$ without determining the evolution of the long wavelength axisymmetric radial electric field on transport time scales. To obtain the toroidal-radial component of $\vec{\pi}_i$ to this order, it is enough to use a distribution function good to order $\delta_i f_{Mi}$ in (5.4). For long wavelengths and to the

order of interest, the gyrophase dependent piece of the distribution function is given by (4.7), i.e., it is proportional to $(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi$. Then, the first integral in (5.4) vanishes. Additionally, I find that for this gyrophase dependence, $\mathbf{P}_i \simeq p_{i\perp} (\mathbf{I} - \hat{\mathbf{b}} \hat{\mathbf{b}}) + p_{i\parallel} \hat{\mathbf{b}} \hat{\mathbf{b}}$ and $\int d^3 v \, \mathbf{v} \mathbf{v} C\{f_i\} \simeq \int d^3 v \, [(v_{\perp}^2/2)(\mathbf{I} - \hat{\mathbf{b}} \hat{\mathbf{b}}) + v_{\parallel}^2 \hat{\mathbf{b}} \hat{\mathbf{b}}] C\{f_i\}$. With all these simplifications, $\langle R \hat{\boldsymbol{\zeta}} \cdot \vec{\pi}_i \cdot \nabla \psi \rangle_{\psi}$ becomes to $O(\delta_i^2 p_i R |\nabla \psi|)$

$$\langle R\hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_{i} \cdot \nabla \psi \rangle_{\psi} \simeq - \left\langle n_{i}RM(\mathbf{V}_{i} \cdot \hat{\boldsymbol{\zeta}}) \frac{c}{B} (\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \psi \right\rangle_{\psi}$$

$$+ \frac{\partial}{\partial t} \left\langle \frac{1}{2B\Omega_{i}} (p_{i\perp} |\nabla \psi|^{2} + p_{i\parallel} I^{2}) \right\rangle_{\psi} - \left\langle \frac{M}{2B\Omega_{i}} \int d^{3}v \, C\{f_{i}\} \left(\frac{v_{\perp}^{2}}{2} |\nabla \psi|^{2} + v_{\parallel}^{2} I^{2} \right) \right\rangle_{\psi}, (5.5)$$

where I have used $\hat{\mathbf{b}} \cdot \nabla \phi \simeq 0$ and $R\hat{\boldsymbol{\zeta}} = I\hat{\mathbf{b}}/B - (\hat{\mathbf{b}} \times \nabla \psi)/B$ to write $cR\hat{\boldsymbol{\zeta}} \cdot \nabla \phi \simeq -(c/B)(\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \phi$. Equation (5.5) is exactly the transport of momentum found from the gyrokinetic vorticity equation, given in (4.64) and (4.65). For turbulence in statistical equilibrium, the time derivative term can be neglected. If in addition the tokamak is up-down symmetric, the collisional term vanishes as proven in section K.2 of Appendix K, leaving only the Reynolds stress

$$\langle R\hat{\boldsymbol{\zeta}} \cdot \vec{\boldsymbol{\pi}}_{i} \cdot \nabla \psi \rangle_{\psi} = - \left\langle n_{i} R M(\mathbf{V}_{i} \cdot \hat{\boldsymbol{\zeta}}) \frac{c}{B} (\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \psi \right\rangle_{\psi} = - \left\langle \int d^{3} v f_{i} R M(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) \frac{c}{B} (\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \psi \right\rangle_{\psi}.$$
 (5.6)

This Reynolds stress is formally of order $\delta_i^2 p_i R |\nabla \psi|$. If the Reynolds stress were this big, the transport of toroidal angular momentum would be much larger than the gyroBohm estimate. It is more plausible that the Reynolds stress averaged over time is almost zero. Therefore, the Reynolds stress to order $\delta_i^2 p_i$ does not determine the evolution of the long wavelength axisymmetric radial electric field on transport time scales. This possibility does not conflict with possible fast growth and evolution of zonal flow structure, that happens in relatively short times, but does not transport angular momentum through large distances.

It is difficult to prove unarguably that the Reynolds stress (5.6) must vanish to order $\delta_i^2 p_i$. In δf flux tube codes like GS2 [1] and GENE [2], only the gradients of density and temperature enter the equation for the correction to the Maxwellian [recall (3.57)]. The gradient of the velocity and hence of the long wavelength axisymmetric radial electric field is ordered out because the average velocity in the plasma is assumed to be small by δ_i . Then, the system does not have a preferred direction and it is unlikely that there is any transport of angular momentum. Quasilinear calculations suggest that in up-down symmetric tokamaks, δf flux tube formulations must give zero transport [65]. If the average velocity is ordered as large as the thermal velocity, the symmetry in the flux tube is broken and there is a net radial momentum transport [66], but such a description is not relevant in many tokamaks.

It seems more reasonable to assume that, at least in a time averaged sense, the Reynolds stress (5.6) becomes of order $\delta_i^3 p_i$. Therefore, from now on, I consider the fast time average of equation (5.4) to filter the fluctuations in the transport of toroidal angular momentum. Fast time here is an intermediate time between the transit time of the particle motion around the tokamak, L/v_i , and the much slower transport time scale, $\delta_i^{-2}L/v_i$. Since this time average should make the Reynolds stress of order $\delta_i^3 p_i$, the rest of the terms in equation (5.4) must be evaluated to order $\delta_i^2 p_i$. To that end, the ion distribution function and the potential must be known to order $\delta_i^2 f_{Mi}$ and order $\delta_i^2 T_e/e$; an order higher than solved for in gyrokinetic codes. In section 5.3, I will prove that this problem can be circumvented under some simplifying assumptions. The rest of this section is on how to evaluate the first term in (5.4) in a convenient way.

The first term in equation (5.4) only depends on the gyrophase dependent piece of the ion distribution function. For this reason, it can be solved by employing the moment **vvv** of the Fokker-Planck equation, given by

$$\Omega_{i} \int d^{3}v f_{i} M[(\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}\mathbf{v} + \mathbf{v}(\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v} + \mathbf{v}\mathbf{v}(\mathbf{v} \times \hat{\mathbf{b}})] = \frac{\partial}{\partial t} \left(\int d^{3}v f_{i} M \mathbf{v} \mathbf{v}\mathbf{v} \right) + \nabla \cdot \left(\int d^{3}v f_{i} M \mathbf{v} \mathbf{v} \mathbf{v} \right) + Ze \int d^{3}v f_{i} (\nabla \phi \mathbf{v}\mathbf{v} + \mathbf{v} \nabla \phi \mathbf{v} + \mathbf{v} \nabla \phi) - \int d^{3}v C\{f_{i}\} M \mathbf{v} \mathbf{v} \mathbf{v}.$$
(5.7)

Multiplying every index in this tensor by $R\hat{\boldsymbol{\zeta}}$, employing $R(\hat{\mathbf{b}} \times \hat{\boldsymbol{\zeta}}) = \nabla \psi / B$ and flux

surface averaging gives

$$\left\langle M \int d^3 v \left(f_i - \overline{f}_i \right) (\mathbf{v} \cdot \nabla \psi) R^2 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^2 \right\rangle_{\psi} = \frac{M^2 c}{3Z e} \frac{\partial}{\partial t} \left\langle \int d^3 v f_i R^3 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^3 \right\rangle_{\psi} + \frac{M^2 c}{3Z e} \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \int d^3 v f_i (\mathbf{v} \cdot \nabla \psi) R^3 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^3 \right\rangle_{\psi} + c \langle R^3 (\hat{\boldsymbol{\zeta}} \cdot \nabla \phi) (\hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\mathbf{P}}_i \cdot \hat{\boldsymbol{\zeta}}) \rangle_{\psi} - \frac{M^2 c}{3Z e} \left\langle \int d^3 v C \{f_i\} R^3 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^3 \right\rangle_{\psi}.$$
(5.8)

This equation has to be evaluated to order $\delta_i^2 p_i v_i R^2 |\nabla \psi|$ to give terms of order $\delta_i^3 p_i R |\nabla \psi|$ in (5.4). The first term in the right side is then negligible because it has a time derivative. With turbulence that has reached statistical equilibrium and after fast time averaging, the time derivative becomes of the order of the transport time scale at long wavelengths, i.e., $\partial/\partial t \sim D_{gB}/L^2 \sim \delta_i^2 v_i/L$. The contribution of such a time derivative is negligible since it gives a term of order $\delta_i^3 p_i v_i R^2 |\nabla \psi|$. The second term in equation (5.8) is also negligible since $(\mathbf{v} \cdot \nabla \psi)(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^3 = 0$ means that only the gyrophase dependent piece of the distribution function contributes. To the order of interest, the gyrophase dependent piece is given by (4.7), and its contribution vanishes. Then, the only terms left are

$$\left\langle M \int d^3 v \, (f_i - \overline{f}_i) (\mathbf{v} \cdot \nabla \psi) R^2 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^2 \right\rangle_{\psi} = c \langle R^3 (\hat{\boldsymbol{\zeta}} \cdot \nabla \phi) (\hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\mathbf{P}}_i \cdot \hat{\boldsymbol{\zeta}}) \rangle_{\psi} - \frac{M^2 c}{3Ze} \left\langle \int d^3 v \, C\{f_i\} R^3 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^3 \right\rangle_{\psi}.$$
 (5.9)

Substituting this relation into equation (5.4) gives

$$\langle R\hat{\boldsymbol{\zeta}} \cdot \overleftarrow{\boldsymbol{\pi}}_{i} \cdot \nabla \psi \rangle_{\psi} = \frac{Mc}{2Ze} \langle R^{2} \rangle_{\psi} \frac{\partial p_{i}}{\partial t} + \langle Mcn_{i}R^{2}(\hat{\boldsymbol{\zeta}} \cdot \nabla \phi)(\mathbf{V}_{i} \cdot \hat{\boldsymbol{\zeta}}) \rangle_{\psi} - \frac{M^{2}c}{2Ze} \left\langle \int d^{3}v C\{f_{i}\}R^{2}(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2} \right\rangle_{\psi} + \frac{Mc^{2}}{2Ze} \frac{1}{V'} \frac{\partial}{\partial \psi} V' \langle R^{3}(\hat{\boldsymbol{\zeta}} \cdot \nabla \phi)(\hat{\boldsymbol{\zeta}} \cdot \overrightarrow{\mathbf{P}}_{i} \cdot \hat{\boldsymbol{\zeta}}) \rangle_{\psi} - \frac{M^{3}c^{2}}{6Z^{2}e^{2}} \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \int d^{3}v C\{f_{i}\}R^{3}(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{3} \right\rangle_{\psi}.$$
(5.10)

In the time derivative term, I used that for statistical equilibrium and after fast time averaging, only the slow transport time scales are left. Then, the dominant term in $\partial(\hat{\boldsymbol{\zeta}}\cdot \vec{\mathbf{P}}_i \cdot \hat{\boldsymbol{\zeta}})/\partial t$ is $\partial p_i/\partial t$. This term is determined by the turbulent heat transport

and heating in the plasma. Of the rest of the terms in equation (5.10), the second and the third term require a distribution function good to order $\delta_i^2 f_{Mi}$, and a potential good to $\delta_i^2 T_e/e$. The last two terms only need the distribution function to order $\delta_i f_{Mi}$ and the potential to order $\delta_i T_e/e$.

In the next section, I apply the methodology suggested in this section to solve for the radial electric field in the θ -pinch problem presented in section 4.5. In section 5.3, I will explain how the distribution function can be found to second order in tokamak geometry. Notice that this piece of the distribution function is only needed for the irreversible transport of angular momentum since the transport of angular momentum to order $\delta_i^2 p_i$ is enough to capture the fast evolution of zonal flow.

5.2 Example: the solution of a θ -pinch

In section 4.5, I showed that, in the θ -pinch, the quasineutrality condition applied to the second order solution (4.79), valid to $O(\delta_i^2 f_{Mi})$, does not determine the radial electric field. However, I will show in this section that the electrostatic potential can be obtained from the conservation of azimuthal angular momentum, equivalent to the conservation of toroidal angular momentum in tokamaks since both momentums are in the direction of symmetry. The momentum equation has the advantage of showing how quasineutrality depends on the long wavelength axisymmetric potential without having to calculate the distribution function to higher order than $O(\delta_i^2 f_{Mi})$. The methodology I use here is presented for screw pinches and dipolar configurations in [67].

For a steady state solution, the azimuthal angular momentum must be conserved, giving

$$\frac{1}{r}\frac{\partial}{\partial r}(r^{2}\hat{\mathbf{r}}\cdot\overleftrightarrow{\boldsymbol{\pi}}_{i}\cdot\hat{\boldsymbol{\theta}}) = 0, \qquad (5.11)$$

where $\hat{\mathbf{r}}$ and $\hat{\boldsymbol{\theta}}$ are the unit vectors in the radial and azimuthal directions, with $\hat{\mathbf{r}} \times \hat{\boldsymbol{\theta}} = \hat{\mathbf{b}}$ [recall figure 4-4], and $\vec{\pi}_i$ is the ion viscosity, given by (2.6). In a case without sources or sinks of momentum, the final equation for the potential is $r^2 \hat{\mathbf{r}} \cdot \vec{\pi}_i \cdot \hat{\boldsymbol{\theta}} = 0$. Finding $r^2 \hat{\mathbf{r}} \cdot \vec{\pi}_i \cdot \hat{\boldsymbol{\theta}}$ directly from the distribution function requires a higher order solution than the one provided by the $O(\delta_i^2 f_{Mi} v_i/L)$ gyrokinetic equation (4.73) used so far. However, this problem can be circumvented by using the equivalent to equation (5.10) for θ -pinches, given by

$$r^{2}\hat{\mathbf{r}}\cdot\overrightarrow{\boldsymbol{\pi}}_{i}\cdot\hat{\boldsymbol{\theta}} = \frac{Mr^{2}}{2\Omega_{i}}\int d^{3}v \,C\{f_{i}\}(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{2} - \frac{M}{6r\Omega_{i}}\frac{\partial}{\partial r}\left[\frac{r^{3}}{\Omega_{i}}\int d^{3}v \,C\{f_{i}\}(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{3}\right].$$
 (5.12)

To obtain this equation, I have used that $\hat{\boldsymbol{\theta}} \cdot \nabla \phi = 0$ due to axisymmetry. In this particular case, this expression can be reduced to integrals of the gyrophase dependent piece of the distribution function, terms much simpler to obtain to order $\delta_i^2 f_{Mi}$. To see this, I use $\overline{(\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^2} = v_{\perp}^2/2$, $(\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^2 - \overline{(\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^2} = (1/2)\hat{\boldsymbol{\theta}} \cdot [\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})] \cdot \hat{\boldsymbol{\theta}} = (1/2)[(\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^2 - (\mathbf{v} \cdot \hat{\mathbf{r}})^2]$ and $\overline{(\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^3} = 0$ to write

$$r^{2}\hat{\mathbf{r}}\cdot\overrightarrow{\boldsymbol{\pi}}_{i}\cdot\hat{\boldsymbol{\theta}} = -\frac{Mr^{2}}{4\Omega_{i}}\int d^{3}v \left(C\{f_{i}\}-\overline{C\{f_{i}\}}\right)\left[(\mathbf{v}\cdot\hat{\mathbf{r}})^{2}-(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{2}\right]$$
$$+\frac{Mr^{2}}{2\Omega_{i}}\int d^{3}v C\{f_{i}\}\frac{v_{\perp}^{2}}{2}-\frac{M}{6r\Omega_{i}}\frac{\partial}{\partial r}\left[\frac{r^{3}}{\Omega_{i}}\int d^{3}v \left(C\{f_{i}\}-\overline{C\{f_{i}\}}\right)(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{3}\right].$$
 (5.13)

For the Krook operator $C\{f_i\} = -\nu(f_i - f_M)$, with f_M given in (4.75),

$$C\{f_i\} - \overline{C}\{f_i\} = -\nu \left\{ f_i - \overline{f}_i - \frac{M\mathbf{v}_{\perp} \cdot \mathbf{V}_i}{T_i} f_{M0} - \frac{M^2}{2T_i^2} (\mathbf{v}_{\perp}\mathbf{v}_{\perp} - \overline{\mathbf{v}_{\perp}\mathbf{v}_{\perp}}) : (\mathbf{V}_i\mathbf{V}_i)f_{M0} \right\},$$
(5.14)

where, according to (4.80),

$$\mathbf{V}_{i} = \frac{c}{B} \hat{\boldsymbol{\theta}} \left(\frac{1}{Zen_{i}} \frac{\partial p_{i}}{\partial r} + \frac{\partial \phi}{\partial r} \right).$$
(5.15)

The term $\int d^3v C\{f_i\}(Mv_{\perp}^2/2)$ in (5.13) can be found from the equation for the perpendicular pressure, given by

$$\frac{\partial p_{i\perp}}{\partial t} + \frac{1}{r} \frac{\partial}{\partial r} \left[rM \int d^3 v \, f_i(\mathbf{v} \cdot \hat{\mathbf{r}}) \frac{v_{\perp}^2}{2} \right] = -Zen_i \mathbf{V}_{i\perp} \cdot \nabla \phi + M \int d^3 v \, C\{f_i\} \frac{v_{\perp}^2}{2}. \tag{5.16}$$

Since in this case, $\partial/\partial t = 0$ and $\mathbf{V}_{i\perp} \cdot \nabla \phi = (\mathbf{V}_{i\perp} \cdot \hat{\mathbf{r}})(\partial \phi/\partial r) = 0$ [recall (5.15)],
equation (5.13) finally becomes

$$r^{2}\hat{\mathbf{r}}\cdot\overrightarrow{\boldsymbol{\pi}}_{i}\cdot\hat{\boldsymbol{\theta}} = -\frac{Mr^{2}}{4\Omega_{i}}\int d^{3}v \left(C\{f_{i}\}-\overline{C\{f_{i}\}}\right)\left[(\mathbf{v}\cdot\hat{\mathbf{r}})^{2}-(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{2}\right] \\ +\frac{Mr}{2\Omega_{i}}\frac{\partial}{\partial r}\left[r\int d^{3}v \left(f_{i}-\overline{f}_{i}\right)(\mathbf{v}\cdot\hat{\mathbf{r}})\frac{v_{\perp}^{2}}{2}\right] \\ -\frac{M}{6r\Omega_{i}}\frac{\partial}{\partial r}\left[\frac{r^{3}}{\Omega_{i}}\int d^{3}v \left(C\{f_{i}\}-\overline{C\{f_{i}\}}\right)(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^{3}\right].$$
(5.17)

In this equation, only the gyrophase dependent piece of the distribution function enters in the integrals. The corrections \mathbf{R}_1 , \mathbf{R}_2 , E_1 , E_2 and μ_1 depend on the gyrophase. Then $f_i(\mathbf{R}, E, \mu, t)$ must be Taylor expanded around \mathbf{r} , E_0 and μ_0 to get the second order gyrophase dependent piece. The calculation is done in section E.2 of Appendix E, and the final result is

$$(f_{i} - \overline{f}_{i})_{g} = \frac{1}{\Omega_{i}} (\mathbf{v} \cdot \hat{\boldsymbol{\theta}}) \left(\frac{\partial f_{M0}}{\partial r} + \frac{Ze}{T_{i}} \frac{\partial \phi}{\partial r} f_{M0} \right)$$
$$-\frac{r}{4\Omega_{i}} [(\mathbf{v} \cdot \hat{\mathbf{r}})^{2} - (\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^{2}] \frac{\partial}{\partial r} \left[\frac{1}{r\Omega_{i}} \left(\frac{\partial f_{M0}}{\partial r} + \frac{Ze}{T_{i}} \frac{\partial \phi}{\partial r} f_{M0} \right) \right]$$
$$-\frac{Mc}{4B\Omega_{i}} [(\mathbf{v} \cdot \hat{\mathbf{r}})^{2} - (\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^{2}] \frac{\partial \phi}{\partial r} \left[\frac{\partial}{\partial r} \left(\frac{f_{M0}}{T_{i}} \right) + \frac{Ze}{T_{i}^{2}} \frac{\partial \phi}{\partial r} f_{M0} \right], \qquad (5.18)$$

where f_{M0} is given in (4.76), and the subindex g indicates the non-collisional origin of this gyrophase dependence. The gyrophase dependent piece given by (3.38) is also necessary. For the Krook operator it becomes

$$(f_i - \overline{f}_i)_c = -\frac{\nu}{\Omega_i^2} f_{M0} \left(\frac{Mv^2}{2T_i} - \frac{5}{2}\right) \mathbf{v}_{\perp} \cdot \nabla \ln T_i.$$
(5.19)

Employing equations (5.14), (5.15), (5.18) and (5.19), I find

$$\frac{Mr^2}{4\Omega_i} \int d^3 v \left(C\{f_i\} - \overline{C\{f_i\}} \right) \left[(\mathbf{v} \cdot \hat{\mathbf{r}})^2 - (\mathbf{v} \cdot \hat{\boldsymbol{\theta}})^2 \right] = \frac{\nu r^3 p_i}{4\Omega_i^2} \frac{\partial}{\partial r} \left[\frac{c}{rB} \left(\frac{\partial \phi}{\partial r} + \frac{1}{Zen_i} \frac{\partial p_i}{\partial r} \right) \right] + \frac{\nu r^3}{4\Omega_i^2} \frac{\partial}{\partial r} \left(\frac{1}{r} \frac{p_i}{M\Omega_i} \frac{\partial T_i}{\partial r} \right),$$
(5.20)

$$\frac{Mr}{2\Omega_i}\frac{\partial}{\partial r}\left[r\int d^3v \left(f_i - \overline{f}_i\right)(\mathbf{v}\cdot\hat{\mathbf{r}})\frac{v_\perp^2}{2}\right] = -\frac{r}{\Omega_i}\frac{\partial}{\partial r}\left(\frac{\nu p_i r}{M\Omega_i^2}\frac{\partial T_i}{\partial r}\right)$$
(5.21)

and

$$\frac{M}{6r\Omega_i}\frac{\partial}{\partial r}\left[\frac{r^3}{\Omega_i}\int d^3v \left(C\{f_i\}-\overline{C\{f_i\}}\right)(\mathbf{v}\cdot\hat{\boldsymbol{\theta}})^3\right] = -\frac{1}{2r\Omega_i}\frac{\partial}{\partial r}\left(\frac{\nu p_i r^3}{M\Omega_i^2}\frac{\partial T_i}{\partial r}\right).$$
 (5.22)

Substituting these results into (5.17), $r^2 \hat{\mathbf{r}} \cdot \overleftarrow{\boldsymbol{\pi}}_i \cdot \hat{\boldsymbol{\theta}} = 0$ gives

$$c\left(\frac{\partial\phi}{\partial r} + \frac{1}{Zen_i}\frac{\partial p_i}{\partial r}\right) = rB(r)\int_0^r dr' \frac{U(r')}{r'} \left[\frac{\partial}{\partial r'}\ln B(r') - \frac{3}{2}\frac{\partial}{\partial r'}\ln\left(\frac{p_i(r')U(r')}{r'}\right)\right],$$
(5.23)

where $U = (2/M\Omega_i)(\partial T_i/\partial r)$. Notice the difference between this equation and (4.84). In particular, notice that for an isothermal f_{M0} , $\partial T_i/\partial r = 0$, a radial Maxwell-Boltzmann response is recovered from (5.23) as expected, but this is not a feature of the non-physical forms (4.82) and (4.84).

Finally, I remark that equation (5.17) gives a radial transport of toroidal angular momentum $\hat{\mathbf{r}} \cdot \boldsymbol{\pi}_i \cdot \hat{\boldsymbol{\theta}} \sim \delta_i^2(\nu/\Omega_i)p_i$, corresponding to the term in the vorticity equation

$$\nabla \cdot \left[\frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_i)\right] = -\frac{1}{r}\frac{\partial}{\partial r} \left[\frac{c}{rB}\frac{\partial}{\partial r}(r^2\hat{\mathbf{r}}\cdot \overleftarrow{\boldsymbol{\pi}}_i\cdot\hat{\boldsymbol{\theta}})\right] \sim \frac{\nu\delta_i^3}{\Omega_i}en_e v_i/L.$$
(5.24)

The radial current density represented by this term is too small to be recovered with a gyrokinetic equation good only to order $\delta_i^2 f_{Mi} v_i/L$, as already shown in section 4.5.

5.3 Distribution function and potential to second order

To evaluate (5.10), the ion distribution function and the potential have to be found to order $\delta_i^2 f_{Mi}$ and $\delta_i^2 T_e/e$, respectively. In this section, I show how both the distribution function and the potential can be calculated to higher order without the full second order Fokker-Planck and vorticity equations. I take advantage of the usually small ratio $1/q \sim B_p/B \ll 1$, where $q(\psi) = (2\pi)^{-1} \oint d\theta (\mathbf{B} \cdot \nabla \zeta / \mathbf{B} \cdot \nabla \theta)$ is the safety factor, and $B_p = |\nabla \psi|/R$ is the poloidal component of the magnetic field. Expanding in $q \gg 1$, I will find the ion distribution function to order $q\delta_i^2 f_{Mi}$, neglecting terms of order $\delta_i^2 f_{Mi}$. The potential is calculated consistently with this higher order solution for f_i by employing a higher order vorticity equation. Importantly, the vorticity equations obtained in this section only give to higher order the short wavelength, nonaxisymmetric part of the potential. The axisymmetric piece of the potential must be calculated employing another equation. The axisymmetric component is composed of the flux surface averaged piece, given by the conservation of toroidal angular momentum equation (2.21) and the higher order viscosity (5.10), and a poloidally varying modification. The poloidal variation is the Geodesic Acoustic Mode (GAM) response [68, 69]; the initial transient of an axisymmetric perturbation in the potential. The perturbation initially induces poloidal density variations that rapidly Landau damp towards a constant zonal flow known as the Hinton-Rosenbluth residual [10, 11]. This initial decay or GAM is axisymmetric and thus does not drive radial transport. It can, however, shear the turbulence. The lower order vorticity equations (4.45) and (4.53) reproduce the lower order GAM response. The new higher order vorticity equations derived in this subsection will not have, however, higher order corrections to GAMs. It is relatively straightforward to calculate the higher order corrections analytically, but they complicate the vorticity equations unnecessarily since in reality the GAM response is believed to be less important than the Hinton-Rosenbluth residual, that is adequately kept by the gyrokinetic ion Fokker-Planck equation (3.39) and the toroidal angular momentum conservation equation (2.21). For this reason, I will drop the higher order axisymmetric corrections to the gyrokinetic vorticity equations (4.45) and (4.53).

In subsection 5.3.1, I show that equation (3.39) is enough to calculate the distribution function to order $q\delta_i^2 f_{Mi}$. I also argue that the second order corrections \mathbf{R}_2 and E_2 are not needed since they only provide corrections of order $\delta_i^2 f_{Mi}$. Employing these two results in subsection 5.3.2, I extend the gyrokinetic vorticity equations (4.45) and (4.53) to give the electrostatic potential consistent with the higher order f_i . To do so, I develop an extended gyrokinetic equation in the physical phase space, as I did in subsection 4.3.1, but now to order $q\delta_i^2 f_{Mi}v_i/L$. Taking moments of this equation, I obtain the new extended vorticity equations that retain the short wavelength,



Figure 5-1: High flow ordering for the ion parallel velocity with $B_p/B \ll 1$. Notice that the poloidal projection of the $E \times B$ drift must be comparable to the poloidal projection of the ion parallel velocity, giving $\mathbf{v}_E \sim (B_p/B)V_{i||} \sim (B_p/B)v_i \ll v_i$.

non-axisymmetric pieces of the potential to higher order.

5.3.1 Higher order ion distribution function

To find $f_i(\mathbf{R}, E, \mu, t)$ to order $\delta_i^2 f_{Mi}$, it is necessary to solve a higher order gyrokinetic Fokker-Planck equation. Similarly, for the higher order potential, it is necessary to find a higher order gyrokinetic vorticity equation. In this section, I show that under certain assumptions, the second order gyrokinetic Fokker-Planck and vorticity equations can be easily deduced from their first order versions.

There has already been some work in transport of toroidal angular momentum in gyrokinetics. For these studies, it was necessary to realize that the Reynolds stress tends to vanish to order $\delta_i^2 p_i$ in the low flow limit, becoming of order $\delta_i^3 p_i$. In references [65, 66, 70] the revised approach is ordering the parallel velocity as comparable to the ion thermal speed. Since for sonic velocities, the plasma can only rotate toroidally [71, 72], a sonic parallel velocity requires, in general, a sonic $E \times B$ drift to cancel its poloidal component. However, sonic $E \times B$ drifts invalidate the gyrokinetic derivation of chapter 3. To avoid this problem, references [65, 66, 70] do not reach sonic $E \times B$ velocities because they take advantage of the expansion parameter $B_p/B \ll 1$. The perpendicular $E \times B$ drift is small compared to the thermal speed by B_p/B , making the traditional gyrokinetic formulation based on subsonic $E \times B$ motion still valid [see figure 5-1]. Under these assumptions, the toroidal velocity is given by $-cR(\partial\phi/\partial\psi)$, and the radial transport of toroidal angular momentum is of order $\delta_i^2 p_i$, as can be found from equation (5.5). The term with the time derivative in equation (5.5) is still negligible if the turbulence reaches its statistical equilibrium, but the collisional term, proven to vanish for up-down symmetric tokamaks in section K.2 of Appendix K, contributes to order $\delta_i^2 p_i$ because the sonic parallel velocity breaks the up-down symmetry by introducing a preferred direction in the Maxwellian. Similarly, the Reynolds stress (5.6) in this case is of order $\delta_i^2 p_i$. In reference [66], the Reynolds stress is calculated employing the δf code GYRO with sonic parallel velocities, and it does not vanish to order $\delta_i^2 p_i$, as expected.

This approach has the disadvantage of making the toroidal velocity only depend on the radial electric field $\partial \phi / \partial \psi$. Density and temperature gradients cannot compete with the radial electric field, and therefore it is not possible to recover naturally the isothermal radial Maxwell-Boltzmann solution, or the dependence of the velocity on the temperature gradient. I propose an alternative approach with subsonic velocities that at the same time avoids solving a full second order gyrokinetic equation. It exploits the extra expansion parameter $B_p/B \ll 1$ in a different manner.

In the new ordering with $B_p/B \ll 1$, the parallel gradient is of order 1/qR, with R the major radius and $q \sim B/B_p \gg 1$ the safety factor. As in section 4.1, the ion and electron zeroth order distribution functions are assumed to be stationary Maxwellians with only radial dependence, i.e., $f_i \simeq f_{Mi}(\psi)$ and $f_e \simeq f_{Me}(\psi)$. With the new orderings, the size of the first order correction to the Maxwellian h_{i1} changes depending on the nature of the correction, i.e., depending on whether it is turbulent, due to non-axisymmetric potential fluctuations, or neoclassical, given by the long wavelength, axisymmetric pieces. For the neoclassical banana regime pieces, the term $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ is negligible [recall the discussion in section 4.1], and the neoclassical correction h_{i1}^{nc} is determined by a balance between the parallel streaming term $u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} h_{i1}^{\mathrm{nc}} \sim (v_i/qR)h_{i1}^{\mathrm{nc}}$ of order $h_{i1}^{\mathrm{nc}} \sim q \delta_i f_{Mi}$ when the transit average collisional constraint is satisfied. On the other hand, in tokamaks, turbulence is driven by toroidal drift wave modes in which the parallel streaming term $u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} f_i$ is of secondary importance. The turbu-

lent contributions h_{i1}^{tb} are determined by the competition between the magnetic drift term $\mathbf{v}_M \cdot \nabla_{\mathbf{R}} h_{i1}^{\text{tb}}$ and the $E \times B$ drift term $\mathbf{v}_E^{\text{tb}} \cdot \nabla_{\mathbf{R}} h_{i1}^{\text{tb}}$. The orderings of (3.1) still hold, giving that for $k_{\perp} \rho_i \sim 1$, $h_{i1}^{\text{tb}} / f_{Mi} \sim e \phi^{\text{tb}} / T_e \sim \delta_i$. The turbulent contribution h_{i1}^{tb} is then smaller than the neoclassical piece h_{i1}^{nc} by a factor of 1/q.

To calculate the axisymmetric radial electric field in a completely general manner, the gyrokinetic treatment needs to be extended to provide the pieces $h_{i2} \sim \delta_i^2 f_{Mi}$ of the ion distribution function. This would require calculating the gyrokinetic Fokker-Planck equation (3.39) to higher order, i.e., obtaining the time derivatives of the gyrokinetic variables \mathbf{R} , E and μ to an order higher in δ_i . However, as just noted, there are terms that are larger by $q \gg 1$. Instead of calculating the complete $O(\delta_i^2 f_{Mi} v_i/L)$ gyrokinetic Fokker-Planck equation, I will only keep the terms that are larger by q. To identify these terms, I let $f_i = f_{Mi} + h_{i1} + h_{i2} + \ldots$ and then write the gyrokinetic equation for the second order perturbation as

$$\frac{\partial h_{i2}}{\partial t} + [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} h_{i2} - \langle C\{f_i\}\rangle^{(2)} = -\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i1}^{\mathrm{nc}} -\dot{\mathbf{R}}^{(2)} \cdot \nabla_{\mathbf{R}} (f_{Mi} + h_{i1}^{\mathrm{tb}}) + \frac{Ze}{M} [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle \frac{\partial h_{i1}}{\partial E} + \dot{E}^{(2)} \frac{M f_{Mi}}{T_i}, \quad (5.25)$$

with $\langle C\{f_i\}\rangle^{(2)} = \langle C\{f_i\}\rangle - \langle C^{(\ell)}\{f_{Mi} + h_{i1}\}\rangle$, $\dot{\mathbf{R}}^{(2)} = \langle \dot{\mathbf{R}}\rangle - [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d]$ and $\dot{E}^{(2)} = \langle \dot{E}\rangle + (Ze/M)[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M]\cdot\nabla_{\mathbf{R}}\langle\phi\rangle$. Here, $\langle C\{f_i\}\rangle$, $\langle \dot{E}\rangle$ and $\langle \dot{\mathbf{R}}\rangle$ are calculated to order $\delta_i^2 \nu_{ii} f_{Mi}$, $\delta_i^2 v_i^3/L$ and $\delta_i^2 v_i$, respectively; an order higher than in equation (3.39). Notice that the first order correction h_{i1} enters differently depending on its nature. The turbulent short wavelength piece h_{i1}^{tb} has large gradients and it is multiplied by the small quantity $\dot{\mathbf{R}}^{(2)}$, while the gradient of the neoclassical piece h_{i1}^{nc} is small but is multiplied by the lowest order term $\mathbf{v}_d \gg \dot{\mathbf{R}}^{(2)}$.

On the right side of equation (5.25), the dominant terms are $-\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i1}^{\mathrm{nc}}$ and $(Ze/M)[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle (\partial h_{i1}^{\mathrm{nc}} / \partial E)$ because h_{i1}^{nc} is larger than all other terms by a factor of q. The higher order corrections $\dot{\mathbf{R}}^{(2)}$ and $\dot{E}^{(2)}$ are finite gyroradius correction that do not contain any q factors. Since h_{i1}^{nc} determines the parallel velocity and the parallel heat flow, the term $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i1}^{\mathrm{nc}}$ represents the effect of the gradient of the parallel velocity and parallel heat flow on turbulence. This term is not kept

in δf flux tube codes because only the short wavelength pieces of the first order correction to the distribution function are calculated. We see from (5.25) that it is possible to simply add this neoclassical term to the δf gyrokinetic equation along with $(Ze/M)[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle (\partial h_{i1}^{nc} / \partial E)$. In full f codes, the distribution function is solved from the lowest order equation (3.39). In this equation, the terms $-\mathbf{v}_d \cdot$ $\nabla_{\mathbf{R}} h_{i1}^{nc}$ and $(Ze/M)[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle (\partial h_{i1}^{nc} / \partial E)$ are naturally included, so it is not necessary to write the gyrokinetic equation to higher order than (3.39).

As for h_{i1} , the function h_{i2} has a turbulent piece h_{i2}^{tb} , and a neoclassical piece h_{i2}^{nc} . The turbulent piece is given by the balance between the drifts $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i2}^{tb} \sim \delta_i v_i k_\perp h_{i2}^{tb}$ and the driving term $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i1}^{nc} \sim q \delta_i^2 v_i f_{Mi}/a$, giving $h_{i2}^{tb} \sim q \delta_i^2 f_{Mi}$ for $k_\perp \rho_i \sim 1$. The neoclassical piece is a result of a balance between the parallel streaming term $u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} h_{i2}^{nc} \sim (v_i/qR) h_{i2}^{nc}$ and the magnetic drift term $\mathbf{v}_M \cdot \nabla_{\mathbf{R}} h_{i1}^{nc} \sim (\rho_i/R) v_i q \delta_i f_{Mi}/a$, leading to $h_{i2}^{nc} \sim q^2 \delta_i^2 f_{Mi}$.

Since h_{i2}^{tb} is larger than $\delta_i^2 f_{Mi}$ by a factor of q, the second order corrections \mathbf{R}_2 and E_2 give negligible contributions to the second order piece of the distribution function. To see this, Taylor expand $f_i(\mathbf{R}, E, \mu, t)$ around $\mathbf{R}_g = \mathbf{r} + \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$, E_0 and μ_0 . Then, the terms $\mathbf{R}_2 \cdot \nabla_{\mathbf{R}_g} f_{ig}$ and $E_2(\partial f_{Mi}/\partial E_0)$, of order $\delta_i^2 f_{Mi}$, are negligible. This fact simplifies the integration in velocity space in (5.10), since it is enough to keep only the first order corrections \mathbf{R}_1 , E_1 and μ_1 . Finally, the gyrophase dependent piece \tilde{f}_i , defined in (3.38), vanishes $[\tilde{f}_i \simeq -\Omega_i^{-1} \int^{\varphi} d\varphi' (C^{(\ell)} \{h_{i1}^{\mathrm{nc}}\} - \langle C^{(\ell)} \{h_{i1}^{\mathrm{nc}}\} \rangle) = 0$ since $\partial h_{i1}^{\mathrm{nc}}/\partial \varphi_0 = 0$].

5.3.2 Higher order electrostatic potential

In this subsection, I find the gyrokinetic vorticity equations (4.45) and (4.53) to higher order. To simplify the derivation, I limit myself to the short wavelength, nonaxisymmetric contributions to the vorticity equation – the ones responsible for the turbulence. The flux surface averaged component of the potential will be given by the conservation of toroidal angular momentum. For the usually unimportant GAM response [68, 69], it is enough to retain the first order terms, already in the lower order gyrokinetic vorticity equations (4.45) and (4.53). Finding the higher order vorticity equation becomes a simple task because the higher order corrections to the gyrokinetic variables \mathbf{R}_2 and E_2 are negligible. Quasineutrality must be enforced for the gyrokinetic Fokker-Planck equation with the new terms $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} h_{i1}^{\mathrm{nc}}$ and $(Ze/M)[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_M] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle (\partial h_{i1}^{\mathrm{nc}} / \partial E)$, and a $O(q\delta_i^2 f_{Mi})$ distribution function given by

$$f_i(\mathbf{R}, E, \mu, t) \simeq f_{iG} + \frac{Ze\tilde{\phi}}{M} \left[\frac{\partial}{\partial E_0} (f_{Mi} + h_{i1}^{\rm nc}) + \frac{1}{B} \frac{\partial h_{i1}^{\rm nc}}{\partial \mu_0} \right], \qquad (5.26)$$

with $f_{iG} \equiv f_i(\mathbf{R}_g, E_0, \mu_g, t)$, $\mathbf{R}_g = \mathbf{r} + \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$ and $\mu_g = \mu - Ze\tilde{\phi}/MB \neq \mu_0$. In expression (5.26), $h_{i1}^{nc}(\mathbf{R}, E, \mu, t) \simeq h_{i1}^{nc}(\mathbf{r}, E_0, \mu_0, t)$. It is convenient to extract the $Ze\tilde{\phi}/MB$ portion of μ by introducing μ_g so we can take advantage of previous results.

As I did in section 4.3, I will find the vorticity equations by taking moments of the gyrokinetic Fokker-Planck equation (3.39); in this case to order $q\delta_i^2 f_{Mi}v_i/L$. To write (3.39) in physical phase space, I will first find the evolution equation for f_{iG} . Equation (3.39) gives the evolution of f_{iG} if **R** is replaced by \mathbf{R}_g , E by E_0 and μ by μ_g , giving to order $q\delta_i^2 f_{Mi}v_i/L$

$$\frac{\partial f_{iG}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + \left[u_g \hat{\mathbf{b}}(\mathbf{R}_g) + \mathbf{v}_{dg}\right] \cdot \left[\nabla_{\mathbf{R}_g} f_{iG} - \frac{Ze}{M} \nabla_{\mathbf{R}_g} \langle \phi \rangle \frac{\partial}{\partial E_0} (f_{Mi} + h_{i1}^{\mathrm{nc}})\right] = \langle C\{f_i\} \rangle |_{\mathbf{R} \to \mathbf{R}_g, E \to E_0, \mu \to \mu_g}, \quad (5.27)$$

with

$$\mathbf{v}_{dg} \equiv \frac{\mu_g}{\Omega_i(\mathbf{R}_g)} \hat{\mathbf{b}}(\mathbf{R}_g) \times \nabla_{\mathbf{R}_g} B(\mathbf{R}_g) + \frac{u_g^2}{\Omega_i(\mathbf{R}_g)} \hat{\mathbf{b}}(\mathbf{R}_g) \times \boldsymbol{\kappa}(\mathbf{R}_g) - \frac{c}{B(\mathbf{R}_g)} \nabla_{\mathbf{R}_g} \langle \phi \rangle (\mathbf{R}_g, \mu_g, t) \times \hat{\mathbf{b}}(\mathbf{R}_g)$$
(5.28)

and

$$u_g \equiv \sqrt{2[E_0 - \mu_g B(\mathbf{R}_g)]} = \sqrt{2[E - \mu B(\mathbf{R})]} = u,$$
 (5.29)

where $E_1 = Ze\tilde{\phi}/M$ and $\mu - \mu_g = Ze\tilde{\phi}/MB(\mathbf{R}_g)$ cancel exactly to give the second equality. In equation (5.27), I have neglected terms of order $\delta_i^2 f_{Mi}$ by taking the approximation $\partial f_{iG}/\partial E_0 \simeq \partial (f_{Mi} + h_{i1}^{\rm nc})/\partial E_0$. Equation (5.27) needs to be written in the physical phase space variables. To the order of interest, $\nabla_{\mathbf{R}_g} f_{iG} = \nabla_{\mathbf{R}_g} \mathbf{r} \cdot \overline{\nabla} f_{iG} + \nabla_{\mathbf{R}_g} \mu_0 (\partial h_{i1}^{\mathrm{nc}} / \partial \mu_0)$ and $\nabla_{\mathbf{R}_g} \langle \phi \rangle \simeq$ $\nabla_{\mathbf{R}_g} \mathbf{r} \cdot \overline{\nabla} \langle \phi \rangle$, with $\nabla_{\mathbf{R}_g} \mu_0 = -\nabla_{\mathbf{R}_g} \mu_{10} \simeq -\overline{\nabla} \mu_{10}$ and $\mu_g - \mu_0 = \mu_{10} = \mu_1 - Ze\tilde{\phi}/MB$. Then, equation (5.27) becomes

$$\frac{\partial f_{iG}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + \left[u\hat{\mathbf{b}}(\mathbf{R}_g) + \mathbf{v}_{M0} + \mathbf{v}_{E0}\right] \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \cdot \left[\overline{\nabla}f_{iG} - \frac{Ze}{M}\overline{\nabla}\langle\phi\rangle\frac{\partial}{\partial E_0}(f_{Mi} + h_{i1}^{\mathrm{nc}})\right] \\ - v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla}\mu_{10}\frac{\partial h_{i1}^{\mathrm{nc}}}{\partial\mu_0} = \langle C\{f_i\}\rangle. (5.30)$$

Here, the term $(\mathbf{v}_{dg} - \mathbf{v}_{M0} - \mathbf{v}_{E0}) \cdot \nabla_{\mathbf{R}} f_{iG} \sim \delta_i^2 f_{Mi} v_i / L$ has been neglected. In the collisional term $\langle C\{f_i\}\rangle$ in (5.30), it is necessary to consider pieces of order $q\delta_i^2 \nu_{ii} f_{Mi}$ that come from the gyroaverage of $C^{(\ell)}\{h_{i1}^{\mathrm{nc}}\}$ performed holding the higher order gyrokinetic variables \mathbf{R}_g , E and μ fixed, with $C^{(\ell)}$ the linearized collision operator. These terms are not considered in Appendix H, where the gyrokinetic variables are approximated by \mathbf{R}_g , E_0 and μ_0 . Fortunately, the collision frequency is usually small, making these terms negligible. Ignoring these terms and assuming that the collision operator can be treated as in Appendix H is reasonable and simplifies the rest of the derivation. Finally, employing the results in section F.2 of Appendix F, equation (5.30) gives

$$\frac{\partial f_{iG}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + \frac{v_{||}}{B} \Big\{ \overline{\nabla} \cdot \left[\frac{B}{v_{||}} f_{iG} \left(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \right) \right] - \frac{\partial}{\partial \mu_{0}} \left[(f_{Mi} + h_{i1}^{\mathrm{nc}}) \mathbf{B} \cdot \overline{\nabla} \mu_{10} \right] \\ - \frac{\partial}{\partial \varphi_{0}} \left[(f_{Mi} + h_{i1}^{\mathrm{nc}}) \mathbf{B} \cdot \overline{\nabla} \varphi_{10} \right] - \frac{\partial}{\partial E_{0}} \left[\frac{B}{v_{||}} (f_{Mi} + h_{i1}^{\mathrm{nc}}) \frac{Ze}{M} \left(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \right) \cdot \overline{\nabla} \langle \phi \rangle \right] \Big\} \\ = \langle C\{f_{i}\} \rangle, (5.31)$$

with $\mathbf{R} \cdot \nabla_{\mathbf{R}_g} \mathbf{r}$ from (4.15). Notice that equation (5.31) is equivalent to equation (4.23) except for the changes $f_{Mi} \to f_{Mi} + h_{i1}^{\mathrm{nc}}$ and $f_{ig} \to f_{iG}$.

The same moment equations that were obtained with (4.23) can be found for equation (5.31), but now with f_{iG} instead of f_{ig} and $f_{Mi} + h_{i1}^{nc}$ instead of f_{Mi} . The differences between equations (4.23) and (5.31) are enough to invalidate some of the cancellations that were found in Appendix G. For example, in the momentum conservation equation for ions (4.38), the piece of the viscosity $M \int d^3v f_{ig} v_{||0} \mathbf{v}_{\perp}$, with $v_{||0}$ given in (4.16), reduced to $M \int d^3v f_{ig} v_{||} \mathbf{v}_{\perp}$ because the small correction $M(v_{||0} - v_{||})\mathbf{v}_{\perp}$, composed of terms that are either odd in $v_{||}$ or in \mathbf{v}_{\perp} , vanished when integrated over the lowest order Maxwellian. In the new vorticity equation, terms of order $q\delta_i^2 p_i/L$ must be retained in the momentum conservation equation, leading to a new non-vanishing term $M \int d^3 v h_{i1}^{\rm nc}(v_{||0} - v_{||}) \mathbf{v}_{\perp}$. Fortunately, most terms like this one are axisymmetric and only enter in the calculation of the flux surface averaged radial electric field and the higher order GAM response. To calculate the flux surface averaged radial electric field, the toroidal angular momentum conservation equation is to be used, and to simplify the equations I ignore the higher order corrections to the GAM response that is expected to be unimportant, as already discussed. Therefore, the vorticity equation is only employed to solve for the non-axisymmetric, turbulent fluctuations in the potential, and many of the cancellations employed in Appendix G are recovered. Additionally, the difference $f_{iG} - f_{ig} \simeq (\mu_g - \mu_0)(\partial h_{i1}^{\rm nc}/\partial \mu_0) \sim q \delta_i^2 f_{Mi}$ is also a long wavelength, axisymmetric piece, and it will not enter in the equations for the non-axisymmetric electric field. Then, f_{iG} can be approximated by the simpler distribution function $f_{ig} \equiv f_i(\mathbf{R}_g, E_0, \mu_0, t)$. The generalized particle conservation equation can be found following section G.1 of Appendix G by ignoring the purely axisymmetric contributions of h_{i1}^{nc} . Then, the non-axisymmetric component of particle conservation is

$$\frac{\partial}{\partial t} \left(n_i - n_{ip}^{(2)} \right) + \nabla \cdot \left(n_i V_{ig||}^{(2)} \hat{\mathbf{b}} + n_i \mathbf{V}_{igd} + n_i \mathbf{V}_{iE} + n_i \tilde{\mathbf{V}}_i + n_i \mathbf{V}_{iC} \right) = 0, \quad (5.32)$$

with

$$n_{ip}^{(2)} = -\int d^3v \, \frac{Ze\tilde{\phi}}{T_i} f_{Mi} + \int d^3v \, \frac{Ze\tilde{\phi}}{M} \left(\frac{\partial h_{i1}^{\rm nc}}{\partial E_0} + \frac{1}{B} \frac{\partial h_{i1}^{\rm nc}}{\partial \mu_0}\right) \tag{5.33}$$

and

$$n_i V_{ig||}^{(2)} = \int d^3 v \, f_{ig} v_{||} \neq n_i V_{i||}. \tag{5.34}$$

The rest of the terms in (5.32) are as defined in subsection 4.3.3, although now the turbulent second order contribution h_{i2}^{tb} implicitly enters in the integrals via the solution f_i to the full gyrokinetic equation (3.39). The non-axisymmetric piece of the momentum conservation equation can be obtained following section G.2 of Appendix G, finally giving

$$\frac{\partial}{\partial t} (n_i M \mathbf{V}_{ig}) + \hat{\mathbf{b}} [\hat{\mathbf{b}} \cdot \nabla p_{ig||} + (p_{ig||} - p_{ig\perp}) \nabla \cdot \hat{\mathbf{b}} + \nabla \cdot \boldsymbol{\pi}_{ig||}] + \nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{ig\times} = -Zen_i \hat{\mathbf{b}} \left(\hat{\mathbf{b}} + \frac{V_{i||}^{\text{nc}}}{\Omega_i} \hat{\mathbf{b}} \times \boldsymbol{\kappa} \right) \cdot \nabla \phi + \tilde{F}_{iE}^{(2)} \hat{\mathbf{b}} + \mathbf{F}_{iB}^{(2)} + \mathbf{F}_{iC}, \quad (5.35)$$

with $n_i V_{i||}^{\text{nc}} = \int d^3 v \, v_{||} h_{i1}^{\text{nc}}$,

$$\tilde{F}_{iE}^{(2)} = Ze \int d^3v \left(f_{Mi} + h_{i1}^{\rm nc} \right) \left(\hat{\mathbf{b}} + \frac{v_{||}}{\Omega_i} \hat{\mathbf{b}} \times \boldsymbol{\kappa} + \frac{1}{\Omega_i} \overline{\nabla} \times \mathbf{v}_{\perp} \right) \cdot \overline{\nabla} \tilde{\phi}$$
(5.36)

and

$$\mathbf{F}_{iB}^{(2)} = \int d^3 v \, M f_{ig} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} + \int d^3 v \, \frac{Mc}{B} (f_{Mi} + h_{i1}^{\mathrm{nc}}) (\overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}.$$
(5.37)

Again, the rest of the terms are as defined in subsection 4.3.4, but with the higher order piece h_{i2}^{tb} implicitly included.

The moment equations (5.32) and (5.35) can be used to extend the gyrokinetic vorticity equations (4.45) and (4.53) to order $q\delta_i^2 en_e v_i/L$. Combining equation (5.32) with the electron number conservation equation (4.26) gives the vorticity equation

$$\frac{\partial}{\partial t}(Zen_{ip}^{(2)}) = \nabla \cdot \left(J_{g||}^{(2)} \hat{\mathbf{b}} + \mathbf{J}_{gd} + \tilde{\mathbf{J}}_i + Zen_i \tilde{\mathbf{V}}_i + Zen_i \mathbf{V}_{iC} \right),$$
(5.38)

with

$$J_{g||}^{(2)} = Zen_i V_{ig||}^{(2)} - en_e V_{e||},$$
(5.39)

and the rest of the terms as defined in subsection 4.4.1 with h_{i2}^{tb} implicit. Finally, combining (5.38) with (5.35) gives

$$\frac{\partial \varpi_G^{(2)}}{\partial t} = \nabla \cdot \left[J_{g||}^{(2)} \hat{\mathbf{b}} + \mathbf{J}_{gd} + \tilde{\mathbf{J}}_{i\phi}^{(2)} + \frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{iG}) + Zen_i \mathbf{V}_{iC} - \frac{c}{B} \hat{\mathbf{b}} \times \mathbf{F}_{iC} \right], \quad (5.40)$$

with

$$\boldsymbol{\varpi}_{G}^{(2)} = \nabla \cdot \left(\frac{Ze}{\Omega_{i}} \int d^{3}v \ f_{ig} \mathbf{v} \times \hat{\mathbf{b}}\right) - \int d^{3}v \ \frac{Z^{2}e^{2}\tilde{\phi}}{T_{i}} f_{Mi} + \int d^{3}v \ \frac{Z^{2}e^{2}\tilde{\phi}}{M} \left(\frac{\partial h_{i1}^{\mathrm{nc}}}{\partial E_{0}} + \frac{1}{B} \frac{\partial h_{i1}^{\mathrm{nc}}}{\partial \mu_{0}}\right), \tag{5.41}$$

$$\tilde{\mathbf{J}}_{i\phi}^{(2)} = \tilde{\mathbf{J}}_{i} - \frac{Zec}{B\Omega_{i}} \hat{\mathbf{b}} \times \int d^{3}v \ (f_{Mi} + h_{i1}^{\mathrm{nc}}) (\overline{\nabla}\tilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}, \tag{5.42}$$

and the rest of the terms as defined in subsection 4.4.2. Both equations (5.38) and (5.40) can be used to find the short wavelength non-axisymmetric pieces of the potential consistent with the higher order f_i . The flux surface averaged component of the potential is given by the conservation equation of toroidal angular momentum. Finally, it is important to remember that dropping the higher order axisymmetric terms in equations (5.38) and (5.40) implies dropping the higher order corrections to the GAM response [68, 69]. I expect this response to be unimportant for core turbulence based on previous experience with tokamak core simulations.

5.4 Discussion

The fast time average of equation (5.10) provides the irreversible transport of toroidal angular momentum across the tokamak. This irreversible transport determines the toroidal rotation profile and hence the self-consistent radial electric field.

It is expected that the fast time average of equation (5.10) is of order $\delta_i^3 p_i$, requiring then a distribution function good to order $\delta_i^2 f_{Mi}$ and a potential consistent with this higher order distribution function. It is in principle possible (if not in practice) to obtain a gyrokinetic equation able to provide such accurate results, but in this thesis I propose an alternative approach. To simplify the problem, I take advantage of the expansion in $B_p/B \ll 1$ to prove that the gyrokinetic Fokker-Planck equation (3.39) is enough to obtain the distribution function up to order $q\delta_i^2 f_{Mi}$. In δf flux tube codes, the distribution function cannot be calculated to higher order than $\delta_i f_{Mi}$ because of the present implementation technique, but these codes can be adapted by adding terms that contain the first order neoclassical correction h_{i1}^{nc} . In full fcodes, the gyrokinetic equation (3.39) is fully implemented. Once the turbulence has



Figure 5-2: Procedure proposed to obtain the long wavelength axisymmetric radial electric field.

reached statistical equilibrium, the long wavelength axisymmetric flows must be close to the neoclassical solution, and the orderings described in this chapter hold, leading to solutions valid up to $q\delta_i^2 f_{Mi}$.

Finally, I have extended the gyrokinetic vorticity equations (4.45) and (4.53), and written them in (5.38) and (5.40). These vorticity equations have only been found for non-axisymmetric pieces of the potential, giving then the non-axisymmetric turbulent fluctuations consistent with the higher order f_i . The flux surface averaged component of the potential cannot be calculated from these vorticity equations, but it can be solved from the conservation of toroidal angular momentum (2.21).

The method proposed to self-consistently solve for the long wavelength axisymmetric radial electric field in the presence of drift wave turbulence is summarized in figure 5-2. To be specific I employ the higher order gyrokinetic vorticity equation (5.40). The solution must be reached by time evolution of ϕ , f_i and \overline{f}_e . Employing the gyrokinetic Fokker-Planck equation (3.39) and the electron drift kinetic equation (3.49), the ion and electron distribution functions $f_i(\mathbf{R}, E, \mu, t)$ and $\overline{f}_e(\mathbf{r}, E_0, \mu_0, t)$ are evolved in time. To find the corresponding electrostatic potential, we first split it into the flux averaged component and the turbulent piece by, for example, Fourier analyzing it into toroidal and poloidal modes. For the turbulent pieces, we must employ the higher order gyrokinetic vorticity equation (5.40). The flux surface averaged component of the potential is obtained by evolving the axisymmetric toroidal rotation $\mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}}$ with equation (2.21) and the viscosity given in (5.10). Once the toroidal rotation is found, the axisymmetric radial electric field is obtained using the lower order result (4.4) to write

$$Rn_{i}\mathbf{V}_{i}\cdot\hat{\boldsymbol{\zeta}} = \int d^{3}v \, f_{ig}R(\mathbf{v}\cdot\hat{\boldsymbol{\zeta}}) - \int d^{3}v \, \frac{Ze\phi}{T_{i}} f_{Mi}R(\mathbf{v}_{\perp}\cdot\hat{\boldsymbol{\zeta}})$$
$$\xrightarrow{k_{\perp}\rho_{i}\rightarrow 0} IU(\psi) - \frac{cR^{2}}{Ze} \left(\frac{\partial p_{i}}{\partial\psi} + Zen_{i}\frac{\partial\phi}{\partial\psi}\right), \qquad (5.43)$$

where to write this last expression I have used the neoclassical relation (4.2). Here, f_{ig} depends implicitly on $\phi(\psi)$. Finally, the turbulent and flux surface averaged components of the electrostatic potential are added to obtain the total electric field, and the distribution functions f_i and \overline{f}_e may be evolved in time again. Notice that equation (5.10), employed here to find the radial electric field, requires that the radial wavelengths be longer than the ion gyroradius. This limitation is probably unimportant since the zonal flow is characterized by $k_{\perp}\rho_i \sim 0.1$ [12].

Chapter 6

Conclusions

In this thesis, I have proven that the current gyrokinetic treatments, composed of a gyrokinetic Fokker-Planck equation and a gyrokinetic quasineutrality equation, cannot provide the long wavelength, axisymmetric radial electric field. Employing the vorticity equation (2.9), I first showed that setting the radial current to zero to obtain the axisymmetric radial electric field is equivalent to solving the toroidal angular momentum conservation equation, given in (2.21).

In chapter 3, I present a new derivation of electrostatic gyrokinetics that generalizes the linear treatment of [58, 59]. This derivation is useful in chapter 4 to study the current conservation or vorticity equation in steady state turbulence. To simplify the problem, I assume that, in statistical equilibrium, the turbulent fluctuations within a flux surface must be small by $\delta_i = \rho_i/L$ because of the fast transport along magnetic field lines. Then, the long wavelength axisymmetric flows must remain neoclassical, and the tokamak is intrinsically ambipolar even in the presence of turbulence, i.e., $\langle \mathbf{J} \cdot \nabla \psi \rangle_{\psi} \simeq 0$ for any long wavelength axisymmetric radial electric field. According to the estimate in (4.9), the radial current density associated with transport of toroidal angular momentum is so small that modern gyrokinetic treatments are unable to selfconsistently calculate the long wavelength radial electric field. For most codes, long wavelengths are those above $\sqrt{\rho_i L}$.

To solve this issue, I propose to solve a vorticity equation instead of the gyrokinetic quasineutrality equation. The vorticity equation has the advantage of showing explicitly the dependence on the transport of toroidal angular momentum. I have derived two approximate gyrokinetic vorticity equations. Vorticity equation (4.45) is similar to the gyrokinetic quasineutrality equation, and vorticity equation (4.53), on the other hand, is constructed to resemble the full vorticity equation (2.9). The two gyrokinetic vorticity equations (4.45) and (4.53) are equivalent to the full vorticity equation (2.9) within terms of order δ_i . The long wavelength radial electric field cannot be found from these gyrokinetic vorticity equations because they are missing crucial terms. However, they satisfy a very desirable property explicitly, namely, the long wavelength toroidal velocity tends to be constant for the short turbulence saturation time scales.

To complement the gyrokinetic vorticity equation, I propose using the conservation equation for the toroidal angular momentum (2.21), where the toroidal-radial component of the ion viscosity $\langle R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i \cdot \nabla \psi \rangle_{\psi}$ is given to order $\delta_i^3 p_i$ in (5.10). Unfortunately, expression (5.10) requires a distribution function and a potential of order higher than calculated in gyrokinetic codes. In section 5.3, I show that, for $q \gg 1$, the ion distribution function can be found to high enough order by employing the full f gyrokinetic equation (3.39). The gyrokinetic vorticity equations (4.45) and (4.53), however, have to be extended to determine the higher order potential. Equation (5.38) and (5.40) are the higher order versions of (4.45) and (4.53).

To summarize, to obtain the self-consistent electric field, it is necessary to refrain from using the lower order gyrokinetic quasineutrality equation. Instead, the electric field has to be found by employing a higher order formulation like the proposed vorticity equation (2.9). Moreover, since the axisymmetric contributions to the vorticity equation are equivalent to the conservation of toroidal angular momentum, only the non-axisymmetric pieces of the potential must be found by employing higher order gyrokinetic vorticity equations like equation (5.38) or equation (5.40). The axisymmetric electric field should be found from (2.21) employing the toroidal-radial component of the ion viscosity in (5.10). At the same time, the ion distribution function evolves according to (3.39). In order to implement this methodology, I foresee several steps. First, the lowest order gyrokinetic vorticity equations (4.45) and (4.53) should be implemented in δf flux tube codes. These codes are well understood and easy to study. Equation (4.45) is appealing since it is very similar to the gyrokinetic quasineutrality equation. If the vorticity equations show good numerical behavior in δf flux tube codes, they should be then implemented in full f codes. For runs that stay below transport time scales, these vorticity equations are still valid. Finally, the transport of toroidal angular momentum given in (5.10) must be studied. To do so, it is necessary to calculate the potential and distribution function to higher order, requiring then the higher order versions of the gyrokinetic vorticity equations (5.38) and (5.40). Again, it will probably be easier to study these equations and the transport of toroidal angular momentum in δf flux tube codes first, and then, finally, implement this method in full f models.

•

Appendix A

Derivation of the gyrokinetic variables

In this Appendix the detailed calculation of the gyrokinetic variables is carried out. In section A.1, the gyrokinetic variables are computed to first order in δ_i . In section A.2, the gyrokinetic variables **R** and *E* are extended to second order, and the gyrokinetic magnetic moment μ is proven to be an adiabatic invariant to higher order. Finally, in section A.3, the Jacobian of the transformation from the variables **r**, **v** to the gyrokinetic variables is calculated. The Jacobian is employed to write the gyrokinetic equation in conservative form.

A.1 First order gyrokinetic variables

It is convenient to express any term that contains the electrostatic potential ϕ in gyrokinetic variables, mainly because the electrostatic potential components with $k_{\perp}\rho_i \sim 1$ cannot be Taylor expanded. In order to do so, I will develop some useful relations involving the potential ϕ in subsection A.1.1. With these relations, the first order corrections, \mathbf{R}_1 , E_1 , φ_1 and μ_1 , are derived.

A.1.1 Useful relations for ϕ

I first derive all possible gyrokinetic partial derivatives of ϕ and their relation to one another. To do so, only $\mathbf{R} = \mathbf{r} + \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}} + O(\delta_i^2 L)$ is needed.

The derivative respect to the gyrocenter position is

$$\nabla_{\mathbf{R}}\phi(\mathbf{r}) = \nabla\phi + \nabla_{\mathbf{R}}(\mathbf{r} - \mathbf{R}) \cdot \nabla\phi = \nabla\phi + O(\delta_i T_e/eL) \simeq \nabla\phi.$$
(A.1)

The derivative respect to the energy is

$$\frac{\partial \phi}{\partial E} = \frac{\partial}{\partial E} (\mathbf{r} - \mathbf{R}) \cdot \nabla \phi = O(\delta_i^2 M/e) \simeq 0, \qquad (A.2)$$

since $\mathbf{r} - \mathbf{R}$ only depends on E at $O(\delta_i^2 L)$.

Using $\mathbf{r} - \mathbf{R} \propto \sqrt{\mu} (\hat{\mathbf{e}}_1 \sin \varphi - \hat{\mathbf{e}}_2 \cos \varphi)$, the derivatives with respect to μ and φ are calculated to be

$$\frac{\partial \phi}{\partial \mu} = \frac{\partial}{\partial \mu} (\mathbf{r} - \mathbf{R}) \cdot \nabla \phi \simeq -\frac{Mc}{Zev_{\perp}^2} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi \tag{A.3}$$

and

$$\frac{\partial \phi}{\partial \varphi} = \frac{\partial}{\partial \varphi} (\mathbf{r} - \mathbf{R}) \cdot \nabla \phi \simeq -\frac{1}{\Omega_i} \mathbf{v}_{\perp} \cdot \nabla \phi.$$
(A.4)

I will need more accurate relationship than (A.1) and (A.4) for the second order corrections. They will be developed in subsection A.2.1.

A.1.2 Calculation of R_1

The first order correction \mathbf{R}_1 is given by (3.9), where in this case, $Q_0 = \mathbf{R}_0 = \mathbf{r}$. The total derivative of \mathbf{R}_0 is $d\mathbf{R}_0/dt = \mathbf{v} = v_{||}\hat{\mathbf{b}} + \mathbf{v}_{\perp}$, and its gyroaverage gives $\langle d\mathbf{R}_0/dt \rangle = v_{||}\hat{\mathbf{b}} + O(\delta_i v_i)$. By employing $\mathbf{v}_{\perp} = \partial(\mathbf{v} \times \hat{\mathbf{b}})/\partial\varphi_0$, equation (3.9) gives (3.14).

A.1.3 Calculation of E_1

The first order correction E_1 is given by (3.9), where $Q_0 = E_0 = v^2/2$ and $dQ_0/dt = dE_0/dt = -(Ze/M)\mathbf{v} \cdot \nabla \phi$. It is convenient to write E_1 as a function of \mathbf{R} , E, μ and φ . To do so, I use (A.1) and (A.4) to find

$$-\mathbf{v} \cdot \nabla \phi = -v_{||} \hat{\mathbf{b}} \cdot \nabla \phi - \mathbf{v}_{\perp} \cdot \nabla \phi \simeq -v_{||} \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \phi + \Omega_i \frac{\partial \phi}{\partial \varphi}.$$
 (A.5)

Notice that $\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\phi} \ll \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle$ because $\tilde{\phi}$ is smaller than $\langle \phi \rangle$. As a result, $dE_0/dt \simeq -(Ze/M)v_{||}\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + (Ze\Omega_i/M)\partial\phi/\partial\varphi$ and $\langle dE_0/dt \rangle = -(Ze/M)v_{||}\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + O(\delta_i v_i^3/L)$. Then, equation (3.9) gives (3.25).

A.1.4 Calculation of φ_1

The first order correction φ_1 is given by (3.9), where $Q_0 = \varphi_0$. The zeroth order gyrophase φ_0 is defined by equation (2.2). According to this definition, upon using $\nabla_v \varphi_0 = -v_{\perp}^{-2} \mathbf{v} \times \hat{\mathbf{b}}$ and $\nabla \varphi_0 = (v_{\parallel}/v_{\perp}^2) \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) + \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1$, the total derivative of φ_0 is

$$\frac{d\varphi_{0}}{dt} = -\overline{\Omega}_{i} - \frac{Z^{2}e^{2}}{M^{2}c}\frac{\partial\widetilde{\phi}}{\partial\mu} + (\mathbf{v}\times\hat{\mathbf{b}})\cdot\left[\nabla\ln\Omega_{i} + \frac{v_{||}^{2}}{v_{\perp}^{2}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}} - \hat{\mathbf{b}}\times\nabla\hat{\mathbf{e}}_{2}\cdot\hat{\mathbf{e}}_{1}\right] \\
+ \frac{v_{||}}{2v_{\perp}^{2}}[\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}}) + (\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}]:\nabla\hat{\mathbf{b}}, \quad (A.6)$$

where the potential $\phi(\mathbf{r}, t)$ and the gyrofrequency $\Omega_i(\mathbf{r})$ have been written as functions of the gyrokinetic variables by using (A.3) and $\Omega_i(\mathbf{r}) \simeq \Omega_i(\mathbf{R}) + (\mathbf{r} - \mathbf{R}) \cdot \nabla \Omega_i$, respectively, and I have used the relations $\langle \mathbf{v}_{\perp} \mathbf{v}_{\perp} \rangle \simeq \overline{\mathbf{v}_{\perp} \mathbf{v}_{\perp}} = (v_{\perp}^2/2) (\mathbf{I} - \mathbf{\hat{b}}\mathbf{\hat{b}})$ and

$$\mathbf{v}_{\perp}\mathbf{v}_{\perp} - \langle \mathbf{v}_{\perp}\mathbf{v}_{\perp} \rangle = \frac{1}{2} [\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})].$$
(A.7)

Here, $\overline{(\ldots)}$ is the gyroaverage holding \mathbf{r} , v_{\parallel} , v_{\perp} and t fixed, and $\langle \ldots \rangle$ is the gyroaverage holding \mathbf{R} , E, μ and t fixed. These two gyroaverages are equivalent in this case because the functions involved do not have short wavelengths. A detailed derivation of (A.7) and other velocity relations can be found in Appendix B.

In equation (A.6), the function $\overline{\Omega}_i$ is given by equation (3.31). Upon gyroaveraging (A.6), I obtain $\langle d\varphi_0/dt \rangle = -\overline{\Omega}_i + O(\delta_i^2 \Omega_i)$. Finally, φ_1 is obtained from (3.9) by employing $\mathbf{v} \times \hat{\mathbf{b}} = -\partial \mathbf{v}_{\perp}/\partial \varphi_0$ and

$$\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) = \frac{1}{2} \frac{\partial}{\partial \varphi_0} [\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp}], \quad (A.8)$$

giving equation (3.29). Relation (A.8) is proven in Appendix B.

A.1.5 Calculation of μ_1

Calculating μ_1 requires more work than calculating any of the other first order corrections since we want μ to be an adiabatic invariant to all orders of interest. This requirement imposes two conditions to μ_1 . One of them is similar to the requirements already imposed to \mathbf{R}_1 , E_1 and φ_1 , $d\mu_0/dt - \Omega_i(\partial\mu_1/\partial\varphi) = \langle d\mu_0/dt \rangle = 0$, but there is an additional condition making $\mu_0 + \mu_1$ an adiabatic invariant to first order,

$$\left\langle \frac{d}{dt}(\mu_0 + \mu_1) \right\rangle = O\left(\delta_i^2 \frac{v_i^3}{BL}\right). \tag{A.9}$$

The solution to both conditions is given by

$$\mu_1 = \frac{1}{\Omega_i} \int^{\varphi} d\varphi' \left(\frac{d\mu_0}{dt} - \left\langle \frac{d\mu_0}{dt} \right\rangle \right) + \langle \mu_1 \rangle. \tag{A.10}$$

Notice that the only difference with the result in (3.9) is that the gyrophase independent term, $\langle \mu_1 \rangle$, must be retained, making it possible to satisfy condition (A.9).

Employing $\nabla_v \mu_0 = \mathbf{v}_{\perp}/B$ and $\nabla \mu_0 = -(v_{\perp}^2/2B^2)\nabla B - (v_{\parallel}/B)\nabla \hat{\mathbf{b}} \cdot \mathbf{v}$, I find that the total derivative for $\mu_0 = v_{\perp}^2/2B$ is

$$\frac{d\mu_{0}}{dt} = -\frac{Ze}{MB}\mathbf{v}_{\perp} \cdot \nabla\phi - \frac{v_{\perp}^{2}}{2B^{2}}\mathbf{v}_{\perp} \cdot \nabla B - \frac{v_{\parallel}^{2}}{B}\hat{\mathbf{b}} \cdot \nabla\hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} -\frac{v_{\parallel}}{2B}[\mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})]:\nabla\hat{\mathbf{b}}, \qquad (A.11)$$

where I have used the relations $\langle \mathbf{v}_{\perp} \mathbf{v}_{\perp} \rangle \simeq \overline{\mathbf{v}_{\perp} \mathbf{v}_{\perp}} = (v_{\perp}^2/2) (\mathbf{\hat{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}})$ and (A.7).

Notice that the gyrophase independent terms in (A.11) cancel exactly due to

 $\hat{\mathbf{b}} \cdot \nabla \ln B + \nabla \cdot \hat{\mathbf{b}} = 0$, making μ_0 an adiabatic invariant to zeroth order. The term that contains ϕ in (A.11) is rewritten as a function of the gyrokinetic variables by using (A.4), to give $-(Ze/MB)\mathbf{v}_{\perp} \cdot \nabla \phi = (Ze\Omega_i/MB)\partial \phi/\partial \varphi$.

Applying (A.10), μ_1 is found to be given by (3.34). To get this result, I have employed $\mathbf{v}_{\perp} = \partial(\mathbf{v} \times \hat{\mathbf{b}})/\partial\varphi_0$ and (A.8). The average value $\langle \mu_1 \rangle = -(v_{\parallel}v_{\perp}^2/2B\Omega_i)(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})$ was chosen to ensure that condition (A.9) is satisfied. In previous works [19, 73], it has been noticed that solving (A.9) may be avoided and replaced by imposing the relation $E = [d\mathbf{R}/dt \cdot \hat{\mathbf{b}}(\mathbf{R})]^2/2 + \mu B(\mathbf{R})$ on the gyrokinetic variables. This procedure works in this case, and allows me to find $\langle \mu_1 \rangle$. I will prove that the chosen $\langle \mu_1 \rangle$ satisfies condition (A.9) in subsection A.2.4.

A.2 Second order gyrokinetic variables

To construct the gyrokinetic variables to second order, higher order relations than the ones developed in subsection A.1.1 are needed to express ϕ as a function of the gyrokinetic variables. These extended relations are deduced in subsection A.2.1. Using them, the second order corrections \mathbf{R}_2 and E_2 and the gyrophase independent piece of the first order correction $\langle \mu_1 \rangle$ are calculated. The magnetic moment and the gyrophase are not required to higher order.

A.2.1 More useful relations for ϕ

To calculate the second order correction E_2 and the gyrophase independent piece $\langle \mu_1 \rangle$, the expressions $\hat{\mathbf{b}} \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi)$ and $\mathbf{v} \cdot \nabla \phi$ must be given in gyrokinetic variables to order $\delta_i T_e/eL$ and $\delta_i T_e v_i/eL$, respectively.

For $\hat{\mathbf{b}} \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi)$, I use $\nabla \phi = \nabla \mathbf{R} \cdot \nabla_{\mathbf{R}} \phi + \nabla E(\partial \phi / \partial E) + \nabla \mu (\partial \phi / \partial \mu) + \nabla \varphi (\partial \phi / \partial \varphi)$ to write

$$\hat{\mathbf{b}} \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi) \simeq \hat{\mathbf{b}} \cdot \nabla \mathbf{R}_1 \cdot \nabla_{\mathbf{R}} \phi + \hat{\mathbf{b}} \cdot \nabla \mu_0 \frac{\partial \phi}{\partial \mu} + \hat{\mathbf{b}} \cdot \nabla \varphi_0 \frac{\partial \phi}{\partial \varphi}, \qquad (A.12)$$

where I have neglected higher order terms. Here, it is important that the gradient is

parallel to the magnetic field, since \mathbf{R}_2 , μ_1 and φ_1 have pieces with short perpendicular wavelengths that are important for the perpendicular component of the gradient. Employing $\nabla \mathbf{R}_1 = -\Omega_i^{-1}[(\nabla \ln B)(\mathbf{v} \times \hat{\mathbf{b}}) + \nabla \hat{\mathbf{b}} \times \mathbf{v}], \nabla \mu_0 = -(v_\perp^2/2B^2)\nabla B - (v_{\parallel}/B)\nabla \hat{\mathbf{b}} \cdot \mathbf{v},$ $\nabla \varphi_0 = (v_{\parallel}/v_\perp^2)\nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) + \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1$ and the lowest order relations (A.3) and (A.4) for $\partial \phi/\partial \mu$ and $\partial \phi/\partial \varphi$, I obtain

$$\hat{\mathbf{b}} \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi) = -\frac{1}{2B\Omega_i} (\hat{\mathbf{b}} \cdot \nabla B) (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$$
$$-\frac{1}{\Omega_i} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) (\hat{\mathbf{b}} \cdot \nabla \phi) + \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1 \frac{\partial \phi}{\partial \varphi}.$$
(A.13)

To find this result, I have used

$$-\frac{v_{||}}{B}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot\mathbf{v}_{\perp}\frac{\partial\phi}{\partial\mu}+\frac{v_{||}}{v_{\perp}^{2}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot(\mathbf{v}\times\hat{\mathbf{b}})\frac{\partial\phi}{\partial\varphi}=\frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot(\hat{\mathbf{b}}\times\nabla\phi),\qquad(A.14)$$

where I employ (A.3) and (A.4) for $\partial \phi / \partial \mu$ and $\partial \phi / \partial \varphi$, and the relation $\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) - (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp} = v_{\perp}^{2}(\mathbf{\vec{I}} \times \hat{\mathbf{b}})$. This relation is obtained from the fact that \mathbf{v}_{\perp} and $\mathbf{v} \times \hat{\mathbf{b}}$ expand the vector space perpendicular to the magnetic field, giving $\mathbf{v}_{\perp}\mathbf{v}_{\perp} + (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) = v_{\perp}^{2}(\mathbf{\vec{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}})$.

To calculate $\mathbf{v} \cdot \nabla \phi$, I use that the total time derivative for ϕ in \mathbf{r} , \mathbf{v} variables is

$$\frac{d\phi}{dt} = \left. \frac{\partial\phi}{\partial t} \right|_{\mathbf{r}} + \mathbf{v} \cdot \nabla\phi, \tag{A.15}$$

while as a function of the new gyrokinetic variables it becomes

$$\left. \frac{d\phi}{dt} = \left. \frac{\partial\phi}{\partial t} \right|_{\mathbf{R}, E, \mu, \varphi} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} \phi + \dot{E} \frac{\partial\phi}{\partial E} + \dot{\varphi} \frac{\partial\phi}{\partial\varphi}.$$
(A.16)

Combining these equations gives an equation for $\mathbf{v} \cdot \nabla \phi$,

$$-\mathbf{v}\cdot\nabla\phi = \left(\frac{\partial\phi}{\partial t}\Big|_{\mathbf{r}} - \frac{\partial\phi}{\partial t}\Big|_{\mathbf{R}, E, \mu, \varphi}\right) - \dot{E}\frac{\partial\phi}{\partial E} - \dot{\mathbf{R}}\cdot\nabla_{\mathbf{R}}\phi - \dot{\varphi}\frac{\partial\phi}{\partial\varphi},\tag{A.17}$$

where the left side of the equation is of order $O(T_e v_i/eL)$. I analyze the right side term by term, keeping terms up to order $\delta_i T_e v_i/eL$. Noticing that $\phi(\mathbf{r}, t) = \phi(\mathbf{R} + (\mathbf{r} - \mathbf{R}), t)$, the partial derivatives with respect to time give the negligible contribution $(\partial \phi / \partial t|_{\mathbf{r}} - \partial \phi / \partial t|_{\mathbf{R}, E, \mu, \varphi}) = -\partial (\mathbf{r} - \mathbf{R}) / \partial t \cdot \nabla \phi = O(\delta_i^2 T_e v_i / eL)$, since the time derivative of $\mathbf{r} - \mathbf{R}$ can only be of order $\delta_i^2 v_i$ for a static magnetic field. The partial derivative with respect to E is estimated in (A.2), giving that $\dot{E} \partial \phi / \partial E = O(\delta_i^2 T_e v_i / eL)$ is negligible. The total derivative $\dot{\mathbf{R}}$ has two different components, which I will calculate in detail in subsection A.2.2. These components are the parallel velocity of the gyrocenter, $u\hat{\mathbf{b}}(\mathbf{R})$, of order v_i , and the drift velocity, \mathbf{v}_d , of order $\delta_i v_i$. Using this information, I find $u\hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \phi = O(T_e v_i / eL)$ and $\mathbf{v}_d \cdot \nabla_{\mathbf{R}} \phi = O(\delta_i T_e v_i / eL)$. Finally, the last term in the right side of (A.17) is $\dot{\varphi}(\partial \phi / \partial \varphi) = O(T_e v_i / eL)$, since $\dot{\varphi} \sim \Omega_i$ and $\partial \phi / \partial \varphi = \partial \tilde{\phi} / \partial \varphi \sim \delta_i T_e / e$ according to (3.17). Neglecting all the terms smaller than $\delta_i T_e v_i / eL$, equation (A.17) becomes

$$-\mathbf{v} \cdot \nabla \phi = -u \hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \phi - \mathbf{v}_d \cdot \nabla_{\mathbf{R}} \phi + \overline{\Omega}_i \frac{\partial \phi}{\partial \varphi}.$$
 (A.18)

A.2.2 Calculation of R_2

The second order correction \mathbf{R}_2 is given by (3.10), where $Q_0 = \mathbf{R}_0 = \mathbf{r}$ and $Q_1 = \mathbf{R}_1 = \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$. The total time derivative of $\mathbf{R}_0 + \mathbf{R}_1$ is

$$\frac{d}{dt}(\mathbf{R}_0 + \mathbf{R}_1) = v_{||}\hat{\mathbf{b}} - \mathbf{v} \cdot \nabla\left(\frac{\hat{\mathbf{b}}}{\Omega_i}\right) \times \mathbf{v} - \frac{c}{B}\nabla\phi \times \hat{\mathbf{b}}, \qquad (A.19)$$

and its gyroaverage may be written as $\langle d(\mathbf{R}_0 + \mathbf{R}_1)/dt \rangle = u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d$, where $u = \langle v_{||} \rangle + (v_{\perp}^2/2\Omega_i)(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})$, and \mathbf{v}_d has been already defined in (3.20). The function u can be written as a function of the gyrokinetic variables. I express $v_{||}$ as a function of \mathbf{r} , E_0 and μ_0 , expand around \mathbf{R} , E and μ , and insert \mathbf{R}_1 , μ_1 and E_1 to obtain

$$v_{||} = \sqrt{2(E_0 - \mu_0 B(\mathbf{r}))} \simeq \sqrt{2(E - \mu B(\mathbf{R}))} - \frac{v_{\perp}^2}{2\Omega_i} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$$
$$-\frac{v_{||}}{\Omega_i} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) - \frac{1}{4\Omega_i} [\mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp}] : \nabla \hat{\mathbf{b}}.$$
(A.20)

Finally, gyroaveraging and using $\langle \mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp} \rangle = 0$ [a result that is deduced from (A.7)] give $u = \sqrt{2[E - \mu B(\mathbf{R})]}$, which can be rewritten as (3.23).

Using (A.19) and (A.20), Taylor expanding $\hat{\mathbf{b}}(\mathbf{r})$ about **R** and inserting the result into (3.10) gives (3.15) and (3.19). To integrate over gyrophase, $\mathbf{v} \times \hat{\mathbf{b}} = -\partial \mathbf{v}_{\perp} / \partial \varphi_0$ and (A.8) have been used.

A.2.3 Calculation of E_2

Equation (3.10) gives E_2 , where $Q_0 = E_0 = v^2/2$ and $Q_1 = E_1 = Ze\tilde{\phi}/M$. The total derivative of $E_0 = v^2/2$ can be expressed as a function of the new gyrokinetic variables to the requisite order by using (A.18) to obtain

$$\frac{dE_0}{dt} = -\frac{Ze}{M} \mathbf{v} \cdot \nabla \phi \simeq \frac{Ze}{M} \left\{ \overline{\Omega}_i \frac{\partial \widetilde{\phi}}{\partial \varphi} - [u \hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} \phi \right\}.$$
(A.21)

From the definition of $E_1 = Z e \tilde{\phi}/M$, use of gyrokinetic variables yields

$$\frac{dE_1}{dt} = \frac{Ze}{M} \left\{ \frac{\partial \widetilde{\phi}}{\partial t} + [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} \widetilde{\phi} - \overline{\Omega}_i \frac{\partial \widetilde{\phi}}{\partial \varphi} \right\}.$$
 (A.22)

Adding both contributions together leaves

$$\frac{d}{dt}(E_0 + E_1) = -\frac{Ze}{M} [u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + \frac{Ze}{M} \frac{\partial \tilde{\phi}}{\partial t}.$$
 (A.23)

As a result, E_2 is as shown in (3.26), and to this order, dE/dt is given by (3.27).

A.2.4 Calculation of $\langle \mu_1 \rangle$

In this subsection, I will check that $\langle \mu_1 \rangle = -(v_{\parallel}v_{\perp}^2/2B\Omega_i)\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$ satisfies the condition in (A.9). To do so, it is going to be useful to distinguish between the part of $d(\mu_0 + \mu_1)/dt$ that depends on ϕ and the part that does not depend on ϕ at all since these pieces will vanish independently of each other. I will write the piece of $d(\mu_0 + \mu_1)/dt$ that depends on ϕ as a function of the gyrokinetic variables, finding

$$\left. \frac{d}{dt} (\mu_0 + \mu_1) \right|_{\phi} = -\frac{Ze}{M} \nabla \phi \cdot \nabla_v \mu_0 - \frac{Ze}{M} \nabla \phi \cdot \nabla_v \mu_1 |_{\mathbf{r}, \mathbf{v}} + \frac{d}{dt} \mu_1 |_{\phi}, \tag{A.24}$$

with $\mu_1|_{\phi} = Ze\tilde{\phi}/MB$ and $\mu_1|_{\mathbf{r},\mathbf{v}} = \mu_1 - Ze\tilde{\phi}/MB$. The piece of $d(\mu_0 + \mu_1)/dt$ that does not depend on ϕ is

$$\left. \frac{d}{dt} (\mu_0 + \mu_1) \right|_{\mathbf{r}, \mathbf{v}} = \mathbf{v} \cdot \nabla \mu_0 - \Omega_i \frac{\partial}{\partial \varphi_0} \mu_1 |_{\mathbf{r}, \mathbf{v}} + \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}}.$$
(A.25)

In this equation, the two first terms cancel by definition of $\mu_1|_{\mathbf{r},\mathbf{v}}$, leaving

$$\left. \frac{d}{dt} (\mu_0 + \mu_1) \right|_{\mathbf{r}, \mathbf{v}} = \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}}.$$
(A.26)

I will first prove that the gyroaverage of (A.26), $\langle \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}} \rangle$, vanishes to $O(\delta_i v_i^3 / BL)$ due to the choice of $\langle \mu_1 \rangle$. Afterwards, I will prove that the gyroaverage of (A.24) vanishes to the same order, demonstrating then that $\mu_0 + \mu_1$ satisfies condition (A.9).

To prove that $\langle \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}} \rangle = 0$, the function $\mu_1 |_{\mathbf{r}, \mathbf{v}}$, is conveniently rewritten as

$$\mu_{1}|_{\mathbf{r},\mathbf{v}} = -\frac{v_{\perp}^{2}}{2B^{2}\Omega_{i}}(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla B - \frac{v_{\parallel}^{2}}{B\Omega_{i}}\mathbf{v}_{\perp}\cdot\nabla\times\hat{\mathbf{b}}$$
$$-\frac{v_{\parallel}}{2B\Omega_{i}}(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\hat{\mathbf{b}}\cdot\mathbf{v} - \frac{v_{\parallel}v_{\perp}^{2}}{4B\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}, \qquad (A.27)$$

where I use $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} = \boldsymbol{\kappa}$ and (2.8) to write $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) = \mathbf{v}_{\perp} \cdot \nabla \times \hat{\mathbf{b}}$, and I employ equation (A.7) and $\langle \mathbf{v}_{\perp} \mathbf{v}_{\perp} \rangle = (v_{\perp}^2/2) (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}})$ to find

$$\frac{1}{2}[\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}}) + (\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}]:\nabla\hat{\mathbf{b}} = (\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\hat{\mathbf{b}}\cdot\mathbf{v}_{\perp} - \frac{v_{\perp}^{2}}{2}\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}.$$
 (A.28)

I will examine term by term the gyroaverage of $\mathbf{v} \cdot \nabla$ [expression (A.27)]. Employing $\mathbf{v} \cdot \nabla (v_{\perp}^2/2) = -\mathbf{v} \cdot \nabla (v_{\parallel}^2/2) = -v_{\parallel} \mathbf{v} \cdot \nabla v_{\parallel}, \ \mathbf{v} \cdot \nabla v_{\parallel} = \mathbf{v} \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v} \text{ and } \langle \mathbf{v} \mathbf{v} \rangle = (v_{\perp}^2/2) \stackrel{\leftrightarrow}{\mathbf{I}} + [v_{\parallel}^2 - (v_{\perp}^2/2)] \hat{\mathbf{b}} \hat{\mathbf{b}}, \text{ the first, second and fourth terms in (A.27) give}$

$$\left\langle \mathbf{v} \cdot \nabla \left[\frac{v_{\perp}^{2}}{2B^{2}\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B \right] \right\rangle = -\frac{v_{\parallel}^{2} v_{\perp}^{2}}{2B^{2}\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\hat{\mathbf{b}} \times \nabla B) + \frac{v_{\perp}^{4}}{4} \nabla \cdot \left(\frac{\hat{\mathbf{b}} \times \nabla B}{B^{2}\Omega_{i}} \right) + \frac{v_{\perp}^{2}}{2} \left(v_{\parallel}^{2} - \frac{v_{\perp}^{2}}{2} \right) \hat{\mathbf{b}} \cdot \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla B}{B^{2}\Omega_{i}} \right) \cdot \hat{\mathbf{b}},$$
(A.29)

$$\left\langle \mathbf{v} \cdot \nabla \left(\frac{v_{\parallel}^{2}}{B\Omega_{i}} \mathbf{v}_{\perp} \cdot \nabla \times \hat{\mathbf{b}} \right) \right\rangle = \frac{v_{\parallel}^{2} v_{\perp}^{2}}{B\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}})$$
$$+ \frac{v_{\parallel}^{2} v_{\perp}^{2}}{2} \nabla \cdot \left[\frac{(\nabla \times \hat{\mathbf{b}})_{\perp}}{B\Omega_{i}} \right] + v_{\parallel}^{2} \left(v_{\parallel}^{2} - \frac{v_{\perp}^{2}}{2} \right) \hat{\mathbf{b}} \cdot \nabla \left[\frac{(\nabla \times \hat{\mathbf{b}})_{\perp}}{B\Omega_{i}} \right] \cdot \hat{\mathbf{b}}$$
(A.30)

and

$$\left\langle \mathbf{v} \cdot \nabla \left(\frac{v_{||} v_{\perp}^{2}}{4B\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \right) \right\rangle = \frac{v_{\perp}^{4}}{8B\Omega_{i}} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$$
$$-\frac{v_{||}^{2} v_{\perp}^{2}}{4B\Omega_{i}} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{v_{||}^{2} v_{\perp}^{2}}{4} \hat{\mathbf{b}} \cdot \nabla \left(\frac{\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}}{B\Omega_{i}} \right). \tag{A.31}$$

The contribution to $\langle \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}} \rangle$ of the third term in (A.27) is calculated by using $\mathbf{v} \cdot \nabla v_{||} = \mathbf{v} \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}, \ \langle \mathbf{v} \mathbf{v} \rangle = (v_{\perp}^2/2) \stackrel{\leftrightarrow}{\mathbf{I}} + [v_{||}^2 - (v_{\perp}^2/2)] \hat{\mathbf{b}} \hat{\mathbf{b}}$ and

$$\langle v_{\perp,j} v_{\perp,k} v_{\perp,l} v_{\perp,m} \rangle = \frac{v_{\perp}^4}{8} [(\delta_{jk} - \hat{b}_j \hat{b}_k) (\delta_{lm} - \hat{b}_l \hat{b}_m) + (\delta_{jl} - \hat{b}_j \hat{b}_l) (\delta_{km} - \hat{b}_k \hat{b}_m) + (\delta_{jm} - \hat{b}_j \hat{b}_m) (\delta_{kl} - \hat{b}_k \hat{b}_l)].$$
(A.32)

This result is proven in Appendix B. With these relations, the third term in (A.27) gives

$$\left\langle \mathbf{v} \cdot \nabla \left[\frac{v_{\parallel}}{2B\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v} \right] \right\rangle = \frac{v_{\perp}^{4}}{16B\Omega_{i}} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$$

$$+ \frac{v_{\perp}^{4}}{16B\Omega_{i}} \mathbf{i} : \left[(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \right] + \frac{v_{\perp}^{4}}{16B\Omega_{i}} \mathbf{i} : \left[(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot (\nabla \hat{\mathbf{b}})^{T} \right]$$

$$+ \frac{v_{\parallel}^{2}}{2} \left(v_{\parallel}^{2} - \frac{3}{2} v_{\perp}^{2} \right) \hat{\mathbf{b}} \cdot \left[\hat{\mathbf{b}} \cdot \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}}{B\Omega_{i}} \right) \cdot \hat{\mathbf{b}} \right] + \frac{v_{\parallel}^{2} v_{\perp}^{2}}{4} \hat{\mathbf{b}} \cdot \nabla \left(\frac{\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}}{B\Omega_{i}} \right)$$

$$+ \frac{v_{\parallel}^{2} v_{\perp}^{2}}{4} \nabla \cdot \left(\frac{\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}}{B\Omega_{i}} \right) \cdot \hat{\mathbf{b}} + \frac{v_{\parallel}^{2} v_{\perp}^{2}}{4} \nabla \cdot \left[\frac{(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}})^{T}}{B\Omega_{i}} \right] \cdot \hat{\mathbf{b}}, \quad (A.33)$$

with $(\nabla \hat{\mathbf{b}})^T$ and $(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}})^T$ the transposes of $\nabla \hat{\mathbf{b}}$ and $\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}$.

Several terms in equations (A.29), (A.30), (A.31) and (A.33) simplify because they vanish. In particular, equation (2.8), with $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} = \boldsymbol{\kappa}$, leads to $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}}) = 0$ and $\hat{\mathbf{b}} \cdot \nabla [(\nabla \times \hat{\mathbf{b}})_{\perp} / B\Omega_i] \cdot \hat{\mathbf{b}} = -(B\Omega_i)^{-1} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}}) = 0$. Also, $\hat{\mathbf{b}} \cdot \{\hat{\mathbf{b}} \cdot \nabla [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) / B\Omega_i] \cdot \hat{\mathbf{b}}\} = 0$, $\mathbf{i} : [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot (\nabla \hat{\mathbf{b}})^T] = -\nabla \cdot (\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \hat{\mathbf{b}} = 0$ and $\nabla \cdot [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) / B\Omega_i] \cdot \hat{\mathbf{b}} = 0$.

With these cancellations, I find that equations (A.30) and (A.33) reduce to

$$\left\langle \mathbf{v} \cdot \nabla \left(\frac{v_{||}^2}{B\Omega_i} \mathbf{v}_{\perp} \cdot \nabla \times \hat{\mathbf{b}} \right) \right\rangle = -\frac{v_{||}^2 v_{\perp}^2}{2B\Omega_i} \nabla \cdot (\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) -\frac{v_{||}^2 v_{\perp}^2}{B^2\Omega_i} (\nabla \times \hat{\mathbf{b}})_{\perp} \cdot \nabla B, \qquad (A.34)$$

where I have employed $\nabla \cdot [(\nabla \times \hat{\mathbf{b}})_{\perp}] = -\nabla \cdot (\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})$, and

$$\left\langle \mathbf{v} \cdot \nabla \left[\frac{v_{||}}{2B\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v} \right] \right\rangle = \frac{v_{\perp}^{4}}{8B\Omega_{i}} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + \frac{v_{||}^{2} v_{\perp}^{2}}{4B\Omega_{i}} \nabla \cdot (\hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}),$$
(A.35)

where I have used $B\Omega_i \hat{\mathbf{b}} \cdot \nabla [(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})/B\Omega_i] = \nabla \cdot (\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) - (\nabla \cdot \hat{\mathbf{b}} + 2\hat{\mathbf{b}} \cdot \nabla \otimes \hat{\mathbf{b}}) + (\nabla \cdot \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} + (\nabla \cdot \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \nabla \otimes \hat{\mathbf{b}}, B\Omega_i \nabla \cdot [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}})^T/B\Omega_i] \cdot \hat{\mathbf{b}} = - \overleftrightarrow{\mathbf{I}} : [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}] = -(\nabla \cdot \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \text{ and}$

$$\vec{\mathbf{I}}: [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}] = (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}.$$
(A.36)

To prove this last expression, I write $\mathbf{\hat{I}}$: $[(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}]$ as a divergence, giving

$$\vec{\mathbf{I}}: [(\hat{\mathbf{b}} \times \nabla \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}] = \nabla \cdot [(\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}) \times \hat{\mathbf{b}}] - \vec{\mathbf{I}}: (\hat{\mathbf{b}} \cdot \nabla \nabla \hat{\mathbf{b}} \times \hat{\mathbf{b}}) + \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}}).$$
(A.37)

Then, relation (A.36) is recovered by using $\vec{\mathbf{I}}$: $(\hat{\mathbf{b}} \cdot \nabla \nabla \hat{\mathbf{b}} \times \hat{\mathbf{b}}) = \hat{\mathbf{b}} \cdot \nabla (\nabla \times \hat{\mathbf{b}}) \cdot \hat{\mathbf{b}} = \nabla \cdot (\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) - (\nabla \cdot \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} - \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}})$ and equation (2.8) to write $(\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}) \times \hat{\mathbf{b}} = -(\nabla \times \hat{\mathbf{b}})_{\perp}$ and $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\nabla \times \hat{\mathbf{b}}) = 0$.

In addition to equations (A.34) and (A.35), equation (A.29) can be written as

$$\left\langle \mathbf{v} \cdot \nabla \left[\frac{v_{\perp}^{2}}{2B^{2}\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B \right] \right\rangle = \frac{v_{\parallel}^{2}v_{\perp}^{2}}{B^{2}\Omega_{i}} (\nabla \times \hat{\mathbf{b}})_{\perp} \cdot \nabla B$$
$$-\frac{v_{\perp}^{4}}{4B\Omega_{i}} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}, \tag{A.38}$$

where I use equation (2.8) to write $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\hat{\mathbf{b}} \times \nabla B) = -(\nabla \times \hat{\mathbf{b}})_{\perp} \cdot \nabla B$, and I employ $B^2 \Omega_i \nabla \cdot [(\hat{\mathbf{b}} \times \nabla B)/B^2 \Omega_i] = (\nabla \times \hat{\mathbf{b}}) \cdot \nabla B$, $B^2 \Omega_i \hat{\mathbf{b}} \cdot \nabla [(\hat{\mathbf{b}} \times \nabla B)/B^2 \Omega_i] \cdot \hat{\mathbf{b}} =$

 $-\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot(\hat{\mathbf{b}}\times\nabla B) = (\nabla\times\hat{\mathbf{b}})_{\perp}\cdot\nabla B$ and $\hat{\mathbf{b}}\cdot\nabla B = -B(\nabla\cdot\hat{\mathbf{b}})$. Finally, I add equations (A.31), (A.34), (A.35) and (A.38) to obtain

$$\langle \mathbf{v} \cdot \nabla \mu_1 |_{\mathbf{r}, \mathbf{v}} \rangle = \frac{v_{\parallel}^2 v_{\perp}^2}{4B\Omega_i} \left[\nabla \cdot (\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) + (\nabla \cdot \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} - B\Omega_i \hat{\mathbf{b}} \cdot \nabla \left(\frac{\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}}{B\Omega_i} \right) \right] = 0,$$
 (A.39)

the property I was trying to prove. Notice that the time derivative of $\langle \mu_1 \rangle = -(v_{\parallel}v_{\perp}^2/2B\Omega_i)\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$ was necessary to obtain this result. The derivative $d\langle \mu_1 \rangle/dt$ is in part responsible for the term (A.31).

There is still the piece of $d(\mu_0 + \mu_1)/dt$ that depends explicitly on the potential, given in (A.24). I will next prove that it also gyroaverages to zero, the desired result. The terms in (A.24) must be written as a function of the gyrokinetic variables in order to make the gyroaverage easier. I will do so for each term to the required order. The first term in (A.24) is $-(Ze/M)\nabla\phi\cdot\nabla_v\mu_0 = -(Ze/MB)\mathbf{v}_{\perp}\cdot\nabla\phi$. Using $\mathbf{v}_{\perp}\cdot\nabla\phi = \mathbf{v}\cdot\nabla\phi - v_{||}\hat{\mathbf{b}}\cdot\nabla\phi$ and relation (A.18), I find to order $\delta_i v_i^3/BL$

$$-\frac{Ze}{M}\nabla\phi\cdot\nabla_{v}\mu_{0} = \frac{Ze}{MB}[v_{||}\hat{\mathbf{b}}(\mathbf{r})\cdot\nabla\phi - u\hat{\mathbf{b}}(\mathbf{R})\cdot\nabla_{\mathbf{R}}\phi] \\ -\frac{Ze}{MB}\mathbf{v}_{d}\cdot\nabla_{\mathbf{R}}\phi + \frac{Ze\overline{\Omega}_{i}}{MB(\mathbf{r})}\frac{\partial\phi}{\partial\varphi}.$$
 (A.40)

In the lower order term $[Ze\overline{\Omega}_i/MB(\mathbf{r})](\partial\phi/\partial\varphi)$, the difference $B(\mathbf{r})-B(\mathbf{R}) \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B$ is important, giving

$$\frac{Ze\overline{\Omega}_i}{MB(\mathbf{r})}\frac{\partial\phi}{\partial\varphi} \simeq \frac{Ze\overline{\Omega}_i}{MB(\mathbf{R})}\frac{\partial\phi}{\partial\varphi} - \frac{c}{B^3}(\mathbf{v}_{\perp}\cdot\nabla\phi)[(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla B], \qquad (A.41)$$

where I employ the lowest order result $\partial \phi / \partial \varphi \simeq -\Omega_i^{-1} \mathbf{v}_{\perp} \cdot \nabla \phi$. The term $v_{\parallel} \hat{\mathbf{b}}(\mathbf{r}) \cdot \nabla \phi - u \hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \phi$ in (A.40) is to the order of interest

$$v_{||}\hat{\mathbf{b}}(\mathbf{r}) \cdot \nabla \phi - u\hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \phi \simeq (v_{||} - u)\hat{\mathbf{b}}(\mathbf{R}) \cdot \nabla_{\mathbf{R}} \phi$$
$$+ u[\hat{\mathbf{b}}(\mathbf{r}) - \hat{\mathbf{b}}(\mathbf{R})] \cdot \nabla_{\mathbf{R}} \phi + u\hat{\mathbf{b}}(\mathbf{R}) \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi), \qquad (A.42)$$

where $\hat{\mathbf{b}}(\mathbf{r}) - \hat{\mathbf{b}}(\mathbf{R}) \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}$ and $\hat{\mathbf{b}}(\mathbf{R}) \cdot (\nabla \phi - \nabla_{\mathbf{R}} \phi)$ is given in (A.13). The difference $v_{||} - u$ can be obtained from (A.20), giving

$$v_{||} - u = -\frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) - \frac{1}{4\Omega_{i}}[\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp}] : \nabla \hat{\mathbf{b}}$$
$$-\frac{v_{\perp}^{2}}{2\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} = -\frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) - \frac{1}{2\Omega_{i}}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp}$$
$$-\frac{v_{\perp}^{2}}{4\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}, \qquad (A.43)$$

where I use (A.28) to obtain the second equality. Employing (A.13), (A.41), (A.42) and the definition of $\overline{\Omega}_i$ in (3.31), equation (A.40) becomes

$$-\frac{Ze}{M}\nabla\phi\cdot\nabla_{v}\mu_{0} = \frac{Ze}{MB}(\hat{\mathbf{b}}\cdot\nabla\phi)\left[(v_{||}-u) - \frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot(\mathbf{v}\times\hat{\mathbf{b}})\right] -\frac{cv_{||}}{B^{2}}(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\hat{\mathbf{b}}\cdot\nabla\phi - \frac{cv_{||}}{2B^{3}}[(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\phi](\hat{\mathbf{b}}\cdot\nabla B) -\frac{c}{B^{3}}(\mathbf{v}_{\perp}\cdot\nabla\phi)[(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla B] - \frac{cv_{||}}{2B^{2}}(\mathbf{v}_{\perp}\cdot\nabla\phi)(\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}) -\frac{Ze}{MB}\mathbf{v}_{d}\cdot\nabla_{\mathbf{R}}\phi + \left(\frac{Z^{2}e^{2}}{M^{2}c} + \frac{Z^{3}e^{3}}{M^{3}cB}\frac{\partial\langle\phi\rangle}{\partial\mu}\right)\frac{\partial\phi}{\partial\varphi}.$$
 (A.44)

To obtain this form, I used the lowest order result $\partial \phi / \partial \varphi \simeq -\Omega_i^{-1} \mathbf{v}_{\perp} \cdot \nabla \phi$ in the term $(v_{\parallel}/2)(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})(\partial \phi / \partial \varphi).$

The second term in (A.24) is calculated employing (A.27), giving

$$-\frac{Ze}{M}\nabla\phi\cdot\nabla_{v}\mu_{1}|_{\mathbf{r},\mathbf{v}} = \frac{Ze}{MB}(\hat{\mathbf{b}}\cdot\nabla\phi)\left[\frac{v_{||}}{\Omega_{i}}\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{b}}\cdot(\mathbf{v}\times\hat{\mathbf{b}}) - (v_{||}-u)\right] \\ +\frac{c}{B^{3}}(\mathbf{v}_{\perp}\cdot\nabla\phi)[(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla B] + \frac{Ze}{MB}\mathbf{v}_{M}\cdot\nabla\phi \\ +\frac{cv_{||}}{2B^{2}}(\nabla\phi\times\hat{\mathbf{b}})\cdot\nabla\hat{\mathbf{b}}\cdot\mathbf{v}_{\perp} + \frac{cv_{||}}{2B^{2}}(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\hat{\mathbf{b}}\cdot\nabla\phi \\ +\frac{cv_{||}}{2B^{2}}(\mathbf{v}_{\perp}\cdot\nabla\phi)(\hat{\mathbf{b}}\cdot\nabla\times\hat{\mathbf{b}}), \quad (A.45)$$

with $v_{||} - u$ from (A.43). Substituting equations (A.44) and (A.45) in equation (A.24), the piece of $d(\mu_0 + \mu_1)/dt$ that depends on ϕ is written as

$$\frac{d}{dt}(\mu_0 + \mu_1) \bigg|_{\phi} = \left(\frac{Z^2 e^2}{M^2 c} + \frac{Z^3 e^3}{M^3 c B} \frac{\partial \langle \phi \rangle}{\partial \mu} \right) \frac{\partial \tilde{\phi}}{\partial \varphi}$$

$$+\frac{Zec}{MB^2}(\nabla_{\mathbf{R}}\langle\phi\rangle\times\hat{\mathbf{b}})\cdot\nabla_{\mathbf{R}}\tilde{\phi}+\frac{d}{dt}\mu_1|_{\phi}.$$
(A.46)

Here, I have used

$$(\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \nabla \phi = -(\nabla \cdot \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi.$$
(A.47)

To prove this last expression, I employ $\nabla_{\perp}\phi = v_{\perp}^{-2}[\mathbf{v}_{\perp}\mathbf{v}_{\perp}\cdot\nabla\phi + (\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\phi]$ to find

$$(\nabla \phi \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \nabla \phi = -\frac{1}{v_{\perp}^{2}} [(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi] [\mathbf{v}_{\perp} \mathbf{v}_{\perp} + (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}})] : \nabla \hat{\mathbf{b}}.$$
(A.48)

Upon using $\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}} = v_{\perp}^{-2}[\mathbf{v}_{\perp}\mathbf{v}_{\perp} + (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})]$, equation (A.47) is recovered.

Finally, the gyroaverage of equation (A.46) is zero. The term

$$\frac{d}{dt}\mu_1|_{\phi} = \frac{\partial}{\partial t}\mu_1|_{\phi} + \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}}\mu_1|_{\phi} + \dot{\varphi}\frac{\partial}{\partial\varphi}\mu_1|_{\phi}$$
(A.49)

vanishes when gyroaveraged because $\mu_1|_{\phi} = Ze\tilde{\phi}/MB(\mathbf{R})$ gyroaverages to zero and the gyrokinetic variables are defined such that $\dot{\mathbf{R}}$ and $\dot{\phi}$ are gyrophase independent. Notice that here it is important that $B(\mathbf{R})$ in $\mu_1|_{\phi} = Ze\tilde{\phi}/MB(\mathbf{R})$ depends on \mathbf{R} and not on \mathbf{r} .

A.3 Jacobian of the gyrokinetic transformation

In this section the Jacobian of the transformation from variables \mathbf{r} , \mathbf{v} to variables \mathbf{R} , E, μ , φ is calculated, and the gyroaverage of condition (3.40) is checked.

The inverse of the Jacobian is

$$\frac{1}{J} = \begin{vmatrix}
\ddots & & & \vdots & \vdots & \vdots \\
\nabla \mathbf{R} & \nabla E & \nabla \mu & \nabla \varphi \\
& & \ddots & \vdots & \vdots & \vdots \\
\nabla v \mathbf{R} & \nabla v E & \nabla v \mu & \nabla v \varphi \\
& & \ddots & \vdots & \vdots & \vdots \\
\nabla v \mathbf{R} & \nabla v E & \nabla v \mu & \nabla v \varphi \\
& & \ddots & \vdots & \vdots & \vdots \\
\end{pmatrix} = \begin{vmatrix}
\ddots & & & \vdots & \vdots & \vdots \\
\nabla \mathbf{R} & \nabla E & \nabla \mu & \nabla \varphi \\
& & \ddots & \vdots & \vdots & \vdots \\
\mathbf{0} & \partial E & \partial \mu & \partial \varphi \\
& & \ddots & \vdots & \vdots & \vdots \\
\end{matrix}$$
(A.50)

Employing that the terms in the left columns of the first form are to first approximation $\nabla \mathbf{R} \simeq \vec{\mathbf{I}}$ and $\nabla_v \mathbf{R} \simeq \Omega_i^{-1} \quad \vec{\mathbf{I}} \times \hat{\mathbf{b}}$, the determinant is simplified by combining linearly the rows in the matrix to determine the second form, where

$$\partial(\ldots) = \nabla_v(\ldots) - \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \nabla(\ldots).$$
 (A.51)

The second form of (A.50) can be simplified by noticing that the lower left piece of the matrix is zero. Thus, the determinant may be written as

$$J^{-1} = \det(\nabla \mathbf{R})[\partial E \cdot (\partial \mu \times \partial \varphi)]. \tag{A.52}$$

I analyze the two determinants on the right side independently. The matrix $\nabla \mathbf{R}$ is $\mathbf{i} + \nabla(\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}} + \mathbf{R}_2)$. Hence, $\det(\nabla \mathbf{R}) \simeq 1 + \nabla \cdot (\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}} + \mathbf{R}_2)$. The Jacobian must be obtained to first order only. The only important term to that order in \mathbf{R}_2 is the term that contains the potential ϕ , since its gradient may be large, but $\nabla \cdot \mathbf{R}_2 \simeq -\nabla \cdot [(c/B\Omega_i)\nabla_{\mathbf{R}}\tilde{\Phi} \times \hat{\mathbf{b}}] \simeq 0$. Therefore, the determinant of $\nabla \mathbf{R}$ becomes

$$\det(\nabla \mathbf{R}) = 1 - \mathbf{v} \cdot \nabla \times \left(\frac{\hat{\mathbf{b}}}{\Omega_i}\right). \tag{A.53}$$

For the second determinant in (A.52), I evaluate the columns of the matrix ∂E ,

 $\partial \mu$ and $\partial \varphi$ to the order of interest, using

$$\partial E = \mathbf{v} + \nabla_v E_1 - \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \nabla E,$$
 (A.54)

$$\partial \mu = \frac{\mathbf{v}_{\perp}}{B} + \nabla_{v} \mu_{1} - \frac{1}{\Omega_{i}} \hat{\mathbf{b}} \times \nabla \mu \tag{A.55}$$

and

$$\partial \varphi = -\frac{1}{v_{\perp}^2} \mathbf{v} \times \hat{\mathbf{b}} + \nabla_v \varphi_1 - \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \nabla \varphi.$$
 (A.56)

The determinant becomes

$$\partial E \cdot (\partial \mu \times \partial \varphi) \simeq \frac{v_{||}}{B(\mathbf{r})} + \frac{\hat{\mathbf{b}}}{B} \cdot \nabla_v E_1 + \left(\frac{v_{||}}{v_{\perp}^2} \mathbf{v}_{\perp} - \hat{\mathbf{b}}\right) \cdot \left(\nabla_v \mu_1 - \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \nabla \mu\right) \\ - \frac{v_{||}}{B} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \left(\nabla_v \varphi_1 - \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \nabla \varphi\right).$$
(A.57)

In the lower order term $v_{\parallel}/B(\mathbf{r})$, the difference $B(\mathbf{r})-B(\mathbf{R}) \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B \sim \delta_i B$ is important. From the definitions of E_1 , μ_1 and φ_1 , I find their gradients in velocity space. I need the gradients in velocity space of $\tilde{\phi}$ and $\partial \tilde{\Phi}/\partial \mu$. The gradient $\nabla_v \tilde{\phi}$ is given by

$$\nabla_{v}\tilde{\phi} = \nabla_{v}E\frac{\partial\tilde{\phi}}{\partial E} + \nabla_{v}\mu\frac{\partial\tilde{\phi}}{\partial\mu} + \nabla_{v}\varphi\frac{\partial\tilde{\phi}}{\partial\varphi} + \nabla_{v}\mathbf{R}\cdot\nabla_{\mathbf{R}}\tilde{\phi}$$
$$= \frac{\mathbf{v}_{\perp}}{B}\frac{\partial\tilde{\phi}}{\partial\mu} - \frac{1}{v_{\perp}^{2}}\mathbf{v}\times\hat{\mathbf{b}}\frac{\partial\tilde{\phi}}{\partial\varphi} + \frac{1}{\Omega_{i}}\hat{\mathbf{b}}\times\nabla_{\mathbf{R}}\tilde{\phi}.$$
(A.58)

The gradient $\nabla_v (\partial \tilde{\Phi} / \partial \mu)$ is found in a similar way. The gradients in real space are only to be obtained to zeroth order. However, some terms of the first order quantities that contain ϕ are important because they have steep gradients. Considering this, I find

$$\nabla E = \frac{Ze}{M} \nabla \tilde{\phi}, \tag{A.59}$$

$$\nabla \mu = -\frac{v_{\perp}^2}{2B^2} \nabla B - \frac{v_{\parallel}}{B} \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} + \frac{Ze}{MB} \nabla \tilde{\phi}$$
(A.60)

and

$$\nabla \varphi = \frac{v_{||}}{v_{\perp}^2} \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) + \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1 - \frac{Ze}{MB} \nabla \left(\frac{\partial \tilde{\Phi}}{\partial \mu}\right).$$
(A.61)

Due to the preceding considerations, equation (A.57) becomes

$$\partial E \cdot (\partial \mu \times \partial \varphi) = \frac{u}{B(\mathbf{r})} \left[1 + \frac{1}{\Omega_i} \mathbf{v} \cdot (\hat{\mathbf{b}} \times \boldsymbol{\kappa}) \right], \qquad (A.62)$$

where I have employed (A.43) to express v_{\parallel} as a function of the gyrokinetic variables, and I have used $(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} - \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) = v_{\perp}^2 \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$. This last result is deduced from $v_{\perp}^2 (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) = \mathbf{v}_{\perp}\mathbf{v}_{\perp} + (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})$. In equation (A.62), in the lower order term $u/B(\mathbf{r})$, the difference $B(\mathbf{r}) - B(\mathbf{R}) = -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B$ is important. Combining (A.53) and (A.62), and using (2.8) to write $\mathbf{v} \cdot (\hat{\mathbf{b}} \times \boldsymbol{\kappa}) = \mathbf{v}_{\perp} \cdot \nabla \times \hat{\mathbf{b}}$, the Jacobian of the transformation is found to be as given by (3.44). Notice that to this order $J = \langle J \rangle$ as required by (3.40).

Finally, I prove that J satisfies the gyroaverage of (3.40) to the required order, namely,

$$\frac{\partial J}{\partial t} + \nabla_{\mathbf{R}} \cdot \{J[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d]\} - \frac{Ze}{M} \frac{\partial}{\partial E} \{J[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \cdot \nabla_{\mathbf{R}} \langle \phi \rangle\} = 0, \qquad (A.63)$$

where \mathbf{v}_d is given by (3.20). To first order, I obtain

$$J[u\hat{\mathbf{b}}(\mathbf{R}) + \mathbf{v}_d] \simeq \mathbf{B}(\mathbf{R}) + \frac{Mc\mu}{Zeu}\hat{\mathbf{b}} \times \nabla_{\mathbf{R}}B + \frac{Mcu}{Ze}\nabla_{\mathbf{R}} \times \hat{\mathbf{b}} - \frac{c}{u}\nabla_{\mathbf{R}}\langle\phi\rangle \times \hat{\mathbf{b}}, \quad (A.64)$$

where I have employed (2.8). Inserting (A.64) into (A.63) and recalling that $u = \sqrt{2[E - \mu B(\mathbf{R})]}$ is enough to prove that (A.63), and thus (3.40) gyroaveraged, are satisfied by the Jacobian to first order.

Appendix B

Useful gyroaverages and gyrophase derivatives

The various derivations require various averages and integrals respect to the gyrophase φ_0 . In particular, the integrals for terms that contain several \mathbf{v}_{\perp} are recurrent. In this Appendix, I show how to work out these type of terms.

According to (2.2), the perpendicular velocity, \mathbf{v}_{\perp} , is $\mathbf{v}_{\perp} = v_{\perp}(\hat{\mathbf{e}}_1 \cos \varphi_0 + \hat{\mathbf{e}}_2 \sin \varphi_0)$. In order to find gyroaverages and integrals, it is useful to express the perpendicular velocity as

$$\mathbf{v}_{\perp} = \operatorname{Re}[v_{\perp} \exp(i\varphi_0)\mathbf{u}] = \frac{v_{\perp}}{2}[\exp(i\varphi_0)\mathbf{u} + \exp(-i\varphi_0)\mathbf{u}^*], \quad (B.1)$$

where $i = \sqrt{-1}$, $\mathbf{u} = \hat{\mathbf{e}}_1 - i\hat{\mathbf{e}}_2$, \mathbf{u}^* is the conjugate of the complex vector \mathbf{u} and $\operatorname{Re}(\mathbf{a})$ is the real part of the complex tensor \mathbf{a} . Notice also that

$$v_{\perp} \exp(i\varphi_0) \mathbf{u} = \mathbf{v}_{\perp} + i\mathbf{v} \times \hat{\mathbf{b}}.$$
 (B.2)

It is very common to find tensors composed of tensor products of $\mathbf{v}_{\perp},$

$$(\mathbf{v}_{\perp})^{n} \equiv \underbrace{\mathbf{v}_{\perp} \mathbf{v}_{\perp} \dots \mathbf{v}_{\perp}}_{n \text{ times}}.$$
 (B.3)

It is possible to find a more convenient form for this tensor product by using equation (B.1). It is useful to distinguish between odd and even n. For even n, n = 2m,

$$(\mathbf{v}_{\perp})^{2m} = \frac{v_{\perp}^{2m}}{2^{2m-1}} \operatorname{Re} \left\{ \exp(i2m\varphi_0) \mathbf{W}^{(2m,0)} + \exp[i2(m-1)\varphi_0] \mathbf{W}^{(2m,1)} + \dots + \exp(i2\varphi_0) \mathbf{W}^{(2m,m-1)} \right\} + \frac{v_{\perp}^{2m}}{2^{2m}} \mathbf{W}^{(2m,m)}, \qquad (B.4)$$

and for odd n, n = 2m + 1,

$$(\mathbf{v}_{\perp})^{2m+1} = \frac{v_{\perp}^{2m+1}}{2^{2m}} \operatorname{Re} \Big\{ \exp[i(2m+1)\varphi_0] \mathbf{W}^{(2m+1,0)} + \exp[i(2m-1)\varphi_0] \mathbf{W}^{(2m+1,1)} + \ldots + \exp(i\varphi_0) \mathbf{W}^{(2m+1,m)} \Big\},$$
(B.5)

where the tensor $\mathbf{W}^{(n,p)}$ is the tensor formed by the addition of all the possible different tensor products between (n-p) **u** vectors and p **u**^{*} vectors, i.e.,

$$\mathbf{W}^{(n,p)} \equiv \underbrace{\mathbf{u} \dots \mathbf{u}}_{n-p} \underbrace{\mathbf{u}^* \mathbf{u}^* \dots \mathbf{u}^*}_{p} + \underbrace{\mathbf{u} \dots \mathbf{u}}_{n-p-1} \mathbf{u}^* \mathbf{u} \underbrace{\mathbf{u}^* \dots \mathbf{u}^*}_{p-1} + \dots + \underbrace{\mathbf{u}^* \dots \mathbf{u}^*}_{p} \underbrace{\mathbf{u} \dots \mathbf{u}}_{n-p}.$$
(B.6)

There are n!/[p!(n-p)!] different terms in the summation. For example, $\mathbf{W}^{(5,2)}$ has 10 summands, given by

$$W^{(5,2)} \equiv uuuu^{*}u^{*} + uuu^{*}uu^{*} + uu^{*}uuu^{*} + u^{*}uuuu^{*} + uuu^{*}u^{*}u + u^{*}uuu^{*}u + u^{*}u^{*}uu + u^{*}u^{*}uu + u^{*}u^{*}uu + u^{*}u^{*}uuu.$$
(B.7)

The tensor $\mathbf{W}^{(n,p)}$ can be written in a form in which only $\mathbf{W}^{(n-2p,0)}$ and the matrix $\mathbf{\tilde{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}$ appear. The tensor of the form $\mathbf{W}^{(m,0)}$ is more convenient because it is part of $\operatorname{Re}[v_{\perp}^{m} \exp(im\varphi_{0})\mathbf{W}^{(m,0)}]$ and easy to write in a recognizable manner. For example, for m = 2, employing (B.2), I find

$$\operatorname{Re}[v_{\perp}^{2} \exp(i2\varphi_{0})\mathbf{W}^{(2,0)}] = \mathbf{v}_{\perp}\mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}).$$
(B.8)
For m = 3,

$$\operatorname{Re}[v_{\perp}^{3} \exp(i3\varphi_{0})\mathbf{W}^{(3,0)}] = \mathbf{v}_{\perp}\mathbf{v}_{\perp}\mathbf{v}_{\perp} - \mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}}) - (\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}}) - (\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}})\mathbf{v}_{\perp}.$$
(B.9)

These tensors are not only easy to write but also easy to integrate in φ_0 because

$$\frac{\partial}{\partial\varphi_0} \operatorname{Im}[v_{\perp}^m \exp(im\varphi_0)\mathbf{W}^{(m,0)}] = m\operatorname{Re}[v_{\perp}^m \exp(im\varphi_0)\mathbf{W}^{(m,0)}], \qquad (B.10)$$

with $Im(\mathbf{a})$ the imaginary part of the complex tensor \mathbf{a} . Equation (B.10) for m = 2 is

$$\frac{\partial}{\partial \varphi_0} [\mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp}] = 2 [\mathbf{v}_{\perp} \mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}})], \qquad (B.11)$$

and for m = 3 is

$$\frac{\partial}{\partial \varphi_0} [\mathbf{v}_{\perp} \mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) + \mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp} + (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp} \mathbf{v}_{\perp} - (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}})]$$

= $3 [\mathbf{v}_{\perp} \mathbf{v}_{\perp} \mathbf{v}_{\perp} - \mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) - (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp} (\mathbf{v} \times \hat{\mathbf{b}}) - (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) \mathbf{v}_{\perp}].(B.12)$

Equation (B.11) is used in the derivation of the gyrokinetic variables.

The decomposition of $\mathbf{W}^{(n,p)}$ into $\mathbf{W}^{(n-2p,0)}$ and matrices $\mathbf{\tilde{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}$ is also interesting for gyroaverages. According to equation (B.4), the tensor products of an even number of vectors \mathbf{v}_{\perp} has the gyroaverage

$$\overline{(\mathbf{v}_{\perp})^{2m}} = \frac{v_{\perp}^{2m}}{2^{2m}} \mathbf{W}^{(2m,m)},\tag{B.13}$$

where $\mathbf{W}^{(2m,m)}$ is a summation of tensor products of $p(\mathbf{i} - \mathbf{b}\mathbf{b})$ matrices. At the end of this appendix, I will give the result as a function of $\mathbf{i} - \mathbf{b}\mathbf{b}$. The gyroaverage of an odd number of \mathbf{v}_{\perp} is zero, as shown in (B.5).

I will prove now that $\mathbf{W}^{(n,p)}$ can be written such that it only contains $\mathbf{W}^{(n-2p,0)}$ and the matrix $\mathbf{\hat{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}$. To do so, I will first prove that $\mathbf{W}^{(n,p)}$ can be decomposed into a summation of tensors formed by the product of $\mathbf{W}^{(n-2p,0)}$ and $\mathbf{W}^{(2p,p)}$. After that, I will show that $\mathbf{W}^{(2p,p)}$ is formed by a summation of tensor products of p $(\mathbf{I} - \mathbf{\hat{b}}\mathbf{\hat{b}})$ matrices.

By multiplying $\mathbf{W}^{(n,p)}$ by (n-p)!/[p!(n-2p)!], (n-2p) **u** vectors can be distinguished from the rest of **u** vectors. I will denote them as \mathbf{u}^d (this is a trick to make the derivation easier). Then, a new tensor $\mathbf{T}^{(n,p,p)}$ is defined as the tensor formed by p **u** vectors, p **u**^{*} vectors and (n-2p) **u**^d vectors, giving

$$\frac{(n-p)!}{p!(n-2p)!} \mathbf{W}^{(n,p)} = \mathbf{T}^{(n,p,p)} \equiv \underbrace{\mathbf{u}^d \dots \mathbf{u}^d}_{n-2p} \underbrace{\mathbf{u} \dots \mathbf{u}^u}_{p} \underbrace{\mathbf{u}^* \mathbf{u}^* \dots \mathbf{u}^*}_{p} + \underbrace{\mathbf{u}^d \dots \mathbf{u}^d}_{n-2p-1} \underbrace{\mathbf{u}^u}_{p} \underbrace{\mathbf{u}^* \mathbf{u}^* \dots \mathbf{u}^*}_{p-1} + \dots + \underbrace{\mathbf{u}^* \dots \mathbf{u}^u}_{p} \underbrace{\mathbf{u} \dots \mathbf{u}}_{p} \underbrace{\mathbf{u}^d \dots \mathbf{u}^d}_{n-2p}.$$
(B.14)

For instance, $3\mathbf{W}^{(5,2)} = \mathbf{T}^{(5,2,2)}$ can be deduced from (B.7) to give

$$\mathbf{T}^{(5,2,2)} \equiv \mathbf{u}^{d}\mathbf{u}\mathbf{u}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{d}\mathbf{u}^{*}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{d}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{u}\mathbf{u}^{d}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{u}\mathbf{u}^{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{u}\mathbf{u}^{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{u}\mathbf{u}^{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}^{*}\mathbf{$$

The tensor $\mathbf{T}^{(n,p,p)}$ can be written as a combination of a tensor $\mathbf{W}^{(n-2p,0)}$ formed by $(n-2p) \mathbf{u}^d$ vectors and a tensor $\mathbf{W}^{(2p,p)}$ formed by $p \mathbf{u}$ vectors and $p \mathbf{u}^*$ vectors, leading to

$$T_{j_{1}j_{2}\dots j_{n}}^{(n,p,p)} \equiv W_{j_{1}\dots j_{2p}}^{(2p,p)} W_{j_{2p+1}\dots j_{n}}^{(n-2p,0)} + W_{j_{1}\dots j_{2p-1}j_{2p+1}}^{(2p,p)} W_{j_{2p}j_{2p+2}\dots j_{n}}^{(n-2p,0)} + \dots + W_{j_{n-2p+1}\dots j_{n}}^{(2p,p)} W_{j_{1}\dots j_{n-2p}}^{(n-2p,0)} = \frac{n!}{(n-2p)!(2p)!} W_{(j_{1}\dots j_{2p})}^{(2p,p)} W_{j_{2p+1}\dots j_{n}}^{(n-2p,0)}.$$
 (B.16)

Here, the number of summands is n!/[(n-2p)!(2p)!] because the tensors $\mathbf{W}^{(m,0)}$ and $\mathbf{W}^{(2p,p)}$ are symmetric respect to all of their indexes. Additionally, the tensor $\mathbf{W}^{(2p,p)}$ is always real, since $[\mathbf{W}^{(2p,p)}]^* = \mathbf{W}^{(2p,p)}$ by definition of $\mathbf{W}^{(2p,p)}$. The last expression

in (B.16) is based on the typical tensor notation, where the parenthesis around the indexes indicate symmetrization of the tensor. As an example of the equivalence in (B.16), the tensor $\mathbf{T}^{(5,2,2)}$ in (B.15) may be written as

$$T_{jklmn}^{(5,2,2)} = W_{jklm}^{(4,2)} W_n^{(1,0)} + W_{jkln}^{(4,2)} W_m^{(1,0)} + W_{jkmn}^{(4,2)} W_l^{(1,0)} + W_{jlmn}^{(4,2)} W_k^{(1,0)} + W_{klmn}^{(4,2)} W_j^{(1,0)}, \qquad (B.17)$$

with $\mathbf{W}^{(1,0)} \equiv \mathbf{u} \equiv \mathbf{u}^d$, and

$$\mathbf{W}^{(4,2)} \equiv \mathbf{u}\mathbf{u}\mathbf{u}^{*}\mathbf{u}^{*} + \mathbf{u}\mathbf{u}^{*}\mathbf{u}\mathbf{u}^{*} + \mathbf{u}^{*}\mathbf{u}\mathbf{u}^{*}\mathbf{u} + \mathbf{u}^{*}\mathbf{u}\mathbf{u}^{*}\mathbf{u} + \mathbf{u}^{*}\mathbf{u}^{*}\mathbf{u}\mathbf{u}.$$
 (B.18)

The tensor $\mathbf{W}^{(2p,p)}$ can be rewritten employing yet a third tensor, $\mathbf{I}^{(2p)}$. This tensor is formed by adding all the possible different tensor products that are formed by pmatrices $(\mathbf{\tilde{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}})_{jk} = \delta_{jk} - \hat{b}_j \hat{b}_k = \delta_{jk}^{\perp}$, i.e.,

$$I_{j_{1}j_{2}\dots j_{2p}}^{(2p)} = \delta_{j_{1}j_{2}}^{\perp} \delta_{j_{3}j_{4}}^{\perp} \dots \delta_{j_{2p-1}j_{2p}}^{\perp} + \delta_{j_{1}j_{3}}^{\perp} \delta_{j_{2}j_{4}}^{\perp} \dots \delta_{j_{2p-1}j_{2p}}^{\perp} + \dots = \frac{(2p)!}{2^{p}p!} \delta_{(j_{1}j_{2})}^{\perp} \delta_{j_{3}j_{4}}^{\perp} \dots \delta_{j_{2p-1}j_{2p})}^{\perp}.$$
 (B.19)

This tensor is formed by $(2p)!/(2^pp!)$ summands. For example, $\mathbf{I}^{(4)}$ is

$$I_{jklm}^{(4)} = \delta_{jk}^{\perp} \delta_{lm}^{\perp} + \delta_{jl}^{\perp} \delta_{km}^{\perp} + \delta_{jm}^{\perp} \delta_{kl}^{\perp}.$$
 (B.20)

The tensor $\mathbf{I}^{(2p)}$ can be written in terms of $\mathbf{W}^{(2p,p)}$ because

$$2(\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) = 2(\hat{\mathbf{e}}_1\hat{\mathbf{e}}_1 + \hat{\mathbf{e}}_2\hat{\mathbf{e}}_2) = \mathbf{u}\mathbf{u}^* + \mathbf{u}^*\mathbf{u}.$$
 (B.21)

Using this relation, the tensor $2^{p}\mathbf{I}^{(2p)}$, formed by all possible tensor products of p $2(\mathbf{\vec{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}})$ matrices, is written as a summation of tensor products of p **u** vectors and p \mathbf{u}^* vectors by substituting equation (B.21) in it. Each summand $2^{p}\delta_{j_{1}j_{2}}^{\perp}\delta_{j_{3}j_{4}}^{\perp}\ldots\delta_{j_{2p-1}j_{2p}}^{\perp}$ of $2^{p}\mathbf{I}^{(2p)}$ [recall (B.19)] gives 2^{p} different tensors formed by p **u** vectors and p \mathbf{u}^* vectors. Adding all the terms in $\mathbf{I}^{(2p)}$, there is a total of (2p)!/p! terms formed by tensor products of p **u** vectors and p **u**^{*} vectors. In this summation, each summand of $\mathbf{W}^{(2p,p)}$ is present p! times. According to (B.6), a summand of $\mathbf{W}^{(2p,p)}$ is uniquely determined by the positions that the p **u** vectors occupy, or alternatively, which pindexes of $j_1 j_2 \ldots j_{2p}$ correspond to the p **u** vectors. Without loss of generality, these indexes can be chosen to be $j_1 j_2 \ldots j_p$. A tensor of the form $u_{j_1} u_{j_2} \ldots u_{j_p} u_{j_{p+1}}^* \ldots u_{j_{2p}}^*$ is obtained from every term of $2^p \mathbf{I}^{(2p)}$ of the form $2^p \delta_{j_1 k_1}^{\perp} \delta_{j_2 k_2}^{\perp} \ldots \delta_{j_p k_p}^{\perp}$, with $k_1 k_2 \ldots k_p$ being any permutation of $j_{p+1} j_{p+2} \ldots j_{2p}$. There are p! of these terms, proving that

$$2^{p}\mathbf{I}^{(2p)} = p!\mathbf{W}^{(2p,p)}.$$
 (B.22)

Considering the examples (B.18) and (B.20), equation (B.22) implies that their relation is $4\mathbf{I}^{(4)} = 2\mathbf{W}^{(4,2)}$. Combining equations (B.14), (B.16) and (B.22) gives

$$W_{j_{1}j_{2}\dots j_{n}}^{(n,p)} = \frac{2^{p}(n-2p)!}{(n-p)!} \Big[I_{j_{1}\dots j_{2p}}^{(2p)} W_{j_{2p+1}\dots j_{n}}^{(n-2p,0)} + I_{j_{1}\dots j_{2p-1}j_{2p+1}}^{(2p)} W_{j_{2p}j_{2p+2}\dots j_{n}}^{(n-2p,0)} + \dots \\ + I_{j_{n-2p+1}\dots j_{n}}^{(2p)} W_{j_{1}\dots j_{n-2p}}^{(n-2p,0)} \Big] = \frac{n!}{p!(n-p)!} I_{(j_{1}\dots j_{2p}}^{(2p)} W_{j_{2p+1}\dots j_{n}}^{(n-2p,0)}.$$
(B.23)

As promised, $\mathbf{W}^{(n,p)}$ is written in terms of $\mathbf{W}^{(n-2p,0)}$ and $\mathbf{\tilde{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}$ because $\mathbf{I}^{(2p)}$ is formed by addition of tensor products of $\mathbf{\tilde{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}$. Continuing with the example in (B.7), $\mathbf{W}^{(5,2)}$ gives

$$W_{jklmn}^{(5,2)} = \frac{2}{3} \Big[I_{jklm}^{(4)} u_n + I_{jkln}^{(4)} u_m + I_{jkmn}^{(4)} u_l + I_{jlmn}^{(4)} u_k + I_{klmn}^{(4)} u_j \Big], \tag{B.24}$$

with $\mathbf{I}^{(4)}$ given in (B.20). This tensor appears in $(\mathbf{v}_{\perp})^5 \equiv \mathbf{v}_{\perp} \mathbf{v}_{\perp} \mathbf{v}_{\perp} \mathbf{v}_{\perp} \mathbf{v}_{\perp}$ as [recall (B.5)]

$$\frac{1}{2^4} \operatorname{Re}[v_{\perp}^5 \exp(i\varphi_0) W_{jklmn}^{(5,2)}] = \frac{v_{\perp}^4}{24} \Big[I_{jklm}^{(4)} v_{\perp,n} + I_{jkln}^{(4)} v_{\perp,m} + I_{jkmn}^{(4)} v_{\perp,l} + I_{jlmn}^{(4)} v_{\perp,k} + I_{klmn}^{(4)} v_{\perp,j} \Big].$$
(B.25)

To finish, the gyroaverage of an even number of vectors \mathbf{v}_{\perp} is calculated. Com-

bining equations (B.13) and (B.22) gives

$$\overline{(\mathbf{v}_{\perp})^{2m}} = \frac{v_{\perp}^{2m}}{2^m m!} \mathbf{I}^{(2m)},\tag{B.26}$$

with $\mathbf{I}^{(2m)}$ given in (B.19). For m = 1, the familiar result

$$\overline{\mathbf{v}_{\perp}\mathbf{v}_{\perp}} = \frac{v_{\perp}^2}{2} (\mathbf{\vec{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}})$$
(B.27)

is found. This result, combined with (B.4) for m = 2 and (B.8) gives (A.7). The gyroaverage for m = 4 gives (A.32).

Appendix C

Gyrokinetic equivalence

In this Appendix, a summary of [61], I prove that the gyrokinetic results in chapter 3 are completely consistent with the pioneering results by Dubin *et al* [21], obtained using a Hamiltonian formalism and Lie transforms [25]. This comparison confirms that both approaches, the recursive method in chapter 3 and the Lie transforms, yield the same final gyrokinetic formalism.

In chapter 3, the recursive approach developed in [58, 59] was generalized for nonlinear electrostatic gyrokinetics in a general magnetic field. In reference [21], the nonlinear electrostatic gyrokinetic equation was derived for a constant magnetic field and a collisionless plasma using a Hamiltonian formalism. The asymptotic expansion was carried out to higher order in [21] because the calculation is easier in a constant magnetic field. When the method proposed in chapter 3 is extended to next order, the results are different in appearance, but I will prove that these differences are due to subtleties in some definitions.

Both methods are asymptotic expansions in the small parameter $\delta = \rho/L \ll 1$. Here L is a characteristic macroscopic length in the problem and $\rho = v_{\rm th}/\Omega$ is the gyroradius, with $\Omega = ZeB/Mc$ the gyrofrequency, $v_{\rm th} = \sqrt{2T/M}$ the thermal velocity and T the temperature. In both methods, the phase space $\{\mathbf{r}, \mathbf{v}\}$, with \mathbf{r} and \mathbf{v} the position and velocity of the particles, is expressed in gyrokinetic variables, defined order by order in δ . In reference [21], the gyrokinetic variables are obtained by Lie transform and the gyrokinetic equation is found to second order in δ . In chapter 3, the gyrokinetic variables are found by imposing that their time derivative is gyrophase independent. Here $d/dt \equiv \partial/\partial t + \mathbf{v} \cdot \nabla + (-Ze\nabla\phi/M + \Omega\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla_v$ is the Vlasov operator. In chapter 3, the gyrokinetic equation was only found to first order. In this Appendix, I will calculate the gyrokinetic equation and the gyrokinetic variables to higher order for a constant magnetic field, and I will compare the results with those in [21]. The orderings and assumptions are more general than in section 3.1. The pieces of the distribution function and the potential with short wavelengths also scale as

$$\frac{f_k}{f_s} \sim \frac{e\phi_k}{T} \sim \frac{1}{k_\perp L} \lesssim 1, \tag{C.1}$$

with $k_{\perp}\rho \lesssim 1$. Here f_s is the lowest order distribution function with a slow variation in both **r** and **v**. In chapter 3, the lowest order distribution functions was assumed to be a Maxwellian. In this Appendix, to ease the comparison with reference [21], I relax that assumption. Finally, I order the time derivatives as $\partial/\partial t \sim v_{\rm th}/L$.

C.1 Constant magnetic field results

The general gyrokinetic variables obtained in section 3.2 are $\mathbf{R} = \mathbf{r} + \mathbf{R}_1 + \mathbf{R}_2$, $E = E_0 + E_1 + E_2$, $\mu = \mu_0 + \mu_1 + \mu_2$ and $\varphi = \varphi_0 + \varphi_1 + \varphi_2$. Since the unit vector $\hat{\mathbf{b}}$ is assumed constant in space and time, I can define $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$ in (2.2) so that they are also constant, and I do so to simplify the comparison with [21]. The corrections found in section 3.2 specialized to constant magnetic field are, for the gyrocenter position \mathbf{R} , $\mathbf{R}_1 = \Omega^{-1}\mathbf{v} \times \hat{\mathbf{b}}$ and $\mathbf{R}_2 = -(c/B\Omega)\nabla_{\mathbf{R}}\tilde{\Phi} \times \hat{\mathbf{b}}$; for the kinetic energy E, $E_1 = Ze\tilde{\phi}/M$ and $E_2 = (c/B)(\partial\tilde{\Phi}/\partial t)$; and for the magnetic moment μ and the gyrophase φ , $\mu_1 = Ze\tilde{\phi}/MB$ and $\varphi_1 = -(Ze/MB)(\partial\tilde{\Phi}/\partial\mu)$. The corrections μ_2 and φ_2 were not calculated because they were not needed to obtain the gyrokinetic equation to first order in δ under the assumptions in chapter 3.

I require the gyroaverage of $d\mathbf{R}/dt$ and dE/dt to higher order than in section 3.2, and I need the second order correction μ_2 . For constant magnetic fields, $\langle d\mathbf{R}/dt \rangle$, $\langle dE/dt \rangle$ and the correction μ_2 can be easily calculated by employing the methodology in chapter 3. I will define μ_2 so that the gyroaverage of $d\mu/dt$ is zero to order $\delta^2 v_{\rm th}^3/BL$. The second order correction φ_2 is needed to obtain $d\varphi/dt$ to higher order. However, f is gyroaveraged, making the dependence on φ weak. Thus, φ_2 never enters in the final results and will not be necessary for our purposes. Once I have **R**, E, μ and their derivatives to higher order, I will compare these results to both the gyrokinetic Vlasov equation and the gyrokinetic Poisson's equation in [21].

C.2 Time derivative of R

Employing the definitions of \mathbf{R}_1 and \mathbf{R}_2 , I find

$$\frac{d\mathbf{R}}{dt} = v_{\parallel}\hat{\mathbf{b}} - \frac{c}{B}\nabla\phi \times \hat{\mathbf{b}} + \frac{d\mathbf{R}_2}{dt}.$$
 (C.2)

The gyroaverage of this expression is performed holding the gyrokinetic variables \mathbf{R} , E, μ and t fixed to obtain

$$\left\langle \frac{d\mathbf{R}}{dt} \right\rangle = u\hat{\mathbf{b}} - \frac{c}{B} \langle \nabla \phi \rangle \times \hat{\mathbf{b}},$$
 (C.3)

where $u = \langle v_{\parallel} \rangle$. My gyrokinetic variables are defined so that when the Vlasov operator is applied to a function with a vanishing gyroaverage, like $\mathbf{R}_2 = \mathbf{R}_2(\mathbf{R}, E, \mu, \varphi, t)$, the result also has a zero gyroaverage; namely $\langle d\mathbf{R}_2/dt \rangle = 0$.

The gradient $\nabla \phi$ is written in the gyrokinetic variables by using

$$\nabla \phi = \nabla \mathbf{R} \cdot \nabla_{\mathbf{R}} \phi + \frac{\partial \phi}{\partial \mu} \nabla \mu + \frac{\partial \phi}{\partial \varphi} \nabla \varphi \simeq \nabla_{\mathbf{R}} \phi + \nabla \mathbf{R}_2 \cdot \nabla_{\mathbf{R}} \phi + \frac{\partial \phi}{\partial \mu} \nabla \mu_1 + \frac{\partial \phi}{\partial \varphi} \nabla \varphi_1.$$
(C.4)

Here, I neglect $\partial \phi / \partial E \simeq -(\partial \mathbf{R}_1 / \partial E) \cdot \nabla \phi$ because the function \mathbf{R}_1 does not depend on E to order δL . To obtain the second equality, I use that $\nabla \mathbf{R}_1 = 0 = \nabla \mu_0 = \nabla \varphi_0$. The gyroaverage of equation (C.4), obtained employing the definitions of \mathbf{R}_2 , μ_1 and φ_1 , gives

$$\langle \nabla \phi \rangle \simeq \nabla_{\mathbf{R}} \langle \phi \rangle - \frac{c}{B\Omega} \langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \tilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \phi) \rangle + \frac{Ze}{MB} \left\langle \frac{\partial \phi}{\partial \mu} \nabla_{\mathbf{R}} \tilde{\phi} - \frac{\partial \phi}{\partial \varphi} \nabla_{\mathbf{R}} \left(\frac{\partial \tilde{\Phi}}{\partial \mu} \right) \right\rangle.$$
(C.5)

This equation can be simplified by integrating by parts in φ to obtain

$$\left\langle \frac{\partial \phi}{\partial \mu} \nabla_{\mathbf{R}} \tilde{\phi} - \frac{\partial \phi}{\partial \varphi} \nabla_{\mathbf{R}} \left(\frac{\partial \tilde{\Phi}}{\partial \mu} \right) \right\rangle = \left\langle \frac{\partial \tilde{\phi}}{\partial \mu} \nabla_{\mathbf{R}} \tilde{\phi} + \tilde{\phi} \nabla_{\mathbf{R}} \left(\frac{\partial \tilde{\phi}}{\partial \mu} \right) \right\rangle = \frac{1}{2} \nabla_{\mathbf{R}} \left(\frac{\partial}{\partial \mu} \langle \tilde{\phi}^2 \rangle \right). \quad (C.6)$$

I next demonstrate that

$$\langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \widetilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \phi) \rangle = \frac{1}{2} \nabla_{\mathbf{R}} \langle \nabla_{\mathbf{R}} \widetilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \widetilde{\phi}) \rangle \tag{C.7}$$

by first noticing that

$$\langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \widetilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \phi) \rangle = \nabla_{\mathbf{R}} \langle \nabla_{\mathbf{R}} \widetilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \widetilde{\phi}) \rangle + \langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \widetilde{\phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \widetilde{\Phi}) \rangle. \quad (C.8)$$

Integrating by parts in φ in the second term, I find $\langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \tilde{\phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \tilde{\Phi}) \rangle = -\langle \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \tilde{\Phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \tilde{\phi}) \rangle$, giving the result in (C.7).

Finally, substituting equations (C.6) and (C.7) into equation (C.5) and using the result in (C.3) gives

$$\left\langle \frac{d\mathbf{R}}{dt} \right\rangle = u\hat{\mathbf{b}} - \frac{c}{B}\nabla_{\mathbf{R}}\Psi \times \hat{\mathbf{b}},$$
 (C.9)

with

$$\Psi = \langle \phi \rangle + \frac{Ze}{2MB} \frac{\partial}{\partial \mu} \langle \tilde{\phi}^2 \rangle + \frac{c}{2B\Omega} \langle \nabla_{\mathbf{R}} \tilde{\phi} \cdot (\hat{\mathbf{b}} \times \nabla_{\mathbf{R}} \tilde{\Phi}) \rangle.$$
(C.10)

To find u, I need $v_{||}$ as a function of the gyrokinetic variables. To do so, I use

$$\frac{v_{\parallel}^2}{2} = E_0 - \mu_0 B = E - \mu B - (E_2 - \mu_2 B), \qquad (C.11)$$

where I employ $E_1 - \mu_1 B = 0$. According to this result, the difference between $u = \langle v_{||} \rangle$ and $v_{||}$ is necessarily of order $\delta^2 v_{\text{th}}$. Once I calculate μ_2 , I will be able to find u.

C.3 Time derivative of E

Employing the definitions of E_1 and E_2 , and gyroaveraging, I find

$$\left\langle \frac{dE}{dt} \right\rangle = -\frac{Ze}{M} \langle \mathbf{v} \cdot \nabla \phi \rangle.$$
 (C.12)

Here, I have used that $\langle dE_1/dt \rangle = 0 = \langle dE_2/dt \rangle$.

The term $\mathbf{v} \cdot \nabla \phi$ can be conveniently rewritten by employing

$$\frac{d\phi}{dt} = \left. \frac{\partial\phi}{\partial t} \right|_{\mathbf{r}} + \mathbf{v} \cdot \nabla\phi = \left. \frac{\partial\phi}{\partial t} \right|_{\mathbf{R},E,\mu,\varphi} + \frac{d\mathbf{R}}{dt} \cdot \nabla_{\mathbf{R}}\phi + \frac{d\mu}{dt} \frac{\partial\phi}{\partial\mu} + \frac{d\varphi}{dt} \frac{\partial\phi}{\partial\varphi}.$$
 (C.13)

Here, I neglect $\partial \phi / \partial E$ again. Solving for $\mathbf{v} \cdot \nabla \phi$ and gyroaveraging, I find

$$\langle \mathbf{v} \cdot \nabla \phi \rangle = \left\langle \frac{d\mathbf{R}}{dt} \cdot \nabla_{\mathbf{R}} \phi \right\rangle + \left\langle \frac{d\mu}{dt} \frac{\partial \phi}{\partial \mu} \right\rangle + \left\langle \frac{d\varphi}{dt} \frac{\partial \phi}{\partial \varphi} \right\rangle - \left\langle \frac{\partial \phi}{\partial t} \Big|_{\mathbf{r}} - \frac{\partial \phi}{\partial t} \Big|_{\mathbf{R}, E, \mu, \varphi} \right\rangle. \quad (C.14)$$

To simplify the calculation, I will assume that I know the corrections \mathbf{R}_3 , $\mu_3 - \langle \mu_3 \rangle$, φ_2 and φ_3 (obtaining these corrections is straightforward following the procedure in section 3.2 but will be unnecessary). With these corrections, I find that to the order needed, $d\mathbf{R}/dt = \langle d\mathbf{R}/dt \rangle$, given in (C.9), $d\mu/dt = \langle d\mu/dt \rangle \simeq 0$ and $d\varphi/dt = \langle d\varphi/dt \rangle$. Then, equation (C.14) simplifies to

$$\langle \mathbf{v} \cdot \nabla \phi \rangle = \left(u \hat{\mathbf{b}} - \frac{c}{B} \nabla_{\mathbf{R}} \Psi \times \hat{\mathbf{b}} \right) \cdot \nabla_{\mathbf{R}} \langle \phi \rangle - \left\langle \frac{\partial \phi}{\partial t} \bigg|_{\mathbf{r}} - \frac{\partial \phi}{\partial t} \bigg|_{\mathbf{R}, E, \mu, \varphi} \right\rangle, \quad (C.15)$$

where Ψ is given in equation (C.10) and $\langle \partial \phi / \partial \varphi \rangle = 0$. Notice that assuming that I already have \mathbf{R}_3 , $\mu_3 - \langle \mu_3 \rangle$, φ_2 and φ_3 is only a shortcut to find the result in (C.15). To obtain $\langle \phi \rangle$ to the order required, these higher order corrections are not needed, neither are they necessary for the difference between time derivatives, as I will prove next. The difference between time derivatives is

$$\frac{\partial \phi}{\partial t}\Big|_{\mathbf{r}} - \frac{\partial \phi}{\partial t}\Big|_{\mathbf{R},E,\mu,\varphi} = \frac{\partial \mathbf{R}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} \cdot \nabla_{\mathbf{R}}\phi + \frac{\partial \mu}{\partial t}\Big|_{\mathbf{r},\mathbf{v}}\frac{\partial \phi}{\partial \mu} + \frac{\partial \varphi}{\partial t}\Big|_{\mathbf{r},\mathbf{v}}\frac{\partial \phi}{\partial \varphi}.$$
 (C.16)

The procedure for rewriting (C.16) is analogous to that used on (C.4). Using the

definitions of \mathbf{R}_1 , \mathbf{R}_2 , μ_1 and φ_1 , I find that $\partial \mathbf{R}/\partial t|_{\mathbf{r},\mathbf{v}} \simeq \partial \mathbf{R}_2/\partial t|_{\mathbf{r},\mathbf{v}}$, $\partial \mu/\partial t|_{\mathbf{r},\mathbf{v}} \simeq \partial \mu_1/\partial t|_{\mathbf{r},\mathbf{v}}$ and $\partial \varphi/\partial t|_{\mathbf{r},\mathbf{v}} \simeq \partial \varphi_1/\partial t|_{\mathbf{r},\mathbf{v}}$, giving

$$\left\langle \left. \frac{\partial \phi}{\partial t} \right|_{\mathbf{r}} - \left. \frac{\partial \phi}{\partial t} \right|_{\mathbf{R}, E, \mu, \varphi} \right\rangle = \frac{\partial}{\partial t} (\Psi - \langle \phi \rangle), \tag{C.17}$$

where I use the equivalent to equations (C.6) and (C.7) with $\partial/\partial t$ replacing $\nabla_{\mathbf{R}}$. The final result, obtained by combining equations (C.12), (C.15) and (C.17), is

$$\left\langle \frac{dE}{dt} \right\rangle = \frac{Ze}{M} \left[\frac{\partial}{\partial t} (\Psi - \langle \phi \rangle) - \left(u\hat{\mathbf{b}} - \frac{c}{B} \nabla_{\mathbf{R}} \Psi \times \hat{\mathbf{b}} \right) \cdot \nabla_{\mathbf{R}} \langle \phi \rangle \right].$$
(C.18)

C.4 Second order correction μ_2

The correction μ_2 , according to section 3.2, is given by

$$\mu_2 = \frac{1}{\Omega} \int^{\varphi} d\varphi' \left[\frac{d}{dt} (\mu_0 + \mu_1) - \left\langle \frac{d}{dt} (\mu_0 + \mu_1) \right\rangle \right] + \langle \mu_2 \rangle, \quad (C.19)$$

where $\langle \mu_2 \rangle$ is found by requiring that $\langle d\mu/dt \rangle = 0$ to order $\delta^2 v_{\rm th}^3/BL$.

The time derivative of $\mu_0 + \mu_1$ is given by

$$\frac{d}{dt}(\mu_0 + \mu_1) = \frac{Ze}{MB} \left(-\mathbf{v}_{\perp} \cdot \nabla \phi + \frac{d\tilde{\phi}}{dt} \right).$$
(C.20)

To rewrite $\mathbf{v}_{\perp} \cdot \nabla \phi$ as a function of the gyrokinetic variables, I employ $\mathbf{v}_{\perp} \cdot \nabla \phi = \mathbf{v} \cdot \nabla \phi - v_{\parallel} \hat{\mathbf{b}} \cdot \nabla \phi$ and equation (C.13) to find

$$-\mathbf{v}_{\perp} \cdot \nabla \phi + \frac{d\tilde{\phi}}{dt} = -\frac{d\langle \phi \rangle}{dt} + \frac{\partial \phi}{\partial t}\Big|_{\mathbf{r}} + v_{||}\hat{\mathbf{b}} \cdot \nabla \phi.$$
(C.21)

To the order I am interested in, $d\mathbf{R}/dt \simeq u\hat{\mathbf{b}} - (c/B)\nabla_{\mathbf{R}}\langle\phi\rangle \times \hat{\mathbf{b}}$, giving

$$-\mathbf{v}_{\perp}\cdot\nabla\phi + \frac{d\dot{\phi}}{dt} = -\left.\frac{\partial\langle\phi\rangle}{\partial t}\right|_{\mathbf{R},E,\mu,\varphi} - u\hat{\mathbf{b}}\cdot\nabla_{\mathbf{R}}\langle\phi\rangle + \left.\frac{\partial\phi}{\partial t}\right|_{\mathbf{r}} + v_{||}\hat{\mathbf{b}}\cdot\nabla\phi. \tag{C.22}$$

According to equation (C.11), the difference between $u = \langle v_{||} \rangle$ and $v_{||}$ is higher order,

and according to equation (C.16), the difference between $\partial \phi / \partial t |_{\mathbf{r}}$ and $\partial \phi / \partial t |_{\mathbf{R},E,\mu,\varphi}$ is negligible. Therefore, equations (C.20) and (C.22) give

$$\frac{d}{dt}(\mu_0 + \mu_1) = \frac{Ze}{MB} \left(\frac{\partial \tilde{\phi}}{\partial t} + u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\phi} \right), \qquad (C.23)$$

which in turn, using equation (C.19), yields

$$\mu_2 = \frac{c}{B^2} \left(\frac{\partial \tilde{\Phi}}{\partial t} + u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\Phi} \right) + \langle \mu_2 \rangle.$$
 (C.24)

To find $\langle \mu_2 \rangle$ I require that $\langle d\mu/dt \rangle = 0$ to order $\delta^2 v_{\rm th}^3/BL$. The gyroaverage of $d\mu/dt$ is given by

$$\left\langle \frac{d\mu}{dt} \right\rangle = \left\langle \frac{d}{dt} (\mu_0 + \mu_1 + \mu_2) \right\rangle = -\frac{Ze}{MB} \langle \mathbf{v}_{\perp} \cdot \nabla \phi \rangle + \frac{d\langle \mu_2 \rangle}{dt}, \quad (C.25)$$

where the gyroaverages of $d\mu_1/dt$ and $d(\mu_2 - \langle \mu_2 \rangle)/dt$ vanish. The term $\langle \mathbf{v}_{\perp} \cdot \nabla \phi \rangle$ can be conveniently rewritten to higher order than in (C.22) by employing equation (C.15) to find

$$\langle \mathbf{v}_{\perp} \cdot \nabla \phi \rangle = \left(u \hat{\mathbf{b}} - \frac{c}{B} \nabla_{\mathbf{R}} \Psi \times \hat{\mathbf{b}} \right) \cdot \nabla_{\mathbf{R}} \langle \phi \rangle - \frac{\partial}{\partial t} (\Psi - \langle \phi \rangle) - \langle v_{||} \hat{\mathbf{b}} \cdot \nabla \phi \rangle, \quad (C.26)$$

where I used equation (C.17). Employing equation (C.4) and the fact that the difference between $u = \langle v_{||} \rangle$ and $v_{||}$ is order $\delta^2 v_{\text{th}}$ (C.11), I find

$$\langle v_{||}\hat{\mathbf{b}}\cdot\nabla\phi\rangle\simeq\langle v_{||}\hat{\mathbf{b}}\cdot\nabla_{\mathbf{R}}\phi\rangle+u\hat{\mathbf{b}}\cdot\nabla_{\mathbf{R}}(\Psi-\langle\phi\rangle)\simeq u\hat{\mathbf{b}}\cdot\nabla_{\mathbf{R}}\Psi.$$
 (C.27)

To obtain the second equality, I employ $\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle \gg \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\phi}$, which means that $\langle v_{||} \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \phi \rangle \simeq \langle v_{||} \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle + u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\phi} \rangle = u \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \langle \phi \rangle$ to order $\delta^2 T v_{\text{th}} / eL$. Then, equation (C.26) becomes $\langle \mathbf{v}_{\perp} \cdot \nabla \phi \rangle = -d(\Psi - \langle \phi \rangle)/dt$, where to this order $d/dt = \partial/\partial t + [u \hat{\mathbf{b}} - (c/B) \nabla_{\mathbf{R}} \langle \phi \rangle \times \hat{\mathbf{b}}] \cdot [\nabla_{\mathbf{R}} - (Ze/M) \nabla_{\mathbf{R}} \langle \phi \rangle (\partial/\partial E)]$ and $\partial (\Psi - \langle \phi \rangle)/\partial E = 0$. Finally, imposing $\langle d\mu/dt \rangle = 0$ on equation (C.25), I find

$$\langle \mu_2 \rangle = -\frac{Ze}{MB} \left(\Psi - \langle \phi \rangle \right).$$
 (C.28)

C.5 Comparisons with Dubin et al

To compare with reference [21], I first need to write the gyrokinetic equation in the same variables that are used in that reference, i.e., I need to employ u instead of E. The change is easy to carry out. I substitute E_2 and (C.24) into (C.11) to write

$$v_{||} = \sqrt{2[E - (\mu - \langle \mu_2 \rangle)B]} + \frac{c}{B}\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}}\tilde{\Phi}, \qquad (C.29)$$

where I Taylor expand $E_2 - (\mu_2 - \langle \mu_2 \rangle)B = -(c/B)u\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\Phi}$. Then, gyroaveraging this equation I find

$$\frac{u^2}{2} = E - (\mu - \langle \mu_2 \rangle) B.$$
 (C.30)

Applying the Vlasov operator to this expression and gyroaveraging, I find

$$\left\langle \frac{du}{dt} \right\rangle = -\frac{Ze}{M} \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \Psi,$$
 (C.31)

where I used equations (C.18), (C.28) and $\langle d\mu/dt \rangle = 0$. With this equation, equation (C.9) and the fact that $\langle d\mu/dt \rangle = 0$, I find the same gyrokinetic Vlasov equation as in reference [21], namely

$$\frac{\partial f}{\partial t} + \left(u\hat{\mathbf{b}} - \frac{c}{B}\nabla_{\mathbf{R}}\Psi \times \hat{\mathbf{b}}\right) \cdot \nabla_{\mathbf{R}}f - \frac{Ze}{M}\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}}\Psi\frac{\partial f}{\partial u} = 0, \quad (C.32)$$

with $f(\mathbf{R}, u, \mu, t)$. The differences between my function Ψ of (C.10) and the function ψ in reference [21], given in their equation (19b), come from their introduction of the potential function $\phi(\mathbf{R} + \boldsymbol{\rho}, t) \neq \phi(\mathbf{r}, t)$, leading to subtle differences in the definitions of $\langle \phi \rangle$, $\tilde{\phi}$ and $\tilde{\Phi}$. Here, the vector $\boldsymbol{\rho}(\mu, \theta)$ is

$$\boldsymbol{\rho} = \frac{\sqrt{2\mu B}}{\Omega} (\hat{\mathbf{e}}_1 \cos \theta - \hat{\mathbf{e}}_2 \sin \theta), \qquad (C.33)$$

with θ the gyrokinetic gyrophase as defined in [21]. The relation between the gyrophase θ and my gyrophase φ is $\theta = -\pi/2 - \varphi$. From now on, I will denote the functions $\langle \phi \rangle$, $\tilde{\phi}$ and $\tilde{\Phi}$ as they are defined in [21] with the subindex D. The definitions in [21] are then

$$\overline{\phi}_D \equiv \overline{\phi}_D(\mathbf{R}, \mu, t) = \frac{1}{2\pi} \oint d\theta \,\phi(\mathbf{R} + \boldsymbol{\rho}, t), \tag{C.34}$$

$$\widetilde{\phi}_D \equiv \widetilde{\phi}_D(\mathbf{R}, \mu, \theta, t) = \phi(\mathbf{R} + \boldsymbol{\rho}, t) - \overline{\phi}_D \tag{C.35}$$

and

$$\widetilde{\Phi}_D \equiv \widetilde{\Phi}_D(\mathbf{R}, \mu, \theta, t) = \int^{\theta} d\theta' \, \widetilde{\phi}_D(\mathbf{R}, \mu, \theta', t) \tag{C.36}$$

such that $\langle \tilde{\Phi}_D \rangle = 0$. Notice that these definitions coincide with mine to order $\delta T/e$, except for $\tilde{\Phi}_D$, for which $\tilde{\Phi}_D \simeq -\tilde{\Phi}$. The sign is due to the definition of the gyrophase θ . To second order, however, Taylor expanding $\phi(\mathbf{r}, t) = \phi(\mathbf{R} + \boldsymbol{\rho} - \boldsymbol{\rho} - \mathbf{R}_1 - \mathbf{R}_2, t)$ gives

$$\phi \simeq \phi(\mathbf{R} + \boldsymbol{\rho}, t) - (\boldsymbol{\rho} + \mathbf{R}_1 + \mathbf{R}_2) \cdot \nabla_{\mathbf{R}} \phi, \qquad (C.37)$$

where

$$\boldsymbol{\rho} + \mathbf{R}_{1} \simeq -\frac{Mc\mu_{1}}{Zev_{\perp}^{2}} \mathbf{v} \times \hat{\mathbf{b}} - \frac{\varphi_{1}}{\Omega} \mathbf{v}_{\perp} = O(\delta\rho).$$
(C.38)

To obtain equation (C.38), I Taylor expand $\rho(\mu, \varphi)$ around μ_0 and φ_0 in equation (C.33). Employing the lowest order results $\partial \phi / \partial \varphi \simeq -\Omega^{-1} \mathbf{v}_{\perp} \cdot \nabla \phi$ and $\partial \phi / \partial \mu \simeq -(Mc/Zev_{\perp}^2)(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$, I write equation (C.37) as

$$\phi \simeq \phi(\mathbf{R} + \boldsymbol{\rho}, t) - \frac{Ze}{MB} \left(\tilde{\phi} \frac{\partial \phi}{\partial \mu} - \frac{\partial \tilde{\Phi}}{\partial \mu} \frac{\partial \phi}{\partial \varphi} \right) + \frac{c}{B\Omega} (\nabla_{\mathbf{R}} \tilde{\Phi} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \phi, \qquad (C.39)$$

where I used the definitions of \mathbf{R}_2 , μ_1 and φ_1 . Then, gyroaveraging gives

$$\langle \phi \rangle \simeq \overline{\phi}_D - \frac{Ze}{MB} \frac{\partial}{\partial \mu} \langle \tilde{\phi}^2 \rangle + \frac{c}{B\Omega} \langle (\nabla_{\mathbf{R}} \tilde{\Phi} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \tilde{\phi} \rangle. \tag{C.40}$$

Substituting this equation into the definition (C.10) of Ψ and employing that to lowest order $\tilde{\phi} \simeq \tilde{\phi}_D$ and $\tilde{\Phi} \simeq -\tilde{\Phi}_D$, I find

$$\Psi = \overline{\phi}_D - \frac{Ze}{2MB} \frac{\partial}{\partial \mu} \langle \widetilde{\phi}_D^2 \rangle - \frac{c}{2B\Omega} \langle (\nabla_{\mathbf{R}} \widetilde{\Phi}_D \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \widetilde{\phi}_D \rangle, \qquad (C.41)$$

exactly as in equation (19b) of reference [21].

Finally, I will compare the quasineutrality equations in both methods. Taylor

expanding the ion distribution function around $\mathbf{R}_g = \mathbf{r} + \Omega^{-1} \mathbf{v} \times \hat{\mathbf{b}}$, $v_{||}$, μ_0 and φ_0 , I find

$$f_i(\mathbf{R}, u, \mu, t) \simeq f_{ig} + \mathbf{R}_2 \cdot \nabla_{\mathbf{R}_g} f_{ig} - \frac{c}{B} \hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\Phi} \frac{\partial f_{ig}}{\partial v_{||}} + (\mu_1 + \mu_2) \frac{\partial f_{ig}}{\partial \mu_0} + \frac{\mu_1^2}{2} \frac{\partial^2 f_{ig}}{\partial \mu_0^2}, \quad (C.42)$$

where $f_{ig} \equiv f_i(\mathbf{R}_g, v_{||}, \mu_0, t)$. Here I have used equations (C.29) and (C.30) to obtain that $u \simeq v_{||} - (c/B)\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\Phi}$. The ion density is given by

$$n_i = \int d^3 v f_i \simeq \int d^3 v \left[f_{ig} + \mathbf{R}_2 \cdot \nabla_{\mathbf{R}_g} f_{ig} + (\mu_1 + \langle \mu_2 \rangle) \frac{\partial f_{ig}}{\partial \mu_0} + \frac{\mu_1^2}{2} \frac{\partial^2 f_{ig}}{\partial \mu_0^2} \right]. \quad (C.43)$$

Here, the integrals of $(c/B)(\hat{\mathbf{b}} \cdot \nabla_{\mathbf{R}} \tilde{\Phi})(\partial f_{ig}/\partial v_{||})$ and $(\mu_2 - \langle \mu_2 \rangle)(\partial f_{ig}/\partial \mu_0)$ vanish because $\oint d\varphi_0 \tilde{\Phi} = 0$ and the only gyrophase dependence is in $\tilde{\Phi}$ since f_{ig} is assumed to be a smooth function of \mathbf{r} and \mathbf{v} to lowest order, giving $f_{ig} \equiv f_i(\mathbf{R}_g, v_{||}, \mu_0, t) \simeq$ $f_i(\mathbf{r}, v_{||}, \mu_0, t)$. The integral $\oint d\varphi_0 \tilde{\Phi}$ is performed holding $\mathbf{r}, v_{||}, \mu_0$ and t fixed, and it vanishes to lowest order as proven at the end of this Appendix. On the other hand, the integral of $\mathbf{R}_2 \cdot \nabla_{\mathbf{R}_g} f_{ig}$ does not vanish. Here the gyrophase dependence of f_{ig} contained in its short wavelength contributions becomes important due to the steep gradient [recall the ordering in (C.1)].

In equation (C.43), I can employ $\tilde{\phi} \simeq \tilde{\phi}_D$ and $\tilde{\Phi} \simeq -\tilde{\Phi}_D$ in the higher order terms. However, for μ_1 I need the difference between $\tilde{\phi}$ and $\tilde{\phi}_D$. Subtracting (C.40) from (C.39), I find

$$\widetilde{\phi} \simeq \widetilde{\phi}_D - \frac{Ze}{MB} \left(\widetilde{\phi} \frac{\partial \phi}{\partial \mu} - \frac{\partial \widetilde{\Phi}}{\partial \mu} \frac{\partial \widetilde{\phi}}{\partial \varphi} \right) + \frac{c}{B\Omega} (\nabla_{\mathbf{R}} \widetilde{\Phi} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \phi + \frac{Ze}{MB} \frac{\partial}{\partial \mu} \langle \widetilde{\phi}^2 \rangle - \frac{c}{B\Omega} \langle (\nabla_{\mathbf{R}} \widetilde{\Phi} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}} \widetilde{\phi} \rangle. \quad (C.44)$$

In this equation, $\tilde{\phi}_D = \tilde{\phi}_D(\mathbf{R}, \mu, \varphi, t)$, but for equation (C.43), it is better to use $\tilde{\phi}_{Dg} = \tilde{\phi}_D(\mathbf{R}_g, \mu_0, \varphi_0, t)$. By Taylor expanding, I find that

$$\tilde{\phi}_D \simeq \tilde{\phi}_{Dg} + \mathbf{R}_2 \cdot \nabla_{\mathbf{R}_g} \tilde{\phi}_{Dg} + \mu_1 \frac{\partial \tilde{\phi}_{Dg}}{\partial \mu_0} + \varphi_1 \frac{\partial \tilde{\phi}_{Dg}}{\partial \varphi_0}.$$
(C.45)

This equation, combined with equation (C.44) and the definitions of \mathbf{R}_2 , μ_1 and φ_1 ,

leads to

$$\begin{split} \tilde{\phi} \simeq \tilde{\phi}_{Dg} - \frac{Ze\tilde{\phi}_{Dg}}{MB} \frac{\partial \overline{\phi}_{Dg}}{\partial \mu} - \frac{c}{B\Omega} (\nabla_{\mathbf{R}_g} \tilde{\Phi}_{Dg} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_g} \overline{\phi}_{Dg} + \frac{Ze}{MB} \frac{\partial}{\partial \mu} \langle \tilde{\phi}_{Dg}^2 \rangle \\ + \frac{c}{B\Omega} \langle (\nabla_{\mathbf{R}_g} \tilde{\Phi}_{Dg} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_g} \tilde{\phi}_{Dg} \rangle, \quad (C.46) \end{split}$$

where I have used that, to lowest order, $\tilde{\phi}_D \simeq \tilde{\phi}_{Dg}$, $\bar{\phi}_D \simeq \bar{\phi}_{Dg} \equiv \bar{\phi}_D(\mathbf{R}_g, \mu_0, t)$ and $\tilde{\Phi}_D \simeq \tilde{\Phi}_{Dg} \equiv \tilde{\Phi}_D(\mathbf{R}_g, \mu_0, \varphi_0, t)$. Substituting equation (C.46) into (C.43) yields

$$n_{i} \simeq \int d^{3}v \left\{ f_{ig} + \frac{Ze\tilde{\phi}_{Dg}}{MB} \frac{\partial f_{ig}}{\partial \mu_{0}} + \frac{c}{B\Omega} (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{Dg} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} f_{ig} \right. \\ \left. + \frac{Z^{2}e^{2}\tilde{\phi}_{Dg}^{2}}{2M^{2}B^{2}} \frac{\partial^{2}f_{ig}}{\partial \mu_{0}^{2}} + \frac{Ze}{MB} \left[-\frac{Ze\tilde{\phi}_{Dg}}{MB} \frac{\partial \overline{\phi}_{Dg}}{\partial \mu_{0}} - \frac{c}{B\Omega} (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{Dg} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} \overline{\phi}_{Dg} \right. \\ \left. + \frac{Ze}{2MB} \frac{\partial}{\partial \mu_{0}} \langle \tilde{\phi}_{Dg}^{2} \rangle + \frac{c}{2B\Omega} \langle (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{Dg} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} \tilde{\phi}_{Dg} \rangle \right] \frac{\partial f_{ig}}{\partial \mu_{0}} \right\}, \quad (C.47)$$

where I have used the definitions of \mathbf{R}_2 and $\langle \mu_2 \rangle$. This result is exactly the same as in equation (20) in reference [21]. For comparison, I give n_i to order $\delta^2 n_i$ with the definitions of $\langle \phi \rangle$, $\tilde{\phi}$ and $\tilde{\Phi}$ in equations (3.16), (3.17) and (3.18),

$$n_{i} \simeq \int d^{3}v \left\{ f_{ig} + \frac{Ze\tilde{\phi}_{g}}{MB} \frac{\partial f_{ig}}{\partial \mu_{0}} - \frac{c}{B\Omega} (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{g} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} f_{ig} + \frac{Z^{2}e^{2}\tilde{\phi}_{g}^{2}}{2M^{2}B^{2}} \frac{\partial^{2}f_{ig}}{\partial \mu_{0}^{2}} \right. \\ \left. + \frac{Ze}{MB} \left[\frac{Ze\tilde{\phi}_{g}}{MB} \frac{\partial \tilde{\phi}_{g}}{\partial \mu_{0}} - \frac{Ze\tilde{\Phi}_{g}}{MB} \frac{\partial \tilde{\phi}_{g}}{\partial \varphi_{0}} - \frac{c}{B\Omega} (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{g} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} \tilde{\phi}_{g} \right. \\ \left. - \frac{Ze}{2MB} \frac{\partial}{\partial \mu_{0}} \langle \tilde{\phi}_{g}^{2} \rangle + \frac{c}{2B\Omega} \langle (\nabla_{\mathbf{R}_{g}} \tilde{\Phi}_{g} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_{g}} \tilde{\phi}_{g} \rangle \right] \frac{\partial f_{ig}}{\partial \mu_{0}} \right\}. \quad (C.48)$$

I have found this equation substituting μ_1 , \mathbf{R}_2 and $\langle \mu_2 \rangle$ into (C.43). From the functions $\tilde{\phi}(\mathbf{R}, \mu, \varphi, t)$ and $\tilde{\Phi}(\mathbf{R}, \mu, \varphi, t)$, I have defined $\tilde{\phi}_g = \tilde{\phi}(\mathbf{R}_g, \mu_0, \varphi_0, t)$ and $\tilde{\Phi}_g = \tilde{\Phi}(\mathbf{R}_g, \mu_0, \varphi_0, t)$. The relationships between $\tilde{\phi}$ and $\tilde{\phi}_g$ and between $\tilde{\Phi}$ and $\tilde{\Phi}_g$ are similar to the one given in (C.45).

The methodology and results of chapter 3 are completely consistent with the results of [21] since they give the same gyrokinetic equation (C.32), generalized potential Ψ (C.41) and quasineutrality condition (C.47).

Integral $\oint d\varphi_0 \widetilde{\Phi}$

To prove that $\oint d\varphi_0 \tilde{\Phi} = 0$ vanishes, I Fourier analyze $\phi = (2\pi)^{-3} \int d^3k \, \phi_{\mathbf{k}} \exp(i\mathbf{k} \cdot \mathbf{r})$, giving to lowest order

$$\phi(\mathbf{r},t) \simeq \phi(\mathbf{R} - \Omega^{-1}\mathbf{v} \times \hat{\mathbf{b}}, t) = \frac{1}{(2\pi)^3} \int d^3k \,\phi_{\mathbf{k}} \exp[i\mathbf{k} \cdot \mathbf{R} - iz\sin(\varphi_0 - \varphi_{\mathbf{k}})], \quad (C.49)$$

where $z = k_{\perp} v_{\perp} / \Omega$. Here, I employ $\mathbf{r} \simeq \mathbf{R} - \Omega^{-1} \mathbf{v} \times \hat{\mathbf{b}}$ and I define $\varphi_{\mathbf{k}}$ such that $\mathbf{k}_{\perp} = k_{\perp} (\hat{\mathbf{e}}_1 \cos \varphi_{\mathbf{k}} + \hat{\mathbf{e}}_2 \sin \varphi_{\mathbf{k}})$ to write $\mathbf{k} \cdot \mathbf{r} \simeq \mathbf{k} \cdot \mathbf{R} - z \sin(\varphi_0 - \varphi_{\mathbf{k}})$. Then, I use

$$\exp(iz\sin\varphi) = \sum_{m=-\infty}^{\infty} J_m(z)\exp(im\varphi), \qquad (C.50)$$

with $J_m(z)$ the Bessel function of the first kind, to find

$$\phi \simeq \frac{1}{(2\pi)^3} \int d^3k \,\phi_{\mathbf{k}} \exp(i\mathbf{k} \cdot \mathbf{R}) \sum_{m=-\infty}^{\infty} J_m(z) \exp[-im(\varphi_0 - \varphi_{\mathbf{k}})]. \tag{C.51}$$

Employing this expression, I obtain $\tilde{\phi}$ by subtracting the average in φ_0 (component m = 0), and I find $\tilde{\Phi}$ by integrating $\tilde{\phi}$ over φ_0 , giving

$$\tilde{\Phi} \simeq \frac{1}{(2\pi)^3} \int d^3k \,\phi_{\mathbf{k}} \exp(i\mathbf{k} \cdot \mathbf{R}) \sum_{m \neq 0} \frac{i}{m} J_m(z) \exp[-im(\varphi_0 - \varphi_{\mathbf{k}})], \quad (C.52)$$

where the summation includes every positive and negative m different from 0. To rewrite $\tilde{\Phi}$ as a function of \mathbf{r} , $v_{||}$, μ_0 and φ_0 , I need the expression

$$\exp(i\mathbf{k}\cdot\mathbf{R})\simeq\exp(i\mathbf{k}\cdot\mathbf{r})\sum_{p=-\infty}^{\infty}J_p(z)\exp[ip(\varphi_0-\varphi_{\mathbf{k}})],\qquad(C.53)$$

deduced from $\mathbf{R} \simeq \mathbf{r} + \Omega^{-1} \mathbf{v} \times \hat{\mathbf{b}}$ and (C.50). Then, I find

$$\widetilde{\Phi} \simeq \frac{1}{(2\pi)^3} \int d^3k \,\phi_{\mathbf{k}} \exp(i\mathbf{k} \cdot \mathbf{r}) \sum_{m \neq 0, p} \frac{i}{m} J_m(z) J_p(z) \exp[i(p-m)(\varphi_0 - \varphi_{\mathbf{k}})]. \quad (C.54)$$

Finally, integrating in φ_0 , I obtain

$$\frac{1}{2\pi} \oint d\varphi_0 \,\widetilde{\Phi} = \frac{1}{(2\pi)^3} \int d^3k \,\phi_{\mathbf{k}} \exp(i\mathbf{k} \cdot \mathbf{r}) \sum_{m \neq 0} \frac{i}{m} [J_m(z)]^2 = 0 \tag{C.55}$$

since $J_{-m}(z) = (-1)^m J_m(z)$.

Appendix D

Quasineutrality equation at long wavelengths

In this Appendix, I obtain the gyrokinetic quasineutrality equation at long wavelengths $(k_{\perp}\rho_i \ll 1)$. In section D.1, the polarization density n_{ip} in (3.55) is calculated for the intermediate scales $\delta_i \ll k_{\perp}\rho_i \ll 1$. In section D.2, the polarization density n_{ip} and the ion guiding center density \hat{N}_i are computed up to $O(\delta_i^2 n_e)$ for the extreme case of a non-turbulent plasma.

D.1 Polarization density at long wavelengths

For long wavelengths $\delta_i \ll k_{\perp}\rho_i \ll 1$, the polarization density simplifies to give (3.59). To obtain this result, $\tilde{\phi}$ and hence ϕ must be obtained to $O(\delta_i k_{\perp}\rho_i T_e/e)$. To this order, the potential is

$$\phi(\mathbf{r}) \simeq \phi(\mathbf{R} - \mathbf{R}_1) \simeq \phi(\mathbf{R}) - \mathbf{R}_1 \cdot \nabla_{\mathbf{R}} \phi + \frac{1}{2} \mathbf{R}_1 \mathbf{R}_1 : \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \phi, \qquad (D.1)$$

with $\mathbf{R}_1 = \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$. Here, according to (3.1), $\nabla_{\mathbf{R}} \phi \sim T_e/eL$ and $\nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \phi \sim k_{\perp} T_e/eL$. With the result in (D.1), the function $\tilde{\phi}$ is

$$\widetilde{\phi} = \phi - \langle \phi \rangle \simeq -\mathbf{R}_1 \cdot \nabla_{\mathbf{R}} \phi + \frac{1}{2} (\mathbf{R}_1 \mathbf{R}_1 - \langle \mathbf{R}_1 \mathbf{R}_1 \rangle) : \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \phi, \qquad (D.2)$$

where I neglect $\langle \mathbf{R}_1 \rangle \sim \delta_i \rho_i$. The gyroaverage of the first order correction \mathbf{R}_1 is not zero due to the difference between \mathbf{r} , μ_0 and φ_0 and the gyrokinetic variables \mathbf{R} , μ and φ . This difference will become important in section D.2.

The integral n_{ip} in (3.55) is performed over velocity space holding **r** and *t* fixed. Then, equation (D.2) has to be Taylor expanded around **r** to obtain a function that depends on **r** and not **R**. To $O(\delta_i k_\perp \rho_i T_e/e)$, the result is

$$\tilde{\phi} \simeq -\mathbf{R}_1 \cdot \nabla \phi - \frac{1}{2} (\mathbf{R}_1 \mathbf{R}_1 + \overline{\mathbf{R}_1 \mathbf{R}_1}) : \nabla \nabla \phi, \qquad (D.3)$$

where I use $\nabla_{\mathbf{R}}\phi \simeq \nabla\phi + \mathbf{R}_1 \cdot \nabla\nabla\phi$. In the higher order terms, the gyroaverage $\langle \ldots \rangle$ holding the gyrokinetics variable fixed can be approximated by the gyroaverage $\overline{(\ldots)}$ holding $\mathbf{r}, v_{\parallel}$ and v_{\perp} fixed.

Expression (D.3) can be readily substituted into (3.55) and integrated to give

$$n_{ip} \simeq \frac{cn_i}{B\Omega_i} (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) : \nabla \nabla \phi.$$
 (D.4)

The final result in (3.59) is found by realizing that $\nabla \cdot [(cn_i/B\Omega_i)(\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}})] \cdot \nabla_{\perp} \phi$ is small by a factor $(k_{\perp}L)^{-1}$ compared to the term in (D.4).

D.2 Quasineutrality equation for $k_{\perp}L \sim 1$

In this section, I obtain the long wavelength quasineutrality condition to order $\delta_i^2 n_e$. The derivation is not applicable to turbulent plasmas because I will assume that neither the potential nor the distribution function have short wavelength pieces.

In quasineutrality, the ion distribution function must be written in \mathbf{r} , \mathbf{v} variables. In a non-turbulent plasma, the ion distribution function can be expanded around \mathbf{r} , E_0 and μ_0 up to $O(\delta_i^2 f_{Mi})$, giving

$$f_{i}(\mathbf{R}, E, \mu, t) = f_{i0} + (\mathbf{R}_{1} + \mathbf{R}_{2}) \cdot \overline{\nabla} f_{i0} + (E_{1} + E_{2}) \frac{\partial f_{i0}}{\partial E_{0}} + \mu_{1} \frac{\partial f_{i0}}{\partial \mu_{0}} + \frac{1}{2} \mathbf{R}_{1} \mathbf{R}_{1} : \overline{\nabla} \overline{\nabla} f_{Mi} + E_{1} \mathbf{R}_{1} \cdot \overline{\nabla} \left(\frac{\partial f_{Mi}}{\partial E_{0}} \right) + \frac{E_{1}^{2}}{2} \frac{\partial^{2} f_{Mi}}{\partial E_{0}^{2}}, \quad (D.5)$$

with $f_{i0} \equiv f_i(\mathbf{r}, E_0, \mu_0, t)$ and $\overline{\nabla}$ the gradient holding E_0 , μ_0 , φ_0 and t fixed. In f_i I have not included the collisional gyrophase dependent piece given in (3.38) because it will not contribute to the density. In the higher order terms of the Taylor expansion (D.5), the lowest order distribution function $f_i \simeq f_{Mi}$ must be used.

To find the ion density, the ion distribution function is integrated in velocity space. Some of the terms vanish because the integral over gyrophase is zero; for example, $\int d^3v \left(\mathbf{R}_1 + \mathbf{R}_2\right) \cdot \overline{\nabla} f_{i0} = 0 = \int d^3v \ E_2(\partial f_{i0}/\partial E_0)$. Then, the ion density becomes

$$n_i \simeq \hat{N}_i + \int d^3 v \, \frac{Ze\tilde{\phi}}{M} \left[\frac{\partial f_{i0}}{\partial E_0} + \frac{1}{B} \frac{\partial f_{i0}}{\partial \mu_0} + \frac{Ze\tilde{\phi}}{2M} \frac{\partial^2 f_{Mi}}{\partial E_0^2} + \mathbf{R}_1 \cdot \overline{\nabla} \left(\frac{\partial f_{Mi}}{\partial E_0} \right) \right], \qquad (D.6)$$

where $\hat{N}_i(\mathbf{r}, t)$ is the ion gyrocenter density, defined as the portion of the ion density independent of $\tilde{\phi}$ and given by

$$\hat{N}_{i} = \int d^{3}v f_{i0} - \int d^{3}v \frac{v_{\parallel} v_{\perp}^{2}}{2B\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \frac{\partial f_{i0}}{\partial \mu_{0}} + \int d^{3}v \frac{\mathbf{R}_{1} \mathbf{R}_{1}}{2} : \overline{\nabla} \,\overline{\nabla} f_{Mi}. \tag{D.7}$$

The formula for \hat{N}_i can be simplified. The second term in the right side of the equation is proportional to $\int d^3v (v_{\parallel}v_{\perp}^2/2B)(\partial f_{i0}/\partial \mu_0)$. This integral is simplified by changing to the variables $E_0 = v^2/2$, $\mu_0 = v_{\perp}^2/2B$ and φ_0 and integrating by parts,

$$-\int d^{3}v \, \frac{v_{||} v_{\perp}^{2}}{2B} \frac{\partial f_{i0}}{\partial \mu_{0}} = -B \sum_{\sigma} \int dE_{0} \, d\mu_{0} \, d\varphi_{0} \, \sigma \mu_{0} \frac{\partial f_{i0}}{\partial \mu_{0}} = \int d^{3}v \, v_{||} f_{i0}, \qquad (D.8)$$

where $\sigma = v_{||}/|v_{||}|$ is the sign of the parallel velocity, the summation in front of the integral indicates that the integral must be done for both signs of $v_{||}$, and I have used the equality $d^3v = dE_0 d\mu_0 d\varphi_0 B/|v_{||}|$. The third term in the right side of (D.7) is proportional to

$$M \int d^3 v \left(\mathbf{v} \times \hat{\mathbf{b}} \right) \left(\mathbf{v} \times \hat{\mathbf{b}} \right) : \overline{\nabla} \, \overline{\nabla} f_{Mi} = (\vec{\mathbf{I}} - \hat{\mathbf{b}} \hat{\mathbf{b}}) : \nabla \nabla p_i.$$
(D.9)

The final function \hat{N}_i reduces to the result shown in (3.62).

In equation (D.6), $\tilde{\phi}$ appears in several integrals. The expression can be simplified by integrating first in the gyrophase. Two integrals in (D.6) can be done by using the lowest order result $\tilde{\phi} \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$. The integrals are

$$\int d^3v \, \frac{Z^2 e^2}{2M^2} \tilde{\phi}^2 \frac{\partial^2 f_{Mi}}{\partial E_0^2} = \frac{M c^2 n_i}{2T_i B^2} |\nabla_\perp \phi|^2 \tag{D.10}$$

and

$$\int d^3 v \, \frac{c}{B} \widetilde{\phi}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \left(\frac{\partial f_{Mi}}{\partial E_0} \right) = \frac{c}{B\Omega_i} \nabla n_i \cdot \nabla_\perp \phi. \tag{D.11}$$

For $\int d^3v \,\tilde{\phi}(\partial f_{i0}/\partial E_0 + B^{-1}\partial f_{i0}/\partial\mu_0)$, only the gyrophase integral $\int_0^{2\pi} \tilde{\phi} \, d\varphi_0$ is needed, but $\tilde{\phi}$ must be written as a function of the variables \mathbf{r} , \mathbf{v} to $O(\delta_i^2 T_e/e)$ to be consistent with the order of the Taylor expansion. To do so, first I will write $\phi(\mathbf{r}, t)$ as a function of the gyrokinetic variables by Taylor expansion to $O(\delta_i^2 T_e/e)$ [this Taylor expansion is carried out to higher order than in (D.1)]. The result is

$$\phi(\mathbf{r},t) \simeq \phi(\mathbf{R},t) - \mathbf{R}_1 \cdot \nabla_{\mathbf{R}}\phi + \frac{1}{2}\mathbf{R}_1\mathbf{R}_1 : \nabla_{\mathbf{R}}\nabla_{\mathbf{R}}\phi - \mathbf{R}_2 \cdot \nabla_{\mathbf{R}}\phi.$$
(D.12)

The second term in the right side of the equation needs to be re-expanded in order to express ϕ as a self-consistent function of the gyrokinetic variables to the right order. The function \mathbf{R}_1 is, to $O(\delta_i \rho_i)$,

$$\mathbf{R}_{1} = \frac{1}{\Omega_{i}} \mathbf{v} \times \hat{\mathbf{b}} \simeq \frac{\sqrt{2\mu B(\mathbf{R})}}{\Omega_{i}(\mathbf{R})} [\hat{\mathbf{e}}_{1}(\mathbf{R}) \sin \varphi - \hat{\mathbf{e}}_{2}(\mathbf{R}) \cos \varphi] - \Delta \boldsymbol{\rho}, \qquad (D.13)$$

with $\Delta \rho(\mathbf{R}, E, \mu, \varphi, t) \sim \delta_i \rho_i$. The function $\Delta \rho$ is found by Taylor expanding $\mathbf{R}_1(\mathbf{r}, \mu_0, \varphi_0)$ around \mathbf{R}, μ and φ to $O(\delta_i \rho_i)$, giving

$$\Delta \boldsymbol{\rho} = -\frac{1}{2B\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B + \frac{Mc}{Zev_\perp^2} \mu_1 (\mathbf{v} \times \hat{\mathbf{b}}) + \frac{1}{\Omega_i} \varphi_1 \mathbf{v}_\perp + \frac{v_\perp}{\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot (\sin \varphi_0 \nabla \hat{\mathbf{e}}_1 - \cos \varphi_0 \nabla \hat{\mathbf{e}}_2).$$
(D.14)

Combining (D.12) and (D.13), ϕ is found to be

$$\tilde{\phi} = \phi - \langle \phi \rangle \simeq -\frac{\sqrt{2\mu B(\mathbf{R})}}{\Omega_i(\mathbf{R})} [\hat{\mathbf{e}}_1(\mathbf{R}) \sin \varphi - \hat{\mathbf{e}}_2(\mathbf{R}) \cos \varphi] \cdot \nabla_{\mathbf{R}} \phi - \mathbf{R}_2 \cdot \nabla_{\mathbf{R}} \phi$$

$$+(\Delta \boldsymbol{\rho} - \langle \Delta \boldsymbol{\rho} \rangle) \cdot \nabla_{\mathbf{R}} \phi + \frac{1}{2\Omega_{i}^{2}} [(\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) - \langle (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) \rangle] : \nabla_{\mathbf{R}} \nabla_{\mathbf{R}} \phi. (D.15)$$

The function $\tilde{\phi}$ must be written as a function of \mathbf{r} , \mathbf{v} and t since the integral in velocity space is done for \mathbf{r} and t fixed. Taylor expanding the first term in (D.15) gives

$$\widetilde{\phi} \simeq -\frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi - \langle \Delta \boldsymbol{\rho} \rangle \cdot \nabla \phi - \mathbf{R}_2 \cdot \nabla \phi - \frac{v_{\perp}^2}{4\Omega_i^2} (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) : \nabla \nabla \phi - \frac{1}{2\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) : \nabla \nabla \phi, \quad (D.16)$$

where I use $\langle (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) \rangle = (v_{\perp}^2/2) (\mathbf{\vec{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}})$. Equation (D.16) is more complete than the approximation in (D.3).

Gyroaveraging $\tilde{\phi}$ in (D.16) holding $\mathbf{r}, v_{||}$ and v_{\perp} fixed leads to

$$\frac{1}{2\pi} \int_{0}^{2\pi} \widetilde{\phi} \, d\varphi_0 \simeq -\langle \Delta \boldsymbol{\rho} \rangle \cdot \nabla \phi - \frac{v_{\perp}^2}{2\Omega_i^2} (\mathbf{\vec{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}) : \nabla \nabla \phi.$$
(D.17)

with $\Delta \boldsymbol{\rho}$ the function in (D.14). To simplify equation (D.14), the gradients of the unit vectors $\hat{\mathbf{e}}_1$ and $\hat{\mathbf{e}}_2$ are expressed as $\nabla \hat{\mathbf{e}}_1 = -(\nabla \hat{\mathbf{b}} \cdot \hat{\mathbf{e}}_1)\hat{\mathbf{b}} - (\nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1)\hat{\mathbf{e}}_2$ and $\nabla \hat{\mathbf{e}}_2 = -(\nabla \hat{\mathbf{b}} \cdot \hat{\mathbf{e}}_2)\hat{\mathbf{b}} + (\nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1)\hat{\mathbf{e}}_1$, giving

$$\Delta \boldsymbol{\rho} = -\frac{1}{2B\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B + \frac{Mc}{Zev_\perp^2} \mu_1 (\mathbf{v} \times \hat{\mathbf{b}}) + \frac{1}{\Omega_i} \varphi_1 \mathbf{v}_\perp -\frac{1}{\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) \hat{\mathbf{b}} - \frac{1}{\Omega_i^2} \mathbf{v}_\perp (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1, \qquad (D.18)$$

and its gyroaverage

$$\langle \Delta \boldsymbol{\rho} \rangle = -\frac{c}{B\Omega_i} \nabla_\perp \phi - \frac{v_\perp^2}{B\Omega_i^2} \nabla_\perp B - \frac{v_{\parallel}^2}{\Omega_i^2} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} - \frac{v_\perp^2}{2\Omega_i^2} (\nabla \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}}, \qquad (D.19)$$

where I use

$$\langle \mu_1(\mathbf{v} \times \hat{\mathbf{b}}) \rangle = -\frac{cv_{\perp}^2}{2B^2} \nabla_{\perp} \phi - \frac{v_{\perp}^4}{4B^2 \Omega_i} \nabla_{\perp} B - \frac{v_{\perp}^2 v_{\parallel}^2}{2B\Omega_i} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}$$
(D.20)

and

$$\langle \varphi_1 \mathbf{v}_{\perp} \rangle = -\frac{c}{2B} \nabla_{\perp} \phi - \frac{v_{\perp}^2}{2B\Omega_i} \nabla_{\perp} B - \frac{v_{\parallel}^2}{2\Omega_i} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} + \frac{v_{\perp}^2}{2\Omega_i} \hat{\mathbf{b}} \times \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1.$$
(D.21)

These expressions are found by using the definitions of μ_1 and φ_1 , given by (3.29) and (3.34), and employing the lowest order expressions $\tilde{\phi} \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$, $\tilde{\Phi} = \int^{\varphi} \tilde{\phi} \, d\varphi' \simeq \Omega_i^{-1} \mathbf{v}_{\perp} \cdot \nabla \phi$ and $\partial \tilde{\Phi} / \partial \mu \simeq (Mc/Zev_{\perp}^2) \mathbf{v}_{\perp} \cdot \nabla \phi$.

Substituting (D.19) in (D.17) gives

$$\frac{1}{2\pi} \int_0^{2\pi} \tilde{\phi} \, d\varphi_0 = -\frac{v_\perp^2}{2} \nabla \cdot \left(\frac{1}{\Omega_i^2} \nabla_\perp \phi\right) + \left(v_{||}^2 - \frac{v_\perp^2}{2}\right) \frac{1}{\Omega_i^2} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot \nabla \phi + \frac{c}{B\Omega_i} |\nabla_\perp \phi|^2, \quad (D.22)$$

where I have used

$$\nabla \cdot \left(\frac{1}{\Omega_i^2} \nabla_{\perp} \phi\right) = \frac{1}{\Omega_i^2} (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) : \nabla \nabla \phi - \frac{2}{\Omega_i^2} \nabla B \cdot \nabla_{\perp} \phi - \frac{1}{\Omega_i^2} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot \nabla \phi - \frac{1}{\Omega_i^2} (\hat{\mathbf{b}} \cdot \nabla \phi) \nabla \cdot \hat{\mathbf{b}}. \quad (D.23)$$

Notice that the gyroaverage of $\tilde{\phi}$ is $O(\delta_i^2 T_e/e)$, which means that the integral is $O(\delta_i^2 n_i)$, and the lowest order distribution function, f_{Mi} , can be used to write $\partial f_{i0}/\partial E_0 \simeq -(M/T_i)f_{Mi}$ and $\partial f_{i0}/\partial \mu_0 \simeq 0$. All these simplifications lead to the final result

$$\frac{Ze}{M} \int d^3 v \, \tilde{\phi} \left(\frac{\partial f_{i0}}{\partial E_0} + \frac{1}{B} \frac{\partial f_{i0}}{\partial \mu_0} \right) = n_i \nabla \cdot \left(\frac{c}{B\Omega_i} \nabla_\perp \phi \right) - \frac{M c^2 n_i}{T_i B^2} |\nabla_\perp \phi|^2, \qquad (D.24)$$

Using (D.10), (D.11) and (D.24), equation (D.6) becomes

$$n_i = \hat{N}_i + \nabla \cdot \left(\frac{cn_i}{B\Omega_i} \nabla_\perp \phi\right) - \frac{Mc^2 n_i}{2T_i B^2} |\nabla_\perp \phi|^2, \qquad (D.25)$$

where \hat{N}_i is given by (3.62). Then the quasineutrality condition is as shown in (3.61).

Appendix E

Gyrophase dependent piece of f_i for $k_{\perp}L \sim 1$

In this Appendix, I show how to obtain the $O(\delta_i^2 f_{Mi})$ gyrophase dependent piece of the ion distribution function from gyrokinetics in a non-turbulent plasma. The general gyrophase dependent piece is found in section E.1. This result is useful because it can be compared to the drift kinetic result [74], proving that the higher order gyrokinetic variables allow us to recover the higher order drift kinetic results. In section E.2, the general gyrophase dependent piece is specialized for the θ -pinch, and it is used in section 5.2 to calculate the radial electric field.

E.1 General gyrophase dependent piece

Part of the gyrophase dependence is in the corrections to the gyrokinetic variables \mathbf{R}_1 , \mathbf{R}_2 , E_1 , E_2 and μ_1 . This gyrophase dependence can be extracted for $k_{\perp}L \sim 1$ by Taylor expanding $f_i(\mathbf{R}, E, \mu, t)$ around \mathbf{r} , E_0 and μ_0 , as already done in (D.5). The contribution of the collisional piece \tilde{f}_i , given by (3.38), can be always added later.

Employing (D.5), the gyrophase dependent part of f_i is found to be

$$f_i - \overline{f}_i = \left(\frac{1}{\Omega_i} \mathbf{v} \times \hat{\mathbf{b}} + \mathbf{R}_2\right) \cdot \overline{\nabla} f_{i0} + \left[\frac{Ze}{M}(\tilde{\phi} - \overline{\tilde{\phi}}) + E_2\right] \frac{\partial f_{i0}}{\partial E_0}$$

$$+\frac{1}{4\Omega_{i}^{2}}[(\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})-\mathbf{v}_{\perp}\mathbf{v}_{\perp}]:\overline{\nabla}\,\overline{\nabla}f_{Mi}+\frac{Z^{2}e^{2}}{2M^{2}}(\widetilde{\phi}^{2}-\overline{\phi}^{2})\frac{\partial^{2}f_{Mi}}{\partial E_{0}^{2}}\\+\frac{c}{B}[\widetilde{\phi}(\mathbf{v}\times\hat{\mathbf{b}})-\overline{\widetilde{\phi}(\mathbf{v}\times\hat{\mathbf{b}})}]\cdot\overline{\nabla}\left(\frac{\partial f_{Mi}}{\partial E_{0}}\right)+(\mu_{1}-\overline{\mu}_{1})\frac{\partial f_{i0}}{\partial \mu_{0}},\qquad(E.1)$$

with $f_{i0} \equiv f_i(\mathbf{r}, E_0, \mu_0, t)$. Here, I have employed the lowest order distribution function f_{Mi} in the higher order results, and I have used (A.7) to rewrite $(\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) - \overline{(\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})}$.

The function $\tilde{\phi}$ must be written as a function of the **r**, **v** variables. To do so, I use equation (D.16) to find

$$\tilde{\phi} - \overline{\tilde{\phi}} = -\frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi - \mathbf{R}_2 \cdot \nabla \phi + \frac{1}{4\Omega_i^2} [\mathbf{v}_\perp \mathbf{v}_\perp - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})] : (\nabla \nabla \phi). (E.2)$$

For the higher order terms in (E.1), I can simply use the lowest order result $\tilde{\phi} \simeq -\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \phi$, which leads to

$$\tilde{\phi}^2 - \overline{\tilde{\phi}^2} = -\frac{1}{2\Omega_i^2} [\mathbf{v}_\perp \mathbf{v}_\perp - (\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})] : (\nabla \phi \nabla \phi)$$
(E.3)

and

$$\widetilde{\phi}(\mathbf{v} \times \widehat{\mathbf{b}}) - \overline{\widetilde{\phi}(\mathbf{v} \times \widehat{\mathbf{b}})} = \frac{1}{2\Omega_i} \nabla \phi \cdot [\mathbf{v}_{\perp} \mathbf{v}_{\perp} - (\mathbf{v} \times \widehat{\mathbf{b}})(\mathbf{v} \times \widehat{\mathbf{b}})].$$
(E.4)

Using these expressions, the gyrophase dependent part of the distribution function becomes

$$f_{i} - \overline{f}_{i} = \mathbf{v} \cdot \mathbf{g}_{\perp} + (\mu_{1} - \overline{\mu}_{1}) \frac{\partial f_{i0}}{\partial \mu_{0}} + \mathbf{R}_{2} \cdot \mathbf{G} + E_{2} \frac{\partial f_{i0}}{\partial E_{0}} + \frac{1}{4\Omega_{i}^{2}} [(\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}}) - \mathbf{v}_{\perp} \mathbf{v}_{\perp}] : \left(\overline{\nabla} \mathbf{G} - \frac{Ze}{M} \nabla \phi \frac{\partial \mathbf{G}}{\partial E_{0}}\right),$$
(E.5)

with

$$\mathbf{g}_{\perp} = \frac{1}{\Omega_i} \hat{\mathbf{b}} \times \left(\overline{\nabla} f_{i0} - \frac{Ze}{M} \nabla \phi \frac{\partial f_{i0}}{\partial E_0} \right)$$
(E.6)

and

$$\mathbf{G} = \overline{\nabla} f_{i0} - \frac{Ze}{M} \nabla \phi \frac{\partial f_{i0}}{\partial E_0}.$$
 (E.7)

Thus, $\mathbf{g}_{\perp} = \Omega_i^{-1} \hat{\mathbf{b}} \times \mathbf{G}$. In the long wavelength limit, $\partial/\partial t \ll v_i/L$, so E_2 as given in (3.26) is negligible since it contains a time derivative. Also, the zeroth order Fokker-Planck equation for the ion distribution function is

$$v_{||}\hat{\mathbf{b}} \cdot \mathbf{G} \equiv v_{||}\hat{\mathbf{b}} \cdot \left(\overline{\nabla}f_{i0} - \frac{Ze}{M}\nabla\phi\frac{\partial f_{i0}}{\partial E_0}\right) = C\{f_i\} = 0,$$
(E.8)

since the ion distribution function is assumed to be Maxwellian to zeroth order. This condition is important in (E.5) because it implies that the components of \mathbf{R}_2 that are parallel to the magnetic field do not enter $f_i - \overline{f}_i$. Therefore, employing the definition of \mathbf{R}_2 in (3.15) and using the fact that for long wavelengths $(c/B\Omega_i)\nabla_{\mathbf{R}}\tilde{\Phi} \times \hat{\mathbf{b}} \sim$ $\delta_i^2 k_\perp \rho_i L$ is negligible, I obtain

$$\mathbf{R}_{2} \cdot \mathbf{G} = \frac{1}{\Omega_{i}} \left[\left(v_{||} \hat{\mathbf{b}} + \frac{\mathbf{v}_{\perp}}{4} \right) \mathbf{v} \times \hat{\mathbf{b}} + \mathbf{v} \times \hat{\mathbf{b}} \left(v_{||} \hat{\mathbf{b}} + \frac{\mathbf{v}_{\perp}}{4} \right) \right] : \left[\nabla \left(\frac{\hat{\mathbf{b}}}{\Omega_{i}} \right) \times \mathbf{G} \right] \\ + \frac{v_{||}}{\Omega_{i}^{2}} \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{G}. \quad (E.9)$$

Equation (E.9) can be written in a more recognizable manner by using

$$\begin{aligned} [\hat{\mathbf{b}}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\hat{\mathbf{b}}] &: \hat{\mathbf{h}} = -\frac{1}{\Omega_i} \hat{\mathbf{b}} \cdot \left(\overline{\nabla} \mathbf{G} - \frac{Ze}{M} \nabla \phi \frac{\partial \mathbf{G}}{\partial E_0} \right) \cdot \mathbf{v}_\perp \\ + [\hat{\mathbf{b}}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\hat{\mathbf{b}}] : \left[\nabla \left(\frac{\hat{\mathbf{b}}}{\Omega_i} \right) \times \mathbf{G} \right], \end{aligned} \tag{E.10}$$

where $\overrightarrow{\mathbf{h}}$ is

$$\vec{\mathbf{h}} = \overline{\nabla} \mathbf{g}_{\perp} - \frac{Ze}{M} \nabla \phi \frac{\partial \mathbf{g}_{\perp}}{\partial E_0}.$$
(E.11)

The first term in the right side of (E.10) can be further simplified by using (E.8) to obtain

$$\hat{\mathbf{b}} \cdot \left(\overline{\nabla}\mathbf{G} - \frac{Ze}{M}\nabla\phi\frac{\partial\mathbf{G}}{\partial E_0}\right) \cdot \mathbf{v}_{\perp} = \mathbf{v}_{\perp} \cdot \overline{\nabla}\mathbf{G} \cdot \hat{\mathbf{b}} = -\mathbf{v}_{\perp} \cdot \nabla\hat{\mathbf{b}} \cdot \mathbf{G}.$$
(E.12)

As a result, (E.9) becomes

$$\mathbf{R}_{2} \cdot \mathbf{G} = \frac{v_{\parallel}}{\Omega_{i}} [\hat{\mathbf{b}}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\hat{\mathbf{b}}] : \mathbf{\hat{h}}$$
$$+ \frac{1}{4\Omega_{i}} [\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp}] : \left[\nabla\left(\frac{\hat{\mathbf{b}}}{\Omega_{i}}\right) \times \mathbf{G}\right].$$
(E.13)

The gyrophase dependent part of the ion distribution function can now be explicitly written as

$$(f_{i} - \overline{f}_{i})_{g} = \mathbf{v} \cdot \mathbf{g}_{\perp} - \left\{ \mathbf{v}_{d} \cdot \mathbf{v} + \frac{v_{\parallel}}{4B\Omega_{i}} [\mathbf{v}_{\perp}(\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}})\mathbf{v}_{\perp}] : \nabla \hat{\mathbf{b}} \right\} \frac{\partial f_{i0}}{\partial \mu_{0}} \\ + \frac{1}{\Omega_{i}} \left[\left(v_{\parallel} \hat{\mathbf{b}} + \frac{\mathbf{v}_{\perp}}{4} \right) \mathbf{v} \times \hat{\mathbf{b}} + \mathbf{v} \times \hat{\mathbf{b}} \left(v_{\parallel} \hat{\mathbf{b}} + \frac{\mathbf{v}_{\perp}}{4} \right) \right] : \mathbf{\vec{h}}, \quad (E.14)$$

where the subindex g indicates the non-collisional origin of this gyrophase dependence. Equation (E.14) is exactly the same gyrophase dependent distribution function found in [74].

E.2 Gyrophase dependent piece in a θ -pinch

The solution for f_i found in (4.79) means that for all the terms in (E.14), f_{i0} is approximately f_{M0} from (4.76). Due to the geometry in the θ -pinch, the vector \mathbf{g}_{\perp} defined in (E.6) is

$$\mathbf{g}_{\perp} = \frac{1}{\Omega_i} \hat{\boldsymbol{\theta}} \left(\frac{\partial f_{M0}}{\partial r} + \frac{Ze}{T_i} \frac{\partial \phi}{\partial r} f_{M0} \right), \tag{E.15}$$

and the matrix $\mathbf{\vec{h}}$ defined in (E.11) is

$$\begin{aligned} \vec{\mathbf{h}} &= \hat{\mathbf{r}}\hat{\boldsymbol{\theta}} \left\{ \frac{\partial}{\partial r} \left[\frac{1}{\Omega_i} \left(\frac{\partial f_{M0}}{\partial r} + \frac{Ze}{T_i} \frac{\partial \phi}{\partial r} f_{M0} \right) \right] + \frac{Ze}{\Omega_i} \frac{\partial \phi}{\partial r} \left[\frac{\partial}{\partial r} \left(\frac{f_{M0}}{T_i} \right) + \frac{Ze}{T_i^2} \frac{\partial \phi}{\partial r} f_{M0} \right] \right\} \\ &- \hat{\boldsymbol{\theta}}\hat{\mathbf{r}} \left[\frac{1}{r\Omega_i} \left(\frac{\partial f_{M0}}{\partial r} + \frac{Ze}{T_i} \frac{\partial \phi}{\partial r} f_{M0} \right) \right], (E.16) \end{aligned}$$

where I use $\nabla \hat{\boldsymbol{\theta}} = -\hat{\boldsymbol{\theta}}\hat{\mathbf{r}}/r$. Employing this results and taking into account that $\partial f_{i0}/\partial \mu_0 \simeq 0$ in this case, I find (5.18).

Appendix F

Gyrokinetic equation in physical phase space

This Appendix contains the details needed to write equation (4.13) as equation (4.23). In section F.1, equation (4.14) is derived. Equation (4.14) models the finite gyroradius effects in the particle motion through the modified parallel velocity $v_{\parallel 0}$ and the perpendicular drift $\tilde{\mathbf{v}}_1$. In section F.2, equation (4.14) is written in conservative form, more convenient to obtain moment equations.

F.1 Gyrokinetic equation in r, E_0 , μ_0 and φ_0 variables

In this section, I rewrite part of the gyrokinetic equation as a function of the variables **r**, E_0 , μ_0 and φ_0 . The gyrokinetic equation is only valid to $O(\delta_i f_{Mi} v_i/L)$, and the expansions will be carried out only to that order. In particular, I am interested in $\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E}(\partial f_i/\partial E)$. In the term $\dot{E}(\partial f_i/\partial E)$, I can make use of the lowest order equality $\partial f_i/\partial E \simeq \partial f_{Mi}/\partial E_0$. Then, employing \dot{E} from (3.27) gives

$$\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}} f_i + \dot{E} \frac{\partial f_i}{\partial E} \simeq \dot{\mathbf{R}} \cdot \left(\nabla_{\mathbf{R}} f_i - \frac{Ze}{M} \nabla_{\mathbf{R}} \langle \phi \rangle \frac{\partial f_{Mi}}{\partial E_0} \right).$$
(F.1)

For $\nabla_{\mathbf{R}} f_i$, changing from \mathbf{R} , E, μ and φ to \mathbf{r} , E_0 , μ_0 and φ_0 , I find

$$\nabla_{\mathbf{R}} f_i \simeq \nabla_{\mathbf{R}} \mathbf{r} \cdot \overline{\nabla} f_i + \nabla_{\mathbf{R}} E_0 \frac{\partial f_i}{\partial E_0} + \nabla_{\mathbf{R}} \mu_0 \frac{\partial f_i}{\partial \mu_0} + \nabla_{\mathbf{R}} \varphi_0 \frac{\partial f_i}{\partial \varphi_0}.$$
 (F.2)

Here, $\partial f_i / \partial \mu_0$ and $\partial f_i / \partial \varphi_0$ are small because the zeroth order distribution function is a stationary Maxwellian. The gradient of E_0 is given by $0 = \nabla_{\mathbf{R}} E \simeq \nabla_{\mathbf{R}} (E_0 + E_1) = \nabla_{\mathbf{R}} E_0 + (Ze/M) \nabla_{\mathbf{R}} \tilde{\phi}$. Similarly, $\nabla_{\mathbf{R}} \mu = 0 = \nabla_{\mathbf{R}} \varphi$ give $\nabla_{\mathbf{R}} \mu_0 \simeq -\nabla_{\mathbf{R}} \mu_1$ and $\nabla_{\mathbf{R}} \varphi_0 \simeq -\nabla_{\mathbf{R}} \varphi_1$. Then, to the required order

$$\nabla_{\mathbf{R}} f_i \simeq \nabla_{\mathbf{R}} \mathbf{r} \cdot \overline{\nabla} f_i - \frac{Ze}{M} \nabla_{\mathbf{R}} \widetilde{\phi} \frac{\partial f_i}{\partial E_0}.$$
 (F.3)

Since $\partial f_i/\partial E \simeq (-M/T_i)f_{Mi}$, and $\tilde{\phi} \nabla_{\mathbf{R}} f_{Mi} \ll f_{Mi} \nabla_{\mathbf{R}} \tilde{\phi}$ because the perpendicular gradient of $\tilde{\phi}$ is steeper and the parallel gradient of f_{Mi} is small, I find

$$\nabla_{\mathbf{R}} f_i \simeq \nabla_{\mathbf{R}} \mathbf{r} \cdot \overline{\nabla} \left(f_i + \frac{Z e \widetilde{\phi}}{T_i} f_{Mi} \right) = \nabla_{\mathbf{R}} \mathbf{r} \cdot \overline{\nabla} f_{ig}, \tag{F.4}$$

where I have employed the lowest order result (4.3) and $\nabla_{\mathbf{R}}\tilde{\phi} \simeq \nabla_{\mathbf{R}}\mathbf{r} \cdot \overline{\nabla}\tilde{\phi}$. To prove that $\nabla_{\mathbf{R}}\tilde{\phi} \simeq \nabla_{\mathbf{R}}\mathbf{r} \cdot \overline{\nabla}\tilde{\phi}$ and $\nabla_{\mathbf{R}}\langle\phi\rangle \simeq \nabla_{\mathbf{R}}\mathbf{r} \cdot \overline{\nabla}\langle\phi\rangle$, I follow a similar procedure to the one used for f_i in (F.2). In this case, $\partial\tilde{\phi}/\partial E_0$ and $\partial\langle\phi\rangle/\partial E_0$ are small. Substituting equations (F.4) and $\nabla_{\mathbf{R}}\langle\phi\rangle \simeq \nabla_{\mathbf{R}}\mathbf{r} \cdot \overline{\nabla}\langle\phi\rangle$ into (F.1), I find equation (4.14), where to write $\nabla_{\mathbf{R}}\mathbf{r} \simeq \nabla_{\mathbf{R}_g}\mathbf{r}$ I have used the fact that \mathbf{R} can be replaced by \mathbf{R}_g to lowest order. The only coefficient left to evaluate is $\mathbf{\dot{R}} \cdot \nabla_{\mathbf{R}_g}\mathbf{r}$. Since $\mathbf{r} = \mathbf{R}_g - \Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}}$, and $\mathbf{v} \times \hat{\mathbf{b}} = \sqrt{2\mu_0 B(\mathbf{r})}[\hat{\mathbf{e}}_1(\mathbf{r})\sin\varphi_0 - \hat{\mathbf{e}}_2(\mathbf{r})\cos\varphi_0]$, I find that $\nabla_{\mathbf{R}_g}\mathbf{r} \simeq \mathbf{I} - \overline{\nabla}(\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}})$, where the gradient $\overline{\nabla}(\Omega_i^{-1}\mathbf{v} \times \hat{\mathbf{b}})$ is evaluated holding μ_0 and φ_0 fixed, and it is given by

$$-\overline{\nabla}\left(\frac{1}{\Omega_{i}}\mathbf{v}\times\hat{\mathbf{b}}\right) = \frac{1}{\Omega_{i}}\left[\frac{1}{2B}(\nabla B)(\mathbf{v}\times\hat{\mathbf{b}}) + \nabla\hat{\mathbf{b}}\cdot(\mathbf{v}\times\hat{\mathbf{b}})\hat{\mathbf{b}} + (\nabla\hat{\mathbf{e}}_{2}\cdot\hat{\mathbf{e}}_{1})\mathbf{v}_{\perp}\right].$$
 (F.5)

In $\hat{\mathbf{R}}$, given by (3.19), the terms \mathbf{v}_M and \mathbf{v}_E are an order smaller than $u\hat{\mathbf{b}}(\mathbf{R})$ so I can use $\nabla_{\mathbf{R}_g}\mathbf{r} \simeq \hat{\mathbf{I}}$ for $\mathbf{v}_M \cdot \nabla_{\mathbf{R}_g}\mathbf{r}$ and $\mathbf{v}_E \cdot \nabla_{\mathbf{R}_g}\mathbf{r}$ to find the result in (4.15). In equation (4.15), I have also used $u\hat{\mathbf{b}}(\mathbf{R}) \simeq u\hat{\mathbf{b}}(\mathbf{r}) + (v_{\parallel}/\Omega_i)(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}}$, $\mathbf{v}_M \simeq \mathbf{v}_{M0}$, $\mathbf{v}_E \simeq \mathbf{v}_{E0}$ and

$$\overline{\nabla} \times \mathbf{v}_{\perp} = (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \hat{\mathbf{b}} + \frac{1}{2B} \mathbf{v} \times \hat{\mathbf{b}} (\hat{\mathbf{b}} \cdot \nabla B) + \mathbf{v}_{\perp} (\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1) + \hat{\mathbf{b}} \left[\frac{1}{2B} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla B - \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1 \right], \quad (F.6)$$

where I employ $\overline{\nabla} \times \mathbf{v}_{\perp} = \overline{\nabla} \times [\hat{\mathbf{b}} \times (\mathbf{v} \times \hat{\mathbf{b}})] = \overline{\nabla} \cdot [(\mathbf{v} \times \hat{\mathbf{b}})\hat{\mathbf{b}}] - \overline{\nabla} \cdot [\hat{\mathbf{b}}(\mathbf{v} \times \hat{\mathbf{b}})]$. To find the result in equation (4.16), notice that $v_{||0} = u + (v_{||}/\Omega_i)\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\mathbf{v} \times \hat{\mathbf{b}}) - (v_{||}/\Omega_i)\hat{\mathbf{b}} \cdot \overline{\nabla} \times \mathbf{v}_{\perp}$, and the difference $u - v_{||}$ is given in (A.43).

F.2 Conservative form in r, E_0 , μ_0 and φ_0 variables

To obtain equation (4.20) from (4.14), I just need to prove that

$$\overline{\nabla} \cdot \left(\frac{B}{v_{||}} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r}\right) - \frac{\partial}{\partial \mu_{0}} \left(\mathbf{B} \cdot \overline{\nabla} \mu_{10}\right) - \frac{\partial}{\partial \varphi_{0}} \left(\mathbf{B} \cdot \overline{\nabla} \varphi_{10}\right) \\ - \frac{\partial}{\partial E_{0}} \left(\frac{B}{v_{||}} \frac{Ze}{M} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_{g}} \mathbf{r} \cdot \overline{\nabla} \langle \phi \rangle\right) = 0.$$
(F.7)

Then, equation (4.20) is found by using $\partial f_{ig}/\partial E_0 \simeq \partial f_{Mi}/\partial E_0$, $\partial f_{ig}/\partial \mu_0 \simeq 0$ and $\partial f_{ig}/\partial \varphi_0 \simeq 0$.

To prove (F.7), I use the value of $\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r}$ from (4.15) and the relations $\partial (\mathbf{B} \cdot \overline{\nabla} \mu_{10}) / \partial \mu_0 = \overline{\nabla} \cdot [\mathbf{B}(\partial \mu_{10} / \partial \mu_0)]$ and $\partial (\mathbf{B} \cdot \overline{\nabla} \varphi_{10}) / \partial \varphi_0 = \overline{\nabla} \cdot [\mathbf{B}(\partial \varphi_{10} / \partial \varphi_0)]$, to find

$$\overline{\nabla} \cdot \left\{ \frac{B}{v_{||}} [(v_{||0} - v_{||})\hat{\mathbf{b}} + \mathbf{v}_{M0} + \mathbf{v}_{E0}] - \mathbf{B} \left(\frac{\partial \mu_{10}}{\partial \mu_0} + \frac{\partial \varphi_{10}}{\partial \varphi_0} \right) \right\} - \frac{\partial}{\partial E_0} \left(\frac{B}{v_{||}} \frac{Ze}{M} \mathbf{v}_{M0} \right) \cdot \overline{\nabla} \langle \phi \rangle = 0.$$
(F.8)

Here, I have also employed $\nabla \cdot \mathbf{B} = 0$, $\partial (\mathbf{B} \cdot \overline{\nabla} \langle \phi \rangle) / \partial E = 0$, $\overline{\nabla} \cdot [(B/v_{||})\tilde{\mathbf{v}}_1] = 0$ and $\partial [(B/v_{||})\tilde{\mathbf{v}}_1] / \partial E_0 = 0$ [these last two expressions are easy to prove by using the definition of $\tilde{\mathbf{v}}_1$ from (4.19)]. In the term $\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \cdot \overline{\nabla} \langle \phi \rangle$, $(v_{||0} - v_{||})\hat{\mathbf{b}} \cdot \overline{\nabla} \langle \phi \rangle$ is negligible because $\hat{\mathbf{b}} \cdot \overline{\nabla} \langle \phi \rangle \simeq 0$ (the zeroth order potential is constant along magnetic field lines). In equation (F.8), it is satisfied that

$$\overline{\nabla} \cdot \left(\frac{B}{v_{||}} \mathbf{v}_{E0}\right) - \frac{\partial}{\partial E_0} \left(\frac{B}{v_{||}} \frac{Ze}{M} \mathbf{v}_{M0}\right) \cdot \overline{\nabla} \langle \phi \rangle = 0, \tag{F.9}$$

where I employ relation (2.8) and $\hat{\mathbf{b}} \cdot \overline{\nabla} \langle \phi \rangle \simeq 0$; and

$$\frac{v_{||0} - v_{||}}{v_{||}} - \frac{\partial \mu_{10}}{\partial \mu_{0}} - \frac{\partial \varphi_{10}}{\partial \varphi_{0}} = \frac{v_{||}}{\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}, \qquad (F.10)$$

with $\partial \mathbf{v}_{\perp}/\partial \mu_0 = (B/v_{\perp}^2)\mathbf{v}_{\perp}$, $\partial \mathbf{v}_{\perp}/\partial \varphi_0 = -\mathbf{v} \times \hat{\mathbf{b}}$ and $v_{\parallel 0}$ defined by (4.16). Using these relations, equation (F.8) becomes

$$\frac{Mc}{Ze}\overline{\nabla}\cdot\left[\frac{\mu_{0}}{v_{||}}\hat{\mathbf{b}}\times\nabla B+v_{||}\nabla\times\hat{\mathbf{b}}\right]=\frac{Mc}{Ze}\overline{\nabla}\cdot\left[\overline{\nabla}\times\left(v_{||}\hat{\mathbf{b}}\right)\right]=0,\tag{F.11}$$

where the relation (2.8) is used again.

Appendix G

Details of the particle and momentum transport calculation

This Appendix contains the details of the derivations of the gyrokinetic particle conservation equation (4.28) and the momentum conservation equation (4.38) from equation (4.24).

G.1 Details of the particle transport calculation

Equation (4.24) with G = 1 leads to equation (4.28), with the integral of the gyroaveraged collision operator giving the term $\nabla \cdot (n_i \mathbf{V}_{iC})$ as shown in Appendix H. The only other integral of some difficulty is $\int d^3 v f_{ig} \dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r}$. The integral of $v_{||} \hat{\mathbf{b}}$ is done realizing that $f_i - f_{ig} = (-Ze\tilde{\phi}/T_i)f_{Mi}$ is even in $v_{||}$ to write (4.29). The integral of $(v_{||0} - v_{||})\hat{\mathbf{b}}$ is done by using $\nabla \cdot [\int d^3 v f_{ig}(v_{||0} - v_{||})\hat{\mathbf{b}}] \simeq \mathbf{B} \cdot \nabla [\int d^3 v f_{ig}(v_{||0} - v_{||})/B]$. Then f_{ig} can be replaced by f_{Mi} because the gradient is along the magnetic field line, and the slow parallel gradients make the small pieces of the distribution function unimportant. From all the terms in $v_{||0} - v_{||}$, only the gyrophase independent piece $(v_{\perp}^2/2\Omega_i)\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}$ gives a non-vanishing contribution.

To perform the integral $\nabla \cdot (n_i \tilde{\mathbf{V}}_i) = \nabla \cdot (\int d^3 v f_{ig} \tilde{\mathbf{v}}_1) = \nabla \cdot [\int d^3 v f_{ig}(v_{||}/\Omega_i) \overline{\nabla} \times \mathbf{v}_{\perp}]$, I use the expression of $\overline{\nabla} \times \mathbf{v}_{\perp}$ in equation (F.6). The contribution of the parallel component of $\overline{\nabla} \times \mathbf{v}_{\perp}$ is negligible because its divergence only has parallel

gradients, and they are small compared to the perpendicular gradients. The integral of $(v_{\parallel}/\Omega_i)\mathbf{v}_{\perp}(\hat{\mathbf{b}}\cdot\nabla\hat{\mathbf{e}}_2\cdot\hat{\mathbf{e}}_1)$, on the other hand, vanishes because its divergence becomes

$$\int d^3 v \, \frac{v_{||}}{\Omega_i} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1 \mathbf{v}_\perp \cdot \overline{\nabla} f_{ig} = \int d^3 v \, v_{||} \hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1 \frac{\partial f_{ig}}{\partial \varphi_0} = 0. \tag{G.1}$$

Here, I neglect the gradients of any quantity that is not f_{ig} because they give contributions of order $\delta_i^2 k_\perp \rho_i n_e v_i/L$. To obtain $\partial f_{ig}/\partial \varphi_0 \simeq \Omega_i^{-1} \mathbf{v}_\perp \cdot \overline{\nabla} f_{ig}$, I use that f_{ig} 's only dependence on φ_0 is through \mathbf{R}_g . The final result for $n_i \tilde{\mathbf{V}}_i$ is written in (4.35).

G.2 Details of the momentum transport calculation

Equation (4.24) with $G = \mathbf{v}$ gives equation (4.38). The integral of the gyroaveraged collision operator gives \mathbf{F}_{iC} . The details are given in Appendix H.

To simplify the integral $\int d^3 v f_{ig}(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r}) M \mathbf{v}$, I use that $\nabla \cdot [\int d^3 v f_{ig}(v_{||0} - v_{||}) \hat{\mathbf{b}} \mathbf{v}] \simeq \mathbf{B} \cdot \nabla [\int d^3 v f_{ig} \mathbf{v}(v_{||0} - v_{||}) / B]$. Then f_{ig} can be replaced by f_{Mi} because the gradient is along the magnetic field line, and the next order corrections can be neglected. The integral $\int d^3 v f_{Mi} \mathbf{v}(v_{||0} - v_{||}) / B$ vanishes because all the terms are either odd in $v_{||}$ or odd in \mathbf{v}_{\perp} . The final result is

$$\int d^3 v f_{ig} \left(\dot{\mathbf{R}} \cdot \nabla_{\mathbf{R}_g} \mathbf{r} \right) M \mathbf{v} = p_{ig||} \hat{\mathbf{b}} \hat{\mathbf{b}} + \boldsymbol{\pi}_{ig||} \hat{\mathbf{b}} + \boldsymbol{\ddot{\pi}}_{ig\times}, \qquad (G.2)$$

where I use the definitions of $p_{ig||} = \int d^3v f_{ig} M v_{||}^2$, $\pi_{ig||}$ from (4.39), and $\dot{\pi}_{ig\times}$ from (4.40).

To find the integral $\int d^3 v \, M f_{ig} K\{\mathbf{v}\}$, with the linear operator K given in (4.25), I use $K\{\mathbf{v}\} = K\{v_{||}\hat{\mathbf{b}}\} + K\{\mathbf{v}_{\perp}\}$. For $K\{v_{||}\hat{\mathbf{b}}\}$, I need $\overline{\nabla}v_{||} = -\mu_0 \nabla B/v_{||}, \partial v_{||}/\partial E_0 = v_{||}^{-1}, \partial v_{||}/\partial \mu_0 = -B/v_{||}$ and $\partial v_{||}/\partial \varphi_0 = 0$. Then, using the definitions of $p_{ig||}$ and $\pi_{ig||}$ along with

$$\left(\mathbf{v}_{M0} + \mathbf{v}_{E0}\right) \cdot \left(\frac{\mu_0 \nabla B}{v_{||}} + \frac{Ze}{Mv_{||}} \overline{\nabla}\langle\phi\rangle\right) = \frac{v_{||}}{\Omega_i} (\hat{\mathbf{b}} \times \boldsymbol{\kappa}) \cdot \left(\mu_0 \nabla B + \frac{Ze}{M} \overline{\nabla}\langle\phi\rangle\right), \quad (G.3)$$

I find

$$\int d^{3}v \, M f_{ig} K\{v_{||}\hat{\mathbf{b}}\} = (p_{ig||}\hat{\mathbf{b}} + \pi_{ig||}) \cdot \nabla \hat{\mathbf{b}} - \int d^{3}v \, f_{ig} M \mu_{0} \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla B$$
$$+ \int d^{3}v \, f_{Mi} \frac{M v_{\perp}^{2}}{2\Omega_{i}} \hat{\mathbf{b}} (\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \overline{\nabla} v_{||} + \int d^{3}v \, f_{Mi} M \hat{\mathbf{b}} (\mathbf{B} \cdot \overline{\nabla} \mu_{10})$$
$$- Ze \int d^{3}v f_{ig} \hat{\mathbf{b}} (\hat{\mathbf{b}} + \Omega_{i}^{-1} \overline{\nabla} \times \mathbf{v}_{\perp}) \cdot \overline{\nabla} \langle \phi \rangle.$$
(G.4)

Here, I have used that $(v_{||0} - v_{||})\hat{\mathbf{b}} \cdot \nabla \langle \phi \rangle \simeq 0$ and that, in the integrals that include $(v_{||0} - v_{||})\hat{\mathbf{b}} \cdot \overline{\nabla} (v_{||}\hat{\mathbf{b}})$ and $\tilde{\mathbf{v}}_1 \cdot \overline{\nabla} v_{||}$, only the term $(v_{\perp}^2/2\Omega_i)\hat{\mathbf{b}}(\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})\hat{\mathbf{b}} \cdot \overline{\nabla} v_{||}$ gives a non-vanishing contribution. The integral of $f_{Mi}(v_{||}/\Omega_i)(\hat{\mathbf{b}} \times \kappa) \cdot [\mu_0 \nabla B + (Ze/M)\overline{\nabla} \langle \phi \rangle]$ vanishes because it is odd in $v_{||}$. Equation (G.4) can be further simplified by using

$$\int d^{3}v f_{Mi} \mathbf{B} \cdot \overline{\nabla} \mu_{10} = -\int d^{3}v f_{Mi} \mathbf{B} \cdot \overline{\nabla} \left(\frac{v_{||} v_{\perp}^{2}}{2B\Omega_{i}} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \right) = -\int d^{3}v f_{Mi} \frac{v_{\perp}^{2}}{2\Omega_{i}} (\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}) \hat{\mathbf{b}} \cdot \overline{\nabla} v_{||}, \qquad (G.5)$$

where I have used that in μ_{10} all the terms but the gyrophase independent piece give vanishing integrals. The last form of (G.5) cancels with a term in (G.4). Using $p_{ig\perp} = \int d^3v f_{ig} M v_{\perp}^2/2$, equation (G.4) then becomes

$$\int d^{3}v M f_{ig} K\{v_{||}\hat{\mathbf{b}}\} = -p_{ig\perp}\hat{\mathbf{b}}\hat{\mathbf{b}} \cdot \nabla \ln B + (p_{ig||}\hat{\mathbf{b}} + \boldsymbol{\pi}_{ig||}) \cdot \nabla \hat{\mathbf{b}}$$
$$-Ze \int d^{3}v f_{ig}\hat{\mathbf{b}}(\hat{\mathbf{b}} + \Omega_{i}^{-1}\overline{\nabla} \times \mathbf{v}_{\perp}) \cdot \overline{\nabla}\langle\phi\rangle. \tag{G.6}$$

The integral $\int d^3 v f_{ig} K\{\mathbf{v}_{\perp}\}$ is obtained using $\overline{\nabla} \mathbf{v}_{\perp} = (\nabla B/2B)\mathbf{v}_{\perp} - \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} \hat{\mathbf{b}} + \nabla \hat{\mathbf{e}}_2 \cdot \hat{\mathbf{e}}_1(\mathbf{v} \times \hat{\mathbf{b}}), \ \partial \mathbf{v}_{\perp} / \partial E_0 = 0, \ \partial \mathbf{v}_{\perp} / \partial \mu_0 = (2\mu_0)^{-1} \mathbf{v}_{\perp} \text{ and } \partial \mathbf{v}_{\perp} / \partial \varphi_0 = -\mathbf{v} \times \hat{\mathbf{b}}.$ Then, I find

$$\int d^3 v \, M f_{ig} K\{\mathbf{v}_{\perp}\} = \int d^3 v \, M f_{ig} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} - \frac{Mc}{B} \int d^3 v \, f_{Mi}(\overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}, \ (G.7)$$

where I employ that the integrals of $f_{Mi}[(v_{||0} - v_{||})\hat{\mathbf{b}} + \mathbf{v}_{M0} + \tilde{\mathbf{v}}_1] \cdot \overline{\nabla} \mathbf{v}_{\perp}, f_{Mi}v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla} \mu_{10}(\partial \mathbf{v}_{\perp}/\partial \mu_0)$ and $f_{Mi}v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla} \varphi_{10}(\partial \mathbf{v}_{\perp}/\partial \varphi_0)$ vanish because the terms are either odd in $v_{||}$ or in \mathbf{v}_{\perp} . The integral that includes $\overline{\nabla} \langle \phi \rangle$ can be rewritten by realizing

that $\overline{\nabla}\langle\phi\rangle = \nabla\phi - \overline{\nabla}\tilde{\phi}$, to find

$$\int d^3 v \, M f_{ig} K\{\mathbf{v}_{\perp}\} = \int d^3 v \, M f_{ig} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} + \frac{Mc}{B} \int d^3 v \, f_{Mi}(\overline{\nabla} \widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{v}_{\perp}. \quad (G.8)$$

Combining equations (G.6) and (G.8), I find

$$\int d^3 v \, M f_{ig} K\{\mathbf{v}\} = p_{ig\perp} \hat{\mathbf{b}} (\nabla \cdot \hat{\mathbf{b}}) + (p_{ig\parallel} \hat{\mathbf{b}} + \boldsymbol{\pi}_{ig\parallel}) \cdot \nabla \hat{\mathbf{b}} - Zen_i \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \phi + \tilde{F}_{iE} \hat{\mathbf{b}} + \mathbf{F}_{iB},$$
(G.9)

where I have used $\hat{\mathbf{b}} \cdot \nabla \ln B = -\nabla \cdot \hat{\mathbf{b}}$, and the definitions of \tilde{F}_{iE} from (4.41) and \mathbf{F}_{iB} from (4.42).

Using (G.2) and (G.9) in (4.24) gives

$$\frac{\partial}{\partial t}(n_i M \mathbf{V}_{ig}) + \nabla \cdot (p_{ig||} \hat{\mathbf{b}} \hat{\mathbf{b}} + \boldsymbol{\pi}_{ig||} \hat{\mathbf{b}} + \boldsymbol{\ddot{\pi}}_{ig\times}) = -Zen_i \hat{\mathbf{b}} \hat{\mathbf{b}} \cdot \nabla \phi$$
$$+ p_{ig\perp} \hat{\mathbf{b}} (\nabla \cdot \hat{\mathbf{b}}) + (p_{ig||} \hat{\mathbf{b}} + \boldsymbol{\pi}_{ig||}) \cdot \nabla \hat{\mathbf{b}} + \tilde{F}_{iE} \hat{\mathbf{b}} + \mathbf{F}_{iB} + \mathbf{F}_{iC}.$$
(G.10)

Finally, employing $\nabla \cdot (p_{ig||}\hat{\mathbf{b}}\hat{\mathbf{b}}) = \hat{\mathbf{b}}(\hat{\mathbf{b}} \cdot \nabla p_{ig||} + p_{ig||}\nabla \cdot \hat{\mathbf{b}}) + p_{ig||}\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}$ and $\nabla \cdot (\boldsymbol{\pi}_{ig||}\hat{\mathbf{b}}) = (\nabla \cdot \boldsymbol{\pi}_{ig||})\hat{\mathbf{b}} + \boldsymbol{\pi}_{ig||} \cdot \nabla \hat{\mathbf{b}}$, I am able to recover equation (4.38). Multiplying equation (4.38) by $\hat{\mathbf{b}}$ and taking into account the cancellation of

$$\mathbf{F}_{iB} \cdot \hat{\mathbf{b}} = -M \int d^3 v \, f_{ig} \left(v_{||} \hat{\mathbf{b}} + \frac{c}{B} \overline{\nabla} \tilde{\phi} \times \hat{\mathbf{b}} \right) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp} \tag{G.11}$$

and

$$(\nabla \cdot \, \vec{\boldsymbol{\pi}}_{ig\times}) \cdot \, \hat{\mathbf{b}} \simeq -M \int d^3 v \, f_{ig} \left(v_{||} \hat{\mathbf{b}} + \frac{c}{B} \overline{\nabla} \widetilde{\phi} \times \, \hat{\mathbf{b}} \right) \cdot \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp}, \tag{G.12}$$

I find equation (4.44). To obtain relation (G.12), I have employed $\overleftarrow{\pi}_{ig\times} \cdot \widehat{\mathbf{b}} = 0$ and I have used the lowest order distribution function f_{Mi} for the higher order terms. All the higher order terms, except for $(c/B)\overline{\nabla}\langle\phi\rangle \times \widehat{\mathbf{b}}$, cancel because they are either odd in v_{\parallel} or \mathbf{v}_{\perp} .
Appendix H

Finite gyroradius effects in the like-collision operator

In this Appendix, I show how to treat the gyroaveraged like-collision operator, $\langle C\{f_i\}\rangle$. The like-collision operator is

$$C\{f_i\} = \gamma \nabla_v \cdot \left[\int d^3 v' \, \nabla_g \nabla_g g \cdot (f'_i \nabla_v f_i - f_i \nabla_{v'} f'_i) \right], \tag{H.1}$$

with $f_i = f_i(\mathbf{v})$, $f'_i = f_i(\mathbf{v}')$, $\mathbf{g} = \mathbf{v} - \mathbf{v}'$, $g = |\mathbf{g}|$, $\nabla_g \nabla_g g = (g^2 \quad \mathbf{I} - \mathbf{gg})/g^3$ and $\gamma = 2\pi Z^4 e^4 \ln \Lambda/M^2$. Linearizing this equation for $f_i = f_{Mi} + f_{i1}$, with $f_{i1} \ll f_{Mi}$, I find

$$C^{(\ell)}\{f_{i1}\} = \gamma \nabla_v \cdot \mathbf{\Gamma}\{f_{i1}\},\tag{H.2}$$

with

$$\mathbf{\Gamma}\{f_{i1}\} = \int d^3v' f_{Mi} f'_{Mi} \nabla_g \nabla_g g \cdot \left[\nabla_v \left(\frac{f_{i1}}{f_{Mi}}\right) - \nabla_{v'} \left(\frac{f'_{i1}}{f'_{Mi}}\right)\right].$$
(H.3)

The vector Γ can also be written as in (4.34) because $\Gamma\{f_i\} = \Gamma\{f_{Mi}\} + \Gamma\{f_{i1}\} = \Gamma\{f_{i1}\}$. Using gyrokinetic variables in equation (H.2) and gyroaveraging gives

$$\langle C^{(\ell)}\{f_{i1}\}\rangle = \gamma \frac{u}{B} \bigg[\frac{\partial}{\partial E} \left(\frac{B}{u} \langle \mathbf{\Gamma} \cdot \nabla_{v} E \rangle \right) + \frac{\partial}{\partial \mu} \left(\frac{B}{u} \langle \mathbf{\Gamma} \cdot \nabla_{v} \mu \rangle \right) + \nabla_{\mathbf{R}} \cdot \left(\frac{B}{u} \langle \mathbf{\Gamma} \cdot \nabla_{v} \mathbf{R} \rangle \right) \bigg]. \tag{H.4}$$

Here, $B/u \simeq \partial(\mathbf{r}, \mathbf{v})/\partial(\mathbf{R}, E, \mu, \varphi)$ is the approximate Jacobian, and I have used the

transformation rule for divergences from one reference system $\{x_i\}$ to another $\{y_j\}$:

$$\nabla_x \cdot \mathbf{\Gamma} = \sum_j \frac{1}{J_y} \frac{\partial}{\partial y_j} (J_y \mathbf{\Gamma} \cdot \nabla_x y_j) = \sum_j \frac{1}{J_y} \frac{\partial}{\partial y_j} (J_y \mathbf{\Gamma}_{y_j}), \tag{H.5}$$

where $J_y = \partial(x_i)/\partial(y_j)$ is the Jacobian of the transformation, $\Gamma_{y_j} = \mathbf{\Gamma} \cdot \nabla_x y_j$ and ∇_x is the gradient in the reference system $\{x_i\}$. To rewrite equation (H.4) in terms of the variables \mathbf{r} , E_0 , μ_0 and φ_0 , I need to use (H.5) and the chain rule to find the transformation between the two reference systems $\{y_j\}$ and $\{z_k\}$

$$\frac{1}{J_y} \sum_j \frac{\partial}{\partial y_j} (J_y \Gamma_{y_j}) = \frac{1}{J_z} \sum_k \frac{\partial}{\partial z_k} \left(J_z \sum_j \Gamma_{y_j} \frac{\partial z_k}{\partial y_j} \right).$$
(H.6)

Employing this relation to write equation (H.4) as a function of \mathbf{r} , E_0 , μ_0 and φ_0 gives

$$\langle C^{(\ell)}\{f_{i1}\}\rangle \simeq \gamma \frac{v_{||}}{B} \bigg\{ \frac{\partial}{\partial E_0} \left(\frac{B}{v_{||}} \langle \mathbf{\Gamma} \cdot \nabla_v E \rangle \right) + \frac{\partial}{\partial \mu_0} \left(\frac{B}{v_{||}} \langle \mathbf{\Gamma} \cdot \nabla_v \mu \rangle \right)$$
$$+ \overline{\nabla} \cdot \bigg[\frac{B}{v_{||}} \left(\langle \mathbf{\Gamma} \cdot \nabla_v \mu \rangle \frac{\partial \mathbf{r}}{\partial \mu} + \langle \mathbf{\Gamma} \cdot \nabla_v \mathbf{R} \rangle \bigg) \bigg] \bigg\},$$
(H.7)

where I have used the lowest order gyrokinetic variables \mathbf{R}_g , E_0 , μ_0 and φ_0 . This approximation is justified because the collision operator vanishes to lowest order, and only the zeroth order definitions must be kept. Notice that I keep the first order correction $\mathbf{R}_1 = \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$ only within the spatial divergence because the spatial gradients are steep. Employing $\nabla_v E \simeq \mathbf{v}$, $\nabla_v \mu \simeq \mathbf{v}_\perp / B$, $\partial \mathbf{r} / \partial \mu \simeq -\partial \mathbf{R}_1 / \partial \mu \simeq$ $-(2\mu_0\Omega_i)^{-1}\mathbf{v} \times \hat{\mathbf{b}}$ and $\nabla_v \mathbf{R} \simeq \Omega_i^{-1} \mathbf{i} \times \hat{\mathbf{b}}$, I find

$$\langle C^{(\ell)}\{f_{i1}\}\rangle \simeq \gamma \frac{v_{||}}{B} \bigg\{ \frac{\partial}{\partial E_0} \left(\frac{B}{v_{||}} \langle \mathbf{\Gamma} \cdot \mathbf{v} \rangle \right) + \frac{\partial}{\partial \mu_0} \left(\frac{1}{v_{||}} \langle \mathbf{\Gamma} \cdot \mathbf{v}_\perp \rangle \right)$$

$$+ \overline{\nabla} \cdot \bigg[\frac{Mc}{Zev_{||}} \left(\langle \mathbf{\Gamma} \rangle \times \hat{\mathbf{b}} - \frac{1}{v_\perp^2} \langle \mathbf{\Gamma} \cdot \mathbf{v}_\perp \rangle \mathbf{v} \times \hat{\mathbf{b}} \bigg) \bigg] \bigg\}.$$
 (H.8)

In the main text, there are two integrals that involve the gyroaveraged collision operator, $\nabla \cdot (n_i \mathbf{V}_{iC}) = -\int d^3 v \langle C\{f_i\} \rangle$ and $\mathbf{F}_{iC} = M \int d^3 v \mathbf{v} \langle C\{f_i\} \rangle$. Using equation (H.8), I obtain equations (4.33) and (4.43). To find (4.43), I have integrated by parts using $\partial \mathbf{v} / \partial E_0 = v_{||}^{-1} \hat{\mathbf{b}}$ and $\partial \mathbf{v} / \partial \mu_0 = (B/v_{\perp}^2) \mathbf{v}_{\perp} - (B/v_{||}) \hat{\mathbf{b}}$, and $\overline{\nabla} \mathbf{v}$ has been neglected because the gradient is of order 1/L.

I can prove that the divergence of $n_i \mathbf{V}_{iC}$, given in (4.33), is of order $\delta_i (k_{\perp} \rho_i)^2 \nu_{ii} n_e$ rather than $\delta_i k_{\perp} \rho_i \nu_{ii} n_e$. For $k_{\perp} \rho_i \ll 1$, the function $\mathbf{\Gamma}(\mathbf{r}, E_0, \mu_0, \varphi_0)$ can be Taylor expanded around \mathbf{R}_g to find $\mathbf{\Gamma}(\mathbf{r}, E_0, \mu_0, \varphi_0) \simeq \mathbf{\Gamma}(\mathbf{R}_g, E_0, \mu_0, \varphi_0) - \Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_g} \mathbf{\Gamma}$. Then, the gyroaverage $\langle \ldots \rangle$ holding \mathbf{R}, E, μ and t fixed gives

$$\langle \mathbf{\Gamma} \rangle = \frac{1}{2\pi} \oint d\varphi_0 \left[\mathbf{\Gamma}(\mathbf{R}_g, E_0, \mu_0, \varphi_0) - \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_g} \mathbf{\Gamma} \right], \tag{H.9}$$

where I have employed that to the order of interest holding \mathbf{R}_g , E_0 and μ_0 fixed is approximately equal to holding \mathbf{R} , E and μ fixed. To rewrite equation (H.9) as a function of \mathbf{r} , E_0 , μ_0 and φ_0 , I Taylor expand $\Gamma(\mathbf{R}_q, E_0, \mu_0, \varphi_0, t)$ around \mathbf{r} to find

$$\frac{1}{2\pi} \oint d\varphi_0 \, \boldsymbol{\Gamma}(\mathbf{R}_g, E_0, \mu_0, \varphi_0) \simeq \overline{\boldsymbol{\Gamma}} + \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \, \overline{\boldsymbol{\Gamma}}, \tag{H.10}$$

with $\overline{\Gamma} \equiv \overline{\Gamma}(\mathbf{r}, E_0, \mu_0) = \overline{\Gamma(\mathbf{r}, E_0, \mu_0, \varphi_0)}$ the gyroaverage holding \mathbf{r}, E_0, μ_0 and t fixed. The second term in the right side of (H.9) is higher order and can be simply written as

$$-\frac{1}{2\pi}\oint d\varphi_0 \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla_{\mathbf{R}_g} \Gamma \simeq -\frac{1}{\Omega_i} \overline{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \Gamma}.$$
 (H.11)

Employing equations (H.10) and (H.11) in equation (H.9), I find

$$\langle \mathbf{\Gamma} \rangle = \overline{\mathbf{\Gamma}} + \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \,\overline{\mathbf{\Gamma}} - \frac{1}{\Omega_i} \overline{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{\Gamma}}.$$
 (H.12)

Similarly, letting $\Gamma \to \Gamma \cdot \mathbf{v}_{\perp}$ in (H.12) and ignoring $\overline{\nabla} \mathbf{v}_{\perp}$ corrections as small, I find

$$\langle \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \rangle = \overline{\mathbf{\Gamma} \cdot \mathbf{v}_{\perp}} + \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \left(\overline{\mathbf{\Gamma} \cdot \mathbf{v}_{\perp}} \right) - \frac{1}{\Omega_i} \overline{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{\Gamma} \cdot \mathbf{v}_{\perp}}.$$
 (H.13)

Using these results in (4.33) the integral becomes

$$\nabla \cdot (n_i \mathbf{V}_{iC}) = -\nabla \cdot \left\{ \frac{\gamma}{\Omega_i} \int d^3 v \left[\mathbf{\Gamma} \times \hat{\mathbf{b}} - \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \mathbf{\Gamma} \times \hat{\mathbf{b}} - \frac{1}{2\Omega_i} \overline{\nabla} \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \right] \right\},$$
(H.14)

where I have used that $\int d^3v \, \overline{(\ldots)} = \int d^3v \, (\ldots)$. The integral $\int d^3v \, \Gamma$ is zero, as can

be proven by exchanging the dummy integration variables \mathbf{v} and \mathbf{v}' . The rest of the integral can be written as

$$\nabla \cdot (n_i \mathbf{V}_{iC}) = \nabla \nabla : \left\{ \frac{\gamma}{\Omega_i^2} \int d^3 v \left[(\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{\Gamma} \times \hat{\mathbf{b}}) + \frac{1}{2} (\vec{\mathbf{I}} - \hat{\mathbf{b}} \hat{\mathbf{b}}) (\mathbf{\Gamma} \cdot \mathbf{v}_\perp) \right] \right\}, \quad (\mathrm{H.15})$$

where the spatial gradients of functions different from Γ have been neglected. Using the definition of the linearized collision operator (H.2) and employing $\int d^3 v \left[(\mathbf{v} \times \hat{\mathbf{b}}) + (v_{\perp}^2/2)(\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) \right] \nabla_v \cdot \Gamma = -\int d^3 v \left[(\Gamma \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) + (\mathbf{v} \times \hat{\mathbf{b}}) (\Gamma \times \hat{\mathbf{b}}) + (\Gamma \cdot \mathbf{v}_{\perp}) (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) \right]$, I find

$$\nabla \cdot (n_i \mathbf{V}_{iC}) = -\nabla \nabla : \left\{ \frac{1}{\Omega_i^2} \int d^3 v \, C^{(\ell)} \{f_{i1}\} \left[\frac{v_\perp^2}{4} (\vec{\mathbf{I}} - \hat{\mathbf{b}}\hat{\mathbf{b}}) + \frac{1}{2} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) \right] \right\}, \tag{H.16}$$

where I have also employed $\overline{\nabla} \overline{\nabla} : [(\mathbf{v} \times \hat{\mathbf{b}})(\mathbf{\Gamma} \times \hat{\mathbf{b}})] = \overline{\nabla} \overline{\nabla} : [(\mathbf{\Gamma} \times \hat{\mathbf{b}})(\mathbf{v} \times \hat{\mathbf{b}})]$. This integral, of order $\delta_i (k_{\perp} \rho_i)^2 \nu_{ii} n_e$, can be simplified by employing that, for $k_{\perp} \rho_i \ll 1$, the gyrophase dependent part of the distribution function is proportional to \mathbf{v}_{\perp} to zeroth order [see (4.7)]. Then, the integral becomes

$$\nabla \cdot (n_i \mathbf{V}_{iC}) = -\nabla \nabla : \left[\frac{1}{\Omega_i^2} \int d^3 v \, C^{(\ell)} \{ f_{i1} \} \frac{v_\perp^2}{2} (\mathbf{\vec{I}} - \mathbf{\hat{b}}\mathbf{\hat{b}}) \right], \tag{H.17}$$

where I used that $\overline{\mathbf{v}_{\perp}(\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})} = 0$ and $\overline{(\mathbf{v}\times\hat{\mathbf{b}})(\mathbf{v}\times\hat{\mathbf{b}})} = (v_{\perp}^2/2)(\mathbf{\vec{I}}-\hat{\mathbf{b}}\hat{\mathbf{b}}).$

Finally, I prove that \mathbf{F}_{iC} from (4.43) is of order $\delta_i k_{\perp} \rho_i \nu_{ii} n_e M v_i$. Here, it is important to realize that the first integral in (4.43) must be carried to the next order in $k_{\perp}\rho_i$, but the integrals in the divergence only need the lowest order expressions. Then, using expressions (H.12) and (H.13) in the first integral of (4.43), and the lowest order expressions $\langle \mathbf{\Gamma} \rangle \simeq \overline{\mathbf{\Gamma}}$ and $\langle \mathbf{\Gamma} \cdot \mathbf{v}_{\perp} \rangle \simeq \overline{\mathbf{\Gamma} \cdot \mathbf{v}_{\perp}}$ for the second integral, I find that for $k_{\perp}\rho_i \ll 1$,

$$\mathbf{F}_{iC} \simeq \nabla \cdot \left\{ \frac{M\gamma}{\Omega_i} \int d^3 v \left[(\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{\Gamma} \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} + (\mathbf{\Gamma} \times \hat{\mathbf{b}}) v_{||} \hat{\mathbf{b}} + (\mathbf{\Gamma} \cdot \mathbf{v}_{\perp}) \stackrel{\mathbf{\dot{\mathbf{T}}}}{\mathbf{I}} \times \hat{\mathbf{b}} \right] \right\} = -\nabla \cdot \left\{ \frac{M}{\Omega_i} \int d^3 v \, C^{(\ell)} \{f_{i1}\} \left[(\mathbf{v} \times \hat{\mathbf{b}}) v_{||} \hat{\mathbf{b}} + \frac{v_{\perp}^2}{2} \stackrel{\mathbf{\dot{\mathbf{T}}}}{\mathbf{I}} \times \hat{\mathbf{b}} \right] \right\}. (H.18)$$

To obtain the last result, I have employed that $\int d^3 v \left[(\mathbf{v} \times \hat{\mathbf{b}}) v_{||} \hat{\mathbf{b}} + (v_{\perp}^2/2) \, \vec{\mathbf{I}} \times \hat{\mathbf{b}} \right] \nabla_v \cdot \mathbf{\Gamma} = -\int d^3 v \left[(\mathbf{\Gamma} \times \hat{\mathbf{b}}) v_{||} \hat{\mathbf{b}} + (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{\Gamma} \cdot \hat{\mathbf{b}}) \hat{\mathbf{b}} + (\mathbf{\Gamma} \cdot \mathbf{v}_{\perp}) \, \vec{\mathbf{I}} \times \hat{\mathbf{b}} \right]$ and I have used the definition of the linearized collision operator (H.2). Only the gyrophase dependent part of f_{i1} contributes to the first part of integral (H.18), and for $k_{\perp} \rho_i \ll 1$ the gyrophase dependent part is even in $v_{||}$ [recall (4.7)] so this portion vanishes. As a result, the integral becomes

$$\mathbf{F}_{iC} = -\nabla \cdot \left[\frac{M}{\Omega_i} \int d^3 v \, C^{(\ell)} \{ f_{i1} \} \frac{v_\perp^2}{2} \stackrel{\leftrightarrow}{\mathbf{I}} \times \hat{\mathbf{b}} \right], \tag{H.19}$$

of order $\delta_i k_{\perp} \rho_i \nu_{ii} n_e M v_i$.

Appendix I

Gyrokinetic vorticity

In this Appendix, I explain how to obtain the gyrokinetic vorticity equation (4.53) from equations (4.38) and (4.45).

Before adding equations (4.45) and $\nabla \cdot \{(c/B) [\text{equation } (4.38)] \times \hat{\mathbf{b}}\}$, I simplify the perpendicular component of the current density $Zen_i \tilde{\mathbf{V}}_i$. The perpendicular component of $\tilde{\mathbf{v}}_1$, defined in (4.19), is given by

$$\tilde{\mathbf{v}}_{1\perp} \equiv \hat{\mathbf{b}} \times (\tilde{\mathbf{v}}_1 \times \hat{\mathbf{b}}) = \frac{v_{||}}{\Omega_i} \hat{\mathbf{b}} \times (\hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} + \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp}), \tag{I.1}$$

where I use that $(\overline{\nabla} \times \mathbf{v}_{\perp}) \times \hat{\mathbf{b}} = \hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_{\perp} - \overline{\nabla} \mathbf{v}_{\perp} \cdot \hat{\mathbf{b}}$ and $\overline{\nabla} \mathbf{v}_{\perp} \cdot \hat{\mathbf{b}} = -\nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp}$. Employing $(\nabla \times \hat{\mathbf{b}}) \times \mathbf{v}_{\perp} = \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{b}} - \nabla \hat{\mathbf{b}} \cdot \mathbf{v}_{\perp}$ and $\hat{\mathbf{b}} \times [(\nabla \times \hat{\mathbf{b}}) \times \mathbf{v}_{\perp}] = -\mathbf{v}_{\perp} (\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}})$, I find

$$\tilde{\mathbf{v}}_{1\perp} = \frac{v_{||}}{\Omega_i} \hat{\mathbf{b}} \times (\hat{\mathbf{b}} \cdot \overline{\nabla} \mathbf{v}_\perp + \mathbf{v}_\perp \cdot \nabla \hat{\mathbf{b}}) + \frac{v_{||}}{\Omega_i} \mathbf{v}_\perp (\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}}).$$
(I.2)

Then, the integral $n_i \tilde{\mathbf{V}}_{i\perp}$ becomes

$$n_{i}\tilde{\mathbf{V}}_{i\perp} = \frac{1}{\Omega_{i}}\hat{\mathbf{b}} \times \left(\int d^{3}v \, f_{ig}v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla}\mathbf{v}_{\perp} + \int d^{3}v \, f_{ig}v_{||}\mathbf{v}_{\perp} \cdot \nabla\hat{\mathbf{b}}\right) \\ + \frac{1}{\Omega_{i}}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^{3}v \, f_{ig}v_{||}\mathbf{v}_{\perp}.$$
(I.3)

Adding equations (4.45) and $\nabla \cdot \{(c/B) | \text{equation (4.38)} | \times \hat{\mathbf{b}}\}$, I find

$$\frac{\partial \varpi_G}{\partial t} = \nabla \cdot \left[J_{||} \hat{\mathbf{b}} + \mathbf{J}_{gd} + \tilde{\mathbf{J}}_i + Zen_i \tilde{\mathbf{V}}_i + Zen_i \mathbf{V}_{iC} + \frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_{ig\times}) - \frac{c}{B} \hat{\mathbf{b}} \times \mathbf{F}_{iB} - \frac{c}{B} \hat{\mathbf{b}} \times \mathbf{F}_{iC} \right], \quad (I.4)$$

with ϖ_G defined in (4.51). In this equation, adding $-(c/B)\hat{\mathbf{b}} \times \mathbf{F}_{iB}$ and the expression in (I.3) for $Zen_i \tilde{\mathbf{V}}_{i\perp}$, two terms cancel to give

$$Zen_{i}\tilde{\mathbf{V}}_{i\perp} - \frac{c}{B}\hat{\mathbf{b}} \times \mathbf{F}_{iB} = \frac{Mc}{B}\hat{\mathbf{b}} \times \int d^{3}v \, f_{ig}v_{||}\mathbf{v}_{\perp} \cdot \nabla\hat{\mathbf{b}}$$
$$+ \frac{Mc}{B}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^{3}v \, f_{ig}v_{||}\mathbf{v}_{\perp} - \frac{Zec}{B\Omega_{i}}\hat{\mathbf{b}} \times \int d^{3}v \, f_{Mi}(\overline{\nabla}\widetilde{\phi} \times \hat{\mathbf{b}}) \cdot \overline{\nabla}\mathbf{v}_{\perp}. \tag{I.5}$$

The last term in this equation is absorbed in the definition of $\tilde{\mathbf{J}}_{i\phi}$ in (4.55). I can further simplify by realizing that $\hat{\mathbf{b}} \times \int d^3 v f_{ig} v_{||} \mathbf{v}_{\perp} \cdot \nabla \hat{\mathbf{b}} = \hat{\mathbf{b}} \times [\nabla \cdot (\int d^3 v f_{ig} \mathbf{v}_{\perp} v_{||} \hat{\mathbf{b}})],$ giving

$$\begin{split} \tilde{\mathbf{J}}_{i} + Zen_{i}\tilde{\mathbf{V}}_{i\perp} - \frac{c}{B}\hat{\mathbf{b}} \times \mathbf{F}_{iB} &= \tilde{\mathbf{J}}_{i\phi} + \frac{c}{B}\hat{\mathbf{b}} \times \left[\nabla \cdot \left(\int d^{3}v \, f_{ig}M\mathbf{v}_{\perp}v_{\parallel}\hat{\mathbf{b}}\right)\right] \\ &+ \frac{Mc}{B}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^{3}v \, f_{ig}v_{\parallel}\mathbf{v}_{\perp}. \end{split} \tag{I.6}$$

The integral $\int d^3v f_{ig} M \mathbf{v}_{\perp} v_{\parallel} \hat{\mathbf{b}}$ is part of the definition of $\overleftarrow{\pi}_{iG}$ in (4.54), so I can finally write

$$\tilde{\mathbf{J}}_{i} + Zen_{i}\tilde{\mathbf{V}}_{i\perp} + \frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_{ig\times}) - \frac{c}{B}\hat{\mathbf{b}} \times \mathbf{F}_{iB} = \tilde{\mathbf{J}}_{i\phi} + \frac{c}{B}\hat{\mathbf{b}} \times (\nabla \cdot \vec{\pi}_{iG}) + \frac{Mc}{B}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^{3}v \, f_{ig}v_{||}\mathbf{v}_{\perp}.$$
(I.7)

Employing this result in equation (I.4) and using the fact that the divergence of $Zen_i \tilde{V}_{i||}\hat{\mathbf{b}}$ and $(Mc/B)\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3v f_{ig}v_{||}\mathbf{v}_{\perp}$ is negligible, I recover equation (4.53). The divergence of $Zen_i \tilde{V}_{i||}\hat{\mathbf{b}} \sim \delta_i^2 k_{\perp} \rho_i en_e v_i$ is small because the parallel gradient is only order 1/L, giving $\nabla \cdot (Zen_i \tilde{V}_{i||}\hat{\mathbf{b}}) \sim \delta_i^2 k_{\perp} \rho_i en_e v_i/L$; which is negligible with respect to the rest of the terms, the smallest of which is order $\delta_i (k_{\perp}\rho_i)^2 en_e v_i/L$. The divergence of $(Mc/B)\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3v f_{ig}v_{||}\mathbf{v}_{\perp} \sim \delta_i^2 k_{\perp}\rho_i en_e v_i$ has only one term that is

of order $\delta_i(k_\perp \rho_i)^2 e n_e v_i/L$, given by

$$\nabla \cdot \left(\frac{Mc}{B}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3 v \, f_{ig} v_{||} \mathbf{v}_{\perp}\right) \simeq \frac{Mc}{B} \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3 v \, v_{||} \mathbf{v}_{\perp} \cdot \overline{\nabla} f_{ig}. \tag{I.8}$$

Since the only dependence of f_{ig} on φ_0 is in $\mathbf{R}_g = \mathbf{r} + \Omega_i^{-1} \mathbf{v} \times \hat{\mathbf{b}}$, I find that $\mathbf{v}_{\perp} \cdot \overline{\nabla} f_{ig} = \Omega_i (\partial f_{ig} / \partial \varphi_0)$. Thus, the divergence of $(Mc/B) \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3 v f_{ig} v_{||} \mathbf{v}_{\perp}$ vanishes to the relevant order due to the gyrophase integration

$$\nabla \cdot \left(\frac{Mc}{B}\hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3 v \, f_{ig} v_{||} \mathbf{v}_{\perp}\right) \simeq Z e \hat{\mathbf{b}} \cdot \nabla \times \hat{\mathbf{b}} \int d^3 v \, v_{||} \frac{\partial f_{ig}}{\partial \varphi_0} = 0. \tag{I.9}$$

Appendix J

Flux surface averaged gyrokinetic vorticity equation

In this Appendix, I obtain the long wavelength limit of the flux surface averaged vorticity equation (4.62). The long wavelength limit of ϖ_G is given by $\varpi_G \to \varpi = \nabla \cdot [(Ze/\Omega_i)n_i \mathbf{V}_i \times \hat{\mathbf{b}}]$, as proven in (4.52). Then, upon using (2.16) and integrating once in ψ , equation (4.62) becomes

$$-\frac{\partial}{\partial t} \langle cRn_i M \mathbf{V}_i \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} = \left\langle \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi + \frac{cI}{B} (\nabla \cdot \boldsymbol{\pi}_{ig||} - \tilde{F}_{iE}) -\frac{c}{B} (\nabla \cdot \boldsymbol{\pi}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) + Zen_i \mathbf{V}_{iC} \cdot \nabla \psi - cR\mathbf{F}_{iC} \cdot \hat{\boldsymbol{\zeta}} \right\rangle_{\psi}.$$
 (J.1)

I will evaluate all the terms on the right side of equation (J.1) to order $\delta_i^2 k_\perp \rho_i e n_e v_i |\nabla \psi|$ for $k_\perp \rho_i \to 0$.

J.1 Limit of $\langle (cI/B) \nabla \cdot \boldsymbol{\pi}_{ig||} \rangle_{\psi}$ for $k_{\perp} \rho_i \to 0$

The term $\langle (cI/B) \nabla \cdot \boldsymbol{\pi}_{ig||} \rangle_{\psi}$ is written as

$$\left\langle \frac{cI}{B} \nabla \cdot \boldsymbol{\pi}_{ig||} \right\rangle_{\psi} \simeq \left\langle \nabla \cdot \left(\frac{cI}{B} \boldsymbol{\pi}_{ig||} \right) \right\rangle_{\psi} = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{cI}{B} \boldsymbol{\pi}_{ig||} \cdot \nabla \psi \right\rangle_{\psi}.$$
 (J.2)

The term $\pi_{ig||} \cdot \nabla(cI/B)$ is neglected because it is of order $\delta_i^3 en_e v_i |\nabla \psi|$. In the

definition of $\pi_{ig||}$ in (4.39), one of the terms is $-\int d^3v f_{ig}(c/B)(\overline{\nabla}\langle\phi\rangle \times \hat{\mathbf{b}})Mv_{||}$. The difference between $f_{ig}(c/B)\overline{\nabla}\langle\phi\rangle \times \hat{\mathbf{b}}$ and $f_i(c/B)\nabla\phi \times \hat{\mathbf{b}}$ gives rise to the term

$$\frac{cI}{B^2} \int d^3 v \, M v_{||} [f_{ig}(\overline{\nabla} \langle \phi \rangle \times \hat{\mathbf{b}}) - f_i(\nabla \phi \times \hat{\mathbf{b}})] \cdot \nabla \psi =$$

$$\frac{cI}{B^2} \int d^3 v \, M v_{||} \left[\frac{Ze\tilde{\phi}}{T_i} f_{Mi}(\nabla \phi \times \hat{\mathbf{b}}) - f_{ig} \frac{c}{B}(\overline{\nabla} \tilde{\phi} \times \hat{\mathbf{b}}) \right] \cdot \nabla \psi$$
(J.3)

in $(I/B)\pi_{ig||}\cdot\nabla\psi$, where I use $f_i - f_{ig} = -(Ze\tilde{\phi}/T_i)f_{Mi}$ to obtain the second equality. I will show below that the difference (J.3) is of order $\delta_i^2 k_\perp \rho_i p_i R |\nabla\psi|$ and therefore negligible compared to the other terms in $(I/B)\pi_{ig||}\cdot\nabla\psi$ that are of order $\delta_i^2 p_i R |\nabla\psi|$. Then, in $(I/B)\pi_{ig||}\cdot\nabla\psi$ the difference between $f_{ig}(c/B)\overline{\nabla}\langle\phi\rangle\times\hat{\mathbf{b}}$ and $f_i(c/B)\nabla\phi\times\hat{\mathbf{b}}$ can be neglected to write equation (J.2) as

$$\left\langle \frac{cI}{B} \nabla \cdot \boldsymbol{\pi}_{ig||} \right\rangle_{\psi} \simeq \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{cI}{B} \boldsymbol{\pi}'_{ig||} \cdot \nabla \psi \right\rangle_{\psi}, \qquad (J.4)$$

with

$$\boldsymbol{\pi}_{ig||}^{\prime} = \int d^{3}v \, f_{ig}(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_{1}) M v_{||} - \frac{Mc}{B} \nabla \phi \times \hat{\mathbf{b}} \int d^{3}v \, f_{i} v_{||}. \tag{J.5}$$

Equation (J.4) is then seen to be of order $\delta_i^2 k_\perp \rho_i e n_e v_i |\nabla \psi|$.

I will now prove that the difference (J.3) is of order $\delta_i^2 k_{\perp} \rho_i p_i R |\nabla \psi|$. In equation (J.3), short wavelength components of ϕ , $\tilde{\phi}$ and f_{ig} can beat nonlinearly to give a long wavelength component, and these functions cannot be Taylor expanded around **r**. However, the total long wavelength contribution (J.3) can be expanded. The gyrophase dependence in \mathbf{R}_g then gives a contribution of order $\delta_i k_{\perp} \rho_i p_i R |\nabla \psi|$ that can be ignored when integrating over velocity space to order $\delta_i^2 p_i R |\nabla \psi|$. The result is $(cI/B^2) \int d^3 v (Ze\tilde{\phi}/T_i) f_{Mi} M v_{||} (\nabla \tilde{\phi} \times \hat{\mathbf{b}}) \cdot \nabla \psi$ because the rest of the terms have vanishing gyroaverages to the order of interest. Since both f_{Mi} and $\tilde{\phi}$ are even in $v_{||}$, this integral vanishes, and (J.3) is higher order than $\delta_i^2 p_i R |\nabla \psi|$.

J.2 Limit of $\langle \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi - (c/B) (\nabla \cdot \overleftrightarrow{\pi}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) \rangle_{\psi}$ for $k_{\perp} \rho_i \to 0$

I simplify the terms $\langle \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi \rangle_{\psi}$ and $-\langle (c/B)(\nabla \cdot \overrightarrow{\boldsymbol{\pi}}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) \rangle_{\psi}$ by first calculating the divergence of $\tilde{\mathbf{J}}_{i\phi} + (c/B)\hat{\mathbf{b}} \times (\nabla \cdot \overrightarrow{\boldsymbol{\pi}}_{iG})$. I employ the long wavelength result for $\nabla \cdot \tilde{\mathbf{J}}_{i\phi}$ in (4.56) to obtain

$$\nabla \cdot \left[\tilde{\mathbf{J}}_{i\phi} + \frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}_{iG}) \right] = \nabla \cdot \left[\frac{c}{B} \hat{\mathbf{b}} \times (\nabla \cdot \overleftarrow{\pi}'_{iG} + \nabla \cdot \overleftarrow{\pi}'_{iG}) \right], \qquad (J.6)$$

with

$$\dot{\boldsymbol{\pi}}_{iG}^{\prime} = M \int d^{3}v \, f_{ig} v_{||} (\hat{\mathbf{b}} \mathbf{v}_{\perp} + \mathbf{v}_{\perp} \hat{\mathbf{b}}) \sim \delta_{i} k_{\perp} \rho_{i} p_{i} \tag{J.7}$$

and

$$\vec{\boldsymbol{\pi}}_{iG}^{\prime\prime} = \int d^3 v \, f_{ig}(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) M \mathbf{v}_{\perp} - \frac{Mc}{B} \nabla \phi \times \hat{\mathbf{b}} \int d^3 v \, f_i \mathbf{v}_{\perp} \sim \delta_i^2 p_i. \tag{J.8}$$

Flux surface averaging equation (J.6) and integrating once in ψ , I find the order $\delta_i^2 k_\perp \rho_i e n_e v_i |\nabla \psi|$ term

$$\left\langle \tilde{\mathbf{J}}_{i\phi} \cdot \nabla \psi - \frac{c}{B} (\nabla \cdot \overleftarrow{\boldsymbol{\pi}}_{iG}) \cdot (\hat{\mathbf{b}} \times \nabla \psi) \right\rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{c}{B} \nabla \psi \cdot \overleftarrow{\boldsymbol{\pi}}_{iG}'' \cdot (\hat{\mathbf{b}} \times \nabla \psi) \right\rangle_{\psi}, (J.9)$$

where I neglected $c \ \vec{\pi}_{iG}^{"}$: $\nabla[(\hat{\mathbf{b}} \times \nabla \psi)/B] \sim \delta_i^3 en_e v_i |\nabla \psi|$, I used the definition of $\vec{\pi}_{iG}^{'}$ in (J.7) to obtain $\nabla \psi \cdot \vec{\pi}_{iG}^{'} \cdot (\hat{\mathbf{b}} \times \nabla \psi) = 0$, and I will next prove that $\langle \vec{\pi}_{iG}^{'} : \nabla[(\hat{\mathbf{b}} \times \nabla \psi)/B] \rangle_{\psi}$ vanishes.

To see that $\langle \vec{\pi}'_{iG} : \nabla [(\hat{\mathbf{b}} \times \nabla \psi)/B] \rangle_{\psi} = 0$, the velocity integral $\int d^3v f_{ig} v_{||} \mathbf{v}_{\perp}$ in $\vec{\pi}'_{iG}$ has to be found to order $\delta_i k_{\perp} \rho_i n_e v_i^2$. This integral only depends on the gyrophase dependent piece of f_{ig} , given to the required order by

$$f_i - \overline{f}_i \simeq \frac{1}{\Omega_i} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} f_{i0} + \frac{1}{2\Omega_i^2} (\mathbf{v} \times \hat{\mathbf{b}}) (\mathbf{v} \times \hat{\mathbf{b}}) : \overline{\nabla} \overline{\nabla} f_{i0}, \qquad (J.10)$$

with $f_{i0} \equiv f_i(\mathbf{r}, E_0, \mu_0, \varphi_0)$ as defined in (4.8) and, thus, gyrophase independent. The

integral involving $\overline{\nabla} \overline{\nabla} f_{i0}$ vanishes, leaving

$$\begin{aligned} & \overleftarrow{\boldsymbol{\pi}}_{iG}^{\prime} \colon \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B} \right) = \\ & \int d^{3}v \, \frac{Mv_{||}}{\Omega_{i}} [(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} f_{i0}] (\hat{\mathbf{b}} \mathbf{v}_{\perp} + \mathbf{v}_{\perp} \hat{\mathbf{b}}) : \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B} \right) \simeq \\ & \nabla \cdot \left[\int d^{3}v \, f_{i0} \frac{Mv_{||}}{\Omega_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) (\hat{\mathbf{b}} \mathbf{v}_{\perp} + \mathbf{v}_{\perp} \hat{\mathbf{b}}) : \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B} \right) \right], \end{aligned} \tag{J.11}$$

where terms of order $\delta_i^2 p_i$ are neglected. Integrating in gyrophase and flux surface averaging, equation (J.11) reduces to

$$\left\langle \overleftarrow{\boldsymbol{\pi}}_{iG}^{\prime}: \nabla \left(\frac{\mathbf{b} \times \nabla \psi}{B} \right) \right\rangle_{\psi} = \frac{1}{V^{\prime}} \frac{\partial}{\partial \psi} V^{\prime} \left\langle M \int d^{3}v \, f_{i0} \frac{v_{||} v_{\perp}^{2}}{2\Omega_{i}} [\hat{\mathbf{b}}(\hat{\mathbf{b}} \times \nabla \psi) + (\hat{\mathbf{b}} \times \nabla \psi) \hat{\mathbf{b}}] : \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B} \right) \right\rangle_{\psi}.$$
 (J.12)

This expression vanishes because $[\hat{\mathbf{b}}(\hat{\mathbf{b}} \times \nabla \psi) + (\hat{\mathbf{b}} \times \nabla \psi)\hat{\mathbf{b}}] : \nabla[(\hat{\mathbf{b}} \times \nabla \psi)/B] = 0.$ To prove this, I employ equation (2.16) to write $\nabla[(\hat{\mathbf{b}} \times \nabla \psi)/B] = \nabla(I\hat{\mathbf{b}}/B) - \nabla(R\hat{\boldsymbol{\zeta}})$. The tensor $\nabla(R\hat{\boldsymbol{\zeta}}) = (\nabla R)\hat{\boldsymbol{\zeta}} - \hat{\boldsymbol{\zeta}}(\nabla R)$ gives zero contribution because it is antisymmetric and it is multiplied by the symmetric tensor $[\hat{\mathbf{b}}(\hat{\mathbf{b}} \times \nabla \psi) + (\hat{\mathbf{b}} \times \nabla \psi)\hat{\mathbf{b}}]$. Then, I am only left with $\nabla(I\hat{\mathbf{b}}/B)$, giving

$$\left[\hat{\mathbf{b}}(\hat{\mathbf{b}} \times \nabla \psi) + (\hat{\mathbf{b}} \times \nabla \psi)\hat{\mathbf{b}}\right] : \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B}\right) = \left(\hat{\mathbf{b}} \times \nabla \psi\right) \cdot \left[\nabla \left(\frac{I}{B}\right) + \frac{I}{B}\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}}\right]. \quad (J.13)$$

To simplify, I use relation (2.8), with $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} = \boldsymbol{\kappa}$, to write $\hat{\mathbf{b}} \cdot \nabla \hat{\mathbf{b}} \cdot (\hat{\mathbf{b}} \times \nabla \psi) = -(\nabla \times \hat{\mathbf{b}}) \cdot \nabla \psi = -\nabla \cdot (\hat{\mathbf{b}} \times \nabla \psi)$. Finally, I employ (2.16) and $\hat{\boldsymbol{\zeta}} \cdot \nabla (I/B) = 0 = \nabla \cdot (RB\hat{\boldsymbol{\zeta}})$ due to axisymmetry to obtain

$$[\hat{\mathbf{b}}(\hat{\mathbf{b}} \times \nabla \psi) + (\hat{\mathbf{b}} \times \nabla \psi)\hat{\mathbf{b}}] : \nabla \left(\frac{\hat{\mathbf{b}} \times \nabla \psi}{B}\right) = I\hat{\mathbf{b}} \cdot \nabla \left(\frac{I}{B}\right) - \frac{I}{B}\nabla \cdot (I\hat{\mathbf{b}}) = 0. \quad (J.14)$$

J.3 Limit of $\langle (cI/B)\tilde{F}_{iE}\rangle_{\psi}$ for $k_{\perp}\rho_i \to 0$

The function \tilde{F}_{iE} , defined in (4.41), is written as

$$\tilde{F}_{iE} = Ze \int dE_0 \, d\mu_0 \, d\varphi_0 \frac{\partial v_{||}}{\partial E_0} Bf_{Mi} \left(\hat{\mathbf{b}} \cdot \overline{\nabla} \tilde{\phi} + \frac{1}{\Omega_i} \overline{\nabla} \times \mathbf{v}_\perp \cdot \overline{\nabla} \tilde{\phi} \right), \tag{J.15}$$

where I use $d^3v = (B/v_{||})dE_0 d\mu_0 d\varphi_0$ and $v_{||}^{-1} = \partial v_{||}/\partial E_0$. Integrating by parts in E_0 , and making use of $\partial \tilde{\phi}/\partial E_0 = 0$, $\partial f_{Mi}/\partial E_0 = (-M/T_i)f_{Mi}$ and $\tilde{\mathbf{v}}_1 = (v_{||}/\Omega_i)\overline{\nabla} \times \mathbf{v}_{\perp}$, I find

$$\tilde{F}_{iE} = M \int dE_0 \, d\mu_0 \, d\varphi_0 B \frac{Ze}{T_i} f_{Mi}(v_{||} \hat{\mathbf{b}} + \tilde{\mathbf{v}}_1) \cdot \overline{\nabla} \tilde{\phi}. \tag{J.16}$$

Multiplying equation (J.16) by cI/B and writing the result as a divergence give

$$\frac{cI}{B}\tilde{F}_{iE} \simeq \nabla \cdot \left(\frac{cI}{B}\boldsymbol{\pi}_{iE}'\right) - \int d^3v \, \frac{Z^2 e^2 \tilde{\phi}}{T_i} f_{Mi} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{Iv_{||}}{\Omega_i}\right), \qquad (J.17)$$

with $\boldsymbol{\pi}_{iE}^{\prime} = \int d^3 v \left(Z e \tilde{\phi} / T_i \right) f_{Mi}(v_{||} \hat{\mathbf{b}} + \tilde{\mathbf{v}}_1) M v_{||}$. Here, I employ $\hat{\mathbf{b}} \cdot \overline{\nabla} (f_{Mi} / T_i) = 0$, and neglect the integral $Mc \int dE_0 d\mu_0 d\varphi_0 Z e \tilde{\phi} \overline{\nabla} \cdot (\tilde{\mathbf{v}}_1 I f_{Mi} / T_i)$ because it is of order $\delta_i^3 e n_e v_i |\nabla \psi|$. I will now consider the two integrals in equation (J.17). Upon using (2.15), the flux surface average of the first integral gives

$$\left\langle \nabla \cdot \left(\frac{cI}{B}\boldsymbol{\pi}_{iE}'\right)\right\rangle_{\psi} = \frac{1}{V'}\frac{\partial}{\partial\psi}V'\left\langle Mc\int d^{3}v\,\frac{Ze\tilde{\phi}}{T_{i}}f_{Mi}\frac{Iv_{||}}{B}\tilde{\mathbf{v}}_{1}\cdot\nabla\psi\right\rangle_{\psi}.$$
 (J.18)

Multiplying equation (2.16) by \mathbf{v} , I find $Iv_{\parallel}/B = R\mathbf{v}\cdot\hat{\boldsymbol{\zeta}} + (\mathbf{v}\times\hat{\mathbf{b}})\cdot\nabla\psi/B$. Substituting this result in equation (J.18), I find that the integral of $\mathbf{v}\times\hat{\mathbf{b}}$ vanishes because $\tilde{\phi}$ and f_{Mi} are even in v_{\parallel} , and $\tilde{\mathbf{v}}_1$ is odd. Thus, the first integral of equation (J.17) gives

$$\left\langle \nabla \cdot \left(\frac{cI}{B}\boldsymbol{\pi}_{iE}'\right)\right\rangle_{\psi} = \frac{1}{V'}\frac{\partial}{\partial\psi}V'\left\langle c\int d^{3}v \,\frac{Ze\tilde{\phi}}{T_{i}}f_{Mi}RM(\mathbf{v}\cdot\hat{\boldsymbol{\zeta}})\tilde{\mathbf{v}}_{1}\cdot\nabla\psi\right\rangle_{\psi}.$$
 (J.19)

In the second integral of (J.17), I need to keep $\tilde{\phi}(\mathbf{R}_g, \mu_0, \varphi_0, t) \simeq \tilde{\phi}(\mathbf{r}, \mu_0, \varphi_0, t) +$

 $\Omega_i^{-1}(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \widetilde{\phi}$. Using $\oint d\varphi_0 \, \widetilde{\phi}(\mathbf{r}, \mu_0, \varphi_0, t) = 0$ leaves

$$\int d^{3}v \, \frac{Z^{2} e^{2} \widetilde{\phi}}{T_{i}} f_{Mi} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{I v_{||}}{\Omega_{i}} \right) \simeq$$

$$\frac{Z e}{\Omega_{i}} \int d^{3}v \, f_{Mi} \frac{Z e}{T_{i}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \overline{\nabla} \widetilde{\phi} \left[v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{I v_{||}}{\Omega_{i}} \right) \right] \simeq$$

$$\nabla \cdot \left\{ \frac{M c}{B} \int d^{3}v \, \frac{Z e \widetilde{\phi}}{T_{i}} f_{Mi} (\mathbf{v} \times \hat{\mathbf{b}}) \left[v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{I v_{||}}{\Omega_{i}} \right) \right] \right\}, \qquad (J.20)$$

where terms of order $\delta_i^3 e n_e v_i |\nabla \psi|$ are neglected to obtain the second equality. Using $v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} (I v_{||} / \Omega_i) = \mathbf{v}_{M0} \cdot \nabla \psi$ in equation (J.20) and flux surface averaging, I find

$$\left\langle \int d^{3}v \, \frac{Z^{2} e^{2} \widetilde{\phi}}{T_{i}} f_{Mi} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{I v_{||}}{\Omega_{i}} \right) \right\rangle_{\psi} = \frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle Mc \int d^{3}v \, \frac{Ze \widetilde{\phi}}{T_{i}} f_{Mi} \frac{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi}{B} \mathbf{v}_{M0} \cdot \nabla \psi \right\rangle_{\psi}. \tag{J.21}$$

Here, equation (2.16) multiplied by \mathbf{v} gives $(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi / B = I v_{||} / B - R \mathbf{v} \cdot \hat{\boldsymbol{\zeta}}$. The integral of $I v_{||} / B$ vanishes because both $\tilde{\phi}$ and f_{Mi} are even in $v_{||}$. Thus, equation (J.21) gives to relevant order

$$\left\langle \int d^{3}v \, \frac{Z^{2} e^{2} \widetilde{\phi}}{T_{i}} f_{Mi} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left(\frac{I v_{||}}{\Omega_{i}} \right) \right\rangle_{\psi} \simeq -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle c \int d^{3}v \, \frac{Z e \widetilde{\phi}}{T_{i}} f_{Mi} R M(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) \mathbf{v}_{M0} \cdot \nabla \psi \right\rangle_{\psi}. \tag{J.22}$$

Substituting equations (J.19) and (J.22) into the flux surface average of equation (J.17), I obtain

$$\left\langle \frac{cI}{B}\tilde{F}_{iE}\right\rangle_{\psi} \simeq \frac{1}{V'}\frac{\partial}{\partial\psi}V'\left\langle c\int d^{3}v\left(f_{ig}-f_{i}\right)RM(\mathbf{v}\cdot\hat{\boldsymbol{\zeta}})(\mathbf{v}_{M0}+\tilde{\mathbf{v}}_{1})\cdot\nabla\psi\right\rangle_{\psi},\qquad(J.23)$$

where I use $f_i - f_{ig} = -(Ze\tilde{\phi}/T_i)f_{Mi}$.

J.4 Limit of
$$\langle Zen_i \mathbf{V}_{iC} \cdot \nabla \psi - cR\mathbf{F}_{iC} \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi}$$
 for $k_{\perp} \rho_i \to 0$

This collisional combination vanishes. According to (H.17), for $k_{\perp}\rho_i \ll 1$, $\langle Zen_i \mathbf{V}_{iC} \cdot \nabla \psi \rangle_{\psi}$ is given by

$$\langle Zen_i \mathbf{V}_{iC} \cdot \nabla \psi \rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{c}{B\Omega_i} \int d^3 v \, C\{f_i\} |\nabla \psi|^2 \frac{M v_{\perp}^2}{2} \right\rangle_{\psi}.$$
 (J.24)

Equation (H.19), on the other hand, gives

$$\langle cR\mathbf{F}_{iC} \cdot \hat{\boldsymbol{\zeta}} \rangle_{\psi} = -\frac{1}{V'} \frac{\partial}{\partial \psi} V' \left\langle \frac{c}{B\Omega_i} \int d^3 v \, C\{f_i\} |\nabla \psi|^2 \frac{M v_{\perp}^2}{2} \right\rangle_{\psi}, \qquad (J.25)$$

where I use equation (2.1) to obtain $\nabla \psi \cdot (\hat{\mathbf{b}} \times \hat{\boldsymbol{\zeta}}) = |\nabla \psi|^2 / RB$.

Finally, since the collisional piece vanishes to relevant order, I just need to substitute equations (J.4), (J.9) and (J.23) into equation (J.1) and employ (2.16) to find (4.63).

Appendix K

Gyroviscosity in gyrokinetics

In this Appendix, I show why the gyroviscosity must take the form given in equation (4.65), and later I simplify that expression for up-down symmetric tokamaks by proving that the collisional piece must vanish.

K.1 Evaluation of equation (4.65)

To prove equation (4.65), I employ that $\mathbf{v}_{M0} \cdot \nabla \psi = v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} (Iv_{||}/\Omega_i)$ and $\tilde{\mathbf{v}}_1 \cdot \nabla \psi = -v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} [\Omega_i^{-1} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi]$, proven below in (K.2), to write

$$(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \nabla \psi = \frac{Mc}{Ze} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left[\frac{Iv_{||}}{B} - \frac{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi}{B} \right] = \frac{Mc}{Ze} v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} [R(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})], \quad (K.1)$$

where I use equation (2.16) to find the last equality. Notice that in $\tilde{\mathbf{v}}_1 \cdot \nabla \psi = (v_{\parallel}/\Omega_i)(\overline{\nabla} \times \mathbf{v}_{\perp}) \cdot \nabla \psi = (v_{\parallel}/\Omega_i)\overline{\nabla} \cdot (\mathbf{v}_{\perp} \times \nabla \psi)$, both \mathbf{v}_{\perp} and $\nabla \psi$ are perpendicular to $\hat{\mathbf{b}}$. Then, $\mathbf{v}_{\perp} \times \nabla \psi$ must be parallel to $\hat{\mathbf{b}}$, giving $\mathbf{v}_{\perp} \times \nabla \psi = -\hat{\mathbf{b}}[(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi]$, and

$$\tilde{\mathbf{v}}_{1} \cdot \nabla \psi = -\frac{v_{||}}{\Omega_{i}} \overline{\nabla} \cdot \left[\hat{\mathbf{b}} (\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi \right] = -v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left[\frac{(\mathbf{v} \times \hat{\mathbf{b}}) \cdot \nabla \psi}{\Omega_{i}} \right].$$
(K.2)

Substituting equation (K.1) into equation (4.65) gives

$$\left\langle \int d^3 v \, f_i R M(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) (\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \nabla \psi \right\rangle_{\psi} =$$

$$\frac{M^{2}c}{2Ze} \left\langle \int d^{3}v f_{i}v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla}[R^{2}(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2}] \right\rangle_{\psi} = -\frac{M^{2}c}{2Ze} \left\langle \int d^{3}v [R^{2}(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2}]v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla}f_{i} \right\rangle_{\psi}.$$
(K.3)

Only the axisymmetric, short radial wavelength pieces of f_i contribute to (K.3) because of the flux surface average. To find this portion of f_i , I employ equations (4.10), (4.11), (4.12), (4.13), (4.14) and (4.15) to obtain the gyrokinetic equation

$$\frac{\partial f_{ig}}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + (v_{||}\hat{\mathbf{b}} + \mathbf{v}_{M0} + \tilde{\mathbf{v}}_{1}) \cdot \left(\overline{\nabla}f_{ig} - \frac{Ze}{M}\overline{\nabla}\langle\phi\rangle\frac{\partial f_{Mi}}{\partial E_{0}}\right) + \mathbf{v}_{E0} \cdot \overline{\nabla}f_{ig} = \langle C\{f_{i}\}\rangle.$$
(K.4)

In this equation, I will only keep terms of order $\delta_i f_{Mi} v_i / L$ or larger. These terms will give $R\hat{\boldsymbol{\zeta}} \cdot \boldsymbol{\pi}_i^{(0)} \cdot \nabla \psi \sim \delta_i^2 p_i R |\nabla \psi|$ as seen from (K.3).

In the nonlinear term $\mathbf{v}_{E0} \cdot \overline{\nabla} f_{ig}$, different components of $\langle \phi \rangle$ and f_{ig} beat together to give an axisymmetric, short radial wavelength contribution to (K.4). The term $\mathbf{v}_{E0} \cdot \overline{\nabla} f_{ig}$ can be treated as the term $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ in the axisymmetric piece of equation (4.1), where it was neglected. In equation (K.4), $\mathbf{v}_{E0} \cdot \overline{\nabla} f_{ig} \sim k_{\perp} \rho_i \delta_i f_{Mi} v_i / L \ll$ $\delta_i f_{Mi} v_i / L$ is negligible, too. This is another form of the conclusion of section 4.1, i.e., the long wavelength, axisymmetric flows are neoclassical.

Neglecting $\mathbf{v}_{E0} \cdot \overline{\nabla} f_{ig}$ and using $f_{ig} = f_i + (Ze\tilde{\phi}/T_i)f_{Mi}$ in equation (K.4), I find that the axisymmetric, short radial wavelength portion of f_i must satisfy

$$\frac{\partial f_i}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + v_{||}\hat{\mathbf{b}} \cdot \left(\overline{\nabla}f_i + \frac{Ze}{T_i}f_{Mi}\nabla\phi\right) + \left(\frac{\partial f_i}{\partial\psi} + \frac{Ze}{T_i}f_{Mi}\frac{\partial\phi}{\partial\psi}\right)(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \nabla\psi \\ = \langle C\{f_i\}\rangle. \quad (K.5)$$

Here, the terms that contain $(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \overline{\nabla} \phi$ and $(\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \overline{\nabla} (\phi f_{Mi})$ are neglected because they are smaller than the rest of the terms by a factor $k_{\perp}\rho_i$ since $\phi \sim \delta_i T_e/e \sim k_{\perp}\rho_i \langle \phi \rangle$. Additionally, I used $\nabla_{\perp} \simeq \nabla \psi (\partial/\partial \psi)$ since only short radial wavelength effects matter in (K.4). Finally, substituting equation (K.1) into (K.5) gives

$$\frac{\partial f_i}{\partial t}\Big|_{\mathbf{r},\mathbf{v}} + v_{||}\hat{\mathbf{b}} \cdot \left(\overline{\nabla}f_i + \frac{Ze}{T_i}f_{Mi}\nabla\phi\right) + \frac{Mc}{Ze}\left(\frac{\partial f_i}{\partial\psi} + \frac{Ze}{T_i}f_{Mi}\frac{\partial\phi}{\partial\psi}\right)v_{||}\hat{\mathbf{b}} \cdot \overline{\nabla}[R(\mathbf{v}\cdot\hat{\boldsymbol{\zeta}})] \\ = \langle C\{f_i\}\rangle.(\mathrm{K.6})$$

I substitute $v_{\parallel} \hat{\mathbf{b}} \cdot \nabla f_i$ from equation (K.6) into equation (K.3) to find

$$\left\langle \int d^3 v \, f_i R M(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}}) (\mathbf{v}_{M0} + \tilde{\mathbf{v}}_1) \cdot \nabla \psi \right\rangle_{\psi} = \frac{M^2 c}{2Ze} \left\langle \int d^3 v \left[R^2 (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^2 \right] \left(\left. \frac{\partial f_i}{\partial t} \right|_{\mathbf{r}, \mathbf{v}} - \langle C\{f_i\} \rangle \right) \right\rangle_{\psi}, \tag{K.7}$$

where I use $(Ze/T_i)\hat{\mathbf{b}} \cdot \nabla \phi \int d^3v f_{Mi} v_{||}[R^2(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^2] = 0$ and

$$\frac{M^{3}c^{2}}{2Z^{2}e^{2}}\left\langle \int d^{3}v \left(\frac{\partial f_{i}}{\partial \psi} + \frac{Ze}{T_{i}} f_{Mi} \frac{\partial \phi}{\partial \psi} \right) v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} [R(\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})] R^{2} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{2} \right\rangle_{\psi} = \frac{M^{3}c^{2}}{6Z^{2}e^{2}} \left\langle \int d^{3}v \, v_{||} \hat{\mathbf{b}} \cdot \overline{\nabla} \left[\left(\frac{\partial f_{i}}{\partial \psi} + \frac{Ze}{T_{i}} f_{Mi} \frac{\partial \phi}{\partial \psi} \right) R^{3} (\mathbf{v} \cdot \hat{\boldsymbol{\zeta}})^{3} \right] \right\rangle_{\psi} = 0. \quad (K.8)$$

To obtain the first equality in equation (K.8), terms that contain $\hat{\mathbf{b}} \cdot \overline{\nabla}(\partial f_i / \partial \psi)$ and $\hat{\mathbf{b}} \cdot \nabla(\partial \phi / \partial \psi)$ are neglected.

Finally, using $k_{\perp}\rho_i \ll 1$ in (K.7), the replacements $f_i \simeq \overline{f}_i$ and $\langle C\{f_i\} \rangle \simeq \overline{C\{f_i\}}$ lead to (4.65).

K.2 Collisional piece in up-down symmetric tokamaks

To prove that the collisional contribution to equation (4.65) vanishes for up-down symmetric tokamaks I employ the drift kinetic equation (4.1) for the long wavelength, axisymmetric component of \overline{f}_i . I already proved that the term $\mathbf{v}_E \cdot \nabla_{\mathbf{R}} f_i$ can be safely neglected. Additionally, I assume that the time derivative is small once the statistical equilibrium is reached, and I split the distribution function into $\overline{f}_i = f_{Mi}(\psi, E_0) + \overline{h}_i$, with $\overline{h}_i(\psi, \theta, E_0, \mu_0, t) \ll f_{Mi}$, giving

$$v_{||}\hat{\mathbf{b}} \cdot \nabla \theta \left[\frac{\partial \overline{h}_i}{\partial \theta} + \frac{\partial}{\partial \theta} \left(\frac{Iv_{||}}{\Omega_i} \frac{\partial f_{Mi}}{\partial \psi} \right) \right] = C^{(\ell)} \{ \overline{h}_i \}, \tag{K.9}$$

where I use $\mathbf{v}_{M0} \cdot \nabla \psi = v_{\parallel} \hat{\mathbf{b}} \cdot \overline{\nabla} (Iv_{\parallel}/\Omega_i)$. Since the tokamak is up-down symmetric,

 Iv_{\parallel}/Ω_i is symmetric in θ and its derivative is antisymmetric.

In equation (K.9), replacing θ by $-\theta$, $v_{||}$ by $-v_{||}$ and \overline{h}_i by $-\overline{h}_i$ does not change the equation. Then, \overline{h}_i changes sign if both θ and $v_{||}$ do. Due to this property, the collisional integral in (4.65) is given by

$$\left\langle \frac{M}{2B\Omega_i} \int d^3 v \, C^{(\ell)}\{\overline{h}_i\} \left(|\nabla \psi|^2 \frac{v_\perp^2}{2} + I^2 v_{||}^2 \right) \right\rangle_{\psi} = 0. \tag{K.10}$$

In the contributions to this integral, the piece of the distribution function with positive $v_{||}$ in the upper half ($\theta > 0$) of the tokamak cancels the piece of the distribution function with negative $v_{||}$ in the lower half ($\theta < 0$). Similarly, the piece with negative $v_{||}$ in the upper half cancels the piece with positive $v_{||}$ in the lower half.

Bibliography

- [1] W. Dorland, F. Jenko, M. Kotschenreuther, and B.N. Rogers. Electron temperature gradient turbulence. *Physical Review Letters*, 85:5579, 2000.
- [2] T. Dannert and F. Jenko. Gyrokinetic simulation of collisionless trapped-electron mode turbulence. *Physics of Plasmas*, 12:072309, 2005.
- [3] J. Candy and R.E. Waltz. An Eulerian gyrokinetic-Maxwell solver. Journal of Computational Physics, 186:545, 2003.
- [4] Y. Chen and S.E. Parker. A δf particle method for gyrokinetic simulations with kinetic electrons and electromagnetic perturbations. Journal of Computational *Physics*, 189:463, 2003.
- [5] A.M. Dimits, T.J. Williams, J.A. Byers, and B.I. Cohen. Scalings of iontemperature-gradient-driven anomalous transport in tokamaks. *Physical Review Letters*, 77:71, 1996.
- [6] Z. Lin, S. Ethier, T.S. Hahm, and W.M. Tang. Size scaling of turbulent transport in magnetically confined plasmas. *Physical Review Letters*, 88:195004, 2002.
- [7] H. Biglari, P.H. Diamond, and M.N. Rosenbluth. Toroidal ion-pressure-gradientdriven drift instabilities and transport revisited. *Physics of Fluids B*, 1:109, 1989.
- [8] F. Romanelli and S. Briguglio. Toroidal semicollisional microinstabilities and anomalous electron and ion transport. *Physics of Fluids B*, 2:754, 1990.
- [9] R.J. Fonck, G. Cosby, R.D. Durst, S.F. Paul, N. Bretz, S. Scott, E. Synakowski, and G. Taylor. Long-wavelength density turbulence in TFTR tokamak. *Physical Review Letters*, 70:3736, 1993.
- [10] M.N. Rosenbluth and F.L. Hinton. Poloidal flow driven by ion-temperaturegradient turbulence in tokamaks. *Physical Review Letters*, 80:724, 1998.
- [11] F.L. Hinton and M.N. Rosenbluth. Dynamics of axisymmetric $(E \times B)$ and poloidal flows in tokamaks. *Plasma Physics and Controlled Fusion*, 41:A653, 1999.
- [12] P.H. Diamond, S.-I. Itoh, K. Itoh, and T.S. Hahm. Zonal flows in plasmas a review. *Plasma Physics and Controlled Fusion*, 47:R35, 2005.

- [13] A.M. Dimits, G. Bateman, M.A. Beer, B.I. Cohen, W. Dorland, G.W. Hammett, C. Kim, J.E. Kinsey, M. Kotschenreuther, A.H. Kritz, L.L. Lao, J. Mandrekas, W.M. Nevins, S.E. Parker, A.J. Redd, D.E. Shumaker, R. Sydora, and J. Weiland. Comparisons and physics basis of tokamak transport models and turbulence simulations. *Physics of Plasmas*, 7:969, 2000.
- [14] J.W. Connor and H.R. Wilson. A review of theories of the L-H transition. Plasma Physics and Controlled Fusion, 42:R1, 2000.
- [15] R.A. Moyer, K.H. Burrell, T.N. Carlstrom, S. Coda, R.W. Conn, E.J. Doyle, P. Gohil, R.J. Groebner, J. Kim, R. Lehmer, W.A. Peebles, M. Porkolab, C.-L. Rettig, T.L. Rhodes, R.P. Seraydarian, R. Stockdale, D.M. Thomas, G.R. Tynan, and J. G. Watkins. Beyond paradigm: turbulence, transport, and the origin of the radial electric field in the low to high confinement mode transitions in the DIII-D tokamak. *Physics of Plasmas*, 2:2397, 1995.
- [16] P.H. Rutherford and E.A. Frieman. Drift instabilities in general magnetic field configurations. *Physics of Fluids*, 11:569, 1968.
- [17] J.B. Taylor and R.J. Hastie. Stability of general plasma equilibria I Formal theory. *Plasma Physics*, 10:479, 1968.
- [18] P.J. Catto. Linearized gyro-kinetics. Plasma Physics, 20:719, 1978.
- [19] P.J. Catto, W.M. Tang, and D.E. Baldwin. Generalized gyrokinetics. *Plasma Physics*, 23:639, 1981.
- [20] E.A. Frieman and L. Chen. Nonlinear gyrokinetic equations for low-frequency electromagnetic waves in general plasma equilibria. *Physics of Fluids*, 25:502, 1982.
- [21] D.H.E. Dubin, J.A. Krommes, C. Oberman, and W.W. Lee. Nonlinear gyrokinetic equations. *Physics of Fluids*, 26:3524, 1983.
- [22] T.S. Hahm, W.W. Lee, and A. Brizard. Nonlinear gyrokinetic theory for finitebeta plasmas. *Physics of Fluids*, 31:1940, 1988.
- [23] T.S. Hahm. Nonlinear gyrokinetic equations for tokamak microturbulence. *Physics of Fluids*, 31:2670, 1988.
- [24] R.G. Littlejohn. Hamiltonian formulation of guiding center motion. *Physics of Fluids*, 24:1730, 1981.
- [25] A.J. Brizard and T.S. Hahm. Foundations of nonlinear gyrokinetic theory. Reviews of Modern Physics, 79:421, 2007.
- [26] W.W. Lee. Gyrokinetic approach in particle simulation. *Physics of Fluids*, 26:556, 1983.

- [27] W.W. Lee. Gyrokinetic particle simulation model. *Journal of Computational Physics*, 72:243, 1987.
- [28] M. Kotschenreuther. Numerical simulation. Bulletin of the American Physical Society, 33:2107, 1988.
- [29] V. Grandgirard, M. Brunetti, P. Bertrand, N. Besse, X. Garbet, P. Ghendrih, G. Manfredi, Y. Sarazin, O. Sauter, E. Sonnendruücker, J. Vaclavik, and L. Villard. A drift-kinetic semi-Lagrangian 4D code for ion turbulence simulation. *Journal of Computational Physics*, 217:395, 2006.
- [30] J.A. Heikkinen, J.S. Janhunen, T.P. Kiviniemi, and F. Ogando. Full f gyrokinetic method for particle simulation of tokamak transport. *Journal of Computational Physics*, 227:5582, 2008.
- [31] C.S. Chang and S. Ku. Spontaneous rotation sources in a quiescent tokamak edge plasma. *Physics of Plasmas*, 15:062510, 2008.
- [32] X.Q. Xu, Z. Xiong, M.R. Dorr, J.A. Hittinger, K. Bodi, J. Candy, B.I. Cohen, R.H. Cohen, P. Colella, G.D. Kerbel, S. Krasheninnikov, W.M. Nevins, H. Qin, T.D. Rognlien, P.B. Snyder, and M.V. Umansky. Edge gyrokinetic theory and continuum simulations. *Nuclear Fusion*, 47:809, 2007.
- [33] P.P. Sosenko, P. Bertrand, and V.K. Decyk. Krylov-Boholiubov methods and gyrokinetics. *Physica Scripta*, 64:264, 2001.
- [34] F.L. Hinton and R.D. Hazeltine. Theory of plasma transport in toroidal confinement systems. *Reviews of Modern Physics*, 48:239, 1976.
- [35] P. Helander and D.J. Sigmar. *Collisional transport in magnetized plasmas*. Cambridge Monographs on Plasmas Physics. Cambridge University Press, 2002.
- [36] P.J. Catto, A.N. Simakov, F.I. Parra, and G. Kagan. Electrostatic turbulence in tokamaks on transport time scales. *Plasma Physics and Controlled Fusion*, 50:115006, 2008.
- [37] L.M. Kovrizhnykh. Transport phenomena in toroidal magnetic systems. Soviet Physics JETP, 29:475, 1969.
- [38] P.H. Rutherford. Collisional diffusion in an axisymmetric torus. *Physics of Fluids*, 13:482, 1970.
- [39] P.J. Catto and A.N. Simakov. Evaluation of the neoclassical radial electric field in a collisional tokamak. *Physics of Plasmas*, 12:012501, 2005.
- [40] P.J. Catto and A.N. Simakov. A new, explicitly collisional contribution to the gyroviscosity and the radial electric in a collisional tokamak. *Physics of Plasmas*, 12:114503, 2005.

- [41] P.J. Catto and A.N. Simakov. Erratum: A new, explicitly collisional contribution to the gyroviscosity and the radial electric in a collisional tokamak. *Physics of Plasmas*, 13:129901, 2006.
- [42] S.K. Wong and V.S. Chan. Fluid theory of radial angular momentum flux of plasmas in an axisymmetric magnetic field. *Physics of Plasmas*, 14:112505, 2007.
- [43] M.N. Rosenbluth, P.H. Rutherford, J.B. Taylor, E.A. Frieman, and L.M. Kovrizhnikh. Neoclassical effects on plasma equilibria and rotation. *Plasma Phys. Controlled Nucl. Fusion Res.*, 1:495, 1971.
- [44] S.K. Wong and V.S. Chan. The neoclassical angular momentum flux in the large aspect ratio limit. *Physics of Plasmas*, 12:092513, 2005.
- [45] J.Y. Kim, W. Horton, and J.Q. Dong. Electromagnetic effect on the toroidal ion temperature gradient mode. *Physics of Fluids B*, 5:4030, 1993.
- [46] P.B. Snyder and G.W. Hammett. Electromagnetic effects on plasma mictroturbulence and transport. *Physics of Plasmas*, 8:744, 2001.
- [47] I.H. Hutchinson, R. Boivin, F. Bombarda, P. Bonoli, S. Fairfax, C. Fiore, J. Goetz, S. Golovato, R. Granetz, M. Greenwald, S. Horne, A. Hubbard, J. Irby, B. LaBombard, B. Lipschultz, E. Marmar, G. McCracken, M. Porkolab, J. Rice, J. Snipes, Y. Takase, J. Terry, S. Wolfe, C. Christensen, D. Garnier, M. Graf, T. Hsu, T. Luke, M. May, A. Niemczewski, G. Tinios, J. Schachter, and J. Urbahn. First results from Alcator-C-MOD. *Physics of Plasmas*, 1:1511, 1994.
- [48] J.L. Luxon. A design retrospective of the DIII-D tokamak. Nuclear Fusion, 42:614, 2002.
- [49] R.J. Groebner, K.H. Burrell, and R.P. Seraydarian. Role of edge electric field and poloidal rotation in the L-H transition. *Physical Review Letters*, 64:3015, 1990.
- [50] P. Gohil, T.E. Evans, J.R. Ferron, R.A. Moyer, C.C. Petty, K.H. Burrell, T.A. Casper, A.M. Garofalo, A.W. Hyatt, R.J. Jayakumar, C. Kessel, J.Y. Kim, R.J. La Haye, J Lohr, T.C. Luce, M.A. Makowski, D. Mazon, J. Menard, M. Murakami, P.A. Politzer, R. Prater, and M.R. Wade. Control of plasma profiles in DIII-D discharges. *Plasma Physics and Controlled Fusion*, 48:A45, 2006.
- [51] M. Greenwald, D. Andelin, N. Basse, S. Bernabei, P. Bonoli1, B. Böse, C. Boswell, R. Bravenec, B. Carreras, I. Cziegler, E. Edlund, D. Ernst, C. Fasoli, M. Ferrara, C. Fiore, R. Granetz, O. Grulke, T. Hender, J. Hosea, D.H. Howell, A. Hubbard, J. Hughes, I. Hutchinson, A. Ince-Cushman, J. Irby, B. LaBombard, R. LaHaye, L. Lin, Y. Lin, B. Lipschultz, J. Liptac, S. Lisgo, A. Lynn, E. Marmar, K. Marr, D.R. Mikkelsen, R. McDermott, D. Mossessian, A. Parisot, R. Parker, C. Phillips, P. Phillips, M. Porkolab, M. Redi, J. Rice, W. Rowan, M. Sampsell, G. Schilling, S. Scott, J.T. Scoville, N. Smick, J. Snipes, P. Stangeby, V. Tang,

J. Terry, M. Ulrickson, G. Wallace, D. Whyte, J. Wilson, J. Wright, S. Wolfe, S. Wukitch, B. Youngblood, H. Yuh, K. Zhurovich, and S. Zweben. Overview of the Alcator C-Mod program. *Nuclear Fusion*, 45:S109, 2005.

- [52] R.M. McDermott, B. Lipschultz, J.W. Hughes, P.J. Catto, A.E. Hubbard, I.H. Hutchinson, R.S. Granetz, M. Greenwald, B. LaBombard, K. Marr, M.L. Reinke, J.E. Rice, D. Whyte, and Alcator C-Mod Team. Edge radial electric field structure and its connections to H-mode confinement in Alcator C-Mod plasmas. *Physics of Plasmas*, 16:056103, 2009.
- [53] J.S. deGrassie, J.E. Rice, K.H. Burrell, R.J. Groebner, and W.M. Solomon. Intrinsic rotation in DIII-D. *Physics of Plasmas*, 14:056115, 2007.
- [54] G. Kagan and P.J. Catto. Arbitrary poloidal gyroradius effects in tokamak pedestals and transport barriers. *Plasma Physics and Controlled Fusion*, 50:085010, 2008.
- [55] F. Jenko, W. Dorland, M. Kotschenreuther, and B.N. Rogers. Electron temperature gradient driven turbulence. *Physics of Plasmas*, 7:1904, 2000.
- [56] A. Zeiler, J.F. Drake, and B. Rogers. Nonlinear reduced Braginskii equations with ion thermal dynamics in toroidal plasma. *Physics of Plasmas*, 4:2134, 1997.
- [57] A.N. Simakov and P.J. Catto. Drift-ordered fluid equations for field-aligned modes in low- β collisional plasma with equilibrium pressure pedestals. *Physics of Plasmas*, 10:4744, 2003.
- [58] X.S. Lee, J.R. Myra, and P.J. Catto. General frequency gyrokinetics. *Physics of Fluids*, 26:223, 1983.
- [59] I.B. Bernstein and P.J. Catto. Generalized gyrokinetics. *Physics of Fluids*, 28:1342, 1985.
- [60] F.I. Parra and P.J. Catto. Limitations of gyrokinetics on transport time scales. Plasma Physics and Controlled Fusion, 50:065014, 2008.
- [61] F.I. Parra and P.J. Catto. Gyrokinetic equivalence. Plasma Physics and Controlled Fusion, 51:065002, 2009.
- [62] M. Kotschenreuther, G. Rewoldt, and W.M. Tang. Comparison of initial value and eigenvalue codes for kinetic toroidal plasma instabilities. *Computer Physics Communications*, 88:128, 1995.
- [63] Y. Chen and S.E. Parker. Gyrokinetic turbulence simulations with kinetic electrons. *Physics of Plasmas*, 8:2095, 2001.
- [64] A.N. Simakov and P.J. Catto. Momentum transport in arbitrary mean-free path plasma with a Maxwellian lowest order distribution function. *Plasma Physics* and Controlled Fusion, 49:729, 2007.

- [65] A.G. Peeters and C. Angioni. Linear gyrokinetic calculations of toroidal momentum transport in a tokamak due to the ion temperature gradient mode. *Physics* of Plasmas, 12:072515, 2005.
- [66] R.E. Waltz, G.M. Staebler, J. Candy, and F.L. Hinton. Gyrokinetic theory and simulation of angular momentum transport. *Physics of Plasmas*, 14:122507, 2007.
- [67] A.N. Simakov and P.J. Catto. Self-consistent radial electric field in collisional screw-pinches and axisymmetric closed field line configurations in the absence of fluctuations. *Physics of Plasmas*, 13:112509, 2006.
- [68] N. Winsor, J.L. Johnson, and J.M. Dawson. Geodesic acoustic waves in hydromagnetic systems. *Physics of Fluids*, 11:2448, 1968.
- [69] Z. Gao, K. Itoh, H. Sanuki, and J.Q. Dong. Eigenmode analysis of geodesic acoustic modes. *Physics of Plasmas*, 15:072511, 2008.
- [70] A.G. Peeters, C. Angioni, and D. Strintzi. Toroidal momentum pinch velocity due to the Coriolis drift effect on small scale instabilities in a toroidal plasma. *Physical Review Letters*, 98:265003, 2007.
- [71] F.L. Hinton and S.K. Wong. Neoclassical ion transport in rotating axisymmetric plasmas. *Physics of Fluids*, 28:3082, 1985.
- [72] P.J. Catto, I.B. Bernstein, and M. Tessarotto. Ion transport in toroidally rotating tokamak plasmas. *Physics of Fluids*, 30:2784, 1987.
- [73] T.G. Northrop and J.A. Rome. Extensions of guiding center motion to higher order. *Physics of Fluids*, 21:384, 1978.
- [74] A.N. Simakov and P.J. Catto. Drift kinetic equation exact through second order in gyroradius expansion. *Physics of Plasmas*, 12:012105, 2005.